# The Metropolitan Water District of Southern California



The mission of the Metropolitan Water District of Southern California is to provide its service area with adequate and reliable supplies of high-quality water to meet present and future needs in an environmentally and economically responsible way.

### **L&C Committee**

- M. Luna, Chair
- J. Garza, Vice Chair
- M. Camacho
- G. Cordero
- L. Dick
- C. Douglas
- C. Kurtz
- T. McCov
- C. Miller
- G. Peterson
- M. Ramos
- K. Seckel

## Legal and Claims Committee - Final - Revised 2

Meeting with Board of Directors \*

February 13, 2024

8:30 a.m.

Tuesday, February 13, 2024 Meeting Schedule		
08:30 a.m. L&C		
10:00 a m FOD		

10:00 a.m. EOP 12:00 p.m. Break 12:30 p.m. OWS 02:30 p.m. BOD

Agendas, live streaming, meeting schedules, and other board materials are available here: https://mwdh2o.legistar.com/Calendar.aspx. If you have technical difficulties with the live streaming page, a listen-only phone line is available at 1-877-853-5257; enter meeting ID: 873 4767 0235. Members of the public may present their comments to the Board on matters within their jurisdiction as listed on the agenda via in-person or teleconference. To participate via teleconference 1-833-548-0276 and enter meeting ID: 876 9484 9772 or click https://us06web.zoom.us/j/87694849772? pwd=V3dGZGRYUjJ3allqdUxXTIJRM044Zz09

MWD Headquarters Building • 700 N. Alameda Street • Los Angeles, CA 90012 Teleconference Locations:

Cedars Sinai Medical Center • 8700 Beverly Blvd • Los Angeles, CA 90048
34819 Calle Del Sol • Capistrano Beach CA 92624
Long Beach Water Department • 1800 E Wardlow Road • Long Beach, CA 90807
3008 W. 82nd Place • Inglewood, CA 90305
1005 S Cardiff Street • Anaheim, CA 92806

- \* The Metropolitan Water District's meeting of this Committee is noticed as a joint committee meeting with the Board of Directors for the purpose of compliance with the Brown Act. Members of the Board who are not assigned to this Committee may participate as members of the Board, whether or not a quorum of the Board is present. In order to preserve the function of the committee as advisory to the Board, members of the Board who are not assigned to this Committee will not vote on matters before this Committee.
- 1. Opportunity for members of the public to address the committee on matters within the committee's jurisdiction (As required by Gov. Code Section 54954.3(a))

### 2. MANAGEMENT ANNOUNCEMENTS AND HIGHLIGHTS

**A.** General Counsel's report of monthly activities

**21-3016** 

Attachments: 02132024 LC 2A Report

\*\* CONSENT CALENDAR ITEMS -- ACTION \*\*

### 3. CONSENT CALENDAR OTHER ITEMS - ACTION

A. Approval of the Minutes of the Legal and Claims Committee for November 14, 2023 (Copies have been submitted to each Director, any additions, corrections, or omissions)

**21-3017** 

Attachments: 02132024 LC 3A (11142023) Minutes

### 4. CONSENT CALENDAR ITEMS - ACTION

**7-9** Human Resources Matter. [WITHDRAWN 2.6.2024]

21-3059

**7-10** Approve amendments to the Metropolitan Water District Administrative Code to modify the structure and duties of various committees; the General Manager has determined that the proposed action is exempt or otherwise not subject to CEQA

21-3020

21-3021

Attachments: 02132024 LC 7-10 B-L

7-11 Approve: (1) changes to the Metropolitan Water District 457(b) and 401(k) deferred compensation plans to establish an investment committee and adopt investment committee bylaws; and (2) amendments necessary to conform the plans and the Metropolitan Administrative Code to the proposed changes, current law, practices, and regulations; the General Manager has determined that the proposed action is exempt or otherwise not subject to

Attachments: 02132024 LC 7-11 B-L

02132024 LC 7-11 Presentation

### \*\* END OF CONSENT CALENDAR ITEMS \*\*

### 5. OTHER BOARD ITEMS - ACTION

NONE

### 6. BOARD INFORMATION ITEMS

NONE

### 7. COMMITTEE ITEMS

**CEQA** 

a. Quarterly Report on General Auditor Professional Services Contracts executed between September 1, 2023 and December 31, 2023, as required by Administrative Code Section 6453

21-3018

Page 3

b. Update on litigation challenging the California Department of Water Resources' compliance with the California Environmental Quality Act and other environmental laws regarding its certification of a Final Environmental Impact Report for and approval of the Delta Conveyance Project in the following nine cases: City of Stockton v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000009; County of Butte v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000011; County of Sacramento v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000014; County of San Joaquin et al. v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000010; Sacramento Area Sewer District v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000012; San Francisco Baykeeper, et al. v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000017; Sierra Club, et al. v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000008; South Delta Water Agency and Rudy Mussi Investment L.P. v. California Department of Water Resources, San County Superior Court Case STK-CV-UCP-2024-0000816; Tulare Lake Basin Water Storage District v. California Department of Water Resources, Sacramento County Superior Court Case No. 24WM000006. [ADDED ITEM 2/1/2024]

21-3061

Attachments: 02132024 LC 7b Presentation

c. Report on Sierra Club et al. v. California Department of Water Resources, Sacramento County Superior Court Case No. 34-2020-80003517 (consolidated with California Department of Water Resources v. All Persons Interested in the Matter, etc., Sacramento County Superior Court Case No. 34-2020-00283112) [Conference with legal counsel – existing litigation; to be heard in closed session pursuant to Gov. Code Section 54956.9(d)(1)]

21-3040

### 8. FOLLOW-UP ITEMS

NONE

### 9. FUTURE AGENDA ITEMS

### 10. ADJOURNMENT

Page 4

NOTE: This committee reviews items and makes a recommendation for final action to the full Board of Directors. Final action will be taken by the Board of Directors. Committee agendas may be obtained on Metropolitan's Web site https://mwdh2o.legistar.com/Calendar.aspx. This committee will not take any final action that is binding on the Board, even when a quorum of the Board is present.

Writings relating to open session agenda items distributed to Directors less than 72 hours prior to a regular meeting are available for public inspection at Metropolitan's Headquarters Building and on Metropolitan's Web site https://mwdh2o.legistar.com/Calendar.aspx.

Requests for a disability-related modification or accommodation, including auxiliary aids or services, in order to attend or participate in a meeting should be made to the Board Executive Secretary in advance of the meeting to ensure availability of the requested service or accommodation.



### Office of the General Counsel





### **Metropolitan Cases**

### Robert Ward v. Metropolitan, et al. (Riverside County Superior Court)

On December 18, 2023, temporary employee Robert Ward filed an employment lawsuit against Metropolitan and retired employee Clara Masannat in Riverside County Superior Court. The complaint was served on Metropolitan and Ms. Masannat on January 19. The complaint alleges 12 causes of action. Seven causes of action are under FEHA alleging: discrimination based on disability; harassment based on disability; failure to prevent, investigate, and remedy discrimination; failure to accommodate disability; failure to engage in the interactive process to accommodate disability; aiding abetting, inciting, compelling, or coercing acts forbidden by FEHA; retaliation for opposing practices forbidden by FEHA. In addition, there are causes of action alleging wrongful termination, intentional infliction of emotional distress, negligent infliction of emotional distress, breach of written contract, and breach of implied covenant of good faith and fair dealing. Plaintiff alleges Metropolitan failed to accommodate his request to be exempt from mask-wearing requirements in November 2021 and that Metropolitan unlawfully terminated him during the interactive process. The Legal Department is representing Metropolitan and Ms. Masannat and will meet and confer with Plaintiff to address legal deficiencies in the complaint. Metropolitan anticipates responding to the complaint in February 2024.

## Ryan Tiegs v. Metropolitan, et al. (Riverside County Superior Court)

On December 18, 2023, employee Ryan Tiegs filed an employment lawsuit against Metropolitan in Riverside County Superior Court. Plaintiff served the lawsuit on Metropolitan on January 11, 2024. The complaint alleges six causes of action under the California Fair Employment and Housing Act (FEHA): race discrimination; race harassment; gender discrimination; gender harassment; retaliation; and failure to prevent harassment, discrimination, and retaliation. Plaintiff alleges he was harassed and discriminated against based on his race and gender, Metropolitan approved and ratified this, he was retaliated against because of his complaints, and Metropolitan failed to prevent

these occurrences. Metropolitan's answer or other responsive pleading is due on February 13, 2024.

### **State Water Project Contract Extension Litigation (Sacramento Superior Court)**

On January 5, 2024, the Third District Court of Appeal issued a detailed 50-page opinion affirming the validity of the State Water Project (SWP) Contract Extension Amendments (Amendments).

These Amendments were the product of extensive negotiations between the Department of Water Resources (DWR) and the State Water Contractors (SWCs), which culminated in June 2014 with an agreement in principle (AIP). That AIP, in turn, served as the framework for the actual language of the Amendments, as well as the "proposed project" for purposes of environmental review under California Environmental Quality Act (CEQA).

Among other things, the Amendments extend the SWP contracts to 2085 and make certain changes to their financial provisions aimed at maintaining and improving the SWP's fiscal integrity and management. DWR opted to prepare an EIR in support of the Amendments, which did not identify any potentially significant impacts. DWR certified the EIR in November 2018, approved the Amendments in December 2018, and immediately filed an action seeking to validate them. Shortly thereafter, two additional lawsuits were filed by various environmental groups challenging approval of the Amendments. Metropolitan filed an answer in support of validation and intervened in the other two lawsuits. These cases were consolidated in Sacramento Superior Court and, as previously reported, the trial court issued a decision in favor of DWR in May 2022.

Although a variety of claims were asserted, most centered around alleged violations of CEQA, the Delta Reform Act and the Public Trust Doctrine, with concerns being raised about impacts to the Delta resulting from ongoing and future operations of the SWP, in general, and the proposed Delta Conveyance Project (DCP), in particular.

With respect to the CEQA claims, the court of appeal agreed that the Amendments were largely

financial in nature and that the extension of the SWP Contract would not, in and of itself, alter the way the SWP was operated or maintained. The court also agreed that it was reasonable for DWR to assume that the SWP would continue as an ongoing concern even if the Amendments were not approved. As such, the existing SWP operations constituted the appropriate baseline by which to assess potential impacts, which supported the conclusion that no such impacts would occur. Finally, the court rejected arguments that potential impacts of the proposed DCP should have been examined as part of the environmental review process for the Amendments. The court noted that the Amendments had an independent purpose and that the proposed DCP was speculative as to both its timing and scope.

With respect to the Delta Reform Act claims, the court of appeal agreed that approval of the Amendments did not constitute a "covered action" requiring certification of consistency with the Delta Plan. In reaching this conclusion, the court noted that the Amendments did not involve future developments affecting the Delta or actions physically taking place within the Delta, nor would they have any significant impact to California's water supply or the Delta's ecosystem. The court further noted that this Act does not apply to

"routine maintenance and operation of the SWP."
On this point, the court stated that it was not necessary to decipher the "precise meaning" of this phrase. Rather, "[i]t suffices to note the Legislature's clear intent to exempt the existing State Water Project from a covered action."

Lastly, with respect to the Public Trust Doctrine claims, the court of appeal began by questioning whether the duty to protect the water resources at issue fell upon DWR, rather than the State Water Resources Control Board. The court concluded they did not, since DWR is not responsible for approving the water diversions that form the basis of the SWP's supplies. The court then went on to note that while this duty might apply to other state agencies in certain circumstances, that was not the case here. The "determinative fact" is whether these Amendments would "have an impact on water 'that is imbued with the public trust." Based on the record before it, the court agreed with DWR's determination that they did not.

On January 16, 2024, a petition for rehearing was filed with the court of appeal. That petition was denied on January 25, with the court making just two minor modifications to its opinion. Accordingly, the appellants' last option is to seek review of this opinion by the California Supreme Court.

### **Matters Concluded and/or Terminated**

### AFSCME Local 1902 v. Metropolitan (Public Employment Relations Board)

On March 6, 2020, AFSCME Local 1902 filed an unfair practice charge with the Public Employment Relations Board alleging Metropolitan engaged in an unfair labor practice in violation of the Myers-Milias-Brown Act and PERB regulations by making unilateral changes to Metropolitan's deferred compensation plan documents without negotiation. Metropolitan contended that it attempted to meet and confer with the bargaining unit prior to implementing the modifications but the plan changes were legally mandated to be in place by end of calendar year 2019 to avoid potential IRS penalties and placing the plan itself in legal jeopardy. Consequently, Metropolitan made the

challenged modifications to the deferred compensation plan documents without completing negotiations with the bargaining unit, with the understanding the negotiations would continue.

The parties engaged in multiple rounds of settlement talks. Talks then progressed at the bargaining table between Metropolitan management and Local 1902 during recent MOU negotiations which were fruitful and resulted in a settlement. Under the terms of the settlement, Local 1902 withdrew the charge and the parties agreed that any potential changes to the deferred compensation plan documents going forward will be discussed between the parties. The settlement contains no financial terms or admission of fault. The Legal Department represented Metropolitan in this matter.

### **Matters Received**

Category	Received	<u>Description</u>	
Action in which MWD is a party	2	Disability; (2) Harassment Prevent, Investigate, and I Retaliation; (4) Failure to A Engage in the Interactive I (6) Aiding, Abetting, Incitin Forbidden by FEHA; (7) R Forbidden by FEHA; (8) W Policy; (9) Intentional Infliction of Emotional Dist (12) Breach of Implied Cov	(1) Discrimination on the Basis of on the Basis of Disability; (3) Failure to Remedy Discrimination, Harassment, or Accommodate Disability; (5) Failure to Process to Accommodate Disability; ag, Compelling, or Coercing Acts etaliation for Opposing Practices /rongful Termination in Violation of Public etion of Emotional Distress; (10) Negligent rict; (11) Breach of Written Contract; venant of Good Faith and Fair Dealing, Superior Court, in the case Robert Ward v. 110449
		Harassment; (3) Gender D (5) Retaliation in Violation Harassment, Discrimination	(1) Race Discrimination; (2) Race Discrimination; (4) Gender Harassment; of FEHA; (6) Failure to Prevent on and Retaliation, filed in Riverside the case <i>Ryan Tiegs v. MWD</i> , Case
Government Code Claims	3	(2) damage to an employe	accident involving an MWD vehicle; ee's appliance from an electrical receptacle ling from a bicycle while riding over loose n Mustang Way in Hemet
Subpoenas	3	defendant in the case Hoo Industrial Partners, LLC, L No. 20STCV41618 6712-0 concerning breach of cont plaintiff/commercial tenant commercial tenant subleas MWD for the PCCP replace construction vehicle parking construction-related uses. MWD's testimony at trial communications to MWD master lease agreement.	al Appearance at Trial, served by the man Enterprises, Inc. v. Long Beach are and the defendant/property owner. The sed a portion of its leased property to sement project in the Long Beach area for and, material storage and staging, and other the defendant/property owner seeks oncerning the plaintiff/commercial tenant's regarding the terms and conditions of its MWD is not a party to this lawsuit.
Requests Pursuant to the Public Records	23	Requestor	Documents Requested
Act		Allied Universal	Addendums/modifications to the contract for security guard services and current pricing/rates

Date of Report: January 3, 2024

<u>Requestor</u>	Documents Requested
American Safety Group	Contract awardee for On-Call General Industrial Hygiene and Safety Services
Berkshire Hathaway Homestate Companies	Bid document, certified payroll records, and contract relating to Golden West Arbor Services, Inc.
California Department of Transportation (Caltrans District 11)	GIS or as-built files for MWD utility facilities near project on Route 78
Catalyst Environmental Solutions	Task orders for the contract for Tunnel Design Engineering Consulting Services
CCS Global Tech	Details of contract for Dam Safety Monitoring, Instrumentation and Data Management Services
CHC Consulting	Information on MWD's waterline near AT&T fiber cable installation project in Anaheim
City of Oxnard, Office of the City Attorney	Professional services agreement between MWD and Meyers Nave for legal services
City of Pomona Public Works Department, Engineering Division	Facilities location maps/as-builts for MWD facilities within the city of Pomona's Capital Improvement Project for new streetlights
Reporter, CityWatchLA	Records relating to certain Ongoing Ethics/EEO related investigations, consulting services provided by and hiring of Mohsen Mortada, and MWD Administrative Code changes from January 1, 2023, through the present
Deltek	Awarded contract and bid tabulation relating to Request for Proposal for Data Management and Data Analytics Consulting & Implementation Services
GrayMar Environmental Services	Current contract for Encampment Abatement & General Clean-up Services
Hardy & Hardy	Contract payment documents, including most current pay application, proof of payment, remittance history, notice of completion, bond, change orders relating to general contractor Ameresco's work on Skinner Water Treatment Plant project
Jacobs	Proposals submitted in request for Request for Qualifications for Progressive Design-Build Services for Sepulveda Feeder Pump Stations

Project

<u>Requestor</u>	Documents Requested
Law Offices of Patrick D. Webb	MWD appraisal (submitted to DOI in about 1940) of the valuation of the land to be purchased to create Lake Havasu and the Department of Interior order designating the reservation lands to be taken to create Lake Havasu
Los Angeles County Sanitation Districts	Contract for sodium bisulfite
Private Citizens (2 requests)	(1) Estimates and invoices for the production of the book Care & Maintenance of Southern California Native Plant Gardens; and (2) MWD response to California Department of Industrial Relations, Division of Occupational Safety and Health ("Cal/OSHA") regarding complaint of exposure to lead from removing paint from overhead crane
R.E.Y. Engineers	Documents showing MWD's right-of-way in the Berdoo Canyon area
Sherpa Marketing Solutions	Awarded proposal and budget submitted in response to the Request for Proposal for Multi-Media Placement Consulting Services for Water Awareness and Outreach Campaign
Sierra Club	Report on the 2070 metrics that the Department of Water Resources made for MWD
Student, University of Northern Colorado	Data on number of rebate participants and amount of turf removed during 2023 under the Turf Replacement Program
U.S. Department of the Interior, Bureau of Reclamation	MWD's Colorado River Watershed Sanitary Survey 2020 Update

### PLEASE NOTE

- ADDITIONS ONLY IN THE FOLLOWING TWO TABLES WILL BE SHOWN IN RED.
- ANY CHANGE TO THE *OUTSIDE COUNSEL AGREEMENTS*TABLE WILL BE SHOWN IN REDLINE FORM (I.E., ADDITIONS, REVISIONS, DELETIONS).

### **Bay-Delta and SWP Litigation**

### Subject Status

### **Delta Conveyance Project CEQA Cases**

<u>City of Stockton v. California Department of Water</u> <u>Resources</u>

<u>County of Butte v. California Department of Water</u> <u>Resources</u>

<u>County of Sacramento v. California Department of Water Resources</u>

County of San Joaquin et al. v. California

Department of Water Resources

<u>Sacramento Area Sewer District v. California</u> Department of Water Resources

<u>San Francisco Baykeeper, et al. v. California</u> Department of Water Resources

<u>Sierra Club, et al. v. California Department of Water</u> <u>Resources</u>

<u>South Delta Water Agency and Rudy Mussi</u> <u>Investment L.P. v. California Department of Water</u> Resources

Tulare Lake Basin Water Storage District v. California Department of Water Resources

- DWR is the only named respondent/defendant
- All alleged CEQA violations
- Most allege violations of the Delta Reform Act,
   Public Trust Doctrine and Delta and
   Watershed Protection Acts
- Two allege violations of the fully protected bird statute
- One alleges violations of Proposition 9 (1982)
   and the Central Valley Project Act
- All but South Delta Water Agency's case were filed in Sacramento County Superior Court
- South Delta Water Agency filed in San Joaquin County Superior Court

### Consolidated DCP Revenue Bond Validation Action and CEQA Case

Sierra Club, et al. v. California Department of Water Resources (CEQA, designated as lead case)

DWR v. All Persons Interested (Validation)

Sacramento County Superior Ct. (Judge Kenneth C. Mennemeier)

### Validation Action

- Metropolitan, Mojave Water Agency, Coachella Valley Water District, and Santa Clarita Valley Water Agency have filed answers in support
- Kern County Water Agency, Tulare Lake Basin Water Storage District, Oak Flat Water District, County of Kings, Kern Member Units & Dudley Ridge Water District, and City of Yuba City filed answers in opposition
- North Coast Rivers Alliance et al., Howard Jarvis Taxpayers Association, Sierra Club et al., County of Sacramento & Sacramento County Water Agency, CWIN et al., Clarksburg Fire Protection District, Delta Legacy Communities, Inc, and South Delta Water Agency & Central Delta Water Agency have filed answers in opposition

Subject	Status
	Case ordered consolidated with the DCP Revenue Bond CEQA Case for pre-trial and trial purposes and assigned to Judge Earl for all purposes
	<ul> <li>DWR's motions for summary judgment re GEQA affirmative defenses granted; cross- motions by opponents denied</li> </ul>
	<ul> <li>Dec. 9, 2022 DWR's motion for summary adjudication of Delta Reform Act and public trust doctrine affirmative defenses granted; NCRA's motion for summary judgment re same denied</li> </ul>
	Trial on the merits held May 15-18, 2023
	<ul> <li>Final Judgment and Final Statement of Decision issued January 16, 2024</li> </ul>
	<ul> <li>Judgment in DWR's favor on CEQA, Delta</li> <li>Reform Act, Public Trust Doctrine and jurisictional causes of action or defenses</li> </ul>
	<ul> <li>Bonds ruled not valid based on the broad definition of "Delta Program facilities" in the General Bond Resolution</li> </ul>
	<ul> <li>Supplemental briefing ordered on three issues with final brief due June 30, 2023</li> </ul>
	<ul> <li>Tentative Decision/Proposed Statement of Decision against validity issued Aug. 25</li> </ul>
	<ul> <li>DWR's objections filed September 18, 2023</li> <li>Opponents responses to objections filed September 28, 2023</li> </ul>
	<ul> <li>Court extended the deadline to issue a judgment to January 22, 2024</li> </ul>
	• CEQA Case
	<ul> <li>Sierra Club, Center for Biological Diversity,         Planning and Conservation League,         Restore the Delta, and Friends of Stone         Lakes National Wildlife Refuge filed a     </li> </ul>
	standalone CEQA lawsuit challenging  DWR's adoption of the bond resolutions
	<ul> <li>Alleges DWR violated CEQA by adopting bond resolutions before certifying a Final</li> </ul>
	<ul> <li>EIR for the Delta Conveyance Project</li> <li>Cases ordered consolidated for all purposes</li> </ul>
	<ul> <li>DWR's motion for summary judgment</li> </ul>
	granted; Sierra Club's motion denied
	<ul> <li>Tentative Decision/Proposed Statement of Decision rejecting CEQA challenge issued Aug. 25, 2023</li> </ul>
	<ul> <li>DWR's objections filed September 18, 2023</li> </ul>

Subject	Status
	<ul> <li>Opponents responses to objections filed September 28, 2023</li> <li>Court extended the deadline to issue a judgment to January 22, 2024</li> </ul>
SWP-CVP 2019 BiOp Cases  Pacific Coast Fed'n of Fishermen's Ass'ns, et al. v. Raimondo, et al. (PCFFA)  Calif. Natural Resources Agency, et al. v. Raimondo, et al. (CNRA)  Federal District Court, Eastern Dist. of California, Fresno Division (Judge Thurston)	<ul> <li>SWC intervened in both PCFFA and CNRA cases</li> <li>Federal defendants reinitiated consultation on Oct 1, 2021</li> <li>Nov. 16, 2023 parties filed a joint status report</li> <li>Federal defendants and state plaintiffs seek another 1-year stay and proposed a 2024 Interim Operations Plan (IOP); PCFFA seeks to extend the 2023 IOP until the court rules on the 2024 IOP</li> <li>Briefing on stay extension and 2024 IOP concludes March 6, 2024</li> <li>Dec. 29, 2023 order extended the stay and 2023 IOP until March 2024 or new order, whichever is earlier</li> </ul>
Cesa Incidental Take Permit Cases  Coordinated Case Name CDWR Water Operations Cases, JCCP 5117 (Coordination Trial Judge Gevercer)  Metropolitan & Mojave Water Agency v. Calif. Dept. of Fish & Wildlife, et al. (CESA/CEQA/Breach of Contract)  State Water Contractors & Kern County Water Agency v. Calif. Dept. of Fish & Wildlife, et al. (CESA/CEQA)  Tehama-Colusa Canal Auth., et al. v. Calif. Dept. of Water Resources (CEQA)  San Bernardino Valley Municipal Water Dist. v. Calif. Dept. of Water Resources, et al. (CEQA/CESA/ Breach of Contract/Takings)  Sierra Club, et al. v. Calif. Dept. of Water Resources (CEQA/Delta Reform Act/Public Trust)	<ul> <li>All 8 cases ordered coordinated in Sacramento County Superior Court</li> <li>Stay on discovery issued until coordination trial judge orders otherwise</li> <li>All four Fresno cases transferred to Sacramento to be heard with the four other coordinated cases</li> <li>Certified administrative records lodged March 4, 2022</li> <li>State Water Contractors et al. granted leave to intervene in Sierra Club, North Coast Rivers Alliance, Central Delta Water Agency, and San Francisco Baykeeper cases by stipulation</li> <li>SWC, et al. granted leave to intervene as respondents in Tehama-Colusa Canal Auth., et al. v. Calif. Dept. of Water Resources CEQA case</li> <li>SWC's renewed motion to augment the administrative records granted in part; a courtappointed referee will review withheld records to determine if the deliberative process privilege applies</li> </ul>
North Coast Rivers Alliance, et al. v. Calif. Dept. of Water Resources (CEQA/Delta Reform Act/Public Trust)	<ul> <li>Sept. 8, 2023 hearing on DWR's and CDFW's motion to modify the referral to exclude certain withheld records</li> </ul>

Subject	Status
Central Delta Water Agency, et. al. v. Calif. Dept. of Water Resources (CEQA/Delta Reform Act/Public Trust/ Delta Protection Acts/Area of Origin)  San Francisco Baykeeper, et al. v. Calif. Dept. of Water Resources, et al. (CEQA/CESA)	<ul> <li>CDFW's motion denied, DWR's motion subject to the Court's in camera review of records proposed for exclusion</li> <li>Referee's recommendation is to grant in part, deny in part SWC parties' motion to augment the administrative records</li> <li>Oct. 13, 2023 objections or responses to Referee's recommendation due</li> <li>Oct. 27, 2023 court's ruling granting in part, and denying in part, the SWC parties' motion to augment DWR's and CDFW's administrative records became final</li> <li>Parties are conferring on a merits briefing schedule</li> </ul>
CDWR Environmental Impact Cases Sacramento Superior Ct. Case No. JCCP 4942, 3d DCA Case No. C091771 (20 Coordinated Cases)  Validation Action DWR v. All Persons Interested  CEQA 17 cases  CESA/Incidental Take Permit 2 cases  (Judge Arguelles)	<ul> <li>Cases dismissed after DWR rescinded project approval, bond resolutions, decertified the EIR, and CDFW rescinded the CESA incidental take permit</li> <li>January 10, 2020 – Nine motions for attorneys' fees and costs denied in their entirety</li> <li>Parties have appealed attorneys' fees and costs rulings</li> <li>May 11, 2022, court of appeal reversed the trial court's denial of attorney fees and costs in an unpublished opinion</li> <li>Opinion ordered published</li> <li>Coordinated cases remitted to trial court for re-hearing of fee motions consistent with the court of appeal's opinion</li> <li>Sept.15, 2023 re-hearing on fee motions</li> <li>Dec. 26, 2023 order denying fee motions</li> <li>Jan. 11, 2024 deadline to file motion for reconsideration</li> <li>Feb. 26, 2024 deadline to file notice of appeal</li> <li>Six notices of appeal filed</li> </ul>
COA Addendum/ No-Harm Agreement  North Coast Rivers Alliance v. DWR Sacramento County Superior Ct. (Judge Rockwell)	<ul> <li>Plaintiffs allege violations of CEQA, Delta Reform Act &amp; public trust doctrine</li> <li>USBR Statement of Non-Waiver of Sovereign Immunity filed September 2019</li> <li>Westlands Water District and North Delta Water Agency granted leave to intervene</li> <li>Metropolitan &amp; SWC monitoring</li> </ul>

Subject	Status
	Deadline to prepare administrative record last extended to Nov. 18, 2022
SWP Contract Extension Validation Action Court of Appeal for the Third App. Dist. Case No. C096316  DWR v. All Persons Interested in the Matter, etc.	<ul> <li>DWR seeks a judgment that the Contract Extension amendments to the State Water Contracts are lawful</li> <li>Metropolitan and 7 other SWCs filed answers in support of validity to become parties</li> <li>Jan. 5-7, 2022 Hearing on the merits held with CEQA cases, below</li> <li>Final statement of decision in DWR's favor filed March 9, 2022</li> <li>Final judgment entered and served</li> <li>C-WIN et al., County of San Joaquin et al. and North Coast Rivers Alliance et al. filed notices of appeal</li> <li>Validation and CEQA cases consolidated on appeal</li> <li>Briefing completed May 30, 2023</li> <li>Oral argument held November 15, 2023</li> <li>January 5, 2024 court of appeal affirmed the trial court judgment</li> <li>CWIN et al. NCRA et al.'s petitions for reconsideration denied</li> </ul>
SWP Contract Extension CEQA Cases Court of Appeal for the Third App. Dist. Case Nos. C096384 & C096304  North Coast Rivers Alliance, et al. v. DWR  Planning & Conservation League, et al. v. DWR	<ul> <li>Petitions for writ of mandate alleging CEQA and Delta Reform Act violations filed on January 8 &amp; 10, 2019</li> <li>Deemed related to DWR's Contract Extension Validation Action and assigned to Judge Culhane</li> <li>Administrative Record completed</li> <li>DWR filed its answers on September 28, 2020</li> <li>Metropolitan, Kern County Water Agency and Coachella Valley Water District have intervened and filed answers in the two CEQA cases</li> <li>Final statement of decision in DWR's favor denying the writs of mandate filed March 9, 2022</li> <li>Final judgments entered and served</li> <li>North Coast Rivers Alliance et al. and PCL et al. filed notices of appeal</li> </ul>

Subject	Status
	Appeals consolidated with the validation action above
Delta Conveyance Project Soil Exploration Cases  Central Delta Water Agency, et al. v. DWR Sacramento County Superior Ct. (Judge Chang)  Central Delta Water Agency, et al. v. DWR (II), Sacramento County Super. Ct. (Judge Acquisto)	<ul> <li>Original case filed August 10, 2020; new case challenging the second addendum to the CEQA document filed Aug. 1, 2022</li> <li>Plaintiffs Central Delta Water Agency, South Delta Water Agency and Local Agencies of the North Delta</li> <li>One cause of action alleging that DWR's adoption of an Initial Study/Mitigated Negative Declaration (IS/MND) for soil explorations needed for the Delta Conveyance Project violates CEQA</li> <li>March 24, 2021 Second Amended Petition filed to add allegation that DWR's addendum re changes in locations and depths of certain borings violates CEQA</li> <li>DWR's petition to add the 2020 CEQA case to the <i>Department of Water Resources Cases</i>, JCCP 4594, San Joaquin County Superior Court denied</li> <li>Hearing on the merits held Oct.13, 2022</li> <li>Dec. 2, 2022 ruling on the merits granting the petition with respect to two mitigation measures and denying on all other grounds</li> <li>Dec. 23, 2022 court order directing DWR to address the two mitigation measures within 60 days while declining to order DWR to vacate the IS/MND</li> <li>March 27, 2023 court entered judgment and issued a writ after ordering and considering supplemental briefing</li> <li>May 5, 2023 court granted DWR's motion to discharge the writ and dismiss the case</li> <li>May 18, 2023 Notice of Appeal filed</li> <li>Hearing on motion for attorneys' fees continued to February 29, 2024</li> </ul>
Water Management Tools Contract Amendment California Water Impact Network et al. v. DWR	<ul> <li>Filed September 28, 2020</li> <li>CWIN and Aqualliance allege one cause of action for violation of CEQA</li> </ul>
Sacramento County Superior Ct. (Judge Aquisto)	NCRA et al. allege four causes of action for violations of CEQA, the Delta Reform Act, Public Trust Doctrine and seeking declaratory relief

Page 12 of 23

Subject	Status
North Coast Rivers Alliance, et al. v. DWR Sacramento County Super. Ct. (Judge Aquisto)	<ul> <li>SWC motion to intervene in both cases granted</li> <li>Dec. 20, 2022 DWR filed notice of certification of the administrative record and filed answers in both cases</li> </ul>

San Diego County Water Authority v. Metropolitan, et al.		
Cases	Date	Status
2014, 2016	Aug. 28, 2020	SDCWA served first amended (2014) and second amended (2016) petitions/complaints.
	Sept. 28	Metropolitan filed demurrers and motions to strike portions of the amended petitions/complaints.
	Sept. 28-29	Member agencies City of Torrance, Eastern Municipal Water District, Foothill Municipal Water District, Las Virgenes Municipal Water District, Three Valleys Municipal Water District, Municipal Water District of Orange County, West Basin Municipal Water District, and Western Municipal Water District filed joinders to the demurrers and motions to strike.
	Feb. 16, 2021	Court issued order denying Metropolitan's demurrers and motions to strike, allowing SDCWA to retain contested allegations in amended petitions/complaints.
	March 22	Metropolitan filed answers to the amended petitions/complaints and cross-complaints against SDCWA for declaratory relief and reformation, in the 2014, 2016 cases.
	March 22-23	Member agencies City of Torrance, Eastern Municipal Water District, Foothill Municipal Water District, Las Virgenes Municipal Water District, Three Valleys Municipal Water District, Municipal Water District of Orange County, West Basin Municipal Water District, and Western Municipal Water District filed answers to the amended petitions/complaints in the 2014, 2016 cases.
	April 23	SDCWA filed answers to Metropolitan's cross-complaints.
	Sept. 30	Based on the Court of Appeal's Sept. 21 opinion (described above), and the Board's Sept. 28 authorization, Metropolitan paid \$35,871,153.70 to SDCWA for 2015-2017 Water Stewardship Rate charges under the Exchange Agreement and statutory interest.
2017	July 23, 2020	Dismissal without prejudice entered.
2018	July 28, 2020	Parties filed a stipulation and application to designate the case complex and related to the 2010-2017 cases, and to assign the case to Judge Massullo's court.
	Nov. 13	Court ordered case complex and assigned to Judge Massullo's court.
	April 21, 2021	SDCWA filed second amended petition/complaint.
	May 25	Metropolitan filed motion to strike portions of the second amended petition/complaint.

Cases	Date	Status
2018 (cont.)	May 25-26	Member agencies City of Torrance, Eastern Municipal Water District, Foothill Municipal Water District, Las Virgenes Municipal Water District, Three Valleys Municipal Water District, Municipal Water District of Orange County, West Basin Municipal Water District, and Western Municipal Water District filed joinders to the motion to strike.
	July 19	Court issued order denying Metropolitan's motion to strike portions of the second amended petition/complaint.
	July 29	Metropolitan filed answer to the second amended petition/complaint and cross-complaint against SDCWA for declaratory relief and reformation.
	July 29	Member agencies City of Torrance, Eastern Municipal Water District, Foothill Municipal Water District, Las Virgenes Municipal Water District, Three Valleys Municipal Water District, Municipal Water District of Orange County, West Basin Municipal Water District, and Western Municipal Water District filed answers to the second amended petition/complaint.
	Aug. 31	SDCWA filed answer to Metropolitan's cross-complaint.
	April 11, 2022	Court entered order of voluntary dismissal of parties' WaterFix claims and cross-claims.
2014, 2016, 2018	June 11, 2021	Deposition of non-party witness.
	Aug. 25	Hearing on Metropolitan's motion for further protective order regarding deposition of non-party witness.
	Aug. 25	Court issued order consolidating the 2014, 2016, and 2018 cases for all purposes, including trial.
	Aug. 30	Court issued order granting Metropolitan's motion for a further protective order regarding deposition of non-party witness.
	Aug. 31	SDCWA filed consolidated answer to Metropolitan's cross-complaints in the 2014, 2016, and 2018 cases.
	Oct. 27	Parties submitted to the court a joint stipulation and proposed order staying discovery through Dec. 8 and resetting pre-trial deadlines.
	Oct. 29	Court issued order staying discovery through Dec. 8 and resetting pretrial deadlines, while the parties discuss the prospect of settling some or all remaining claims and crossclaims.
	Jan. 12, 2022	Case Management Conference. Court ordered a 35-day case stay to allow the parties to focus on settlement negotiations, with weekly written check-ins with the court; and directed the parties to meet and confer regarding discovery and deadlines.

Cases	Date	Status
2014, 2016, 2018 (cont.)	Feb. 22	Court issued order resetting pre-trial deadlines as proposed by the parties.
	Feb. 22	Metropolitan and SDCWA each filed motions for summary adjudication.
	April 13	Hearing on Metropolitan's and SDCWA's motions for summary adjudication.
	April 18	Parties filed supplemental briefs regarding their respective motions for summary adjudication, as directed by the court.
	April 18	Court issued order resetting pre-trial deadlines as proposed by the parties.
	April 29	Parties filed pre-trial briefs.
	April 29	Metropolitan filed motions in limine.
	May 4	Court issued order granting Metropolitan's motion for summary adjudication on cross-claim for declaratory relief that the conveyance facility owner, Metropolitan, determines fair compensation, including any offsetting benefits; and denying its motion on certain other cross-claims and an affirmative defense.
	May 11	Court issued order granting SDCWA's motion for summary adjudication on cross-claim for declaratory relief in the 2018 case regarding lawfulness of the Water Stewardship Rate's inclusion in the wheeling rate and transportation rates in 2019-2020; certain cross-claims and affirmative defenses on the ground that Metropolitan has a duty to charge no more than fair compensation, which includes reasonable credit for any offsetting benefits, with the court also stating that whether that duty arose and whether Metropolitan breached that duty are issues to be resolved at trial; affirmative defenses that SDCWA's claims are untimely and SDCWA has not satisfied claims presentation requirements; affirmative defense in the 2018 case that SDCWA has not satisfied contract dispute resolution requirements; claim, crossclaims, and affirmative defenses regarding applicability of Proposition 26, finding that Proposition 26 applies to Metropolitan's rates and charges, with the court also stating that whether Metropolitan violated Proposition 26 is a separate issue; and cross-claims and affirmative defenses regarding applicability of Government Code section 54999.7, finding that section 54999.7 applies to Metropolitan's rates. Court denied SDCWA's motion on certain other cross-claims and affirmative defenses.
	May 13	Pre-trial conference; court denied Metropolitan's motions in limine.
	May 16	Court issued order setting post-trial brief deadline and closing arguments.
	May 16-27	Trial occurred but did not conclude.

Cases	Date	Status
2014, 2016, 2018 (cont.)	May 23, June 21	SDCWA filed motions in limine.
	May 26, June 24	Court denied SDCWA's motions in limine.
	June 3, June 24, July 1	Trial continued, concluding on July 1.
	June 24	SDCWA filed motion for partial judgment.
	July 15	Metropolitan filed opposition to motion for partial judgment.
	Aug. 19	Post-trial briefs filed.
	Sept. 14	Court issued order granting in part and denying in part SDCWA's motion for partial judgment (granting motion as to Metropolitan's dispute resolution, waiver, and consent defenses; denying motion as to Metropolitan's reformation cross-claims and mistake of fact and law defenses; and deferring ruling on Metropolitan's cost causation cross-claim).
	Sept. 21	Metropolitan filed response to order granting in part and denying in part SDCWA's motion for partial judgment (requesting deletion of Background section portion relying on pleading allegations).
	Sept. 22	SDCWA filed objection to Metropolitan's response to order granting in part and denying in part SDCWA's motion for partial judgment.
	Sept. 27	Post-trial closing arguments.
	Oct. 20	Court issued order that it will rule on SDCWA's motion for partial judgment as to Metropolitan's cost causation cross-claim simultaneously with the trial statement of decision.
	Dec. 16	Parties filed proposed trial statements of decision.
	Dec. 21	SDCWA filed the parties' stipulation and proposed order for judgment on Water Stewardship Rate claims for 2015-2020.
	Dec. 27	Court entered order for judgment on Water Stewardship Rate claims for 2015-2020 as proposed by the parties.
	March 14, 2023	Court issued tentative statement of decision (tentatively ruling in Metropolitan's favor on all claims litigated at trial, except for those ruled to be moot based on the rulings in Metropolitan's favor)
	March 14	Court issued amended order granting in part and denying in part SDCWA's motion for partial judgment (ruling that Metropolitan's claims for declaratory relief regarding cost causation are not subject to court review).

Cases	Date	Status
	March 29	SDCWA filed objections to tentative statement of decision
	April 3	Metropolitan filed response to amended order granting in part and denying in part SDCWA's motion for partial judgment (requesting deletion of Background section portion relying on pleading allegations).
	April 25	Court issued statement of decision (ruling in Metropolitan's favor on all claims litigated at trial, except for those ruled to be moot based on the rulings in Metropolitan's favor)
	Jan. 10, 2024	Parties filed joint status report and stipulated proposal on form of judgment
	<u>Jan. 17</u>	Court issued order approving stipulated proposal on form of judgment (setting briefing and hearing)
	<u>Jan. 26</u>	Parties filed opening briefs on proposed form of judgment
	March 13	Hearing on proposed form of judgment
All Cases	April 15, 2021	Case Management Conference on 2010-2018 cases. Court set trial in 2014, 2016, and 2018 cases on May 16-27, 2022.
	April 27	SDCWA served notice of deposition of non-party witness.
	May 13-14	Metropolitan filed motions to quash and for protective order regarding deposition of non-party witness.
	June 4	Ruling on motions to quash and for protective order.

Outside Counsel Agreements					
Firm Name	Matter Name	Agreement No.	Effective Date	Contract Maximum	
Albright, Yee & Schmit, APC	Employment Matter	211923	05/23	\$60,000	
Andrade Gonzalez LLP	MWD v. DWR, CDFW and CDNR Incidental Take Permit (ITP) CESA/CEQA/Contract Litigation	185894	07/20	\$250,000	
Aleshire & Wynder	Oil, Mineral and Gas Leasing	174613	08/18	\$50,000	
Atkinson Andelson Loya Ruud & Romo	Employee Relations	59302	04/04	\$1,277,187	
Loya Rudu & Rollio	Delta Conveyance Project Bond Validation-CEQA Litigation	185899	09/21	\$250,000	
	MWD Drone and Airspace Issues	193452	08/20	\$50,000	
	AFSCME Local 1902 in Grievance No. 1906G020 (CSU Meal Period)	201883	07/12/21	\$30,000	
	AFSCME Local 1902 v. MWD, PERB Case No. LA-CE-1438-M	201889	09/15/21	\$20,000	
	MWD MOU Negotiations**	201893	10/05/21	\$100,000	
Best, Best & Krieger	Bay-Delta Conservation Plan/Delta Conveyance Project (with SWCs)	170697	08/17	\$500,000	
	Environmental Compliance Issues	185888	05/20	\$100,000	
	Grant Compliance Issues	211921	05/23	\$75,000	
	Pure Water Southern California	207966	11/22	\$100,000	
Blooston, Mordkofsky, Dickens, Duffy & Prendergast, LLP	FCC and Communications Matters	110227	11/10	\$100,000	
Buchalter, a Professional Corp.	Union Pacific Industry Track Agreement	193464	12/07/20	\$50,000	

Firm Name	Matter Name	Agreement No.	Effective Date	Contract Maximum
Burke, Williams &	Real Property – General	180192	01/19	\$100,000
Sorensen, LLP	Labor and Employment Matters	180207	04/19	\$75,000
	General Real Estate Matters	180209	08/19	\$200,000
	Rancho Cucamonga Condemnation Actions (Grade Separation Project)	207970	05/22	\$100,000
Law Office of Alexis S.M. Chiu*	Bond Counsel	200468	07/21	N/A
Cislo & Thomas LLP	Intellectual Property	170703	08/17	\$100,000
Curls Bartling P.C.*	Bond Counsel	200470	07/21	N/A
Duane Morris LLP	SWRCB Curtailment Process	138005	09/14	\$615,422
Duncan, Weinberg, Genzer & Pembroke	Power Issues	6255	09/95	\$3,175,000
Ellison, Schneider, Harris & Donlan	Colorado River Issues	69374	09/05	\$175,000
Tiams & Doman	Issues re SWRCB	84457	06/07	\$200,000
Erin Joyce Law, PC	Employment Matter	216039	11/23	\$100,000
Greines, Martin, Stein & Richland LLP	SDCWA v. MWD	207958	10/22	\$100,000
& Richand LLP	Colorado River Matters	207965	11/22	\$100,000
Haden Law Office	Real Property Matters re Agricultural Land	180194	01/19	\$50,000
Hanna, Brophy, MacLean, McAleer & Jensen, LLP	Workers' Compensation	211926	06/23	\$100,000
Hanson Bridgett LLP	SDCWA v. MWD	124103	03/12	\$1,100,000

Firm Name	Matter Name	Agreement No.	Effective Date	Contract Maximum
	Finance Advice	158024	12/16	\$100,000
	Deferred Compensation/HR	170706	10/17	\$500,000
	Tax Issues	180200	04/19	\$50,000
	Alternative Project Delivery (ADP)	207961	10/22	\$250,000
	Ad Valorem Property Taxes	216042	11/23	\$100,000
Hausman & Sosa, LLP	MOU Hearing Officer Appeal	<del>201892</del>	09/21	<del>\$95,000</del>
	MOU Hearing Officer Appeal	<del>207949</del>	<del>07/22</del>	<del>\$25,000</del>
Hawkins Delafield & Wood LLP*	Bond Counsel	193469	07/21	N/A
Hemming Morse, LLP	Baker Electric v. MWD	211933	08/23	\$100,000
Horvitz & Levy	SDCWA v. MWD	124100	02/12	\$1,250,000
	General Appellate Advice	146616	12/15	\$200,000
	Colorado River	203464	04/22	\$100,000
Innovative Legal Services, P.C.	Employment Matter	211915	01/19/23	\$125,000 \$100,000
Internet Law Center	Cybersecurity and Privacy Advice and Representation	200478	04/13/21	\$100,000
	Systems Integrated, LLC v. MWD	201875	05/17/21	\$100,000
Amira Jackmon, Attorney at Law*	Bond Counsel	200464	07/21	N/A
Jackson Lewis P.C.	Employment: Department of Labor Office of Contract Compliance	137992	02/14	\$45,000
Jones Hall, A Professional Law Corp*	Bond Counsel	200465	07/21	N/A
Kronenberger Rosenfeld, LLP	Systems Integrated, LLC v. MWD	211920	04/23	\$250,000
Kutak Rock LLP	Delta Islands Land Management	207959	10/22	\$10,000
	Labor and Employment	158032	02/17	\$229,724

Firm Name	Matter Name	Agreement No.	Effective Date	Contract Maximum
Liebert Cassidy Whitmore	FLSA Audit	180199	02/19	\$50,000
willunore	EEO Advice	<u>216041</u>	12/23	<u>\$100,000</u>
Manatt, Phelps & Phillips	SDCWA v. MWD rate litigation	146627	06/16	\$4,400,000
Fillilips	Raftelis-Subcontractor of Manatt, Agr. #146627: Per 5/2/22 Engagement Letter between Manatt and Raftelis, MWD paid Raftelis Financial Consultants, Inc.	Invoice No. 23949		\$56,376.64 for expert services & reimbursable expenses in SDCWA v. MWD
Marten Law LLP	PFAS Multi-District Litigation	216034	09/23	\$400,000
Martenson, Hasbrouck & Simon LLP	Employment Matter	211932	08/23	\$50,000
Meyers Nave Riback Silver & Wilson	Pure Water Southern California	207967	11/22	\$100,000
Sliver & Wilson	PFAS Compliance Issues	207968	11/14/22	\$100,000
Miller Barondess, LLP	SDCWA v. MWD	138006	12/14	\$600,000
Morgan, Lewis & Bockius	SDCWA v. MWD	110226	07/10	\$8,750,000
DOCKIUS	Project Labor Agreements	200476	04/21	\$100,000
Musick, Peeler & Garrett LLP	Colorado River Aqueduct Electric Cables Repair/Contractor Claims	193461	11/20	\$2,500,000
	Arvin-Edison v. Dow Chemical	203452	01/22	\$100,000
	Semitropic TCP Litigation	207954	09/22	\$75,000
Nixon Peabody LLP*	Bond Counsel [re-opened]	193473	07/21	\$100,000
	Special Finance Project	207960	10/22	\$50,000
Norton Rose Fulbright US LLP*	Bond Counsel	200466	07/21	N/A

Firm Name	Matter Name	Agreement No.	Effective Date	Contract Maximum
Olson Remcho LLP	Government Law	131968	07/14	\$400,000
	Executive Committee/Ad Hoc Committees Advice	207947	08/22	\$60,000
	Public Records Act	207950	08/22	\$54,000 \$45,000
	Advice/Assistance re Proposition 26/Election Issues	211922	05/23	\$100,000
Pearlman, Brown & Wax, L.L.P.	Workers' Compensation	216037	10/23	\$100,000
Rains Lucia Stern St. Phalle & Silver, PC	Employment Matter	211919	4/23	\$60,000
Renne Public Law Group, LLP	ACE v. MWD (PERB Case No. LA-CE-1574-M)	203466	05/22	\$100,000
	ACE v. MWD (PERB Case No. LA-CE-1611-M)	207962	10/22	\$50,000
	Employee Relations and Personnel Matters	<u>216045</u>	<u>01/24</u>	<u>\$50,000</u>
Ryan & Associates	Leasing Issues	43714	06/01	\$200,000
	Oswalt v. MWD	211925	05/23	\$100,000
Seyfarth Shaw LLP	Claim (Contract #201897)	201897	11/04/21	\$350,000
	Claim (Contract #203436)	203436	11/15/21	\$350,000
	Claim (Contract #203454)	203454	01/22	\$210,000
	Reese v. MWD	207952	11/22	\$750,000
	General Labor/Employment Advice	211917	3/23	\$100,000
	Civil Rights Department Complaint	211931	07/23	\$100,000
	Crawford v. MWD	216035	09/23	\$100,000
	<u>Tiegs v. MWD</u>	<u>216043</u>	12/23	<u>\$250,000</u>
	Zarate v. MWD	216044	01/24	<u>\$250,000</u>
	Rivers v. MWD	207946	07/22	\$250,000

Firm Name	Matter Name	Agreement No.	Effective Date	Contract Maximum
Sheppard Mullin Richter & Hampton	Lorentzen v. MWD	216036	09/23	\$100,000
Stradling Yocca Carlson & Rauth*	Bond Counsel	200471	07/21	N/A
Theodora Oringher PC	Construction Contracts - General Conditions Update	185896	07/20	\$100,000
Thompson Coburn LLP	NERC Energy Reliability Standards	193451	08/20	\$300,000
Van Ness Feldman, LLP	General Litigation	170704	07/18	\$50,000
LLP	Colorado River MSHCP	180191	01/19	\$50,000
	Bay-Delta and State Water Project Environmental Compliance	193457	10/15/20	\$50,000
	Colorado River Issues	211924	05/23	\$100,000

<sup>\*</sup>Expenditures paid by Bond Proceeds/Finance \*\*Expenditures paid by another group

### THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA

### **MINUTES**

### LEGAL AND CLAIMS COMMITTEE

### November 14, 2023

Chair Luna called the in-person and teleconference meeting to order at 8:30 a.m.

Director Peterson indicated he is participating under AB 2449 "emergency circumstances." Director Peterson appeared by audio and on camera.

Director Seckel made a motion, seconded by Director Kurtz, to approve AB 2449 emergency circumstances for Director Peterson. The vote was:

Ayes: Directors Camacho, Dick, Kurtz, Luna, Miller, and Seckel

Noes: None Abstentions: None

Absent: Directors Cordero, Douglas, Garza, McCoy, and Ramos

The motion for AB 2449 emergency circumstances for Director Peterson passed by a vote of 6 ayes, 0 noes, 0 recusals, 0 abstention, and 5 absent.

Members present: Directors Camacho, Cordero (entered after roll call), Dick, Garza (entered after roll call), Kurtz, Luna, McCoy (entered after roll call), Miller, Peterson (AB 2449 emergency circumstance), Ramos (entered after roll call), and Seckel.

Member absent: Director Douglas.

Other Directors present: Directors Ackerman, Armstrong, Bryant, Dennstedt, Erdman, Fong-Sakai, Goldberg, Gray (teleconference posted location), Morris, Ortega, Pressman (teleconference posted location), and Smith.

Committee Staff present: Beatty, Hagekhalil, Miyashiro, and Scully.

## 1. OPPORTUNITY FOR MEMBERS OF THE PUBLIC TO ADDRESS THE COMMITTEE ON MATTERS WITHIN THE COMMITTEE'S JURISDICTION

None

#### 2. MANAGEMENT ANNOUNCEMENTS AND HIGHLIGHTS

a. Subject: General Counsel's report of monthly activities

General Counsel Scully had nothing to add to her written report.

### **CONSENT CALENDAR ITEMS – ACTION**

### 3. CONSENT CALENDAR OTHER ITEMS – ACTION

**A.** Subject: Approval of the Minutes of the Legal and Claims Committee for

October 10, 2023.

#### 4. CONSENT CALENDAR ITEMS – ACTION

**7-11** Subject Approve amendments to the Metropolitan Water District

Administrative Code to conform to current law, practices, and regulations; the General Manager has determined that the proposed

action is exempt or otherwise not subject to CEQA

Motion: Approve amendments to the Metropolitan Water District

Administrative Code to conform to current law, practices, and

regulations.

7-12 Subject Approve amendments to Metropolitan Water District

Administrative Code sections 6431 and 6453 to increase the

authority of the General Counsel to obtain legal and related services to an amount not to exceed \$250,000 and the General Auditor to obtain professional services to an amount not to exceed \$100,000; the General Manager has determined that the proposed action is

exempt or otherwise not subject to CEQA

Motion: Approve amendments to Metropolitan Water District

Administrative Code sections 6431 and 6453 to increase the

authority of the General Counsel to obtain legal and related services to an amount not to exceed \$250,000 and the General Auditor to obtain professional services to an amount not to exceed \$100,000.

No presentations were given, Director Camacho made a motion, seconded by Director Dick, to approve items 3A, 7-11, and 7-12. The vote was:

Ayes: Directors Camacho, Dick, Kurtz, Luna, Miller, Peterson, and Seckel

Noes: None Abstentions: None

Absent: Directors Cordero, Douglas, Garza, McCoy, and Ramos

The motion for Items 3A, 7-11 and 7-12 passed by a vote of 7 ayes, 0 noes, 0 recusals,

0 abstention, and 5 absent.

Director Peterson announced during roll call that no one was in the room with him 18 years of age or older.

### END OF CONSENT CALENDAR ITEMS

-3-

### 5. OTHER BOARD ITEMS – ACTION

Director Dick recused himself on Item 8-4 because he owns stock in Bayer, 3M, and Cordova companies.

Director Erdman recused himself on Item 8-4 because he owns stock in Raytheon and 3M companies.

Director Miller recused himself on Item 8-4 because he owns stock in 3M company.

Board Chair Ortega recused himself on Item 8-4 because Honeywell is a source of income.

8-4 Subject: Receive report on litigation in In re: Aqueous Film Forming Foams

Products Liability Litigation, Master Docket No.:

2:18mn2873RMG, consider options, and provide direction on action in response to proposed settlements in: (1) City of Camden, et al. v. 3M Company, Civil Action No.: 2:23cv03147RMG; and

(2) City of Camden, et al. v. E.I. DuPont De Nemours and

Company (n/k/a EIDP, Inc.) et al., Civil Action No.:

2:23cv03230RMG; the General Manager has determined that the proposed action is exempt or otherwise not subject to CEQA [Conference with legal counsel – existing litigation; may be heard

in closed session pursuant to Government Code Section

54956.9(d)(1)]

Presented by: Senior Deputy General Counsel Jill Teraoka and outside counsel,

Jeff Kray of Marten Law, presented the item in both open and

closed sessions.

Director McCoy entered the meeting. Director Cordero entered the meeting.

Director Ramos entered the meeting.

Vice Chair Garza entered the meeting.

Chair Luna reported there were no reportable actions taken in closed session.

8-5 Subject: Report on litigation in Systems Integrated, LLC v. Metropolitan

Water District of Southern California, Los Angeles County

Superior Court Case No. 21STCV18292; authorize an increase in maximum amount payable under contract for legal services with Internet Law Center, Ltd. in the amount of \$150,000 for a total amount not to exceed \$250,000; and authorize an increase in maximum amount payable under contract for legal services with Kronenberger Rosenfeld, LLP in the amount of \$150,000 for a total

amount not to exceed \$250,000; the General Manager has determined that the proposed action is exempt or otherwise not subject to CEQA. [Conference with legal counsel – existing litigation; to be heard in closed session pursuant to Gov. Code

Section 54956.9(d)(1)]

Presented by: Senior Deputy General Counsel Anthony Zepeda presented the

item in closed session.

Motion: Authorize an increase in maximum amount payable under contract

> for legal services with Internet Law Center, Ltd. in the amount of \$150,000 for a total amount not to exceed \$250,000; and authorize an increase in maximum amount payable under contract for legal services with Kronenberger Rosenfeld, LLP in the amount of

\$100,000 for a total amount not to exceed \$250,000

Director Ramos left the meeting.

In open session, Director Camacho made a motion, seconded by Director Dick, to approve the contract increases.

Director Peterson announced during roll call that no one was in the room with him 18 years of age or older.

The vote in open session was:

Directors Camacho, Cordero, Dick, Garza, Kurtz, Luna, McCoy, Miller, Ayes:

Peterson, and Seckel

None Noes: Abstentions: None

Absent: **Directors Douglas and Ramos** 

The motion for Item 8-5 passed by a vote of 10 ayes, 0 noes, 0 recusals, 0 abstentions, and

2 absent.

### 6. BOARD INFORMATION ITEMS

None

### 7. COMMITTEE ITEMS

None

### 8. FOLLOW-UP ITEMS

None

### 9. FUTURE AGENDA ITEMS

None

Next meeting will be held on January 9, 2024.

Meeting adjourned at 10:01 a.m.

Miguel Luna Chair



## **Board Action**

## Board of Directors Legal and Claims Committee

2/13/2024 Board Meeting

7-10

### Subject

Approve amendments to the Metropolitan Water District Administrative Code to modify the structure and duties of various committees; the General Manager has determined that the proposed action is exempt or otherwise not subject to CEQA

### **Executive Summary**

The proposed amendments to Administrative Code Division II, Chapters 2, 4, and 7, Division IV, Chapters 3 and 5, Division V, Chapters 1 and 3, Division VI, Chapter 4, and Division VIII, Chapter 1 will update the Code to modify the structure and duties of various committees to provide clarity and conform the committee structure to the current workload and priorities of the Board.

### Proposed Action(s)/Recommendation(s) and Options

#### **Staff Recommendation: Option #1**

#### Option #1

Approve amendments to the Metropolitan Water District Administrative Code to modify the structure and duties of various committees.

Fiscal Impact: None

**Business Analysis:** Will provide clarity and conform the committee structure to the current workload and priorities of the Board.

### Option #2

Do not approve amendments to the Metropolitan Water District Administrative Code to modify the structure and duties of various committees.

Fiscal Impact: None

**Business Analysis:** Will not provide clarity or conform the committee structure to the current workload and priorities of the Board.

#### **Alternatives Considered**

Not applicable

### **Applicable Policy**

Metropolitan Water District Administrative Code Section 2451: Duties and Functions [Legal and Claims Committee]

Metropolitan Water District Administrative Code Section 11104: Delegation of Responsibilities

### Related Board Action(s)/Future Action(s)

Future changes will be based on the workload and priorities of the Board as they change.

### California Environmental Quality Act (CEQA)

### **CEQA** determination for Option #1:

The proposed action is not defined as a project under CEQA because it involves organizational, maintenance, or administrative activities; personnel-related actions; and/or general policy and procedure making that will not result in direct or indirect physical changes in the environment. (Public Resources Code Section 21065; State CEQA Guidelines Section 15378(b)(2) and (5).)

### **CEQA determination for Option #2:**

None required

### **Details and Background**

This letter sets forth recommendations by the Chair of the Board for amendments to Metropolitan's Administrative Code Division II, Chapters 2, 4, and 7, Division IV, Chapters 3 and 5, Division V, Chapters 1 and 3, Division VI, Chapter 4, and Division VIII, Chapter 1 relating to the Board committee structure and duties to provide clarity and consistency with the Board's current workload and priorities. The recommendations include the termination and creation of certain ad hoc and subcommittees, as set forth below. The proposed amendments creating new ad hoc and subcommittees are contingent upon the acceptance and implementation of the recommended changes herein.

The proposed changes include:

- Eliminate the Subcommittee on Audits and the Ad Hoc Committee on the Implementation of the State Audit, and transfer their audit-related responsibilities to a new Audit Subcommittee of the Executive Committee.
- Rename the Finance, Audit, Insurance, and Real Property Committee (FAIRP) as the Finance and Asset Management Committee. Delete the audit-related duties from FAIRP and transfer them to the new Audit Subcommittee of the Executive Committee.
- Increase the number of nonofficer members from three (3) to four (4) serving on the Executive Committee.
- Rename the Legislation, Regulatory Affairs and Communications Committee (LRAC) as the Legislation and Communications Committee. Delete regulatory matters from the duties of this committee; such matters will be addressed by the appropriate committee based upon the substance of the regulation. Further, eliminate the LRAC Subcommittee on Public Affairs Engagement.

The proposed amendments, and the amendments authorized by the Board at the January 9, 2024 meeting are set forth in **Attachment 1**, with overstrikes reflecting deletions and underlining reflecting additions. **Attachment 2** sets forth the sections as they will appear in the Administrative Code if the changes are approved.

The Administrative Code is proposed to be amended as follows:

1. Eliminate FAIRP's Subcommittee on Audits in Section 2204(3)(a) and the Board's Ad Hoc Committee on the Implementation of the State Audit. Recommend that the current audit responsibility of these two committees be transferred to the proposed Audit Subcommittee of the Executive Committee at Section 2416(f)(6). Add Section 2416(f)(6) to require the Executive Committee to form an Audit Subcommittee, whose membership will include the at-large members of the Executive Committee. This subcommittee

shall have the responsibility to monitor compliance with the recommendations of the California State Audit (April 2022) and, with respect to the duties and functions of the General Auditor, shall be responsible to:

- i. Study, advise and make recommendations with regard to:
  - a. All reports of the General Auditor and external auditors, including the audited financial statements of the District.
  - b. The Audit Department's annual business plan and biennial budget.
  - c. Requests from other committees of the Board and individual Board members for audits and review not included in the Audit Department's annual business plan.
- ii. Monitor and oversee the duties and responsibilities of the Audit Department and the external auditors as those duties and responsibilities relate to the effectiveness of the District's internal control system.
- iii. Review and approve the Audit Department business plan containing the key priorities for the coming year of the General Auditor and the Audit Department in advance of the July Board meeting.
- iv. As part of the Department Head's annual evaluation process, be responsible for engaging in periodic performance expectations discussions, including progress checks, with the General Auditor.

Make conforming, non-substantive amendments to Administrative Code Sections 2703 and 2722 transferring the General Auditor's annual and quarterly reporting functions from FAIRP to the Executive Committee. Make further conforming, non-substantive amendments to Administrative Code Division VI, Sections 6416, 6436(b), 6450(a), (b), (d), (e), and (f), 6451(b), (d), and (e), and 6453 transferring the General Auditor's duties and responsibilities outlined in Article 4 (General Auditor) from FAIRP to the Executive Committee. Rename the Vice Chair for Finance, Audit, and Planning as the Vice Chair for Finance and Planning at Section 2204(3).

- 2. Amend Administrative Code Sections 2204(3)(a), 2400, 2440, and 2441 to rename FAIRP as the Finance and Asset Management Committee. Delete Administrative Code Sections 2441(h), (i), and (j) and (o), in relevant part, and transfer these audit duties from FAIRP to the Executive Committee, and renumber the remaining provisions therein.
  - Make conforming, non-substantive amendments to rename FAIRP as the Finance and Asset Management Committee at Administrative Code Division II, Sections 2701(b), 2720(b) and (c), and 2750 and Division V, Sections 5104(b), 5107(a), and 5305 concerning reporting duties of the General Manager, Treasurer, and Controller and the biennial budget process, Division IV, Sections 4304(a), (b), (c), (d), 4305(a) and (b) and 4507(g)(3) concerning water service policies, and Division VIII, Sections 8248(b) and 8257 concerning real property reporting duties of the General Manager.
- 3. Amend Administrative Code Sections 2410 and 2411 to increase the number of nonofficer members from three (3) to four (4) serving on the Executive Committee.
- 4. Amend Administrative Code Sections 2460 and 2461 to rename LRAC as the Legislation and Communications Committee, delete references to regulations from Sections 2461(a), (b), and (c), and amend Administrative Code Section 2204(2)(a) to eliminate the LRAC Subcommittee on Public Affairs Engagement.

5. Amend Administrative Code Section 2204(1)(b) to delete the Subcommittee on Bay-Delta to conform to the Board's January 2024 approval of this action.

2/7/2024 Set Ortega, Jr. Date

Chair of the Board

- Attachment 1 The Administrative Code of The Metropolitan Water District of Southern California (with changes marked)
- Attachment 2 The Administrative Code of The Metropolitan Water District of Southern California (clean copy)

Ref# I12692328

#### **Division II**

# PROCEDURES PERTAINING TO BOARD, COMMITTEES AND DIRECTORS

# Chapter 2

#### BOARD OFFICERS

# § 2204. Delegation of Duties to Vice Chairs.

The Vice Chairs selected by the Chair shall act in the Chair's absence, failure or inability to act. The Vice Chairs shall have ex efficio membership on standing committees, subcommittees, or special committees as designated in this Code. The Chair assigns the following duties to the designated Vice Chairs as follows:

- (1) Vice Chair for Climate Action will provide guidance to the following committees:
  - (a) Engineering, Operations and Technology Committee and the Subcommittee on Pure Water Southern California and Regional Conveyance; and
  - (b) One Water and Stewardship Committee and the Subcommittee on Bay Delta and the Subcommittee on Demand Management and Conservation Programs and Priorities.
- (2) Vice Chair for Strategic Communications and Stakeholder Engagement Policy will provide guidance to the following committees:
  - (a) Legislation, Regulatory Affairs, and Communications Committee and the Subcommittee on Public Affairs Engagement.
- (3) Vice Chair for Finance, Audit, and Planning will provide guidance to the following committees:
  - (a) Finance, Audit, Insurance, and Real Property and Asset Management Committee and the Subcommittee on Audits, and the Subcommittee on Long-Term Regional Planning Processes and Business Modeling.
- (4) Vice Chair for Organizational Integrity and Accountability will provide guidance to the following committees:
  - (a) Ethics, Organization and Personnel Committee;
  - (b) Legal and Claims Committee; and
  - (c) Equity, Inclusion, and Affordability Committee.

# Chapter 4

# STANDING COMMITTEES

Article		Sec.
1	General	2400
2	Executive	2410
3	Engineering, Operations and Technology Committee	2430
4	Finance, Audit, Insurance and Real Property and Asset Management Committee	2440
5	Legal and Claims Committee	2450
6	Legislation, Regulatory Affairs and Communications Committee	2460
7	Ethics, Organization and Personnel Committee	2470
8	One Water and Stewardship Committee	2480
9	Equity, Inclusion and Affordability Committee	2490

#### **Article 1**

#### **GENERAL**

# § 2400. Identification of Standing Committees.

The Standing Committees of the Board of Directors are:

**Executive Committee** 

Engineering, Operations and Technology Committee

Finance, Audit, Insurance and Real Property and Asset Management Committee Legal and

Claims Committee

Legislation, Regulatory Affairs and Communications Committee Ethics, Organization and

Personnel Committee

One Water and Stewardship Committee

Equity, Inclusion and Affordability Committee

# **Article 2**

# **EXECUTIVE COMMITTEE**

# § 2410. Membership.

The Executive Committee shall consist of the Chair, Vice Chairs, Secretary, all past Chairs of the Board who are directors of the District, and the Chairs of the standing committees in addition to the Executive Committee, and three four (4) additional directors as nonofficer members.

# § 2411. Selection of Nonofficer Members.

The three four (4) nonofficer members of the Executive Committee shall be nominated from the floor and elected at the January meeting of the Board.

# § 2416. Duties and Functions. [Executive Committee]

- (a) The Executive Committee shall study, advise, and make recommendations with regard to:
- (1) Public information for governmental and other entities and officials, and for the citizens of California regarding matters affecting the District's interests;
- (2) Official dealings with the United States Government, the State of California or other states, member public agencies or their sub-agencies, foreign governments and other entities or persons in matters of public policy or other activities as deemed appropriate;
- (3) Policies and procedures to be considered by the Board or committees thereof, except for policy matters within the jurisdiction of a specific standing committee;
  - (4) Matters relating to the Colorado River Board of California;
- (5) Major policy issues to be considered by the Board, including proposed amendments to the Metropolitan Water District Act;
- (6) Questions raised by the officers and staff in intervals between meetings of the Board and in unexpected situations and emergencies.
- (7) The terms and conditions of employment of all consultants and advisors not within the jurisdiction of other committees;
  - (8) Resolution of conflicting committee recommendations pursuant to § Section 2314;
- (9) The progress of, and propose modifications to, the Board's goals in light of then existing and projected future conditions; and
  - (10) Such other matters as may be required by Division II of this Code.
  - (b) The Executive Committee shall:
- (1) Review and approve board and committee agendas and, notwithstanding the jurisdiction of the other standing committees in the Code, have the authority to direct which committee shall consider an item;
  - (2) Review and approve the scheduling of board and committee meetings;
- (3) Be responsible for the oversight and management of the organization including, but not limited to, the form of the District's organization and the flow of the authority and responsibility. This includes monitoring and overseeing the duties and responsibilities of management; and
- (4) Consider the effectiveness of the District's internal control system, including information technology security and control.
- (c) The Executive Committee shall retain ultimate responsibility for those duties as are specifically assigned to the subcommittees of the Executive Committee.
- (d) The Executive Committee shall be responsible for reviewing and approving the annual business plan containing the General Manager's key priorities for the coming year.

- (e) As part of the Department Head annual evaluation process, the Executive Committee shall be responsible for engaging in periodic performance expectations discussions, including progress checks, with the General Manager.
  - (f) The Executive Committee shall also:
- (1) Act on behalf of the Board in unexpected situations and emergencies, subject to subsequent approval or ratification of the actions taken whenever such approval or ratification is required by law.
- (2) Provide policy guidance where appropriate to those directors and District staff members who are associated with organizations in which the District has membership.
- (3) Conduct hearings on appeals of protest denials involving Purchasing Contracts and Professional and Technical Services Contracts pursuant to Section 8150(b).
  - (i) Hearings shall be held by the committee at its next regular meeting to be held at least 72 hours after the filing of the notice of appeal of the General Manager's determination under Section 8150(b). The decision of the committee shall be final unless the committee chooses to refer the notice of protest to the Board.
  - (ii) The Chair of the Executive Committee may re-delegate duties provided for under subparagraph (i) above to a minimum of three members of the Executive Committee who shall act in place of the committee.
- (4) Resolve disputes over inspection dates and monitor conduct of inspection trips to assure maximum effectiveness.
- (5) Address substantiated allegations of discrimination, harassment, and retaliation against directors, the General Manager, General Counsel, General Auditor, and Ethics Officer:
  - (i) The Executive Committee shall create an ad hoc subcommittee of three members and two alternates that will serve for a period of one year to address substantiated findings of violations determined as a result of Equal Employment Opportunity (EEO) investigations conducted for alleged violations of Section 2131 made against a director and for alleged violations of Section 6305 made against the General Manager, General Counsel, General Auditor, or Ethics Officer. If any director serving on the three-member subcommittee is an involved party to an EEO investigation, or has a conflict of interest with any involved party, the conflicted director shall recuse themselves from the matter. The Chief EEO Officer (EEO Officer) will select an alternate director to fulfill all subcommittee duties related to the particular matter. If the EEO Officer has a conflict of interest in the same matter, the Ethics Officer and/or General Counsel will select an alternate director. Directors serving as alternates will not participate in ad hoc committee matters unless or until called to serve.
  - (ii) The Executive Committee will also select an external law firm to serve as counsel to the ad hoc subcommittee to provide guidance, as needed, for post investigation actions.
  - (iii) The ad hoc subcommittee shall delegate to the EEO Officer the responsibility to designate an external investigator to conduct a fact-finding EEO investigation pursuant to this section. ALL EEO investigations will be fair, impartial, timely, and promptly initiated and completed by qualified personnel. Detailed EEO investigative procedures,

pursuant to this section can be found in *EEO Investigative Procedures for the Board and its Direct Reports*. The EEO Officer shall refer substantiated findings of EEO investigations to the ad hoc subcommittee to determine recommended appropriate action. At its discretion, the ad hoc subcommittee may consult with the EEO Officer, Ethics Officer, and/or General Counsel on appropriate action regarding a director or department head. The ad hoc subcommittee shall report a substantiated finding of an EEO violation by a director or department head and recommend appropriate action for the Board's consideration. Appropriate action for directors may include, but is not limited to, counseling, training, a private warning letter, public censure, temporary or permanent removal from committee assignments, or referral to the Director's appointing authority requesting appropriate action. Appropriate action for department heads may include, but is not limited to, counseling, training, performance review, or the imposition of discipline, as deemed appropriate under the circumstances.

- (iv) When the General Manager is a party to the complaint or when in the judgment of the EEO Officer that matter should be handled differently to avoid real or perceived conflicts of interest, or to avoid potential bias or threats to impartiality, the EEO Officer shall delegate to the Ethics Officer the responsibility to retain an external investigator to conduct a fact-finding EEO investigation pursuant to this section. If the Ethics Officer has a conflict of interest in the particular case, the EEO Officer would delegate to the General Counsel the responsibility to retain an external investigator to conduct a fact-finding investigation pursuant to this section. Substantiated EEO findings under this subsection shall be referred directly to the ad hoc subcommittee to determine recommended appropriate action for the Board's consideration.
- (v) A deviation of this investigation protocol by the EEO Officer may occur, in certain circumstances, with a written justification and approval of the ad hoc subcommittee responsible for EEO investigations of directors and department heads
- (vi) On a quarterly basis, the Chief EEO Officer will report to the Executive Committee EEO case statistics regarding EEO complaints filed against the Board and its direct reports.
- (6) Audit Subcommittee. The Executive Committee shall create an Audit Subcommittee, whose membership shall include the at-large members of the Executive Committee. This committee shall have the responsibility to monitor compliance with the recommendations of the California State Audit (April 2022) and, with respect to the duties and functions of the General Auditor, shall be responsible to:
  - (i) Study, advise and make recommendations with regard to:
    - a. All reports of the General Auditor and external auditors, including the audited financial statements of the District;
    - b. The Audit Department's annual business plan and biennial budget; and,
    - c. Requests from other committees of the Board and individual Board members for audits and review not included in the Audit Department's annual business plan.
  - (ii) Monitor and oversee the duties and responsibilities of the Audit Department and the external auditors as those duties and responsibilities relate to the effectiveness of the District's internal control system.

- (iii) Review and approve, in advance of the July Board meeting, the Audit

  Department annual business plan containing the key priorities of the General Auditor and the Audit Department.
- (iv) As part of the Department Head annual evaluation process, be responsible for engaging in periodic performance expectations discussions, including progress checks, with the General Auditor.

#### **Article 4**

# FINANCE<del>, AUDIT, INSURANCE AND REAL PROPERTY</del> AND ASSET MANAGEMENT COMMITTEE

# § 2440. Day of Regular Meetings.

The regular meetings of the Finance, Audit, Insurance and Real Property and Asset Management Committee shall be held on the Monday preceding regular Board meetings or on the Tuesday of regular meetings.

# § 2441. Duties and Functions.

The Finance, Audit, Insurance and Real Property and Asset Management Committee shall study, advise and make recommendations with regard to:

- (a) Preparation of budgets;
- (b) Policies and procedures related to budget development and cost containment;
- (c) Sale of bonds and borrowing and repayment of money;
- (d) Disposition and investment of funds;
- (e) Authorization of appropriations, except appropriations for capital projects;
- (f) The determination of revenues to be obtained through water transactions, including, but not limited to, sales, exchanges, and wheeling of water, water standby or availability of service charges, and the levying of taxes;
  - (g) The financial impact and requirements of policies concerning annexation;
- (h) With respect to the duties and functions of the General Auditor, this Committee shall study, advise and make recommendations with regard to:
  - i. All reports of the General Auditor and external auditors, including the audited financial statements of the District:
  - ii. The Audit Department's annual business plan and biennial budget; and
  - iii. Requests from other committees of the Board for audits and reviews not included in the Audit Department's annual business plan.
- (i) Monitoring and overseeing the duties and responsibilities of the Audit Department and the external auditors as those duties and responsibilities relate to the effectiveness of the District's internal control system. It shall review and approve the Audit Department business plan containing the key

priorities for the coming year of the General Auditor and the Audit Department in advance of the July Board meeting.

- (j) As part of the Department Head annual evaluation process, be responsible for engaging in periodic performance expectations discussions, including progress checks, with the General Auditor.
  - (hk) The financial aspects of the District's risk management program;
  - (il) Questions pertaining to insurance coverage and self-insurance;
- (jm) The selection of financial and insurance consultants and the determination of the scope of their assignments;
  - (kn) Form and contents of accounts, financial reports, and financial statements;
- (le) Proposed amendments to the Metropolitan Water District Act affecting the finance, audit, insurance and real property and asset management functions of the District;
- (mp) Goals and objectives related to financial planning for Metropolitan, including but not limited to revenues, operating expenses, reserve policies, internally funded construction, debt management, investments and capital financing strategies;
- (ne) The prices and conditions governing water transactions, including, but not limited to, sales, exchanges, and wheeling sales of water;
- (or) Costs and accounting procedures relating to the District's and other state water service contracts:
- (ps) Policies regarding water transactions, including, but not limited to, the sale, exchange, and wheeling of water for various uses;
- (qt) Policies regarding allocation of water standby or availability of service revenue requirements among member public agencies;
  - (14) Water standby or availability of service charges within the District;
- (<u>s</u>\*) Determinations by the General Manager with respect to appeals concerning charges for water service, including readiness-to-serve charges and capacity charges, and report its recommendations, as appropriate, to affirm or reverse the General Manager's determinations;
- (tw) Appeals from determinations by the General Manager to deny or qualify an application for exemption from the water standby charge, and report its recommendations, as appropriate, to affirm or reverse the General Manager's determinations;
- (<u>ux</u>) The purchase, management and disposition of personal property assets such as equipment and vehicles;
- (vy) Facility master plans, including budgeting for capital improvements and long-term facilities commitments;
- (wz) Proposed rules and proposals regarding business development opportunities for real property;
  - (xaa) Policies for the acquisition of rights-of-way;

- (ybb) The purchase, sale, and leasing of land and buildings, including the District's various office and garage space needs;
  - (zee) The incidental use of land in farming operations and otherwise;
  - (aadd) The operation and maintenance of buildings;
- (<u>bbee</u>) The development, oversight, and coordination of recreational facilities at Diamond Valley Lake and Lake Skinner;
- (ccff) The use of proceeds from the sale or disposition of surplus property related to Diamond Valley Lake and Lake Skinner for recreational purposes;
- (<u>ddgg</u>) Annexations and annexation policies including the requirements, procedures, terms and conditions for annexation.

#### **Article 6**

# THE LEGISLATION, REGULATORY AFFAIRS AND COMMUNICATIONS COMMITTEE § 2461. Duties and Functions.

The Legislation, Regulatory Affairs and Communications Committee shall study, advise and make recommendations to the Board with regard to:

- (a) Proposals of the General Manager, other committees, and board members concerning State and Federal legislation—and regulations, or amendments thereto, that may affect the District;
- (b) Recommendations for new legislation-or regulations identified by members of the Board or the General Manager;
- (c) Opportunities for members of the Board to assist in outreach activities, including efforts to inform members of the Legislature or the Congress of the District's position with regard to proposed legislation—and regulations;
- (d) The effectiveness of legislative and administrative advocacy efforts by staff and members of the Board:
- (e) The development and implementation of Directors' inspection trips, including the expectations and goals for these trips;
- (f) The development and implementation of school education programs, including the expectations and goals for these programs;
- (g) The effectiveness of Metropolitan's external affairs programs and general communications efforts directed at member agencies and the general public; and
  - (h) The selection of public information consultants and the scope of their assignments.

# Chapter 7

#### PERIODIC STAFF REPORTS TO BOARD AND COMMITTEES

#### Article 1

#### ANNUAL REPORTS

# § 2701. Treasurer's Reports

- (a) The Treasurer shall annually report to the Board on the investment policy of the District.
- (b) The Treasurer shall report to the Finance, Audit, Insurance and Real Property and Asset Management Committee:
  - (1) All payments after loss of a bond interest coupon made pursuant to Section 5104; and
  - (2) All issuances of duplicate bonds made pursuant to Section 5105.

# § 2703. General Auditor's Report

The General Auditor shall annually report to the Finance, Audit, Insurance and Real Property

Executive Committee a business plan containing the General Auditor's key priorities for the coming year for the Audit Department as required by Sections 2416 2501 and 6451.

#### Article 2

# **QUARTERLY REPORTS**

# § 2720. General Manager's Quarterly Reports.

The General Manager shall quarterly make the following reports:

- (a) To the Engineering, Operations and Technology Committee: a report on the Capital Investment Plan, including service connections approved by the General Manager pursuant to Sections 4700-4708 with the estimated cost and approximate location of each and the execution of any relocation agreements involving an amount in excess of \$100,000 under authority of Section 8122(c);
- (b) To the Finance, Audit, Insurance and Real Property and Asset Management Committee: A summary of financial performance including Comparative Statements of Operations and Comparative Balance Sheets and variances thereof from estimates;
  - (c) To the Finance, Audit, Insurance and Real Property and Asset Management Committee:
    - (1) Deeds or grants accepted during the preceding quarter;
- (2) Easements, or similar rights, granted during the preceding quarter under the authority of Section 8220, and shall also include in such report any relocation or protection agreement made in connection therewith;
- (3) All leases made during the preceding quarter under the authority of Sections 8222, 8223, 8230 and 8232;
- (4) The details of any transactions during the preceding quarter in which an improvement was disposed of in such a manner as to make the improvement available for subsequent use by a party other than the District; and

- (5) Property sold pursuant to the authority granted by Section 8240 et seq.
- (d) To the Legal and Claims Committee, jointly with the General Counsel, the exercise of any power delegated to them by Sections 6431, 6433 and 6434, and, in addition any and all other personnel-related settlements that invoke confidentiality or have any financial impact including paid and reinstated leave regardless of settlement type. The reporting on personnel-related settlements should include whether equal employment opportunity issues were implicated, whether the employee is still employed by the District, the existence and type of any financial or confidentiality terms, and whether the District has taken any corrective action in response to the alleged issues.
  - (e) To the Engineering, Operations and Technology Committee:
    - (1) The status of all information technology projects throughout the organization.
- (2) The employment of any professional and technical consultant, the extension of any professional and technical consulting agreement, on the exercise of authority under Section 8121(c) and 8122(h) and the execution of any contract authorized pursuant to Section 8122(g) during the preceding calendar quarter. The report covering the last calendar quarter of the year may be combined with and included in the annual report. Each such report shall indicate when a consultant is a former employee of the District.

# § 2722. General Auditor's Quarterly Reports.

The General Auditor shall report to the Finance, Audit, Insurance and Real Property Executive Committee the exercise of any power delegated to the General Auditor by Section 6453.

#### **Article 4**

#### MISCELLANEOUS REPORTS

#### § 2750. Escheat to District of Unclaimed Funds.

The Controller shall file a report with the Finance, <u>Audit, Insurance and Real Property and Asset Management</u> Committee after each publication made pursuant to Section 5113 regarding moneys unclaimed in the District treasury.

#### **Division IV**

#### WATER SERVICE POLICIES

#### Chapter 3

# WATER TRANSACTIONS REVENUE

# § 4304. Apportionment of Revenues and Setting of Water Rates.

- (a) Not later than at its February meeting the General Manager shall present to the Finance <u>and</u> Asset Management, Audit, Insurance and Real Property Committee of the Board:
  - (1) Determinations of the revenue requirements and cost of service analysis supporting the rates and charges required during the biennial period beginning the following July 1, as determined by the General Manager in accordance with current Board policies, and,

- (2) Recommendations of rates including, but not limited to, the System Access Rate, Water Stewardship Rate, System Power Rate, Treatment Surcharge, and the Supply Rates for the various classes of water service to become effective each January 1 of the biennial period. These recommended rates shall be the General Manager's determination, made in accordance with current Board policies, of the rates necessary to produce substantially the revenues to be derived from water transactions, including, but not limited to, sales, exchanges, and wheeling, during the biennial period beginning the following July 1.
- (b) Not later than at its February meeting, the General Manager shall also present to the Finance and Asset Management, Audit, Insurance and Real Property Committee recommendations regarding the continuation of a water standby charge or the imposition of an availability of service charge (such as the readiness-to-serve charge and capacity charge), which shall be the General Manager's determination, made in accordance with current Board policies, of the charge necessary to produce substantially the revenues to be derived from fixed revenue sources, if any, exclusive of taxes, during the biennial period beginning the following July 1 which the Finance and Asset Management, Audit, Insurance, and Real Property Committee has determined to be necessary.
- (c) Not later than its February meeting the Finance and Asset Management, Audit, Insurance and Real Property Committee shall set a time or times for, and shall thereafter hold, one or more meetings of the Finance and Asset Management, Audit, Insurance and Real Property Committee, to be held prior to its regular April meeting, at which interested parties may present their views regarding the proposed water rates and availability of service charges to said committee. The Finance and Asset Management, Audit, Insurance and Real Property Committee shall direct the General Manager to cause the publication of a notice of such public hearing to be published in newspapers of general circulation within the District's service area. Such notice shall be published not less than 10 days prior to the public hearing.
- (d) Not later than its regular April meeting the Finance and Asset Management, Audit, Insurance and Real Property Committee shall make its determination regarding the revenue requirement to be paid from water rates and the water rates to become effective each January 1 of the biennial period and shall recommend said water rates to the Board no later than the Board's regular April meeting.
- (e) Not later than its April meeting, the Board shall establish water rates for deliveries beginning each January 1 of the biennial period.
- (f) Proposals for changes in water rates to become effective at times other than on January 1 shall require adequate notice to the public and a hearing before such proposals are acted upon by the Board, unless the Board finds that an immediate change in water rates is urgent.

# § 4305. Setting of Charges to Raise Fixed Revenue.

- (a) Not later than its regular May meeting each year, the Finance <u>and Asset Management, Audit, Insurance</u>, and Real Property Committee shall make its final determination regarding the water standby charge or other fixed revenue charge, if any, for the fiscal year beginning the following July 1, and shall recommend such charge, if any, to the Board at its regular May meeting.
- (b) Not later than such May meeting, the Board shall consider and take action upon the recommendations, if any, of the Finance and Asset Management, Audit, Insurance, and Real Property Committee regarding a fixed revenue source, exclusive of taxes, to become effective the following January 1 or for the fiscal year beginning the following July 1, as determined by the Board for each fixed revenue source.

# Chapter 5

# WATER SERVICE REGULATIONS - GENERAL

# § 4507. Billing and Payment for Water Deliveries.

- (a) Timeframe for Billing and Payment. Except as noted herein below, invoices shall be mailed electronically, or, if requested by the member agency, by hardcopy via United States mail, not later than the tenth day of the month following delivery to a member public agency. Each such invoice shall indicate the date of mailing and the date on which the payment thereunder becomes delinquent and shall show the total amount of water delivered for each class of service, the charges for water sold and delivered for each class, the readiness-to-serve and capacity charges, as applicable, and the total amount due and owing, all as determined by the General Manager. Payment of the amount shown on any such invoice shall be due on the last business day of that month and shall be delinquent if not received by the Treasurer of the District before the close of crediting activity on the last business day of the first month following such date of mailing. When making any such payment the member public agency shall specify the invoice or invoices to which the payment shall be credited by the District.
- (1) For purposes of Section 4507(a), "business day" shall mean any day other than a Saturday, a Sunday, or a Holiday (as defined in Section 1106).
- (2) For purposes of Section 4507(a), "received by the Treasurer of the District" shall mean receipt either (1) in the office of the Treasurer or (2) by crediting pursuant to advance agreement with the Treasurer to the District's general demand account at the District's principal depository bank, in such form that the funds are immediately available for investment or other use or disposal by the District.
- (3) For purposes of Section 4507(a), "crediting activity" shall mean either (1) 2:00 p.m. if payment is delivered to the office of the Treasurer, or (2) the cutoff time for crediting by the District's principal depository bank of that day's transactions if payment is initiated by wire transfer, automated clearinghouse transfer, interbranch transfer, direct deposit, or by other means pursuant to advance agreement with the Treasurer.

If, under advance agreement with the Treasurer, a member agency has authorized payment of any invoice by automated clearinghouse transfer initiated by the Treasurer, the Treasurer shall initiate such transfer for processing two business days prior to the business day on which such payment shall be delinquent. Failure of such transfer shall not relieve such member agency from liability for such payment or charges in the event such payment should become delinquent, except as specifically provided under advance agreement with the Treasurer.

- (b) Full Service and Emergency Storage Program Facility. In cases where water through a particular facility is delivered during any month for full service or Emergency Storage Program Service, the bill for water delivered in such month will be prepared by applying the rates for water sold and delivered in full service to the total quantity of water delivered. If the member public agency desires to receive credit for water used in Emergency Storage Program Service, the facts concerning the quantities of water so used must be certified to the District via the District's electronic certification and billing system by an authorized user for the member public agency purchasing such water as provided for in Section 4507 (c). The amount of such credits shall be based on the difference in water rates in effect at the time the water is used.
- (c) Late Certifications. Based on available information, the District will notify a member agency for any certification that it has not received, if known, three months from the end of the month for

which the agency would normally certify. No certification received after six months following the end of any month in which such a credit is claimed will be accepted. Certifications must be received by Metropolitan before 3:30 p.m. on the third working day after the end of the month to receive credit for any preceding month on the next bill, subject to the provisions with respect to late certifications in this Section. This Section applies to all cases where a certification is required to receive a credit, whether or not specifically named in this Section, unless otherwise provided by this Code.

- (d) **Determination by General Manager as to Type of Delivery.** In the event the respective quantities of water sold and delivered in any month on order of any member public agency for use therein in any water program or contract requiring certification, are not determinable to the satisfaction of the General Manager in time for preparing regular monthly bills, then billing and payment for all water sold and delivered in such month to such member public agency shall be made at the rates prescribed for water used in full service in Section 4401(a)(1) hereof. Upon the determination by the General Manager of the correct quantities of water sold and delivered and used in any water program or contract requiring certification, any adjustment which is necessary to give effect to the applicable credit for the water used in any water program or contract requiring certification, shall be made by application of credits on subsequent purchases of water from the District by such member public agency. Such adjustments shall not be made in cases where a claim for the applicable credit is not submitted within the period provided in Section 4507(c).
- (e) **Obligation to Pay for Appropriate Class of Service.** If water has been sold and delivered at the rates prescribed for water sold in any water program or contract and appropriate certifications have been submitted for the water so used, but the water has in fact been used in full service or another class of service, the member public agency shall be obligated to pay the difference between the rates prescribed for water sold for the applicable water program or contract and the rates prescribed for the class of service actually used.
- (f) Submission of Documentation by Member Agency. With respect to water sold and delivered at the rates prescribed for water sold under water programs or contract (unless otherwise specified in an agreement with the District), original documentation supporting the use of such water as certified must be submitted no later than December 31 following the end of the fiscal year for which a certification is submitted, unless otherwise specified in an agreement with the District. If the documentation is not submitted by December 31 following the end of the fiscal year for which a certification was submitted, an agency will receive a late penalty of \$2,500. If the agency does not submit documentation by February 28/29 following the end of the fiscal year for which a certification was submitted, it shall be conclusively presumed that:
- (1) The water sold from the District was used for full service, and the District's next monthly billing shall reflect such adjustment; or
- (2) The yield was not produced as certified and the District's next monthly billing shall reflect such adjustment.

This provision will apply individually to each program or agreement that an agency or sub-agency participates in separately.

(g) **Review Process.** With respect to water sold and delivered at the rates prescribed for water sold under water programs or contract (unless otherwise specified in an agreement with the District) the District will complete its review within twelve months from date of receipt of the original supporting documentation.

- (1) Should the District not complete its review within twelve months of the submittal of all source documentation, the review will be considered complete and the certifications final.
- (2) When the review is completed, the District will notify the member agency of its initial findings for its comments. The member agency will provide its comments within 60 days. Metropolitan staff and the agency will work together to reconcile any differences.
- (3) If the member agency and Metropolitan staff cannot reconcile the differences, Metropolitan's Water System Operations' Group Manager has the responsibility to consult with the member agency and make a final ruling, subject to the General Manager's oversight. If the ruling is unsatisfactory to the agency, it can be appealed to Metropolitan's Finance and Asset Management, Audit, Insurance and Real Property Committee.
- (4) If the member agency does not provide further documentation correcting Metropolitan staff findings within the 60-day comment period as specified in (g) (2), then it shall be conclusively presumed that the District's findings are correct and the District's next monthly billing shall reflect such adjustment.
- (h) **Discovery of Mistakes or Errors.** In the event a mistake or error is discovered in a District water sales record, the General Manager shall initiate appropriate corrective action. No mistake or error made more than three years prior to its discovery shall be corrected unless otherwise specified in an agreement with the District. In the event a mistake or error is discovered by a member agency in its water sales record or certifications, no mistake or error made more than three years prior to its discovery shall be corrected unless otherwise specified in an agreement with the District.
- (1) A District water sales record shall include a water billing invoice, or district invoice for other water-related charges.
- (2) If the District finds the mistake or error, the discovery of the mistake or error shall be documented in writing to the member agency. The date of discovery for corrective action purposes shall be the date notice is sent to the member agency.
- (3) If the member agency discovers the mistake or error, the discovery of the mistake or error shall be documented in writing to the District by either a revised certification form or letter, whichever is applicable. The date of discovery for corrective action purposes shall be the date the certification or letter is received by the District.
- (4) If an incorrect invoice has been issued to a member public agency, the General Manager shall notify the affected agency of any adjustment and the manner of making any required credit or charge, neither of which shall bear interest.
- (5) Mistakes or errors shall also include but are not limited to mistakes or errors in metering or recording deliveries to member agencies, entry or calculation errors in fixed charges, discovery of errors in either a member agency or sub-agency submitted certification(s), or processing of a certification(s) for the Local Projects Program, the Local Resources Program, the Groundwater Recovery Program, Conservation Credit Program, or any other water management program or storage programs or agreements unless specified otherwise in the contract.
- (6) Any mistakes or error for a fiscal year period that is less than five acre-feet cumulative by agency or sub-agency, by program or agreement, shall be waived.

(i) Rate Change. In the event that deliveries of water are made by the District to member public agencies over a billing period during which the District's water rates change, the General Manager may cause the meters recording deliveries of water during such period to be read at the end of the period and the statement of charges for such deliveries of water may be based on a proration between the previous and new water rates for the periods of time during which each were in effect as determined by the General Manager.

#### **Division V**

#### FINANCIAL MATTERS

#### Chapter 1

#### **ADMINISTRATIVE MATTERS**

# § 5104. Payment After Loss of Bond Interest Coupon.

- (a) The Treasurer of the District is authorized to effectuate payment, without action of the Board, of a claim arising from the loss of a bond interest coupon that has been detached from a District bond or from the destruction of a bond interest coupon at any time after the date of its maturity, and the Treasurer has received:
- (1) An affidavit or affidavits establishing the ownership of the coupon and reciting therein the circumstances under which it was lost or destroyed; and
- (2) An indemnity bond in a penal sum which is at least the amount of the claim, said sum being specifically stated in said bond, said bond to be approved by the General Counsel and then filed with the Treasurer. The indemnity bond must include a rider substantially in the form hereinafter set forth:

#### (i) Rider.

This Rider is attached to and is a part of the Bond of Indemnity executed by the (enter name of insurance company) respecting the loss of coupons due coupons at \$ per coupon - total \$, detached from \$ Bonds of THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA, numbered to , %, maturing , at \$ each bond in bearer form.

It is understood and agreed that in the event the balance in the coupon account respecting the above-described issue of bonds, maintained by the Treasurer of said District, should hereafter not be sufficient as a result of the payment of coupon(s) to provide for outstanding unpaid coupons, (enter name of insurance company) will reimburse The Metropolitan Water District of Southern California, up to the face amount of the coupon(s) paid under this indemnity contingent upon presentation by said District of (a) evidence that said District has paid the afore-described coupon(s) or (b) a certificate from the Treasurer of said District that there is a deficiency in said coupon account balance.

It is understood that within the first year after the due date of the afore-described coupon(s) that (enter name of insurance company) may request that the Treasurer of said District search the District's records to ascertain if in fact the afore-described coupon(s) have been paid, but any such request shall only be honored by said Treasurer upon payment by (enter name of insurance company) of any fee required by said Treasurer to cover costs of such search.

Executed this day of  $\frac{1920}{}$ .

(enter name of insurance company)

(b) The Treasurer shall report to the Finance, Audit, Insurance and Real Property and Asset Management Committee all payments made pursuant to this Section 5104.

# § 5107. Biennial Budget Process.

- (a) There shall be prepared each even-numbered year, under the direction of the General Manager, a proposed biennial budget covering District operations for the following two fiscal years. The proposed biennial budget shall be submitted to the Board no later than the date of the regular Board meeting in June immediately preceding the first fiscal year of the biennium to which the budget applies. The proposed biennial budget shall indicate by fund all anticipated expenses and required reserves and the source of revenues to be used to meet such expenses and provide such reserves. The proposed biennial budget will at a minimum include a five-year financial forecast. At least one Board Workshop on the proposed biennial budget will be conducted prior to submission of the proposed biennial budget for Board approval. The Finance, Audit, Insurance and Real Property and Asset Management Committee shall review the proposed biennial budget in its entirety, together with the recommendations from the Board workshop, and report its recommendations to the Board.
- (b) After considering the proposed biennial budget and making any revisions thereto that it may deem advisable, the Board shall adopt the biennial budget before the beginning of the biennial period to which the budget applies. The amounts provided in the adopted budget for the biennial period for total expenses for operations and maintenance, including minimum and variable operations and maintenance charges under water or power contracts with the State, for capital charges under such contracts, and for debt service shall be deemed to be appropriated from the funds indicated in the budget.
- (c) The adoption of the budget shall have no effect upon appropriations for capital projects and continuing expenditures not susceptible to immediate direct allocation, as described in Section 5108 hereof, and shall not establish any limitations on expenditures for such purposes.
- (d) The total operations and maintenance budget shall be measured against the regional rate of inflation as measured by the five-year rolling average change in the Consumer Price Index (CPI) for the Los Angeles-Riverside-range County area, not seasonally adjusted, for all items as reported by the U.S. Bureau of Labor Statistics. The budget will include explanations of increases greater than the CPI due to unique conditions, growth or expansion of services.

# Chapter 3

#### SHORT-TERM REVENUE CERTIFICATES

# § 5305. Report of Exercise of Authority.

The General Manager shall report to the next following meeting of the Finance, Audit, Insurance and Real Property and Asset Management Committee of the Board any exercise of authority pursuant to this Chapter.

#### **Division VI**

#### PERSONNEL MATTERS

Chapter 4

# **OFFICERS**

#### Article 2

# **General Manager**

# § 6416. Annual Report to Executive Committee

The General Manager shall annually submit to the Executive Committee a business plan containing the General Manager's key priorities for the coming year. The business plan shall be submitted in conjunction with similar plans by the General Auditor to the <a href="ExecutiveFinance">ExecutiveFinance</a>, Audit, Insurance, and Real Property Committee and the Ethics Officer to the Ethics, Organization and Personnel Committee and the General Counsel to the Legal and Claims Committee.

#### **Article 3**

# **General Counsel**

# § 6436. Annual and Quarterly Reports to Legal and Claims Committee.

- (a) The General Manager and General Counsel shall report quarterly to the Legal and Claims Committee the exercise of any power delegated to them by Sections 6433 and 6434. The General Counsel shall report quarterly to the Legal and Claims Committee the exercise of any power delegated to them by Section 6431.
- (b) The General Counsel shall annually, in advance of the July Board meetings, submit to the Legal and Claims Committee a business plan containing the Legal Department's key priorities for the coming year for review and approval. The business plan shall be submitted in conjunction with similar plans by the General Manager to the Executive Committee and the General Auditor to the Finance, Audit, Insurance, and Real Property Executive Committee, and the Ethics Officer to the Ethics, Organization and Personnel Committee.

#### **Article 4**

# **GENERAL AUDITOR**

# § 6450. Powers and Duties.

- (a) The District's independent internal auditing function is governed by provisions of the California Government Code and by policies established by the Board of Directors. The Finance, Audit, Insurance and Real Property Executive Committee is responsible for the oversight of the internal auditing function, approving the Audit Department charter (subject to review and approval of the Board of Directors), selecting and overseeing the work of external auditors, and reviewing reports issued by both the internal auditors.
- (b) The General Auditor manages the District's Audit Department and is responsible for formulating departmental policies and procedures; directing and evaluating the performance of work done by employees within the department, administering the internal records of the department; and

administering the District's contract for external audit services. The General Auditor shall, annually in advance of the July Board meetings, submit to the <a href="ExecutiveFinance">ExecutiveFinance</a>, Audit, Insurance and Real Property Committee an Audit business plan containing key priorities for the coming year for review and approval. The business plan shall be submitted in conjunction with similar plans by the General Manager to the Executive Committee, the General Counsel to the Legal and Claims Committee and Ethics Officer to the Finance, Audit, Insurance and Real Property <a href="Ethics, Organization">Ethics, Organization</a>, and <a href="Personnel">Personnel</a> <a href="Committee">Committee</a>.

- (c) The General Auditor shall report the findings, opinions, and recommendations which result from the performance of the duties outlined in paragraph 6450(b) to the General Manager, General Counsel and Ethics Officer for their information and appropriate actions. Whenever an audit report contains recommendations for corrective actions or changes in current practices, the General Manager, General Counsel, Ethics Officer or their designees shall respond to the General Auditor in an appropriate manner and within a reasonable time, indicating their views on the recommendations and proposed actions to be taken, if any.
- (d) The General Auditor's reports on internal audit assignments shall be addressed to the Finance, Audit, Insurance and Real Property Executive Committee. The General Auditor shall have the discretion to determine the form and content of such audit reports, subject to guidance by the Finance, Audit, Insurance and Real Property Executive Committee. With the exception of those reports which the General Auditor deems to be urgent or confidential in nature, copies of all audit reports addressed to the Finance, Audit, Insurance and Real Property Executive Committee shall be submitted to the General Manager and General Counsel for review and comment simultaneously to their submittal to the Finance Audit, Insurance and Real Property Executive Committee.
- (e) The General Auditor shall transmit all reports issued by the District's external auditors to the Finance, Audit, Insurance and Real Property Executive Committee and any other committees of the Board as may be applicable. Such transmittal letters should include any comments on the external auditor's reports that the General Auditor deems necessary.
- (f) The General Auditor may receive requests from time to time from the other executive officers or committees of the Board to perform audit assignments which are not included in the approved annual Audit Business Plan. Similarly, the General Auditor may identify a need to include new assignments in the Audit Business Plan during the year. The General Auditor shall have sufficient latitude and discretion to include those new assignments in the annual Audit Business Plan as the General Auditor deems necessary based upon their professional judgment and available resources. Requests from other committees of the Board and individual Board members desiring specific audit assignments shall be submitted to the Subcommittee on Audits Audit Subcommittee for study, advise, and recommendation, or if such subcommittee is not currently in place, the Finance, Audit, Insurance, and Real Property Executive Committee. Once the audit assignment is approved by the Board, the General Auditor reserves the right to determine how to best fit the directed audit assignment into the Audit Business Plan. The reporting process for assignments requested by either executive management, by committees of the Board, or by individual Board members shall generally follow the process outlined in paragraphs 6450(c) or (d) previously.
- (g) The General Auditor shall manage the work of the Audit Department in accordance with the Audit Department Charter. The General Auditor shall assess annually whether the purpose, authority and responsibility, as defined in this Charter, continue to be adequate to enable the Audit Department to accomplish its objectives.

# § 6451. Audit Department Charter.

- (a) Mission and Scope of Work The mission of the Audit Department is to provide independent, professional, objective assurance and consulting services designed to add value and improve Metropolitan's operations. It helps the organization accomplish its objectives by using a proactive, systematic approach to evaluate and improve the effectiveness of risk management, control and governance processes. The scope of work of the Audit Department is to determine whether the organization's network of risk management, internal control and governance processes, as designed and represented by management, is adequate and functioning in a manner to ensure:
  - (1) Risks are appropriately identified, managed and monitored;
  - (2) Significant financial, managerial and operating information is accurate, reliable and
- (3) Employees' actions are in compliance with policies, standards, procedures and applicable laws and regulations;
  - (4) Resources are acquired economically, used efficiently, and adequately protected;
  - (5) Programs, plans, and objectives are achieved;
- (6) Quality and continuous improvement are fostered in the organization's control process;
- (7) Significant legislative or regulatory issues impacting the organization are recognized and addressed appropriately;

Opportunities for improving management internal control, efficiency and the organization's image may be identified during audits. They will be communicated to the appropriate level of management.

- (b) Accountability The General Auditor shall be accountable to the Board of Directors and the Finance, Audit, Insurance and Real Property Executive Committee to:
- (1) Advise on the adequacy and effectiveness of the organization's processes for controlling its activities and managing its risks;
- (2) Report significant issues related to the processes for controlling the activities of the organization, including potential improvements to those processes, and provide information concerning such issues through to resolution;
- (3) Coordinate with other Metropolitan control and monitoring functions (risk management, legal, finance, ethics, security and environmental);
- (c) Professional Standards The Audit Department shall govern itself by adherence to The Institute of Internal Auditors' mandatory guidance, including the Definition of Internal Auditing, the Code of Ethics, and the *International Standards for the Professional Practice of Internal Auditing (Standards)*. This mandatory guidance constitutes principles of the fundamental requirements for the professional practice of internal auditing and for evaluating the effectiveness of the Audit Department's performance. These documents constitute the operating procedures for the department and constitute an addendum to the charter.

The Institute of Internal Auditors' Practice Advisories, Practice Guides, and Position Papers shall also be adhered to as applicable. In addition, Audit Department staff shall adhere to Metropolitan's policies and procedures, the California Government Code; Government Auditing Standards (GAGAS); and the Audit Department's Policies and Procedures Manual.

- (d) Responsibilities The Audit Department shall carry out the following responsibilities:
- (1) Develop and present a flexible annual audit plan to the Finance, Audit, Insurance and Real Property Executive Committee for review and approval. This plan should be developed utilizing a risk-based methodology and should include risks or internal control concerns identified by Management or the Board of Directors;
- (2) Report periodically to the Finance, Audit, Insurance and Real Property Executive Committee and Management the status of the current year's audit plan and the sufficiency of department resources;
  - (3) Issue an opinion on internal controls over financial reporting on an annual basis;
- (4) Submit audit reports to the Finance, Audit, Insurance and Real Property Executive Committee and Management communicating the auditor's opinion regarding the internal control structure, identifying significant control issues and providing related recommendations;
- (5) Evaluate the adequacy and timeliness of Management's responses to, and the corrective action taken on, all significant control issues noted in such reports. Conduct follow-up reviews as necessary and periodically report to the Finance, Audit, Insurance and Real Property Executive

  Committee the status of Management's progress;
- (6) Ensure the selection, development and supervision of competent and professional audit staff;
- (7) Perform a quality assurance program by which the General Auditor evaluates internal auditing activities against professional standards. Obtain external quality assurance review as required by GAGAS and the Standards;
- (8) Perform consulting services to assist management in meeting its objectives. Examples may include facilitation, process design, training, and advisory services;
- (9) Evaluate additions or changes in internal control processes coincident with their development and implementation;
- (10) Keep the Finance, Audit, Insurance and Real Property Executive Committee informed of significant emerging trends and best practices in internal auditing and governance;
- (11) Assist in the investigation of significant suspected fraudulent activities within the organization. Assure reporting to the Finance, Audit, Insurance and Real Property Executive Committee on the results, as appropriate;
- (12) Coordinate with external auditors to minimize duplication of effort and to ensure that issues raised, as a result of their review, are appropriately addressed.
  - (e) Authority The General Auditor and Audit Department staff members are authorized to:
- (1) Have unrestricted access to all functions, records, property, and personnel, subject to the requirements of safekeeping, confidentiality and applicable process;
- (2) Have full and free access to the Finance, Audit, Insurance and Real Property Executive Committee, subject to applicable law,
- (3) Allocate resources, set frequencies, and select subject, determine scopes of work, and apply the techniques required to accomplish audit objectives;
- (4) Obtain the necessary assistance of personnel in units of the organization where they perform audit, as well as other specialized services from within or outside the organization;

The General Auditor and Audit Department staff members are not authorized to:

(1) Perform any operational duties for the organization;

- (2) Initiate or approve accounting transactions external to the internal auditing department;
- (3) Direct the activities of any organization employee not employed by the internal auditing department, except to the extent such employees have been appropriately assigned to auditing teams or to otherwise assist the internal auditors.

This Charter shall be reviewed at least annually by the Finance, Audit, Insurance and Real Property Executive Committee and Board of Directors.

# § 6453. Authority to Obtain Professional Services.

The General Auditor is authorized to employ the services of independent auditors or other professional or technical consultants to advise or assist them in performing their assigned duties as may be required or as they deem necessary provided that the amount to be expended in fees, costs, and expenses under any one contract in any one year shall not exceed \$100,000. External auditors, while employed as external auditors for Metropolitan, are prohibited from performing any other consulting work for Metropolitan or performing any work for other clients that conflicts, or may conflict, with their responsibilities as Metropolitan's external auditors. These prohibitions shall be included in Metropolitan's agreements with external auditors. The General Auditor shall inform the Finance, Audit, Insurance and Real Property Executive Committee whenever they exercise the authority granted under this section and they shall further report quarterly to the Legal and Claims Committee concerning any agreements entered into under this section.

#### **Division VIII**

#### CONTRACTS/DISTRICT PROPERTY

#### Chapter 1

# **CONTRACTS**

#### § 8248. Disposal of Unnecessary Improvements.

- (a) The General Manager is authorized to dispose of, in the manner the General Manager deems to be in the best interest of the District, any improvements that must be removed to make land acquired for District operations suitable for District use.
- (b) The General Manager shall report quarterly to the Finance, Audit, Insurance and Real Property and Asset Management Committee the details of any transactions during the preceding quarter in which an improvement was disposed of in such a manner as to make the improvement available for subsequent use by a party other than the District.

# § 8257. Quarterly Reports.

The General Manager shall report to the Finance, Audit, Insurance and Real Property and Asset Management Committee quarterly on any real property sold pursuant to this Article.

#### **Division II**

# PROCEDURES PERTAINING TO BOARD, COMMITTEES AND DIRECTORS

# Chapter 2

# **BOARD OFFICERS**

# § 2204. Delegation of Duties to Vice Chairs.

The Vice Chairs selected by the Chair shall act in the Chair's absence, failure or inability to act. The Vice Chairs shall have ex efficio membership on standing committees, subcommittees, or special committees as designated in this Code. The Chair assigns the following duties to the designated Vice Chairs as follows:

- (1) Vice Chair for Climate Action will provide guidance to the following committees:
  - (a) Engineering, Operations and Technology Committee and the Subcommittee on Pure Water Southern California and Regional Conveyance; and
  - (b) One Water and Stewardship Committee and the Subcommittee on Demand Management and Conservation Programs and Priorities.
- (2) Vice Chair for Strategic Communications and Stakeholder Engagement Policy will provide guidance to the following committees:
  - (a) Legislation and Communications Committee.
  - (3) Vice Chair for Finance and Planning will provide guidance to the following committees:
    - (a) Finance and Asset Management Committee and the Subcommittee on Long-Term Regional Planning Processes and Business Modeling.
- (4) Vice Chair for Organizational Integrity and Accountability will provide guidance to the following committees:
  - (a) Ethics, Organization and Personnel Committee;
  - (b) Legal and Claims Committee; and
  - (c) Equity, Inclusion, and Affordability Committee.

# Chapter 4

# STANDING COMMITTEES

Artic	le	Sec.
1	General	2400
2	Executive	2410
3	Engineering, Operations and Technology Committee	2430
4	Finance and Asset Management Committee	2440
5	Legal and Claims Committee	2450
6	Legislation and Communications Committee	2460
7	Ethics, Organization and Personnel Committee	2470
8	One Water and Stewardship Committee	2480
9	Equity, Inclusion and Affordability Committee	2490

# **Article 1**

#### **GENERAL**

# § 2400. Identification of Standing Committees.

The Standing Committees of the Board of Directors are:

**Executive Committee** 

Engineering, Operations and Technology Committee

Finance and Asset Management Committee

Legal and Claims Committee

Legislation and Communications Committee

Ethics, Organization and Personnel Committee

One Water and Stewardship Committee

Equity, Inclusion and Affordability Committee

# **Article 2**

# **EXECUTIVE COMMITTEE**

# § 2410. Membership.

The Executive Committee shall consist of the Chair, Vice Chairs, Secretary, all past Chairs of the Board who are directors of the District, and the Chairs of the standing committees in addition to the Executive Committee, and four (4) additional directors as nonofficer members.

# § 2411. Selection of Nonofficer Members.

The four (4) nonofficer members of the Executive Committee shall be nominated from the floor and elected at the January meeting of the Board.

# § 2416. Duties and Functions. [Executive Committee]

- (a) The Executive Committee shall study, advise, and make recommendations with regard to:
- (1) Public information for governmental and other entities and officials, and for the citizens of California regarding matters affecting the District's interests;
- (2) Official dealings with the United States Government, the State of California or other states, member public agencies or their sub-agencies, foreign governments and other entities or persons in matters of public policy or other activities as deemed appropriate;
- (3) Policies and procedures to be considered by the Board or committees thereof, except for policy matters within the jurisdiction of a specific standing committee;
  - (4) Matters relating to the Colorado River Board of California;
- (5) Major policy issues to be considered by the Board, including proposed amendments to the Metropolitan Water District Act;
- (6) Questions raised by the officers and staff in intervals between meetings of the Board and in unexpected situations and emergencies.
- (7) The terms and conditions of employment of all consultants and advisors not within the jurisdiction of other committees;
  - (8) Resolution of conflicting committee recommendations pursuant to Section 2314;
- (9) The progress of, and propose modifications to, the Board's goals in light of then existing and projected future conditions; and
  - (10) Such other matters as may be required by Division II of this Code.
  - (b) The Executive Committee shall:
- (1) Review and approve board and committee agendas and, notwithstanding the jurisdiction of the other standing committees in the Code, have the authority to direct which committee shall consider an item;
  - (2) Review and approve the scheduling of board and committee meetings;
- (3) Be responsible for the oversight and management of the organization including, but not limited to, the form of the District's organization and the flow of the authority and responsibility. This includes monitoring and overseeing the duties and responsibilities of management; and
- (4) Consider the effectiveness of the District's internal control system, including information technology security and control.
- (c) The Executive Committee shall retain ultimate responsibility for those duties as are specifically assigned to the subcommittees of the Executive Committee.
- (d) The Executive Committee shall be responsible for reviewing and approving the annual business plan containing the General Manager's key priorities for the coming year.

- (e) As part of the Department Head annual evaluation process, the Executive Committee shall be responsible for engaging in periodic performance expectations discussions, including progress checks, with the General Manager.
  - (f) The Executive Committee shall also:
- (1) Act on behalf of the Board in unexpected situations and emergencies, subject to subsequent approval or ratification of the actions taken whenever such approval or ratification is required by law.
- (2) Provide policy guidance where appropriate to those directors and District staff members who are associated with organizations in which the District has membership.
- (3) Conduct hearings on appeals of protest denials involving Purchasing Contracts and Professional and Technical Services Contracts pursuant to Section 8150(b).
  - (i) Hearings shall be held by the committee at its next regular meeting to be held at least 72 hours after the filing of the notice of appeal of the General Manager's determination under Section 8150(b). The decision of the committee shall be final unless the committee chooses to refer the notice of protest to the Board.
  - (ii) The Chair of the Executive Committee may re-delegate duties provided for under subparagraph (i) above to a minimum of three members of the Executive Committee who shall act in place of the committee.
- (4) Resolve disputes over inspection dates and monitor conduct of inspection trips to assure maximum effectiveness.
- (5) Address substantiated allegations of discrimination, harassment, and retaliation against directors, the General Manager, General Counsel, General Auditor, and Ethics Officer:
  - (i) The Executive Committee shall create an ad hoc subcommittee of three members and two alternates that will serve for a period of one year to address substantiated findings of violations determined as a result of Equal Employment Opportunity (EEO) investigations conducted for alleged violations of Section 2131 made against a director and for alleged violations of Section 6305 made against the General Manager, General Counsel, General Auditor, or Ethics Officer. If any director serving on the three-member subcommittee is an involved party to an EEO investigation, or has a conflict of interest with any involved party, the conflicted director shall recuse themselves from the matter. The Chief EEO Officer (EEO Officer) will select an alternate director to fulfill all subcommittee duties related to the particular matter. If the EEO Officer has a conflict of interest in the same matter, the Ethics Officer and/or General Counsel will select an alternate director. Directors serving as alternates will not participate in ad hoc committee matters unless or until called to serve.
  - (ii) The Executive Committee will also select an external law firm to serve as counsel to the ad hoc subcommittee to provide guidance, as needed, for post investigation actions.
  - (iii) The ad hoc subcommittee shall delegate to the EEO Officer the responsibility to designate an external investigator to conduct a fact-finding EEO investigation pursuant to this section. ALL EEO investigations will be fair, impartial, timely, and promptly initiated and completed by qualified personnel. Detailed EEO investigative procedures,

pursuant to this section can be found in *EEO Investigative Procedures for the Board and its Direct Reports*. The EEO Officer shall refer substantiated findings of EEO investigations to the ad hoc subcommittee to determine recommended appropriate action. At its discretion, the ad hoc subcommittee may consult with the EEO Officer, Ethics Officer, and/or General Counsel on appropriate action regarding a director or department head. The ad hoc subcommittee shall report a substantiated finding of an EEO violation by a director or department head and recommend appropriate action for the Board's consideration. Appropriate action for directors may include, but is not limited to, counseling, training, a private warning letter, public censure, temporary or permanent removal from committee assignments, or referral to the Director's appointing authority requesting appropriate action. Appropriate action for department heads may include, but is not limited to, counseling, training, performance review, or the imposition of discipline, as deemed appropriate under the circumstances.

- (iv) When the General Manager is a party to the complaint or when in the judgment of the EEO Officer that matter should be handled differently to avoid real or perceived conflicts of interest, or to avoid potential bias or threats to impartiality, the EEO Officer shall delegate to the Ethics Officer the responsibility to retain an external investigator to conduct a fact-finding EEO investigation pursuant to this section. If the Ethics Officer has a conflict of interest in the particular case, the EEO Officer would delegate to the General Counsel the responsibility to retain an external investigator to conduct a fact-finding investigation pursuant to this section. Substantiated EEO findings under this subsection shall be referred directly to the ad hoc subcommittee to determine recommended appropriate action for the Board's consideration.
- (v) A deviation of this investigation protocol by the EEO Officer may occur, in certain circumstances, with a written justification and approval of the ad hoc subcommittee responsible for EEO investigations of directors and department heads
- (vi) On a quarterly basis, the Chief EEO Officer will report to the Executive Committee EEO case statistics regarding EEO complaints filed against the Board and its direct reports.
- (6) Audit Subcommittee. The Executive Committee shall create an Audit Subcommittee, whose membership shall include the at-large members of the Executive Committee. This committee shall have the responsibility to monitor compliance with the recommendations of the California State Audit (April 2022) and, with respect to the duties and functions of the General Auditor, shall be responsible to:
  - (i) Study, advise and make recommendations with regard to:
    - a. All reports of the General Auditor and external auditors, including the audited financial statements of the District;
    - b. The Audit Department's annual business plan and biennial budget; and,
    - c. Requests from other committees of the Board and individual Board members for audits and review not included in the Audit Department's annual business plan.
  - (ii) Monitor and oversee the duties and responsibilities of the Audit Department and the external auditors as those duties and responsibilities relate to the effectiveness of the District's internal control system.

- (iii) Review and approve, in advance of the July Board meeting, the Audit Department annual business plan containing the key priorities of the General Auditor and the Audit Department.
- (iv) As part of the Department Head annual evaluation process, be responsible for engaging in periodic performance expectations discussions, including progress checks, with the General Auditor.

#### Article 4

#### FINANCE AND ASSET MANAGEMENT COMMITTEE

# § 2440. Day of Regular Meetings.

The regular meetings of the Finance and Asset Management Committee shall be held on the Monday preceding regular Board meetings or on the Tuesday of regular meetings.

# § 2441. Duties and Functions.

The Finance and Asset Management Committee shall study, advise and make recommendations with regard to:

- (a) Preparation of budgets;
- (b) Policies and procedures related to budget development and cost containment;
- (c) Sale of bonds and borrowing and repayment of money;
- (d) Disposition and investment of funds;
- (e) Authorization of appropriations, except appropriations for capital projects;
- (f) The determination of revenues to be obtained through water transactions, including, but not limited to, sales, exchanges, and wheeling of water, water standby or availability of service charges, and the levying of taxes;
  - (g) The financial impact and requirements of policies concerning annexation;
  - (h) The financial aspects of the District's risk management program;
  - (i) Questions pertaining to insurance coverage and self-insurance;
- (j) The selection of financial and insurance consultants and the determination of the scope of their assignments;
  - (k) Form and contents of accounts, financial reports, and financial statements;
- (l) Proposed amendments to the Metropolitan Water District Act affecting the finance and asset management functions of the District;
- (m) Goals and objectives related to financial planning for Metropolitan, including but not limited to revenues, operating expenses, reserve policies, internally funded construction, debt management, investments and capital financing strategies;
- (n) The prices and conditions governing water transactions, including, but not limited to, sales, exchanges, and wheeling sales of water;

- (o) Costs and accounting procedures relating to the District's and other state water service contracts;
- (p) Policies regarding water transactions, including, but not limited to, the sale, exchange, and wheeling of water for various uses;
- (q) Policies regarding allocation of water standby or availability of service revenue requirements among member public agencies;
  - (r) Water standby or availability of service charges within the District;
- (s) Determinations by the General Manager with respect to appeals concerning charges for water service, including readiness-to-serve charges and capacity charges, and report its recommendations, as appropriate, to affirm or reverse the General Manager's determinations;
- (t) Appeals from determinations by the General Manager to deny or qualify an application for exemption from the water standby charge, and report its recommendations, as appropriate, to affirm or reverse the General Manager's determinations;
- (u) The purchase, management and disposition of personal property assets such as equipment and vehicles;
- (v) Facility master plans, including budgeting for capital improvements and long-term facilities commitments;
- (w) Proposed rules and proposals regarding business development opportunities for real property;
  - (x) Policies for the acquisition of rights-of-way;
- (y) The purchase, sale, and leasing of land and buildings, including the District's various office and garage space needs;
  - (z) The incidental use of land in farming operations and otherwise;
  - (aa) The operation and maintenance of buildings;
- (bb) The development, oversight, and coordination of recreational facilities at Diamond Valley Lake and Lake Skinner;
- (cc) The use of proceeds from the sale or disposition of surplus property related to Diamond Valley Lake and Lake Skinner for recreational purposes;
- (dd) Annexations and annexation policies including the requirements, procedures, terms and conditions for annexation.

#### Article 6

# THE LEGISLATION AND COMMUNICATIONS COMMITTEE

# § 2461. Duties and Functions.

The Legislation and Communications Committee shall study, advise and make recommendations to the Board with regard to:

- (a) Proposals of the General Manager, other committees, and board members concerning State and Federal legislation or amendments thereto, that may affect the District;
- (b) Recommendations for new legislation identified by members of the Board or the General Manager;
- (c) Opportunities for members of the Board to assist in outreach activities, including efforts to inform members of the Legislature or the Congress of the District's position with regard to proposed legislation;
- (d) The effectiveness of legislative and administrative advocacy efforts by staff and members of the Board;
- (e) The development and implementation of Directors' inspection trips, including the expectations and goals for these trips;
- (f) The development and implementation of school education programs, including the expectations and goals for these programs;
- (g) The effectiveness of Metropolitan's external affairs programs and general communications efforts directed at member agencies and the general public; and
  - (h) The selection of public information consultants and the scope of their assignments.

# Chapter 7

#### PERIODIC STAFF REPORTS TO BOARD AND COMMITTEES

#### Article 1

# **ANNUAL REPORTS**

# § 2701. Treasurer's Reports

- (a) The Treasurer shall annually report to the Board on the investment policy of the District.
- (b) The Treasurer shall report to the Finance and Asset Management Committee:
  - (1) All payments after loss of a bond interest coupon made pursuant to Section 5104; and
  - (2) All issuances of duplicate bonds made pursuant to Section 5105.

# § 2703. General Auditor's Report

The General Auditor shall annually report to the Executive Committee a business plan containing the General Auditor's key priorities for the coming year for the Audit Department as required by Sections 2416 and 6451.

#### Article 2

# **QUARTERLY REPORTS**

# § 2720. General Manager's Quarterly Reports.

The General Manager shall quarterly make the following reports:

- (a) To the Engineering, Operations and Technology Committee: a report on the Capital Investment Plan, including service connections approved by the General Manager pursuant to Sections 4700-4708 with the estimated cost and approximate location of each and the execution of any relocation agreements involving an amount in excess of \$100,000 under authority of Section 8122(c);
- (b) To the Finance and Asset Management Committee: A summary of financial performance including Comparative Statements of Operations and Comparative Balance Sheets and variances thereof from estimates;
  - (c) To the Finance and Asset Management Committee:
    - (1) Deeds or grants accepted during the preceding quarter;
- (2) Easements, or similar rights, granted during the preceding quarter under the authority of Section 8220, and shall also include in such report any relocation or protection agreement made in connection therewith;
- (3) All leases made during the preceding quarter under the authority of Sections 8222, 8223, 8230 and 8232;
- (4) The details of any transactions during the preceding quarter in which an improvement was disposed of in such a manner as to make the improvement available for subsequent use by a party other than the District; and
  - (5) Property sold pursuant to the authority granted by Section 8240 et seq.
- (d) To the Legal and Claims Committee, jointly with the General Counsel, the exercise of any power delegated to them by Sections 6431, 6433 and 6434, and, in addition any and all other personnel-related settlements that invoke confidentiality or have any financial impact including paid and reinstated leave regardless of settlement type. The reporting on personnel-related settlements should include whether equal employment opportunity issues were implicated, whether the employee is still employed by the District, the existence and type of any financial or confidentiality terms, and whether the District has taken any corrective action in response to the alleged issues.
  - (e) To the Engineering, Operations and Technology Committee:
    - (1) The status of all information technology projects throughout the organization.
- (2) The employment of any professional and technical consultant, the extension of any professional and technical consulting agreement, on the exercise of authority under Section 8121(c) and 8122(h) and the execution of any contract authorized pursuant to Section 8122(g) during the preceding calendar quarter. The report covering the last calendar quarter of the year may be combined with and included in the annual report. Each such report shall indicate when a consultant is a former employee of the District.

# § 2722. General Auditor's Quarterly Reports.

The General Auditor shall report to the Executive Committee the exercise of any power delegated to the General Auditor by Section 6453.

#### **Article 4**

#### MISCELLANEOUS REPORTS

# § 2750. Escheat to District of Unclaimed Funds.

The Controller shall file a report with the Finance and Asset Management Committee after each publication made pursuant to Section 5113 regarding moneys unclaimed in the District treasury.

#### **Division IV**

#### WATER SERVICE POLICIES

# Chapter 3

# WATER TRANSACTIONS REVENUE

# § 4304. Apportionment of Revenues and Setting of Water Rates.

- (a) Not later than at its February meeting the General Manager shall present to the Finance and Asset Management Committee of the Board:
  - (1) Determinations of the revenue requirements and cost of service analysis supporting the rates and charges required during the biennial period beginning the following July 1, as determined by the General Manager in accordance with current Board policies, and,
  - (2) Recommendations of rates including, but not limited to, the System Access Rate, Water Stewardship Rate, System Power Rate, Treatment Surcharge, and the Supply Rates for the various classes of water service to become effective each January 1 of the biennial period. These recommended rates shall be the General Manager's determination, made in accordance with current Board policies, of the rates necessary to produce substantially the revenues to be derived from water transactions, including, but not limited to, sales, exchanges, and wheeling, during the biennial period beginning the following July 1.
- (b) Not later than at its February meeting, the General Manager shall also present to the Finance and Asset Management Committee recommendations regarding the continuation of a water standby charge or the imposition of an availability of service charge (such as the readiness-to-serve charge and capacity charge), which shall be the General Manager's determination, made in accordance with current Board policies, of the charge necessary to produce substantially the revenues to be derived from fixed revenue sources, if any, exclusive of taxes, during the biennial period beginning the following July 1 which the Finance and Asset Management Committee has determined to be necessary.
- (c) Not later than its February meeting the Finance and Asset Management Committee shall set a time or times for, and shall thereafter hold, one or more meetings of the Finance and Asset Management Committee, to be held prior to its regular April meeting, at which interested parties may present their views regarding the proposed water rates and availability of service charges to said committee. The Finance and Asset Management Committee shall direct the General Manager to cause the publication of a notice of such public hearing to be published in newspapers of general circulation within the District's service area. Such notice shall be published not less than 10 days prior to the public hearing.
- (d) Not later than its regular April meeting the Finance and Asset Management Committee shall make its determination regarding the revenue requirement to be paid from water rates and the water rates

to become effective each January 1 of the biennial period and shall recommend said water rates to the Board no later than the Board's regular April meeting.

- (e) Not later than its April meeting, the Board shall establish water rates for deliveries beginning each January 1 of the biennial period.
- (f) Proposals for changes in water rates to become effective at times other than on January 1 shall require adequate notice to the public and a hearing before such proposals are acted upon by the Board, unless the Board finds that an immediate change in water rates is urgent.

# § 4305. Setting of Charges to Raise Fixed Revenue.

- (a) Not later than its regular May meeting each year, the Finance and Asset Management Committee shall make its final determination regarding the water standby charge or other fixed revenue charge, if any, for the fiscal year beginning the following July 1, and shall recommend such charge, if any, to the Board at its regular May meeting.
- (b) Not later than such May meeting, the Board shall consider and take action upon the recommendations, if any, of the Finance and Asset Management Committee regarding a fixed revenue source, exclusive of taxes, to become effective the following January 1 or for the fiscal year beginning the following July 1, as determined by the Board for each fixed revenue source.

# Chapter 5

#### WATER SERVICE REGULATIONS – GENERAL

# § 4507. Billing and Payment for Water Deliveries.

- (a) Timeframe for Billing and Payment. Except as noted herein below, invoices shall be mailed electronically, or, if requested by the member agency, by hardcopy via United States mail, not later than the tenth day of the month following delivery to a member public agency. Each such invoice shall indicate the date of mailing and the date on which the payment thereunder becomes delinquent and shall show the total amount of water delivered for each class of service, the charges for water sold and delivered for each class, the readiness-to-serve and capacity charges, as applicable, and the total amount due and owing, all as determined by the General Manager. Payment of the amount shown on any such invoice shall be due on the last business day of that month and shall be delinquent if not received by the Treasurer of the District before the close of crediting activity on the last business day of the first month following such date of mailing. When making any such payment the member public agency shall specify the invoice or invoices to which the payment shall be credited by the District.
- (1) For purposes of Section 4507(a), "business day" shall mean any day other than a Saturday, a Sunday, or a Holiday (as defined in Section 1106).
- (2) For purposes of Section 4507(a), "received by the Treasurer of the District" shall mean receipt either (1) in the office of the Treasurer or (2) by crediting pursuant to advance agreement with the Treasurer to the District's general demand account at the District's principal depository bank, in such form that the funds are immediately available for investment or other use or disposal by the District.
- (3) For purposes of Section 4507(a), "crediting activity" shall mean either (1) 2:00 p.m. if payment is delivered to the office of the Treasurer, or (2) the cutoff time for crediting by the District's principal depository bank of that day's transactions if payment is initiated by wire transfer, automated

clearinghouse transfer, interbranch transfer, direct deposit, or by other means pursuant to advance agreement with the Treasurer.

If, under advance agreement with the Treasurer, a member agency has authorized payment of any invoice by automated clearinghouse transfer initiated by the Treasurer, the Treasurer shall initiate such transfer for processing two business days prior to the business day on which such payment shall be delinquent. Failure of such transfer shall not relieve such member agency from liability for such payment or charges in the event such payment should become delinquent, except as specifically provided under advance agreement with the Treasurer.

- (b) Full Service and Emergency Storage Program Facility. In cases where water through a particular facility is delivered during any month for full service or Emergency Storage Program Service, the bill for water delivered in such month will be prepared by applying the rates for water sold and delivered in full service to the total quantity of water delivered. If the member public agency desires to receive credit for water used in Emergency Storage Program Service, the facts concerning the quantities of water so used must be certified to the District via the District's electronic certification and billing system by an authorized user for the member public agency purchasing such water as provided for in Section 4507 (c). The amount of such credits shall be based on the difference in water rates in effect at the time the water is used.
- (c) Late Certifications. Based on available information, the District will notify a member agency for any certification that it has not received, if known, three months from the end of the month for which the agency would normally certify. No certification received after six months following the end of any month in which such a credit is claimed will be accepted. Certifications must be received by Metropolitan before 3:30 p.m. on the third working day after the end of the month to receive credit for any preceding month on the next bill, subject to the provisions with respect to late certifications in this Section. This Section applies to all cases where a certification is required to receive a credit, whether or not specifically named in this Section, unless otherwise provided by this Code.
- (d) **Determination by General Manager as to Type of Delivery.** In the event the respective quantities of water sold and delivered in any month on order of any member public agency for use therein in any water program or contract requiring certification, are not determinable to the satisfaction of the General Manager in time for preparing regular monthly bills, then billing and payment for all water sold and delivered in such month to such member public agency shall be made at the rates prescribed for water used in full service in Section 4401(a)(1) hereof. Upon the determination by the General Manager of the correct quantities of water sold and delivered and used in any water program or contract requiring certification, any adjustment which is necessary to give effect to the applicable credit for the water used in any water program or contract requiring certification, shall be made by application of credits on subsequent purchases of water from the District by such member public agency. Such adjustments shall not be made in cases where a claim for the applicable credit is not submitted within the period provided in Section 4507(c).
- (e) **Obligation to Pay for Appropriate Class of Service.** If water has been sold and delivered at the rates prescribed for water sold in any water program or contract and appropriate certifications have been submitted for the water so used, but the water has in fact been used in full service or another class of service, the member public agency shall be obligated to pay the difference between the rates prescribed for water sold for the applicable water program or contract and the rates prescribed for the class of service actually used.

- (f) Submission of Documentation by Member Agency. With respect to water sold and delivered at the rates prescribed for water sold under water programs or contract (unless otherwise specified in an agreement with the District), original documentation supporting the use of such water as certified must be submitted no later than December 31 following the end of the fiscal year for which a certification is submitted, unless otherwise specified in an agreement with the District. If the documentation is not submitted by December 31 following the end of the fiscal year for which a certification was submitted, an agency will receive a late penalty of \$2,500. If the agency does not submit documentation by February 28/29 following the end of the fiscal year for which a certification was submitted, it shall be conclusively presumed that:
- (1) The water sold from the District was used for full service, and the District's next monthly billing shall reflect such adjustment; or
- (2) The yield was not produced as certified and the District's next monthly billing shall reflect such adjustment.

This provision will apply individually to each program or agreement that an agency or sub-agency participates in separately.

- (g) **Review Process.** With respect to water sold and delivered at the rates prescribed for water sold under water programs or contract (unless otherwise specified in an agreement with the District) the District will complete its review within twelve months from date of receipt of the original supporting documentation.
- (1) Should the District not complete its review within twelve months of the submittal of all source documentation, the review will be considered complete and the certifications final.
- (2) When the review is completed, the District will notify the member agency of its initial findings for its comments. The member agency will provide its comments within 60 days. Metropolitan staff and the agency will work together to reconcile any differences.
- (3) If the member agency and Metropolitan staff cannot reconcile the differences, Metropolitan's Water System Operations' Group Manager has the responsibility to consult with the member agency and make a final ruling, subject to the General Manager's oversight. If the ruling is unsatisfactory to the agency, it can be appealed to Metropolitan's Finance and Asset Management Committee.
- (4) If the member agency does not provide further documentation correcting Metropolitan staff findings within the 60-day comment period as specified in (g) (2), then it shall be conclusively presumed that the District's findings are correct and the District's next monthly billing shall reflect such adjustment.
- (h) **Discovery of Mistakes or Errors.** In the event a mistake or error is discovered in a District water sales record, the General Manager shall initiate appropriate corrective action. No mistake or error made more than three years prior to its discovery shall be corrected unless otherwise specified in an agreement with the District. In the event a mistake or error is discovered by a member agency in its water sales record or certifications, no mistake or error made more than three years prior to its discovery shall be corrected unless otherwise specified in an agreement with the District.
- (1) A District water sales record shall include a water billing invoice, or district invoice for other water-related charges.

- (2) If the District finds the mistake or error, the discovery of the mistake or error shall be documented in writing to the member agency. The date of discovery for corrective action purposes shall be the date notice is sent to the member agency.
- (3) If the member agency discovers the mistake or error, the discovery of the mistake or error shall be documented in writing to the District by either a revised certification form or letter, whichever is applicable. The date of discovery for corrective action purposes shall be the date the certification or letter is received by the District.
- (4) If an incorrect invoice has been issued to a member public agency, the General Manager shall notify the affected agency of any adjustment and the manner of making any required credit or charge, neither of which shall bear interest.
- (5) Mistakes or errors shall also include but are not limited to mistakes or errors in metering or recording deliveries to member agencies, entry or calculation errors in fixed charges, discovery of errors in either a member agency or sub-agency submitted certification(s), or processing of a certification(s) for the Local Projects Program, the Local Resources Program, the Groundwater Recovery Program, Conservation Credit Program, or any other water management program or storage programs or agreements unless specified otherwise in the contract.
- (6) Any mistakes or error for a fiscal year period that is less than five acre-feet cumulative by agency or sub-agency, by program or agreement, shall be waived.
- (i) Rate Change. In the event that deliveries of water are made by the District to member public agencies over a billing period during which the District's water rates change, the General Manager may cause the meters recording deliveries of water during such period to be read at the end of the period and the statement of charges for such deliveries of water may be based on a proration between the previous and new water rates for the periods of time during which each were in effect as determined by the General Manager.

# **Division V**

# FINANCIAL MATTERS

# Chapter 1

# **ADMINISTRATIVE MATTERS**

# § 5104. Payment After Loss of Bond Interest Coupon.

- (a) The Treasurer of the District is authorized to effectuate payment, without action of the Board, of a claim arising from the loss of a bond interest coupon that has been detached from a District bond or from the destruction of a bond interest coupon at any time after the date of its maturity, and the Treasurer has received:
- (1) An affidavit or affidavits establishing the ownership of the coupon and reciting therein the circumstances under which it was lost or destroyed; and
- (2) An indemnity bond in a penal sum which is at least the amount of the claim, said sum being specifically stated in said bond, said bond to be approved by the General Counsel and then filed with the Treasurer. The indemnity bond must include a rider substantially in the form hereinafter set forth:

(i) Rider.

This Rider is attached to and is a part of the Bond of Indemnity executed by the (enter name of insurance company) respecting the loss of coupons due coupons at \$ per coupon - total \$, detached from \$ Bonds of THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA, numbered to , %, maturing , at \$ each bond in bearer form.

It is understood and agreed that in the event the balance in the coupon account respecting the above-described issue of bonds, maintained by the Treasurer of said District, should hereafter not be sufficient as a result of the payment of coupon(s) to provide for outstanding unpaid coupons, (enter name of insurance company) will reimburse The Metropolitan Water District of Southern California, up to the face amount of the coupon(s) paid under this indemnity contingent upon presentation by said District of (a) evidence that said District has paid the afore-described coupon(s) or (b) a certificate from the Treasurer of said District that there is a deficiency in said coupon account balance.

It is understood that within the first year after the due date of the afore-described coupon(s) that (enter name of insurance company) may request that the Treasurer of said District search the District's records to ascertain if in fact the afore-described coupon(s) have been paid, but any such request shall only be honored by said Treasurer upon payment by (enter name of insurance company) of any fee required by said Treasurer to cover costs of such search.

Executed this day of , 20.

(enter name of insurance company)

(b) The Treasurer shall report to the Finance and Asset Management Committee all payments made pursuant to this Section 5104.

# § 5107. Biennial Budget Process.

- (a) There shall be prepared each even-numbered year, under the direction of the General Manager, a proposed biennial budget covering District operations for the following two fiscal years. The proposed biennial budget shall be submitted to the Board no later than the date of the regular Board meeting in June immediately preceding the first fiscal year of the biennium to which the budget applies. The proposed biennial budget shall indicate by fund all anticipated expenses and required reserves and the source of revenues to be used to meet such expenses and provide such reserves. The proposed biennial budget will at a minimum include a five-year financial forecast. At least one Board Workshop on the proposed biennial budget will be conducted prior to submission of the proposed biennial budget for Board approval. The Finance and Asset Management Committee shall review the proposed biennial budget in its entirety, together with the recommendations from the Board workshop, and report its recommendations to the Board.
- (b) After considering the proposed biennial budget and making any revisions thereto that it may deem advisable, the Board shall adopt the biennial budget before the beginning of the biennial period to which the budget applies. The amounts provided in the adopted budget for the biennial period for total expenses for operations and maintenance, including minimum and variable operations and maintenance charges under water or power contracts with the State, for capital charges under such contracts, and for debt service shall be deemed to be appropriated from the funds indicated in the budget.

- (c) The adoption of the budget shall have no effect upon appropriations for capital projects and continuing expenditures not susceptible to immediate direct allocation, as described in Section 5108 hereof, and shall not establish any limitations on expenditures for such purposes.
- (d) The total operations and maintenance budget shall be measured against the regional rate of inflation as measured by the five-year rolling average change in the Consumer Price Index (CPI) for the Los Angeles-Riverside-range County area, not seasonally adjusted, for all items as reported by the U.S. Bureau of Labor Statistics. The budget will include explanations of increases greater than the CPI due to unique conditions, growth or expansion of services.

#### Chapter 3

#### **SHORT-TERM REVENUE CERTIFICATES**

#### § 5305. Report of Exercise of Authority.

The General Manager shall report to the next following meeting of the Finance and Asset Management Committee of the Board any exercise of authority pursuant to this Chapter.

#### **Division VI**

#### PERSONNEL MATTERS

**Chapter 4** 

#### **OFFICERS**

#### Article 2

#### **General Manager**

#### § 6416. Annual Report to Executive Committee

The General Manager shall annually submit to the Executive Committee a business plan containing the General Manager's key priorities for the coming year. The business plan shall be submitted in conjunction with similar plans by the General Auditor to the Executive Committee and the Ethics Officer to the Ethics, Organization and Personnel Committee and the General Counsel to the Legal and Claims Committee.

#### Article 3

#### **General Counsel**

#### § 6436. Annual and Quarterly Reports to Legal and Claims Committee.

- (a) The General Manager and General Counsel shall report quarterly to the Legal and Claims Committee the exercise of any power delegated to them by Sections 6433 and 6434. The General Counsel shall report quarterly to the Legal and Claims Committee the exercise of any power delegated to them by Section 6431.
- (b) The General Counsel shall annually, in advance of the July Board meetings, submit to the Legal and Claims Committee a business plan containing the Legal Department's key priorities for the coming year for review and approval. The business plan shall be submitted in conjunction with similar

plans by the General Manager to the Executive Committee and the General Auditor to the Executive Committee, and the Ethics Officer to the Ethics, Organization and Personnel Committee.

#### **Article 4**

#### **GENERAL AUDITOR**

#### § 6450. Powers and Duties.

- (a) The District's independent internal auditing function is governed by provisions of the California Government Code and by policies established by the Board of Directors. The Executive Committee is responsible for the oversight of the internal auditing function, approving the Audit Department charter (subject to review and approval of the Board of Directors), selecting and overseeing the work of external auditors, and reviewing reports issued by both the internal and external auditors.
- (b) The General Auditor manages the District's Audit Department and is responsible for formulating departmental policies and procedures; directing and evaluating the performance of work done by employees within the department, administering the internal records of the department; and administering the District's contract for external audit services. The General Auditor shall, annually in advance of the July Board meetings, submit to the Executive Committee an Audit business plan containing key priorities for the coming year for review and approval. The business plan shall be submitted in conjunction with similar plans by the General Manager to the Executive Committee, the General Counsel to the Legal and Claims Committee and Ethics Officer to the Ethics, Organization, and Personnel Committee.
- (c) The General Auditor shall report the findings, opinions, and recommendations which result from the performance of the duties outlined in paragraph 6450(b) to the General Manager, General Counsel and Ethics Officer for their information and appropriate actions. Whenever an audit report contains recommendations for corrective actions or changes in current practices, the General Manager, General Counsel, Ethics Officer or their designees shall respond to the General Auditor in an appropriate manner and within a reasonable time, indicating their views on the recommendations and proposed actions to be taken, if any.
- (d) The General Auditor's reports on internal audit assignments shall be addressed to the Executive Committee. The General Auditor shall have the discretion to determine the form and content of such audit reports, subject to guidance by the Executive Committee. With the exception of those reports which the General Auditor deems to be urgent or confidential in nature, copies of all audit reports addressed to the Executive Committee shall be submitted to the General Manager and General Counsel for review and comment simultaneously to their submittal to the Executive Committee.
- (e) The General Auditor shall transmit all reports issued by the District's external auditors to the Executive Committee and any other committees of the Board as may be applicable. Such transmittal letters should include any comments on the external auditor's reports that the General Auditor deems necessary.
- (f) The General Auditor may receive requests from time to time from the other executive officers or committees of the Board to perform audit assignments which are not included in the approved annual Audit Business Plan. Similarly, the General Auditor may identify a need to include new assignments in the Audit Business Plan during the year. The General Auditor shall have sufficient latitude and discretion to include those new assignments in the annual Audit Business Plan as the General Auditor deems necessary based upon their professional judgment and available resources. Requests from other

committees of the Board and individual Board members desiring specific audit assignments shall be submitted to the Audit Subcommittee for study, advise, and recommendation, or if such subcommittee is not currently in place, the Executive Committee. Once the audit assignment is approved by the Board, the General Auditor reserves the right to determine how to best fit the directed audit assignment into the Audit Business Plan. The reporting process for assignments requested by either executive management, by committees of the Board, or by individual Board members shall generally follow the process outlined in paragraphs 6450(c) or (d) previously.

(g) The General Auditor shall manage the work of the Audit Department in accordance with the Audit Department Charter. The General Auditor shall assess annually whether the purpose, authority and responsibility, as defined in this Charter, continue to be adequate to enable the Audit Department to accomplish its objectives.

#### § 6451. Audit Department Charter.

- (a) Mission and Scope of Work The mission of the Audit Department is to provide independent, professional, objective assurance and consulting services designed to add value and improve Metropolitan's operations. It helps the organization accomplish its objectives by using a proactive, systematic approach to evaluate and improve the effectiveness of risk management, control and governance processes. The scope of work of the Audit Department is to determine whether the organization's network of risk management, internal control and governance processes, as designed and represented by management, is adequate and functioning in a manner to ensure:
  - (1) Risks are appropriately identified, managed and monitored;
  - (2) Significant financial, managerial and operating information is accurate, reliable and
- (3) Employees' actions are in compliance with policies, standards, procedures and applicable laws and regulations;
  - (4) Resources are acquired economically, used efficiently, and adequately protected;
  - (5) Programs, plans, and objectives are achieved;
- (6) Quality and continuous improvement are fostered in the organization's control process;
- (7) Significant legislative or regulatory issues impacting the organization are recognized and addressed appropriately;

Opportunities for improving management internal control, efficiency and the organization's image may be identified during audits. They will be communicated to the appropriate level of management.

- (b) Accountability The General Auditor shall be accountable to the Board of Directors and the Executive Committee to:
- (1) Advise on the adequacy and effectiveness of the organization's processes for controlling its activities and managing its risks;
- (2) Report significant issues related to the processes for controlling the activities of the organization, including potential improvements to those processes, and provide information concerning such issues through to resolution;
- (3) Coordinate with other Metropolitan control and monitoring functions (risk management, legal, finance, ethics, security and environmental);

(c) Professional Standards – The Audit Department shall govern itself by adherence to The Institute of Internal Auditors' mandatory guidance, including the Definition of Internal Auditing, the Code of Ethics, and the *International Standards for the Professional Practice of Internal Auditing (Standards)*. This mandatory guidance constitutes principles of the fundamental requirements for the professional practice of internal auditing and for evaluating the effectiveness of the Audit Department's performance. These documents constitute the operating procedures for the department and constitute an addendum to the charter.

The Institute of Internal Auditors' Practice Advisories, Practice Guides, and Position Papers shall also be adhered to as applicable. In addition, Audit Department staff shall adhere to Metropolitan's policies and procedures, the California Government Code; Government Auditing Standards (GAGAS); and the Audit Department's Policies and Procedures Manual.

- (d) Responsibilities The Audit Department shall carry out the following responsibilities:
- (1) Develop and present a flexible annual audit plan to the Executive Committee for review and approval. This plan should be developed utilizing a risk-based methodology and should include risks or internal control concerns identified by Management or the Board of Directors;
- (2) Report periodically to the Executive Committee and Management the status of the current year's audit plan and the sufficiency of department resources;
  - (3) Issue an opinion on internal controls over financial reporting on an annual basis;
- (4) Submit audit reports to the Executive Committee and Management communicating the auditor's opinion regarding the internal control structure, identifying significant control issues and providing related recommendations;
- (5) Evaluate the adequacy and timeliness of Management's responses to, and the corrective action taken on, all significant control issues noted in such reports. Conduct follow-up reviews as necessary and periodically report to the Executive Committee the status of Management's progress;
- (6) Ensure the selection, development and supervision of competent and professional audit staff;
- (7) Perform a quality assurance program by which the General Auditor evaluates internal auditing activities against professional standards. Obtain external quality assurance review as required by GAGAS and the Standards;
- (8) Perform consulting services to assist management in meeting its objectives. Examples may include facilitation, process design, training, and advisory services;
- (9) Evaluate additions or changes in internal control processes coincident with their development and implementation;
- (10) Keep the Executive Committee informed of significant emerging trends and best practices in internal auditing and governance;
- (11) Assist in the investigation of significant suspected fraudulent activities within the organization. Assure reporting to the Executive Committee on the results, as appropriate;
- (12) Coordinate with external auditors to minimize duplication of effort and to ensure that issues raised, as a result of their review, are appropriately addressed.
  - (e) Authority The General Auditor and Audit Department staff members are authorized to:
- (1) Have unrestricted access to all functions, records, property, and personnel, subject to the requirements of safekeeping, confidentiality and applicable process;
  - (2) Have full and free access to the Executive Committee, subject to applicable law,

- (3) Allocate resources, set frequencies, and select subject, determine scopes of work, and apply the techniques required to accomplish audit objectives;
- (4) Obtain the necessary assistance of personnel in units of the organization where they perform audit, as well as other specialized services from within or outside the organization;

The General Auditor and Audit Department staff members are not authorized to:

- (1) Perform any operational duties for the organization;
- (2) Initiate or approve accounting transactions external to the internal auditing department;
- (3) Direct the activities of any organization employee not employed by the internal auditing department, except to the extent such employees have been appropriately assigned to auditing teams or to otherwise assist the internal auditors.

This Charter shall be reviewed at least annually by the Executive Committee and Board of Directors.

#### § 6453. Authority to Obtain Professional Services.

The General Auditor is authorized to employ the services of independent auditors or other professional or technical consultants to advise or assist them in performing their assigned duties as may be required or as they deem necessary provided that the amount to be expended in fees, costs, and expenses under any one contract in any one year shall not exceed \$100,000. External auditors, while employed as external auditors for Metropolitan, are prohibited from performing any other consulting work for Metropolitan or performing any work for other clients that conflicts, or may conflict, with their responsibilities as Metropolitan's external auditors. These prohibitions shall be included in Metropolitan's agreements with external auditors. The General Auditor shall inform the Executive Committee whenever they exercise the authority granted under this section and they shall further report quarterly to the Legal and Claims Committee concerning any agreements entered into under this section.

#### **Division VIII**

#### CONTRACTS/DISTRICT PROPERTY

#### Chapter 1

#### **CONTRACTS**

#### § 8248. Disposal of Unnecessary Improvements.

- (a) The General Manager is authorized to dispose of, in the manner the General Manager deems to be in the best interest of the District, any improvements that must be removed to make land acquired for District operations suitable for District use.
- (b) The General Manager shall report quarterly to the Finance and Asset Management Committee the details of any transactions during the preceding quarter in which an improvement was disposed of in such a manner as to make the improvement available for subsequent use by a party other than the District.

#### § 8257. Quarterly Reports.

The General Manager shall report to the Finance and Asset Management Committee quarterly on any real property sold pursuant to this Article.



# **Board Action**

# Board of Directors Legal and Claims Committee

2/13/2024 Board Meeting

7-11

#### **Subject**

Approve: (1) changes to the Metropolitan Water District 457(b) and 401(k) deferred compensation plans to establish an investment committee and adopt investment committee bylaws; and (2) amendments necessary to conform the plans and the Metropolitan Administrative Code to the proposed changes, current law, practices, and regulations; the General Manager has determined that the proposed action is exempt or otherwise not subject to CEOA

#### **Executive Summary**

The proposed changes will establish a new Deferred Compensation Investment Committee (DCIC) with governing bylaws (DCIC bylaws) to determine investment-related matters affecting plan assets of the Metropolitan Water District 457(b) and 401(k) deferred compensation plans (the Plans). Proposed amendments will conform the Plans and the Metropolitan Water District Administrative Code (MWD Administrative Code) to the proposed changes in governance of the Plans and conform the Plans to current law, practices, and regulations. The Plan Administrator's duties will be realigned to non-investment-related matters. The Board will continue as the plan fiduciary retaining its ultimate authority over the Plans. These changes will necessarily result in dissolving certain components of the current governance structure, namely the Deferred Compensation Advisory Committee (DCAC), the Trustee Committee, and the Plans' trust agreements.

#### Time/Urgency

Delayed adoption of the proposed changes would fail to establish an investment committee and realign the Plan Administrator's duties and would postpone updating the Plans and MWD Administrative Code to change the plan governance and conform the Plans to current law, practices, and regulations.

#### Proposed Action(s)/Recommendation(s) and Options

#### **Staff Recommendation:**

#### Option #1

Approve: (1) changes to the Metropolitan Water District 457(b) and 401(k) deferred compensation plans to establish an investment committee and adopt investment committee bylaws; and (2) amendments necessary to conform the plans and the Metropolitan Administrative Code to the proposed changes, current law, practices, and regulations.

**Fiscal Impact:** No significant fiscal impacts are expected. Implementation of after-tax Roth accounts could attract new participants, thus increasing the gross match on 401(k) plan accounts. However, as the 401(k) plan participation rate exceeds 90% and the pre-tax option is the primary advantage sought by most participants, fiscal impacts should be negligible.

**Business Analysis:** Approval would establish an investment committee and realign the Plan Administrator's duties and would update the Plans and MWD Administrative Code to change the plan governance and conform the Plans to current law, practices, and regulations.

#### Option #2

Do not approve: (1) changes to the Metropolitan Water District 457(b) and 401(k) deferred compensation plans to establish an investment committee and adopt investment committee bylaws; and (2) amendments necessary to conform the plans and the Metropolitan Administrative Code to the proposed changes, current law, practices, and regulations.

Fiscal Impact: None

**Business Analysis:** Not adopting the proposed changes would fail to establish an investment committee and realign the Plan Administrator's duties and would postpone updating the Plans and MWD Administrative Code to change the plan governance and conform the Plans to current law, practices, and regulations.

#### **Alternatives Considered**

An alternative governance restructure was considered, comprised of three components. First, still maintaining the DCAC as an advisory body, but adopting bylaws clarifying its roles and functions. Second, delegating the role of financial fiduciary from the Plan Administrator to the Trust Committee (a three-member, management-level committee established in 2107 by the General Manager), which would have its own bylaws and meetings; meanwhile, each member would continue to serve in an advisory capacity on the DCAC. Third, as with the current proposal, realigning the Plan Administrator's duties to the role of administrative fiduciary. This alternative was rejected because it would not give members of the District's employee bargaining units, who serve on the DCAC, a vested stake in managing the Plans' assets and because having two committees would significantly increase administrative burdens across the board for staff, committee members, and third-party service providers.

#### **Applicable Policy**

Metropolitan Water District Administrative Code Section 6810: Employee Deferred Compensation and Savings Plans

The Metropolitan Water District of Southern California 401(k) Savings Plan

The Metropolitan Water District of Southern California Deferred Compensation Plan

Metropolitan Water District Administrative Code Section 11104: Delegation of Responsibilities

#### Related Board Action(s)/Future Action(s)

There are no related board actions or future actions as part of the proposed restructure of governance changes or the proposed amendments to the Plans or the MWD Administrative Code.

#### **Summary of Outreach Completed**

The current Interim Plan Administrator reached out to Metropolitan's bargaining units to clarify and obtain support for the current proposed changes as presented in this board letter.

#### California Environmental Quality Act (CEQA)

#### **CEQA determination for Option #1:**

The proposed action is not defined as a project under CEQA because it involves organizational, maintenance, or administrative activities; personnel-related actions; and/or general policy and procedure making that will not result in direct or indirect physical changes in the environment. (Public Resources Code Section 21065; State CEQA Guidelines Section 15378(b)(2) and (5).)

#### **CEQA determination for Option #2:**

None required

#### **Details and Background**

#### **Background**

#### A. Deferred Compensation Investment Committee

Board approval of MWD Administrative Code amendments and DCIC bylaws is required to establish the DCIC. The following summarizes these proposed actions:

**MWD Code Amendments:** Approve the following proposed amendments to the MWD Administrative Code, Division VI, Section 8 to establish the DCIC.

- Update the Plans' effective date to January 1, 2024. (Sections 6811(a) and (b); 6819.)
- Delete the Trustee Committee. (Sections 6814, 6815, 6816(a).)
- Affirm that the Plan Administrator is indemnified by Metropolitan with respect to liability or loss except due to willful misconduct or gross negligence; this is a minor cleanup amendment. (Section 6815.)
- Underscore the right of Metropolitan's bargaining units to meet and confer if the Board of Directors ever were to consider terminating the Plans. (Section 6816(a).)
- Substitute "DCIC" for "DCAC" throughout Division VII, Section 8. (Sections 6818, 6814, 6815.)

See **Attachment 1** - MWD Administrative Code (with changes marked) and **Attachment 2** - MWD Administrative Code (clean copy).

**DCIC Bylaws:** Adopt the proposed DCIC bylaws to govern the new committee. The bylaws cover committee composition, fiduciary duties, officers, meetings, investment-related duties, committee policies and procedures, advisory duties, disclosure of conflicts, and indemnification by Metropolitan. (**Attachment 3**)

These board actions will realign the Plan Administrator's duties to non-investment-related matters. These duties mainly include determining employee eligibility, establishing non-investment plan features, deciding death benefit claims, procuring plan service providers, designing communication campaigns, reviewing and updating the Plans, paying invoices, maintaining plan records, and directing the external corporate Trustee, who holds the Plan's assets. The Board will retain its authority to delegate its fiduciary duties, establish the Plans' terms and specifications, and terminate the Plans. These changes in governance of the Plans will necessarily result in dissolving the DCAC, the Trustee Committee, and the Plans' 1985 and 1997 trust agreements, originally executed to establish and operate the Plans in-house. (Attachments 4 and 5)

#### **B.** 2024 Plan Amendments

Board approval of the proposed amendments to the Plans is required to conform the Plans to the proposed changes in governance of the Plans and to current law, practices, and regulations. The following summarizes the proposed amendments:

- Adopting Setting Every Community Up for Retirement Security Act provisions increasing the required beginning date from age 70½#to 72 and allowing \$5,000 distributions for new births and adoptions.
- Adopting a Bipartisan Miner's Act provision lowering the age from 70½ to 59½ for 457(b) in-service distributions, thus aligning with the 401(k) plan age in-service age minimum for plan participants.
- Adopting Coronavirus Aid, Relief, and Economic Security Act provisions that allowed hardship distributions and loans of up to \$100,000 during the COVID-19 crisis.
- Establishing after-tax Roth option for elective deferrals and rollover accounts and loan availability from Roth accounts. The new Roth feature is expected to be implemented in the April-May 2024 timeframe.
- Aligning the standard for indemnifying the Plans' fiduciaries to MWD Administrative Code Section 6815.
- Clarifying the Plan Administrator's role to interpret the Plans and determine administrative claims.

- Affirming the General Counsel's role with respect to plan-related litigation in alignment with MWD Administrative Code Section 6433.
- Clarifying the Plans' operation rules for ordering excess deferral corrections, minimum rollovers, default beneficiaries, requiring spousal consent, student intern hours, and former employee roll-ins.
- Conforming the Plans to the proposed DCIC bylaws.

See **Attachment 6** - MWD 457(b) Deferred Compensation Plan (with changes marked) and **Attachment 7** – 457(b) MWD Deferred Compensation Plan (clean copy). See **Attachment 8** - MWD 401(k) Deferred Compensation Plan (with changes marked) and **Attachment 9** - MWD 401(k) Deferred Compensation Plan (clean copy). See also **Attachment 10** – Summary Charts of Proposed Plan Amendments.

Katano Kasaine

2/8/2024 Date

Assistant General Manager/Chief Financial

Kegm

Officer

2/8/2024

Adel Hagekhlil General Manager Date

Attachment 1 – MWD Administrative Code excerpts (with changes marked)

Attachment 2 – MWD Administrative Code excerpts (clean copy)

Attachment 3 - MWD Bylaws of the Deferred Compensation Investment Committee

Attachment 4 – MWD 401(k) Trust Agreement (November 18,1997)

Attachment 5 – MWD 457(b) Trust Agreement (July 1,1985)

Attachment 6 – MWD 457(b) Deferred Compensation Plan (with changes marked)

Attachment 7 – MWD 457(b) Deferred Compensation Plan (clean copy)

Attachment 8 – MWD 401(k) Deferred Compensation Plan (with changes marked)

Attachment 9 – MWD 401(k) Deferred Compensation Plan (clean copy)

Attachment 10 - Summary Charts of Proposed Plan Amendments

Ref# I12694607LC

#### **Chapter 8**

#### EMPLOYEE DEFERRED COMPENSATION AND SAVINGS PLANS

#### Article 1

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 457(b) DEFERRED COMPENSATION PLAN AND

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 401(k) SAVINGS PLAN

- 6810. Official Names.
- 6811. Establishment of the Plans.
- 6812. Purposes.
- 6813. Administration by the General Manager.
- 6814. Immunities.
- 6815. Indemnification.
- 6816. Amendment, Termination or Merger.
- 6817. Definitions.
- 6818. <u>Deferred Compensation Investment Advisory</u> Committee
- 6819. Effective Dates.

#### § 6810. Official Names.

- (a) The official name of the District's 401(k) defined contribution profit-sharing plan is "The Metropolitan Water District of Southern California 401(k) Savings Plan" ("401(k) Plan").
- (b) The official name of the District's 457(b) deferred compensation plan is "The Metropolitan Water District of Southern California Deferred Compensation Plan" ("457(b) Plan").

#### § 6811. Establishment of the Plans.

(a) The Board of Director's established a profit-sharing plan with a cash or deferral arrangement known as "The Metropolitan Water District of Southern California Savings Plan" ("Savings Plan I") effective July 1, 1985 (M.I. 35632). The Board of Director's also established another profit-sharing plan with a cash or deferral arrangement known as "Savings Plan II" applicable to a separate group of employees effective October 1, 1985 (M.I. 35816). As a result, the Board of Director's merged "Savings Plan I" into "Savings Plan II" and renamed it "The Metropolitan Water District of Southern California Consolidated Savings Plan" effective January 1, 2006 (M.I.

- 46592)("Consolidated Plan") and incorporated it into Division VI, Chapter 7, Article 6 of this code. The Consolidated Plan has been amended and restated from time to time. It was most recently restated and renamed "The Metropolitan Water District of Southern California 401(k) Savings Plan" (the "401(k) Plan") effective August 1, 2018. It was most recently restated effective January 1, 2024.
- The Board of Director's established the "Deferred Compensation Plan" effective
  November 1, 1976 (Reso. No. 7683)("the Plan"). The Plan was restated by Board of
  Director's effective April 1, 1983 (M.I. 34625). Board of Director's caused the Plan to be
  codified in Division VI, Chapter 7, Article 8 of the this code and named the
  "Metropolitan Water District of Southern California 1997 Deferred Compensation Plan"
  ("1997 Plan") effective November 18, 1997 (M.I. 42712). The 1997 Plan has been
  amended from time to time. The plan was most recently restated and renamed "The
  Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan"
  (the "457(b) Plan") effective August 1, 2018. It was most recently restated effective
  January 1, 2024.
- (b)(c) The District is authorized to and shall establish the governing terms and specifications of the 401(k) Plan in a written plan document outside the Administrative Code. A copy of the 401(k) Plan is retained by the General Manager and will be made available upon request to Employees and other interested parties.
- (e)(d) The District is authorized to and shall establish the governing terms and specifications of the 457(b) Plan in a written plan document outside the Administrative Code. A copy of the 457(b) Plan is retained by the General Manager and will be made available upon request to Employees and other interested parties.

#### § 6812. Purposes.

- (a) The 401(k) Plan is a tax-qualified governmental defined contribution plan under Internal Revenue Code ("IRC") Sections 401(a) and 414(d). The 401(k) Plan is a profit sharing plan pursuant to IRC Section 401(a)(27) that has a grandfathered cash or deferral arrangement under Code § 401(k).
- (b) The 457(b) Plan was established, and is maintained, as an eligible governmental deferred compensation plan meeting the requirements of IRC Sections 457(b) and 457(e)(1)(A) and Treas. Reg. Sections 1.457-3 through 1.457-10. Further, the Plan meets the definition of "qualified plan" under Code Section 409A(d)(2)(A), and is therefore exempt from the nonqualified deferred compensation requirements of Code Section 409A.

(c) The purpose of the 401(k) Plan and the 457(b) Plan (referred to herein after as "the 401(k) and 457(b) Plans" or "the 401(k) or 457(b) Plans") is to attract and retain officers and employees by permitting them to enter into agreements with the District which will provide benefits on retirement, death, disability, separation from service or as otherwise permitted by law.

#### § 6813. Administration by General Manager.

The District is the Administrator of the 401(k) and 457(b) Plans and has designated the General Manager to carry out the powers, duties and responsibilities of the Plan Administrator. The General Manager shall represent the District in all matters concerning administration of the 401(k) and 457(b) Plans. Further, the General Manager shall have full power and authority: 1) to adopt rules and regulations for the administration of the 401(k) and 457(b) Plans provided they are not inconsistent with either plan, respectively; 2) to interpret, alter, amend or revoke any rules and regulations so adopted; 3) to appoint such administrative agents, or persons as they deem advisable or desirable to carry out the terms and conditions of the 401(k) and 457 (b) Plans, as applicable; and, 4) to exercise any other powers, duties and responsibilities as are provided in the 401(k) and 457(b) Plans, respectively, for purposes of their orderly administration. However, the exercise of such power and authority does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to Section 6102 (Employee Rights).

#### § 6814. Immunities.

- (a) Neither the District, its Directors, nor its Employees shall be liable: 1) for any action taken or not taken with respect to the 401(k) or 457(b) Plans, except for its or their own gross negligence or willful misconduct;
- (b) Neither the District, nor its Directors, nor its Employees shall be personally liable upon any contract, agreement or other instrument made or executed for administration of the 401(k) or 457(b) Plans.

The District, its Board of Directors, each Employee thereof, specifically including the General Manager, and any other persons to whom the General Manager delegates or assigns any duty with respect to the 401(k) or 457(b) Plans, may rely and shall be fully protected in acting in good faith upon: 1) the advice of counsel, who may be counsel for the District; 2) the records of the District; 3) the opinion, certificate, valuation, report, recommendation, or determination of the Trustee Committee or any member thereof, the Investment Advisor, the Deferred Compensation Investment Advisory Committee or any member thereof, or the District's General Auditor; or, 4) any certificate, statement or other representation made by or any information furnished by an

Employee, Participant, Beneficiary, Alternate Payee, Trustee Committee or any member thereof, the Deferred Compensation Investment Advisory Committee or any member thereof, or Investment Advisor, concerning any fact required to be determined under any of the provisions of the 401(k) or 457(b) Plans.

#### § 6815. Indemnification.

The District will indemnify, defend and hold harmless members of the Board of Directors, the General Manager, the Plan Administrator, or any Employees who have administrative responsibility under the 401(k) or 457(b) Plans, the Trustee Committee or any member thereof, and the Deferred Compensation Investment Advisory Committee or any member thereof, with respect to any liability, loss, damage or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacities in the administration or management of the assets of the 401(k) or 457(b) Plans, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.

#### § 6816. Amendment, Termination or Merger.

- The District reserves the right at any time to terminate the 401(k) or 457(b) Plans, in whole or in part, or as to any designated group of Employees, Participants, Beneficiaries, or to amend the 401(k) or 457(b) Plans, in whole or in part. Any termination of the 401(k) Plan or 457(b) Plan shall be expressed in an instrument executed by the District on the order of its Board of Directors and filed with the Trustee Committee and filed with the Plan Administrator, and shall become effective as of the date designated in such instrument or, if no date is so designated, on its execution. If either of the 401(k) Plan or 457(b) Plan is terminated by the District, accounts in the terminating plan shall be distributed, to Participants, Beneficiaries, and Alternate Payees, as applicable. Distributions will be made in accordance with the applicable tax law. However, the exercise of such right to terminate the 401(k) or 457(b) Plans does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to Section 6102 (Employee Rights).
- (a)(b) The General Manager may, at any time and from time to time, amend the Plans in whole or in part to ensure compliance with applicable laws and regulations and further the administration of the Plans. However, no Plan amendment may authorize or permit any portion of the assets held under the Plans to be used for or diverted to a purpose other than the exclusive benefit of Participants or Beneficiaries, except to the extent such assets are used to pay taxes or administrative expenses of the Plans. An amendment also may not cause or permit any portion of the assets held under the Plans to revert to or become

property of the District. Nor may the General Manager amend the Plans to effectuate a termination.

(b)(c) In the case of any merger or consolidation with, or transfer of assets or liabilities to, any other plan, each Participant, Beneficiary, or Alternate Payee in the 401(k) or 457(b) Plans, as applicable, will receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit the Participant, Beneficiary, or Alternate Payee would have been entitled to receive immediately before the merger, consolidation, or transfer.

#### § 6817. Definitions.

In the event of any inconsistencies between the provisions of this Chapter 8 and the IRS Code of the 401(k) and 457(b) Plans, the terms of the IRS Code and the 401(k) and 457(b) Plans shall govern and control.

#### § 6818. Deferred Compensation Investment Advisory Committee.

A Deferred Compensation Investment Advisory Committee may be established to manage the assets of the 401(k) and 457 Plans and, when reasonable and appropriate, may advise the General Manager concerning policies and guidelines relating to the administration of the 401(k) and 457(b) Plans and provide input on 401(k) Plan and 457(b) Plan amendments. Representatives on the Deferred Compensation Investment Advisory Committee so qualified shall be allowed time off without loss of compensation or benefits when they are participating in committee meetings. The General Manager shall give due consideration to said advice, but shall not be bound by any such advice. Members of the Deferred Compensation Investment Advisory Committee shall serve without compensation.

#### § 6819. Effective Dates.

The effective date of the 401(k) Plan restatement shall be <u>January 1, 2024.</u> August 1, 2018. The effective date of the 457(b) Plan restatement shall be <u>January 1, 2024August 1, 2018</u>.

#### **Chapter 8**

#### EMPLOYEE DEFERRED COMPENSATION AND SAVINGS PLANS

#### Article 1

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 457(b) DEFERRED COMPENSATION PLAN AND

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 401(k) SAVINGS PLAN

- 6810. Official Names.
- 6811. Establishment of the Plans.
- 6812. Purposes.
- 6813. Administration by the General Manager.
- 6814. Immunities.
- 6815. Indemnification.
- 6816. Amendment, Termination or Merger.
- 6817. Definitions.
- 6818. Deferred Compensation Investment Committee
- 6819. Effective Dates.

#### § 6810. Official Names.

- (a) The official name of the District's 401(k) defined contribution profit-sharing plan is "The Metropolitan Water District of Southern California 401(k) Savings Plan" ("401(k) Plan").
- (b) The official name of the District's 457(b) deferred compensation plan is "The Metropolitan Water District of Southern California Deferred Compensation Plan" ("457(b) Plan").

#### § 6811. Establishment of the Plans.

(a) The Board of Director's established a profit-sharing plan with a cash or deferral arrangement known as "The Metropolitan Water District of Southern California Savings Plan" ("Savings Plan I") effective July 1, 1985 (M.I. 35632). The Board of Director's also established another profit-sharing plan with a cash or deferral arrangement known as "Savings Plan II" applicable to a separate group of employees effective October 1, 1985 (M.I. 35816). As a result, the Board of Director's merged "Savings Plan I" into "Savings Plan II" and renamed it "The Metropolitan Water District of Southern California Consolidated Savings Plan" effective January 1, 2006 (M.I. 46592) ("Consolidated Plan") and incorporated it into Division VI, Chapter 7, Article 6 of

this code. The Consolidated Plan has been amended and restated from time to time. It was restated and renamed "The Metropolitan Water District of Southern California 401(k) Savings Plan" (the "401(k) Plan") effective August 1, 2018. It was most recently restated effective January 1, 2024.

- (b) The Board of Director's established the "Deferred Compensation Plan" effective November 1, 1976 (Reso. No. 7683) ("the Plan"). The Plan was restated by Board of Director's effective April 1, 1983 (M.I. 34625). Board of Director's caused the Plan to be codified in Division VI, Chapter 7, Article 8 of this code and named the "Metropolitan Water District of Southern California 1997 Deferred Compensation Plan" ("1997 Plan") effective November 18, 1997 (M.I. 42712). The 1997 Plan has been amended from time to time. The plan was restated and renamed "The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan" (the "457(b) Plan") effective August 1, 2018. It was most recently restated effective January 1, 2024.
- (c) The District is authorized to and shall establish the governing terms and specifications of the 401(k) Plan in a written plan document outside the Administrative Code. A copy of the 401(k) Plan is retained by the General Manager and will be made available upon request to Employees and other interested parties.
- (d) The District is authorized to and shall establish the governing terms and specifications of the 457(b) Plan in a written plan document outside the Administrative Code. A copy of the 457(b) Plan is retained by the General Manager and will be made available upon request to Employees and other interested parties.

#### § 6812. Purposes.

- (a) The 401(k) Plan is a tax-qualified governmental defined contribution plan under Internal Revenue Code ("IRC") Sections 401(a) and 414(d). The 401(k) Plan is a profit sharing plan pursuant to IRC Section 401(a)(27) that has a grandfathered cash or deferral arrangement under Code § 401(k).
- (b) The 457(b) Plan was established, and is maintained, as an eligible governmental deferred compensation plan meeting the requirements of IRC Sections 457(b) and 457(e)(1)(A) and Treas. Reg. Sections 1.457-3 through 1.457-10. Further, the Plan meets the definition of "qualified plan" under Code Section 409A(d)(2)(A), and is therefore exempt from the nonqualified deferred compensation requirements of Code Section 409A.
- (c) The purpose of the 401(k) Plan and the 457(b) Plan (referred to herein after as "the 401(k) and 457(b) Plans" or "the 401(k) or 457(b) Plans") is to attract and retain officers and employees by permitting them to enter into agreements with the District which will

provide benefits on retirement, death, disability, separation from service or as otherwise permitted by law.

#### § 6813. Administration by General Manager.

The District is the Administrator of the 401(k) and 457(b) Plans and has designated the General Manager to carry out the powers, duties and responsibilities of the Plan Administrator. The General Manager shall represent the District in all matters concerning administration of the 401(k) and 457(b) Plans. Further, the General Manager shall have full power and authority: 1) to adopt rules and regulations for the administration of the 401(k) and 457(b) Plans provided they are not inconsistent with either plan, respectively; 2) to interpret, alter, amend or revoke any rules and regulations so adopted; 3) to appoint such administrative agents, or persons as they deem advisable or desirable to carry out the terms and conditions of the 401(k) and 457 (b) Plans, as applicable; and, 4) to exercise any other powers, duties and responsibilities as are provided in the 401(k) and 457(b) Plans, respectively, for purposes of their orderly administration. However, the exercise of such power and authority does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to Section 6102 (Employee Rights).

#### § 6814. Immunities.

- (a) Neither the District, its Directors, nor its Employees shall be liable: 1) for any action taken or not taken with respect to the 401(k) or 457(b) Plans, except for its or their own gross negligence or willful misconduct;
- (b) Neither the District, nor its Directors, nor its Employees shall be personally liable upon any contract, agreement or other instrument made or executed for administration of the 401(k) or 457(b) Plans.

The District, its Board of Directors, each Employee thereof, specifically including the General Manager, and any other persons to whom the General Manager delegates or assigns any duty with respect to the 401(k) or 457(b) Plans, may rely and shall be fully protected in acting in good faith upon: 1) the advice of counsel, who may be counsel for the District; 2) the records of the District; 3) the opinion, certificate, valuation, report, recommendation, or determination of the Investment Advisor, the Deferred Compensation Investment Committee or any member thereof, or the District's General Auditor; or, 4) any certificate, statement or other representation made by or any information furnished by an Employee, Participant, Beneficiary, Alternate Payee, the Deferred Compensation Investment Committee or any member thereof, or Investment Advisor, concerning any fact required to be determined under any of the provisions of the 401(k) or 457(b) Plans.

#### § 6815. Indemnification.

The District will indemnify, defend and hold harmless members of the Board of Directors, the General Manager, the Plan Administrator, or any Employees who have administrative responsibility under the 401(k) or 457(b) Plans, and the Deferred Compensation Investment Committee or any member thereof, with respect to any liability, loss, damage or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacities in the administration or management of the assets of the 401(k) or 457(b) Plans, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.

#### § 6816. Amendment, Termination or Merger.

- (a) The District reserves the right at any time to terminate the 401(k) or 457(b) Plans, in whole or in part, or as to any designated group of Employees, Participants, Beneficiaries, or to amend the 401(k) or 457(b) Plans, in whole or in part. Any termination of the 401(k) Plan or 457(b) Plan shall be expressed in an instrument executed by the District on the order of its Board of Directors and filed with the Plan Administrator, and shall become effective as of the date designated in such instrument or, if no date is so designated, on its execution. If either of the 401(k) Plan or 457(b) Plan is terminated by the District, accounts in the terminating plan shall be distributed, to Participants, Beneficiaries, and Alternate Payees, as applicable. Distributions will be made in accordance with the applicable tax law. However, the exercise of such right to terminate the 401(k) or 457(b) Plans does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to Section 6102 (Employee Rights).
- (b) The General Manager may, at any time and from time to time, amend the Plans in whole or in part to ensure compliance with applicable laws and regulations and further the administration of the Plans. However, no Plan amendment may authorize or permit any portion of the assets held under the Plans to be used for or diverted to a purpose other than the exclusive benefit of Participants or Beneficiaries, except to the extent such assets are used to pay taxes or administrative expenses of the Plans. An amendment also may not cause or permit any portion of the assets held under the Plans to revert to or become property of the District. Nor may the General Manager amend the Plans to effectuate a termination.
- (c) In the case of any merger or consolidation with, or transfer of assets or liabilities to, any other plan, each Participant, Beneficiary, or Alternate Payee in the 401(k) or 457(b)

Plans, as applicable, will receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit the Participant, Beneficiary, or Alternate Payee would have been entitled to receive immediately before the merger, consolidation, or transfer.

#### § 6817. Definitions.

In the event of any inconsistencies between the provisions of this Chapter 8 and the IRS Code of the 401(k) and 457(b) Plans, the terms of the IRS Code and the 401(k) and 457(b) Plans shall govern and control.

#### § 6818. Deferred Compensation Investment Committee.

A Deferred Compensation Investment Committee may be established to manage the assets of the 401(k) and 457 Plans and, when reasonable and appropriate, may advise the General Manager concerning policies and guidelines relating to the administration of the 401(k) and 457(b) Plans and provide input on 401(k) Plan and 457(b) Plan amendments. Representatives on the Deferred Compensation Investment Committee so qualified shall be allowed time off without loss of compensation or benefits when they are participating in committee meetings. The General Manager shall give due consideration to said advice, but shall not be bound by any such advice. Members of the Deferred Compensation Investment Committee shall serve without compensation.

#### § 6819. Effective Dates.

The effective date of the 401(k) Plan restatement shall be January 1, 2024. The effective date of the 457(b) Plan restatement shall be January 1, 2024.

#### **BYLAWS**

of the

# DEFERRED COMPENSATION INVESTMENT COMMITTEE

of

The Metropolitan Water District of Southern California 401(k) Savings Plan and The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan

Effective February 13, 2024

#### ARTICLE I. ESTABLISHMENT AND PURPOSE

<u>Section 1: Establishment</u>. The Board of Directors (the "Board") of The Metropolitan Water District of Southern California (the "District") has, by its approval of these Bylaws of the Deferred Compensation Investment Committee ("Bylaws") at its meeting held on the 13th day of February 2024, formally established the Deferred Compensation Investment Committee ("Committee").

Section 2: Purpose. The District established and maintains The Metropolitan Water District of Southern California 401(k) Savings Plan ("401(k) Plan") and The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan ("457(b) Plan"), referred to collectively herein as "the Plans<sub>3</sub>" for the purpose of providing retirement benefits to its eligible employees.

The District established the Committee to invest the assets of the Plans. The Committee has the authority and responsibility to invest the assets of the Plans, subject to the duties set forth in Article VI, Section 1, and the terms of the Plans. In addition, the Committee has such authority and responsibilities as the Board may delegate to it from time to time.

The Committee has adopted these Bylaws effective January 1, 2024, to set forth the governance structure and the guidelines it will follow to fulfill its responsibilities under the Plans. These Bylaws will be interpreted in a manner consistent with Plans and, in the event of conflict, the terms of the Plans will govern.

#### ARTICLE II. COMPOSITION OF COMMITTEE

<u>Section 1: Number, Appointment, and Term.</u> The Committee consists of 11 members, as follows: six ex-officio members:

- (1) Human Resources Group Manager;
- (2) Assistant General Manager, Operations or his or her designee;

- (3) Treasury & Debt Manager;
- (4) Assistant General Manager, Chief Financial Officer or Finance Group Manager, as his or her designee;
- (5) General Auditor or his or her designee; and
- (6) Risk Manager;

and five appointed members, one appointed by each of the four bargaining units:

- (1) the Employees' Association of the Metropolitan Water District, AFSCME Local 1902 (aka "AFSCME Local 1902);
- (2) the Management and Professional Employees Association of the Metropolitan Water District, AFSCME Local 1001 (aka "MAPA");
- (3) the Supervisors Association of the Metropolitan Water District of Southern California (aka "SA");
- (4) the Association of Confidential Employees (aka "ACE"); and
- (5) one Retiree who is a participant in the Plans, and is appointed by the Committee.

Each ex officio Committee member will serve for his or her term of office. Each appointed Committee member will serve until his or her resignation, removal, or ceasing to qualify for membership, except that the Plans' retiree member will serve for a limited term of no more than five (5) years, as established by the Committee at the time of his or her appointment.

The Human Resources Benefits Manager and the Human Resource Principal Analyst will attend all Committee meetings and provide the Committee with administrative support. A representative of the District's Office of the General Counsel shall attend all meetings and provide the Committee with legal support and advice.

<u>Section 2: Resignation and Removal</u>. An appointed Committee member may resign or be removed by the appointing authority. Such member's resignation will be effective 30 days after the appointing authority's receipt of written notice of such resignation, or sooner at the appointing authority's discretion. Such member's removal will be effective upon receipt by every individual Committee member of written notice from the appointing authority, or later at the appointing authority's discretion.

<u>Section 3: Qualifications of Members</u>. Each Committee member, except the Plans' retiree member, must be a District employee. In addition, the Committee will, when evaluating candidates for appointment to Committee membership as the Plans' retiree member, consider, among other things, the following:

- (a) Recommendations from District executives;
- (b) The candidate's availability to regularly attend meetings;

- (c) The candidate's experience, training, or education working with retirement plans;
- (d) The candidate's experience, training, or education with regard to the management of investments; and
- (e) The length of the candidate's relationship with the District and the likelihood that such relationship will continue for an extended period of time.

<u>Section 4: Vacancies</u>. Upon a member's resignation, removal or ceasing to qualify for membership, the appointing authority may appoint a successor.

#### ARTICLE III. OFFICERS OF THE COMMITTEE

<u>Section 1: Appointment of Officers</u>. The Human Resources Group Manager or, in his or her absence, the Assistant General Manager, Chief Financial Officer (or Finance Group Manager as his or her designee) will serve as the Committee's Chairperson. The Chairperson will preside at all Committee meetings, will make subcommittee appointments, and perform such other duties as the Committee may delegate to him or her. The Human Resources Group Manager will be the Committee Secretary and will be responsible for the following functions:

- (a) Oversee the keeping of the minutes of the Committee;
- (b) Execute documents on behalf of the Committee;
- (c) Maintain a permanent record of all Committee minutes and activities;
- (d) Issue formal written correspondence on behalf of the Committee with other entities, groups, or persons; and
- (e) Perform such other administrative support duties as the Committee may delegate to him or her.

<u>Section 2</u>: Term of Office. The Chairperson will hold his or her office as the Chairperson for his or her term of office as Human Resources Group Manager or Assistant General Manager, Chief Financial Officer (or Finance Group Manager as his or her designee). The Secretary will hold his or her office for the term for which he or she was appointed, or until his or her earlier resignation or removal from office by the General Manager.

#### ARTICLE IV. COMMITTEE MEETINGS

<u>Section 1: Meetings</u>. The Committee will meet at least quarterly. The Chairperson shall set the dates of the regular quarterly meetings. In addition, the Committee may hold special meetings, as well, at such other times and places as it may fix.

<u>Section 2: Quorum.</u> The presence of a majority of the Committee members at a meeting constitutes a quorum, and a quorum is required before the Committee may conduct business at a meeting. The Committee may act by the majority votes cast at a meeting at which there is a

quorum. In the absence of a quorum, the Committee may meet, but may not conduct a vote or decide any issue.

<u>Section 3: Action By Written Consent Without a Meeting</u>. The majority of the full Committee may consent in writing (including e-mail) to any action required or permitted to be taken at any Committee meeting. Any such consent shall have the same force and effect as if such action were taken at a Committee meeting.

<u>Section 4: Meetings by Conference Telephone or Similar Communication Equipment.</u>
Committee members may participate in a meeting by means of conference telephone or by other interactive mediums providing for real-time participation. Participation in a meeting in this manner constitutes attendance at such meeting.

<u>Section 5: Agenda</u>. Committee members must submit items for consideration by the Committee to the Chairperson. The Chairperson will compile all agenda items and related information and present such items and information to the Committee members before the Committee meets to discuss them.

#### ARTICLE V. COMMITTEE POWERS AND DUTIES

<u>Section 1: Investment Powers and Duties.</u> The Committee will be responsible for and perform any duties with respect to the management of the Plans' assets, and will have all of the discretionary authority necessary to perform those duties, including, but not be limited to, all of the following:

- (a) To establish, maintain, periodically review and approve any amendments the Committee deems necessary or advisable to the Plans' Investment Policy Statement ("IPS"). (i) The purpose of the IPS is to define the Plans' investment goals and objectives, describe the criteria and procedures for initially selecting investment managers, establish measurement procedures and monitor criteria for ongoing evaluation of investment managers, describe ways to address investment managers who fail to satisfy the established criteria, provide appropriate diversification within and among various investment options, and define how any default investment alternatives are selected and when they will be utilized. (ii) The Committee will follow all policies and procedures set forth in the IPS. (iii) In case of any conflict between the terms of these Bylaws and the terms of the IPS, the terms of the IPS will govern.
- (b) To select, periodically evaluate the financial performance of and the reasonableness of fees charged by, and approve any changes to the Plan's investment options in accordance with the IPS.
- (c) To establish and maintain proxy voting policies, cast proxy votes, and direct the Plan Administrator to execute such proxy votes cast.

- (d) To determine investment-related Plan design features.
- (e) To review and approve service provider fees and expenses for appropriateness and reasonableness at least annually.
- (f) To select, retain and monitor investment advisors for the purpose of obtaining second opinions, and to pay such persons' reasonable expenses and reasonable compensation for services from the Plans' trust.

<u>Section 2</u>: <u>Administrative Powers and Duties</u>. The Plan Administrator is the Plan fiduciary with the primary authority and responsibility for the administration of the Plans. However, in addition to the investment powers and duties enumerated in the preceding sections, the Committee will, subject to the Plan Administrator's ultimate authority and responsibility for Plan administration, have the following powers and duties for the administration of the Plans:

- (a) To provide input to the Plan Administrator regarding non-investment related plan features.
- (b) To review and provide input on the selection, performance, and retention of the record keeper, investment advisors, service providers, advisors, consultants, and auditors when requested by the Plan Administrator.
- (c) To review and provide input on plan amendments when requested by the Plan Administrator.

#### Section 3: General Powers and Duties.

- (a) To establish and maintain such policies and procedures as the Committee determines, are necessary or appropriate for it to carry out its duties hereunder or to otherwise ensure the proper investment of the Plans' assets including, but not limited to, procedures governing plan investments.
- (b) To take all other actions requested or directed by the Board in the furtherance of the duties and responsibilities hereunder, consistent with these Bylaws and governing law as the Board deem necessary or appropriate.

#### ARTICLE VI. FIDUCIARY DUTY AND CONFLICTS OF INTEREST

<u>Section 1: Fiduciary Duties.</u> The members of the Committee are fiduciaries under Article XVI, Section 17 of the California Constitution. As such, the Committee members must discharge their duties with respect to the Plans:

- (a) Solely in the interest of, and for the exclusive purposes of providing benefits to, the Plans' participants and their beneficiaries and defraying reasonable expenses of administering the Plans;
- (b) With the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person acting in a like capacity and familiar with these

- matters would use in the conduct of an enterprise of a like character and with like aims; and
- (c) By diversifying the investments of the Plans so as to minimize the risk of loss and to maximize the rate of return, unless under the circumstances it is clearly not prudent to do so.

<u>Section 2</u>: <u>Statement of Economic Interests</u>. Committee members are required to file a Statement of Economic Interests pursuant to the Political Reform Act.

#### ARTICLE VII. COMPENSATION AND EXPENSES

<u>Section 1: Compensation</u>. Committee members will not receive compensation directly or indirectly from the Plans' trust for their services rendered with respect to the Plans under these Bylaws.

<u>Section 2: Expenses.</u> Reasonable and direct expenses properly and actually incurred by Committee members and their delegates in connection with the performance of their duties with respect to the Plans will be paid from the Plans' trust if not otherwise reimbursed by the District.

#### ARTICLE VIII. MISCELLANEOUS

<u>Section 1: Indemnification</u>. The District will indemnify, defend, and hold harmless the Committee members with respect to any liability, loss, damage, or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacity as Committee members, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.

<u>Section 2</u>: <u>Amendments</u>. The Committee may, with Board approval, amend these Bylaws from time to time as it deems appropriate.

#### ARTICLE IX. ACKNOWLEDGMENT OF APPROVAL

<b>O</b> ,	mpensation Committee Secretary, does hereby acknowly approved by the Committee on	acknowledge that, 2024.
Date:	By:	
	Katano Kassaine Assistant General Man	nager/Chief
	Financial Officer and Interim Plan Administr	

#### THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA

#### TRUST AGREEMENT

Effective November 18, 1997, this trust agreement is made and entered into by and between The Metropolitan Water District of Southern California (hereinafter referred to as the "District"), and Roger N. Marumoto, the Trustee. This Trust is established to hold the assets of the Metropolitan Water District of Southern California 1997 Deferred Compensation Plan (the "Plan"). This Trust forms a part of such Plan.

#### WITNESSETH

WHEREAS, the District has established the Metropolitan Water District of Southern California 1997 Deferred Compensation Plan; and

WHEREAS, said Plan provides for the establishment of one or more trusts to carry out its purposes by the creation of a fund to which contributions paid under the Plan will be paid and from which benefits under such Plan will be paid; and

WHEREAS, the parties hereto desire to execute a trust agreement which sets forth the rights and duties of the Trustee and the terms and conditions under which said fund is to be established and administered;

NOW, THEREFORE, it is mutually understood and agreed as follows:

#### ARTICLE I ACCEPTANCE OF TRUST

#### 1.1 - Acceptance of Trust by Trustee.

The Trustee hereby accepts the Trust created hereunder and agrees to perform the obligations imposed by this Trust Agreement.

#### ARTICLE II DEFINITIONS

#### 2.1 - Definitions.

Unless the context or subject matter otherwise requires, all terms herein shall be defined in accordance with the Metropolitan Water District of Southern California 1997 Deferred Compensation Plan.

#7273

1

#### ARTICLE IV CONTRIBUTIONS

#### 4.1 - Contributions.

The Trustee shall receive all contributions paid under the Plan in cash or other property acceptable to the Trustee. All contributions so received, together with the income therefrom and any increment thereon, shall be held, managed and administered by the Trustee as a single Trust pursuant to the terms of the Plan without distinction between principal and income. The Trustee shall have no duty to require any contributions to be made to it by the District and shall have no authority to compute any amount to be paid to it by the District nor to determine whether the amounts received by it from the District comply with the Plan or to determine that the assets of the Trust comply with the Plan or to determine that the assets of the Trust are adequate to provide any benefits payable pursuant to the Plan.

#### ARTICLE V PAYMENTS FROM TRUST FUND

#### 5.1 - Payments by Trustee to Participants and Beneficiaries.

All distributions from the Trust shall be made by the Trustee to such persons, in such manner, at such times and in such amounts as the General Manager shall direct and the Trustee shall be under no duty to make inquiry as to whether any distribution directed by the General Manager is made pursuant to the provisions of the Plan. In the event that the General Manager directs that any payment be made only during or until the time a certain condition exists, it shall be the responsibility of the General Manager to notify the Trustee in writing as to the future existence or nonexistence of such condition and any payments made by the Trustee prior to such notification shall as to the Trustee proper.

#### 5.2 - Trustee Compensation and Expenses.

The Trustee shall be paid no additional compensation for his services as Trustee. The Trustee shall be reimbursed for his expenses that are reasonably necessary and incident to his administration of the Trust. Such expenses shall include counsel fees, if any, incurred by the Trustee for the purpose of determining his responsibilities under the Trust. All such amounts shall be paid to the Trustee from the Trust unless such expenses are paid by the District. The District shall indemnify the Trustee, to the fullest extent permitted by law, against any liability or expense resulting

3

#### ARTICLE III ACTIONS OF GENERAL MANAGER

#### 3.1 - General Manager

The General Manager shall administer the Plan. The General Manager may appoint an Investment Manager to direct the investment of Plan assets. If no Investment Manager is appointed, the Trustee shall have sole discretion with respect to investing Plan assets.

#### 3.2 - General Manager Directions to Trustee.

All directions by the General Manager to the Trustee shall be in writing and signed by the General Manager.

#### 3.3 - General Manager Sole Responsibility.

The General Manager shall have sole responsibility for the exercise of his rights and duties as set forth in the Plan, specifically including the determination of the existence, nonexistence, nature and amount of rights and interests of all persons in the Trust.

#### 3.4 - Maintenance of Records.

The General Manager shall have the duty and responsibility to maintain the individual Participant records and to prepare and file all reports and other information required by any federal or state law or regulation relating to the Plan, except for such reports or information returns as are required by law to be prepared and filed by the Trustee.

#### 3.5 - Designation of Agents.

The General Manager shall in his sole discretion have the right to appoint such agents as he may deem necessary to carry out his duties pursuant to the provisions of the Plan and this Trust. The General Manager may notify the Trustee in writing that one or more delegates have been appointed by him to carry out all or a portion of his duties under this Trust. Upon such notification, the Trustee shall be entitled to rely upon actions of such delegate (to the extent of the duties delegated) to the same extent as if the actions were performed directly by the General Manager.

W7273

2

from his service as Trustee, except for liability or expense incurred by reason of his own willful misconduct.

#### 5.3 - Taxes.

The Trustee is authorized, with or without direction from the General Manager, to pay from the Trust assets all real and personal property taxes, income taxes and other taxes of any kind levied or assessed under existing or future laws against the Trust assets. The Trustee shall not be personally liable for any such taxes.

#### 5.4 - Alienation.

Other than as expressly permitted by the Plan, the benefits, proceeds, payments or claims of any Participant or Beneficiary of the Plan payable from the Trust assets shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, charge, garnishment, execution or levy of any kind, either voluntary or involuntary. Any attempt to anticipate, alienate sell, transfer, assign, pledge, encumber, garnish, levy or otherwise dispose of or execute upon any right or benefit payable hereunder shall be void. The Trust assets shall not in any manner be liable for or subject to the debts, contracts, liabilities, engagements or torts of any Participant entitled to benefits hereunder and such benefits shall not be considered an asset of the Participant in the event of his or her insolvency or bankruptcy.

#### 5.5 - General Manager Expenses.

No expenses or fees of the General Manager or the District for the administration of the Plan and services in relation thereto for legal and accounting and other similar expenses shall be paid from the Trust assets and all such expenses shall be paid by the District, unless the District determines that any or all of such expenses will be paid by the Trust and such payment does not violate any law.

#### ARTICLE VI INVESTMENT OF TRUST ASSETS

#### 6.1 - Fiduciary Responsibilities

Unless an Investment Manager is appointed by the General Manager, the Trustee shall have full discretion in the exercise of investment authority with regard to the Trust assets and earnings thereon.

#### 6.2 - Relationship of Parties.

It is the intent of the District, Trustee, General Manager, and any other person or persons designation under the provisions of Article III that each shall be solely responsible for its own acts or omissions. Except to the extent imposed by the Code, no such party shall have the duty to question whether any other party is fulfilling any or all of the responsibilities imposed upon it by the Plan or the Trust. No such party shall have any liability for a breach of responsibility of another party with respect to the Plan and this Trust.

#### ARTICLE VII POWERS OF TRUSTEE

#### 7.1 - Powers of Trustee.

Except as otherwise provided by law, the Trustee is authorized and empowered, but not by way of limitation except as specifically set forth:

- (a) To purchase, hold, sell, invest and reinvest assets of the Trust, together with the income therefrom, in stock, (including common, preferred, and convertible preferred stock), evidences of indebtedness (including debentures, convertible debentures and bonds), mortgages, notes (including loans to Participants to the extent authorized under the Plan), mutual fund shares, interests in unit trusts, interests in partnerships, interests in money market funds, time certificates of deposit, commercial paper and other evidences of indebtedness), other securities, and property (personal, real or mixed, and tangible or intangible) the value of which are readily ascertainable and seadily mentaetable.
- readily marketable;
  (b) To retain in eash or other property unproductive of income, without liability for interest, so much of the Trust assets as may be determined; to deposit cash in any bank and select any bank as custodian; and to cause securities or other property to be registered and / or held in its individual name, either with or without disclosing fiduciary capacity, or in the name of its nominee:
- (c) To manage, control, sell, convey, exchange, partition, divide, subdivide, improve, and repair; to grant options and to sell upon deferred payments; to lease for terms within or extending beyond the duration of this Trust for any purpose, including exploration for and removal of gas or other minerals; to enter

#7273

Trust, showing carrying and market values of such assets, at the close of the period covered by such account; and (2) such information which is within the control of the Trustee, and a certification by the Trustee of the accuracy thereof, as shall be necessary or appropriate for the General Manager to complete all required disclosures and reporting with respect to the Plan and Trust.

#### 8.3 - Value of Trust Assets.

For the purposes of this Article, the fair market value of assets in the Trust shall be determined by the Trustee or his delegate based upon such sources of information as it may deem reliable including, but not limited to, information reported in (1) newspapers of general circulation, (2) standard financial periodicals or publications, (3) statistical and valuation services, (4) the records of securities exchanges or brokerage firms deemed by the Trustee to be reliable, (5) independent valuation, or any combination thereof.

#### ARTICLE IX RESIGNATION AND REMOVAL OF TRUSTEE

#### 9.1 - Method and Procedure.

The Trustee may resign at any time by delivering to the General Manager a written notice or resignation, to take effect at a date specified therein, which shall not be less than 30 days after the delivery thereof, unless such notice shall be waived.

The General Manager shall have the right to remove the Trustee by delivery of a written notice of removal to take effect at a specified time therein, which shall not be less than 30 days after the delivery thereof, unless such notice shall be waived.

In the event the Trustee notifies the General Manager of his intention to resign, or the General Manager removes the Trustee, in accordance with the foregoing provisions of this Article IX, the General Manager shall appoint a successor trustee and in default thereof, such successor trustee may be appointed by a court of competent jurisdiction. The Trustee hereunder shall thereupon deliver to the successor trustee all property of this Trust, less such amounts as said Trustee shall deem necessary to provide for any expenses, compensation and taxes or advances, together with such records as may be reasonably required to enable the successor trustee to properly administer the Trust.

In the case of his resignation or removal, the Trustee shall have the right to a settlement of his account, which may be made, at the option of the Trustee, either (1) by a judicial statement in an action instituted by the Trustee in a court of

- into community oil leases; to create restrictions, easements and other servitudes;
- (d) To compromise, arbitrate or otherwise adjust claims in favor of or against the Trust; and to institute, compromise and defend actions and proceedings;
   (e) To transfer part or all of the money constituting assets of the
- (e) To transfer part or all of the money constituting assets of the Trust to and as an investment in any type of interest-bearing account, including those evidenced by time certificates of deposit;
- To appoint agents to carry out those Trustee responsibilities that can be efficiently carried out through an agent; and
   Generally to do all such acts, execute all such instruments, take
- (g) Generally to do all such acts, execute all such instruments, take all such proceedings, and exercise all such rights and privileges with relation to property constituting the Trust assets as if the Trustee were the absolute owner thereof.

#### 7.2 - Controversy or Disagreement

Except as otherwise provided by law, if any controversy arises with respect to this Trust, the Trustee shall take such action as directed by the General Manager or, in the absence of such directions, as it deems advisable, whether by legal proceedings, compromise or otherwise. The Trustee may retain the funds or property involved without liability pending settlement of the controversy.

#### ARTICLE VIII ACCOUNTS OF THE TRUSTEE

#### 8.1 - Records.

The Trustee or his delegate shall maintain accurate records and accounts of all transactions hereunder, which shall be available at all reasonable times for inspection or audit by any person or persons designated by the General Manager.

#### 8.2 - Report

If the General Manager so directs, the Trustee or his delegate shall submit to the General Manager such interim valuations, reports or other information as the General Manager may reasonably require. Within 90 days following (a) the close of each Plan Year or (b) the effective date of the removal or resignation of the Trustee, the Trustee or his delegate shall file with the General Manager (1) a written account settling forth all transactions effected by it subsequent to the end of the period covered by its last previous account, and listing the assets of the

73

competent jurisdiction, or (2) by agreement of settlement between the Trustee and the General Manager.

Upon such settlement, all right, title and interest of such Trustee in the assets of the Trust and all rights and privileges under the Plan and this Trust theretofore vested in such Trustee shall vest in the successor trustee, and thereupon all future liability of such Trustee shall terminate; provided, however, that the Trustee shall execute, acknowledge and deliver all documents and written instruments which are necessary to transfer and convey the right, title and interest in the Trust assets, and all rights and privileges, to the successor trustee.

#### 9.2 - Successor Trustee.

In the event the Trustee notifies the General Manager of his intention to resign or the General Manager removes the Trustee, unless the Trust assets attributable to the Participants employed by the District are distributed to or held by another trust qualifying under the applicable provisions of Sections 457 and 501(a) of the Code, the District shall terminate the Plan and the General Manager shall terminate this Trust in accordance with the provisions of the Plan.

#### ARTICLE X AMENDMENT AND TERMINATION

#### 10.1 - Amendment and Termination.

The General Manager shall have the right at any time and from time to time (1) to modify or amend in whole or in part, by written agreement with the Trustee, any or all of the provisions of this Trust Agreement, or (2) to terminate this Trust Agreement upon 30 days prior notice in writing to the Trustee; and the General Manager hereby covenants and agrees with the Trustee as follows:

- (a) No such termination, modification, or amendment shall have any retroactive effect so as to deprive any Participant of any benefit already vested, except such changes, if any, as may be required to permit this Trust Agreement to meet the applicable requirements of the Code;
- (b) No such termination, modification, or amendment shall permit, prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries, any part of the principal or income of this Trust to be used for, or diverted to, purposes other than the exclusive benefit of Participants and their Beneficiaries. In the event of termination of this Trust, all cash, securities and other property then constituting the assets of the Trust, less any amounts constituting charges and

8

#727

#7273

expenses payable from the assets of the Trust, shall be paid over or delivered by the Trustee to or on order of the General Manager

#### ARTICLE XI MISCELLANEOUS

#### 11.1 - Contribution Not Recoverable.

It shall be impossible for any part of the principal or income of this Trust to be used for, or diverted to, purposes other than the exclusive benefit of Participants or their Beneficiaries; provided, however, that notwithstanding this or any other provision of this Trust, if the District Director of Internal Revenue determines that the Plan does not initially meet the requirements of the Code and Regulations, the District shall be entitled to withdraw all contributions theretofore made, in which event this Trust shall thereupon terminate and all right, if any, of the Participants thereunder, shall cease and come to an end with the same effect as if the Trust had not been adopted.

#### 11.2 - Limitation on Participants' Rights.

Participation in this Trust shall not give any employee the right to be retained in the District's employ or any right to interest in this Trust other than as herein provided. The District reserves the right to dismiss any employee without any liability for any claim either against this Trust, except to the extent provided herein, or against the District. All benefits payable hereunder shall be provided solely from the assets of the Trust.

#### 11.3 - Receipt or Release

Any payment to any Participant or his Beneficiary in accordance with the provisions of this Trust shall, to the extent thereof, be in full satisfaction of all claims against the Trustee, the General Manager and the District and the Trustee may require such Participant or Beneficiary, as a condition precedent to such payment, to execute a receipt and release to such effect.

#### 11.4 - Governing Law.

This Trust Agreement and the Trust hereby created shall be construed. administered and governed in all respects under the laws of the State of California, except to the extent federal law controls; provided, however, that if any provision is susceptible to more than one interpretation, such interpretation shall be given thereto as is consistent with the Trust being a

IN WITNESS WHEREOF, the undersigned have executed this Trust Agreement effective as of the later day set forth below

Approved as to Form:

N. Gregory Taylor General Counsel

Sr. Deputy General Counsel

THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA

EXECUTED November 19, 1997

General Manager

TRUSTEE:

EXECUTED November 19, 1997
BY Roger Marunoto

qualified trust within the meaning of Sections 457 of the Code. If any provision of this instrument shall be held by a court of competent jurisdiction to be invalid or unenforceable, the remaining provisions hereof shall continue to be fully effective.

#### 11.5 - Headings, etc., Not Part of Agreement.

Headings and subheadings in this Trust Agreement are inserted for convenience of reference only and are not to be considered in the construction of the provisions hereof.

#### 11.6 - Instrument in Counterparts.

This Trust Agreement has been executed in several counterparts, each of which shall be deemed an original and said counterparts shall constitute but one and the same instrument, which may be sufficiently evidenced by any one counterpart.

#### 11.7 - Successors and Assigns.

#7273

This Trust Agreement shall inure to the benefit of and be binding upon the parties hereto and their successors and assigns

#### 11.8 - Masculine Gender Includes Feminine and Neuter.

10

As used in this Trust Agreement, the masculine gender shall include the feminine and neuter genders.

11

# 2nd Amdmt to 1985 Savings Plan II Trust Agmt – Att A.

#### **SECOND AMENDMENT TO**

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA SAVINGS PLAN II TRUST AGREEMENT

The Metropolitan Water District of Southern California Savings Plan II Trust Agreement ("Agreement")(Attachment A) is hereby amended in the following respects:

1. Subsection 2.1 of Article II – <u>Definitions</u> of the Agreement is hereby amended to read as follows:

#### Strikeout/Underline Version:

#### 2.1 – Definitions

Unless the context or subject matter otherwise requires, all terms herein shall be defined in accordance with The Metropolitan Water District of Southern California Savings Plan II.

"Trustee" shall mean any person or joint committee of persons appointed by the General Manager to hold in trust and administer some or all of the assets of this Savings Plan.

#### **Final Version:**

#### 2.1 – Definitions

Unless the context or subject matter otherwise requires, all terms herein shall be defined in accordance with The Metropolitan Water District of Southern California Savings Plan II.

"Trustee" shall mean any person or joint committee of persons appointed by the General Manager to hold in trust and administer some or all of the assets of this Savings Plan.

THE METROPOLITAN W SOUTHERN CALIFORNIA		Approved	as to Form:	
		Marcia L. S	cully	
		General Co	ounsel	
By:		By: Ma	Carmain	Grana
Jeffrey k	Cightlinger	Marianne Greene Senior Deputy General Counsel		
General	Manager			
Executed:	. 2017	Date:	6/29	. 2017



## The Metropolitan Water District of Southern California

November 27, 1985

To: Acting Controller David I. Furukawa

From: Deputy General Counsel James W. Mountain

Subject: Savings Plan II Trust Agreement

Enclosed for your custody is the original Trust

Agreement for the MWD Savings Plan II executed by the General

Manager and Richard D. Sjoberg, Trustee.

James W. Mountain

JWM:gld Encl.

cc w/encl.:

R. D. Sjoberg (2)

G. D. Leddy

P. M. Moyer

cc w/o encl.:

W. J. Abbott

RECEIVED NOV 27 1985

CONTROLLER'S OFFICE
THE METROPOLITAN WATER DISTRICT
OF SOUTHERN CALIFORNIA

# METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA SAVINGS PLAN II TRUST AGREEMENT

#### METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA SAVINGS PLAN II

**INDEX** 

			<u>Page</u>
١.	ACC	EPTANCE OF TRUST	
	1.1	Acceptance of Trust by Trustee	1
п.	DEF	INITIONS	
	2.1	Definitions	1
111.	ACT	IONS OF GENERAL MANAGER	
	3.1	General Manager	1
	3.2	General Manager Delegates Directions to Trustee	2
	3.3	General Manager Sole Responsibility	2
	3.4	Maintenance of Records	2
	3.5	Designation of Agents	2
IV.	CONTRI	BUTIONS	
	4.1	Contributions	2
V.	PAYMEN	TS FROM TRUST FUND	
	5.1	Payments by Trustee to Participants and Beneficiaries	3
	5.2	Trustee Compensation and Expenses	3
	5.3	Taxes	3
	5.4	Alienation	3
	5.5	General Manager Expenses	3

			<u>Page</u>
VI.	INVESTM	MENT OF TRUST ASSETS	
	6.1	Responsibilities	4
	6.2	Relationship of Parties	4
VII.	POWERS	OF TRUSTEE	
	7.1	Powers of Trustee	4
	7.2	Funds	5
	7.3	Controversy or Disagreement	5
VIII.	ACCOUN	TS OF THE TRUSTEE	
	8.1	Records	5
	8.2	Reports	6
	8.3	Value of Trust Assets	6
ıx.	RESIGNA	ATION AND REMOVAL OF TRUSTEE	
	9.1	Method and Procedure	6
	9.2	Successor Trustee	7
x.	AMENDM	ENT AND TERMINATION	
	10.1	Amendment and Termination	7
XI.	MISCELL	ANEOUS	
	11.1	Contribution Not Recoverable	8
	11.2	Limitation on Participants' Rights	8
	11.3	Receipt or Release	8
	11.4	Governing Law	8
	11.5	Headings, Etc., Not Part of Agreement	8

	<u>Page</u>
11.6 Instrument in Counterparts	9
11.7 Successors and Assigns	9
11.8 Masculine Gender Includes	9

#### **SIGNATURES**

## METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA

Effective October 1, 1985, this trust agreement is made and entered into by and between The Metropolitan Water District of Southern California, (hereinafter referred to as the "District"), and Richard D. Sjoberg, as Trustee. This Trust is established to hold the assets of the Metropolitan Water District of Southern California Savings Plan II (the "Plan"). This Trust forms a part of such Plan.

#### WITNESSETH:

WHEREAS, the District has established the Metropolitan Water District of Southern California Savings Plan II; and

WHEREAS, said Plan provides for the establishment of one or more trusts to carry out its purposes by the creation of a fund to which contributions paid under the Plan will be paid and from which benefits under such Plan will be paid; and

WHEREAS, the parties hereto desire to execute a trust agreement which sets forth the rights and duties of the Trustee and the terms and conditions under which said fund is to be established and administered;

NOW, THEREFORE, it is mutually understood and agreed as follows:

## ARTICLE I ACCEPTANCE OF TRUST

#### 1.1 - Acceptance of Trust by Trustee.

The Trustee hereby accepts the Trust created hereunder and agrees to perform the obligations imposed by this Trust Agreement.

### ARTICLE II DEFINITIONS

#### 2.1 - Definitions.

Unless the context or subject matter otherwise requires, all terms herein shall be defined in accordance with the Metropolitan Water District of Southern California Savings Plan II.

## ARTICLE III ACTIONS OF GENERAL MANAGER

#### 3.1 - General Manager.

The General Manager shall administer the Plan. The General Manager, in conjunction with any Investment Manager selected by the General Manager, shall have sole discretion with respect to choosing the

Investment Vehicles in which Participants may cause their Tax Deferred Election Contributions to be invested.

#### 3.2 General Manager Directions to Trustee.

All directions by the General Manager to the Trustee shall be in writing and signed by the General Manager.

#### 3.3 - General Manager Sole Responsibility.

The General Manager shall have sole responsibility for the exercise of his rights and duties as set forth in the Plan, specifically including the determination of the existence, nonexistence, nature and amount of rights and interests of all persons in the Trust.

#### 3.4 - Maintenance of Records.

The General Manager shall have the duty and responsibility to maintain the individual Participant records and to prepare and file all reports and other information required by any federal or state law or regulation relating to the Plan, except for such reports or information returns as are required by law to be prepared and filed by the Trustee.

#### 3.5 - Designation of Agents.

The General Manager shall in his sole discretion have the right to appoint such agents as it may deem necessary to carry out his duties pursuant to the provisions of the Plan and this Trust. The General Manager may notify the Trustee in writing that one or more delegates have been appointed by him to carry out all or a portion of his duties under this Trust. Upon such notification the Trustee shall be entitled to rely upon actions of such delegate (to the extent of the duties delegated) to the same extent as if the actions were performed directly by the General Manager.

## ARTICLE IV CONTRIBUTIONS

#### 4.1 - Contributions.

The Trustee shall receive all contributions paid under the Plan in cash or other property acceptable to the Trustee. All contributions so received, together with the income therefrom and any increment thereon, shall be held, managed and administered by the Trustee as a single Trust pursuant to the terms of the Plan without distinction between principal and income. The Trustee shall have no duty to require any contributions to be made to it by the District and shall have no authority to compute any amount to be paid to it by the District nor to determine whether the amounts received by it from the District comply with the Plan or to determine that the assets of the Trust are adequate to provide any benefits payable pursuant to the Plan.

## ARTICLE V PAYMENTS FROM TRUST FUND

#### 5.1 - Payments by Trustee to Participants and Beneficiaries.

All distributions from the Trust shall be made by the Trustee to such persons, in such manner, at such times and in such amounts as the General Manager shall direct and the Trustee shall be under no duty to make inquiry as to whether any distribution directed by the General Manager is made pursuant to the provisions of the Plan. In the event that the General Manager directs that any payment be made only during or until the time a certain condition exists, it shall be the responsibility of the General Manager to notify the Trustee in writing as to the future existence or non-existence of such condition and any payments made by the Trustee prior to such notification shall as to the Trustee be proper.

#### 5.2 - Trustee Compensation and Expenses.

The Trustee shall be paid no additional compensation for his services as Trustee. The Trustee shall be reimbursed for his expenses that are reasonably necessary and incident to its administration of the Trust. Such expenses shall include counsel fees, if any, incurred by the Trustee for the purpose of determining his responsibilities under the Trust. All such amounts shall be paid to the Trustee from the Trust unless such expenses are paid by the District.

#### 5.3 - Taxes.

The Trustee is authorized, with or without direction from the General Manager, to pay from the Trust assets all real and personal property taxes, income taxes and other taxes of any kind levied or assessed under the existing or future laws against the Trust assets. The Trustee shall not be personally liable for any such taxes.

#### 5.4 - Alienation.

Other than as expressly permitted by the Plan, the benefits, proceeds, payments or claims of any Participant or Beneficiary of the Plan payable from the Trust assets shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, charge, garnishment, execution or levy of any kind, either voluntary or involuntary. Any attempt to anticipate, alienate, sell, transfer, assign, pledge, encumber, garnish, levy or otherwise dispose of or execute upon any right or benefit payable hereunder shall be void. The Trust assets shall not in any manner be liable for or subject to the debts, contracts, liabilities, engagements or torts of any Participant entitled to benefits hereunder and such benefits shall not be considered an asset of the Participant in the event of his insolvency or bankruptcy.

#### 5.5 - General Manager Expenses.

No expenses or fees of the General Manager or the District for the administration of the Plan and services in relation thereto for legal and accounting and other similar expenses shall be paid from the Trust assets and all such expenses shall be paid by the District, unless the District determines that any or all of such expenses will be paid by the Trust and such payment does not violate any law.

#### ARTICLE VI INVESTMENT OF TRUST ASSETS

#### 6.1 - Fiduciary Responsibilities.

The Trustee shall have no discretion in the exercise of investment authority with regard to the Trust assets and earnings thereon. Individual participants in the Plan shall have the authority and responsibility to choose from the Investment Vehicles which are made available by the District those Investment Vehicles into which their Tax Deferred Elective Contributions shall be invested. The Trustee's investment responsibility is limited to owning the Trust assets.

#### 6.2 - Relationship of Parties.

It is the intent of the District, Trustee, General Manager, and any other person or persons designated under the provisions of Article III that each shall be solely responsible for its own acts or omissions. Except to the extent imposed by the Code, no such party shall have the duty to question whether any other party is fulfilling any or all of the responsibilities imposed upon it by the Plan or the Trust. No such party shall have any liability for a breach of responsibility of another party with respect to the Plan and this Trust.

## ARTICLE VII POWERS OF TRUSTEE

#### 7.1 - Powers of Trustee.

Except as otherwise provided by law, the Trustee is authorized and empowered, but not by way of limitation except as specifically set forth:

- (a) To purchase, hold, sell, invest and reinvest assets of the Trust, together with the income therefrom, in stock, (including common, preferred, and convertible preferred stock), evidences of indebtedness (including debentures, convertible debentures and bonds), mortgages, notes (including loans to Participants to the extent authorized under the Plan), mutual fund shares, interests in unit trusts, interests in partnerships, interests in money market funds, time certificates of deposit, commercial paper and other evidences of indebtedness), other securities, and property (personal, real or mixed, and tangible or intangible) the value of which are readily ascertainable and readily marketable;
- (b) To retain in cash or other property unproductive of income, without liability for interest, so much of the Trust assets as may be determined; to deposit cash in any bank and select any bank as custodian; and to cause securities or other property to be registered

and/or held in its individual name, either with or without disclosing fiduciary capacity, or in the name of its nominee;

- (c) To manage, control, sell, convey, exchange, partition, divide, subdivide, improve, and repair; to grant options and to sell upon deferred payments; to lease for terms within or extending beyond the duration of this Trust for any purpose, including exploration for and removal of gas or other minerals; to enter into community oil leases; to create restrictions, easements and other servitudes;
- (d) To compromise, arbitrate or otherwise adjust claims in favor of or against the Trust; and to institute, compromise and defend actions and proceedings;
- (e) To transfer part or all of the money constituting assets of the Trust to and as an investment in any type of interest-bearing account, including those evidenced by time certificates of deposit;
- (f) To appoint agents to carry out those Trustee responsibilities that can be efficiently carried out through an agent; and
- (g) Generally to do all such acts, execute all such instruments, take all such proceedings, and exercise all such rights and privileges with relation to property constituting the Trust assets as if the Trustee were the absolute owner thereof.

#### 7.2 - Funds.

The Trust Assets shall be held, administered, invested and reinvested among such separate Investment Vehicles as shall be made available for investment of Trust Assets in accordance with the instructions from the Participants to the Trustee or his delegate.

#### 7.3 - Controversy or Disagreement.

Except as otherwise provided by law, if any controversy arises with respect to this Trust, the Trustee shall take such action as directed by the General Manager or, in the absence of such directions, as it deems advisable, whether by legal proceedings, compromise or otherwise. The Trustee may retain the funds or property involved without liability pending settlement of the controversy.

## ARTICLE VIII ACCOUNTS OF THE TRUSTEE

#### 8.1 - Records.

The Trustee or his delegate shall maintain accurate records and accounts of all transactions hereunder, which shall be available at all reasonable times for inspection or audit by any person or persons designated by the General Manager.

#### 8.2 - Reports.

If the General Manager so directs, the Trustee or his delegate shall submit to the General Manager such interim valuations, reports or other information as the General Manager may reasonably require. Within 90 days following (a) the close of each Plan Year or (b) the effective date of the removal or resignation of the Trustee, the Trustee or his delegate shall file with the General Manager (1) a written account setting forth all transactions effected by it subsequent to the end of the period covered by its last previous account, and listing the assets of the Trust, showing carrying and market values of such assets, at the close of the period covered by such account; and (2) such information which is within the control of the Trustee, and a certification by the Trustee of the accuracy thereof, as shall be necessary or appropriate for the General Manager to complete all required disclosures and reporting with respect to the Plan and Trust.

#### 8.3 - Value of Trust Assets.

For the purposes of this Article, the fair market value of assets in the Trust shall be determined by the Trustee or his delegate based upon such sources of information as it may deem reliable including, but not limited to, information reported in (1) newspapers of general circulation, (2) standard financial periodicals or publications, (3) statistical and valuation services, (4) the records of securities exchanges or brokerage firms deemed by the Trustee to be reliable, or any combination thereof.

## ARTICLE IX RESIGNATION AND REMOVAL OF TRUSTEE

#### 9.1 - Method and Procedure.

The Trustee may resign at any time by delivering to the General Manager a written notice of resignation, to take effect at a date specified therein, which shall not be less than 30 days after the delivery thereof, unless such notice shall be waived.

The General Manager shall have the right to remove the Trustee by delivery of a written notice of removal to take effect at a specified time therein, which shall not take effect in less than 30 days after the delivery thereof, unless such notice shall be waived.

In the event the Trustee notifies the General Manager of his intention to resign, or the General Manager removes the Trustee, in accordance with the foregoing provisions of this Article IX, the General Manager shall appoint a successor trustee and in default thereof, such successor trustee may be appointed by a court of competent jurisdiction. The Trustee hereunder shall thereupon deliver to the successor trustee all property of this Trust, less such amounts as said Trustee shall deem necessary to provide for any expenses, compensation and taxes or advances, together with such records as may be reasonably required to enable the successor trustee to properly administer the Trust.

In the case of his resignation or removal, the Trustee shall have the right to a settlement of his account, which may be made, at the option of the Trustee, either (1) by a judicial statement in an action instituted by the Trustee in a court of competent jurisdiction, or (2) by agreement of settlement between the Trustee and the General Manager.

Upon such settlement, all right, title and interest of such Trustee in the assets of the Trust and all rights and privileges under the Plan and this Trust theretofore vested in such Trustee shall vest in the successor trustee, and thereupon all future liability of such Trustee shall terminate; provided, however, that the Trustee shall execute, acknowledge and deliver all documents and written instruments which are necessary to transfer and convey the right, title and interest in the Trust assets, and all rights and privileges, to the successor trustee.

#### 9.2 - Successor Trustee.

In the event the Trustee notifies the General Manager of his intention to resign or the General Manager removes the Trustee, unless the Trust assets attributable to the Participants employed by the District are distributed to or held by another trust qualifying under the applicable provisions of Sections 401 and 501 of the Code, the District shall terminate the Plan and the General Manager shall terminate this Trust in accordance with the provisions of the Plan.

## ARTICLE X AMENDMENT AND TERMINATION

#### 10.1 - Amendment and Termination.

The General Manager shall have the right at any time and from time to time (1) to modify or amend in whole or in part, by written agreement with the Trustee, any or all of the provisions of this Trust Agreement, or (2) to terminate this Trust Agreement upon 30 days prior notice in writing to the Trustee; and the General Manager hereby covenants and agrees with the Trustee as follows:

- (a) No such termination, modification, or amendment shall have any retroactive effect so as to deprive any Participant of any benefit already vested, except such changes, if any, as may be required to permit this Trust Agreement to meet the applicable requirements of the Code; and
- (b) No such termination, modification, or amendment shall permit, prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries, any part of the principal or income of this Trust to be used for, or diverted to, purposes other than the exclusive benefit of Participants and their Beneficiaries. In the event of termination of this Trust, all cash, securities and other property then constituting the assets of the Trust, less any amounts constituting charges and expenses payable from the assets of the Trust, shall be paid over or delivered by the Trustee to or on order of the General Manager.

## ARTICLE XI MISCELLANEOUS

#### 11.1 - Contribution Not Recoverable.

It shall be impossible for any part of the principal or income of this Trust to be used for, or diverted to, purposes other than the exclusive benefit of Participants or their Beneficiaries; provided, however, that notwithstanding this or any other provision of this Trust, if the District Director of Internal Revenue determines that the Plan and this Trust do not initially meet the requirements of the Code and Regulations, the District shall be entitled to withdraw all contributions theretofore made, in which event this Trust and the Plan associated therewith shall thereupon terminate and all rights, if any, of the Participants thereunder, shall cease and come to an end with the same effect as if the Plan and Trust had not been adopted.

#### 11.2 - Limitation on Participants' Rights.

Participation in this Trust shall not give any employee the right to be retained in the District's employ or any right to interest in this Trust other than as herein provided. The District reserves the right to dismiss any employee without any liability for any claim either against this Trust, except to the extent provided herein, or against the District. All benefits payable hereunder shall be provided solely from the assets of the Trust.

#### 11.3 - Receipt or Release.

Any payment to any Participant or his Beneficiary in accordance with the provisions of this Trust shall, to the extent thereof, be in full satisfaction of all claims against the Trustee, the General Manager and the District and the Trustee may require such Participant or Beneficiary, as a condition precedent to such payment, to execute a receipt and release to such effect.

#### 11.4 - Governing Law.

This Trust Agreement and the Trust hereby created shall be construed, administered and governed in all respects under the laws of the State of California, except to the extent federal law controls; provided, however, that if any provision is susceptible to more than one interpretation, such interpretation shall be given thereto as is consistent with the Trust being a qualified trust within the meaning of Sections 401 and 501 of the Code. If any provision of this instrument shall be held by a court of competent jurisdiction to be invalid or unenforceable, the remaining provisions hereof shall continue to be fully effective.

#### 11.5 - Headings, etc., Not Part of Agreement.

Headings and subheadings in this Trust Agreement are inserted for convenience of reference only and are not to be considered in the construction of the provisions hereof.

#### 11.6 - <u>Instrument in Counterparts</u>.

This Trust Agreement has been executed in several counterparts, each of which shall be deemed an original and said counterparts shall constitute but one and the same instrument, which may be sufficiently evidenced by any one counterpart.

#### 11.7 - Successors and Assigns.

This Trust Agreement shall inure to the benefit of and be binding upon the parties hereto and their successors and assigns.

#### 11.8 - Masculine Gender Includes Feminine and Neuter.

As used in this Trust Agreement, the masculine gender shall include the feminine and neuter genders.

IN WITNESS WHEREOF, the undersigned have executed this Trust Agreement effective as of the later day set forth below.

THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA

Approved as to Form:

Warren J. Abbott General Counsel

By Deputy General Counsel

**EXECUTED** 

General Manager

TRUSTEE

EXECUTEL

DV

# The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan

<u>January 1, 2024</u>

## THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 457(bb) DEFERRED COMPENSATION PLAN

#### **ABBREVIATED**

#### **TABLE OF CONTENTS**

BACKGROUND 1	L
ARTICLE I: DEFINITIONS 1	L
ARTICLE II: ELIGIBILITY AND PARTICIPATION 7	7
ARTICLE III: DEFERRAL CONTRIBUTIONS AND LIMITATIONS 8	}
ARTICLE IV: TIME AND METHOD OF PAYMENT OF BENEFITS 1	4
ARTICLE V: PLAN ADMINISTRAT <u>IONOR—DUTIES WITH RESPECT TO PARTICIPANTS' ACCOUNTS</u>	5
25	
ARTICLE VI: BENEFICIARY DESIGNATIONS AND ENROLLMENT 3	80
ARTICLE VII: TRUST PROVISIONS TRUST FUNDS 3	32
ARTICLE VIII: AMENDMENT, TERMINATION AND TRANSFERS 3	87
ARTICLE IX: MISCELLANEOUS 3	88

APPENDIX A: SECOND AMENDMENT TO THE METROPOLITAN WATER DISTRICT OF SOUTHERN-CALIFORNIA

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 457(bb) DEFERRED COMPENSATION PLAN

# DETAILED TABLE OF CONTENTS

		Page(s)
BACK	GROUND	1
ARTIC	CLE I. DEFINITIONS	1
1.1	Account	
1.2	Accounting Date	
1.3	Beneficiary	
1.4	Code	
1.5	Compensation	
1.0	(A) Compensation Taken Into Account	
	(B) Elective Contributions	
	(C) Post-Severance Compensation	
	(1) Timing	
	(a) Regular Pay	
	(b) Leave Cash-outs	
	(D) Salary Continuation for Disabled Participants	
	(E) Differential Wage Payments	
1.6	Deferred Compensation	
1.7	Deferred Compensation InvestmentAdvisory Committee (DCIAC)	
1.8	Effective Date	
1.9	Eligible Employee	4
1.10	Employee	
1.11	Employer or District	
1.12	Excess Deferrals	
1.13	Excluded Employee	
1.14	Includible Compensation	
1.15	Independent Contractor	
1.16	Leased Employee	4
1.17	Normal Retirement Age	4
1.18	Participant	4
1.19	Plan	4
1.20	Plan Administrator	5
1.21	Plan Year	5
1.22	Qualified Birth or Adoption Distribution (QBOAD)	5
1.23	Recurrent Employee	5
<u>1.24</u>	Rehired Annuitant	5
<u>1.25</u>	Rollover Contribution	
1.26	Roth Elective Deferrals	
1.27	Salary Reduction Agreement	5

<u>1.28</u>	Salary Red	uction Contribution	5
1.29	Service		6
	(A) Qua	alified Military Service	6
	(B) Cor	ntinuous Service	6
	(C) Sev	erance from Employment	6
	(1)	1 7	
	(2)		
1.30	. ,		
1.31		ar	
1.32		r Employee	
1.33			
1.34			
1.35		ement	
1.3 <del>5</del> 6			
1.000	Trustee		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
ARTIC	LE II. ELIGII	BILITY AND PARTICIPATION	7
2.1			
2.2		ment of Salary Reduction Contributions	
2.3		an Restatement	
2.4		on Upon Re-Employment	
ARTIC	LE III. DEFE	RRAL CONTRIBUTIONS AND LIMITATIONS	8
3.1			
		ntribution Formula	
	. ,	urn of Contributions	
	. ,	ne of Payment of Contribution	
3.2	. ,	uction Contributions	
J.2		erral from Sick, Vacation and Back Pay	
		plication to Leave of Absence and Disability	
	(C) Pos	t-Severance Deferrals Limited to Post-Severance Compensation	9
3.3		nitation	
3.4		ch-Up Contribution	
3.5		ch-Up Contributionch	
0.0	(1)	•	
	(2)		
	(3)	Pre-2002 Coordination	
3.6	. ,	ontributions	
5.0		erational Administration	
		parate Accounting	
		h Elective Deferrals	
3.7	Nictributio	n of Excess Deferralsn	11 11
5.7		gible Governmental 457(b) Plan	
		ividual Limitation	
3.8	Ranafit Acc	ruals for Qualified Military Service	
3.9		ve Deferrals	
<u>J.7</u>		ctive Deferrals	
	(u) File	CUYC DCICITAIS	<u></u> 12

		(1)	Pre-T	'ax Elec	tive Deferrals	<u></u> 12
		(2)			e Deferrals	
	(B)	Order	ing Rul	es for D	Distributions	12
	(C)	Loans				12
3.10	In-Pla	n Roth (	Convers	sion		12
	(A)	In-Pla	n Roth	Conver	sion Account	12
	(B)	Eligib	ility to I	Elect In	-Plan Roth Rollover	13
	(C)				er Procedure	
	(D)	Plan L	oans			13
	<u>(E)</u>	Treati	ment of	In-Plar	n Roth Conversions	<u></u> 13
	<u>(F)</u>	Spous	al Cons	ent		13
	(G)	Tax T	<u>reatmer</u>	nt of In-	-Plan Roth Conversions	<u></u> 13
	(H)	In-Pla	n Roth	Conver	sion Definitions	14
		(1)	Direc	t In-Pla	n Roth Rollover	14
		(2)	In-Pla	an Roth	Rollover	14
		(3)	In-Pla	an Roth	Rollover Account	14
ARTI(	CLE IV.	TIME AN	ND MET	O don'	F PAYMENT OF BENEFITS	14
4.1	Distri	bution F	Restricti	ions		14
4.2	Time	and Met	hod of	Paymer	nt of Account	14
	(A)	Partic	ipant E	lection	of Time and Method	14
	(B)	Auton	natic Ca	sh-Out	Distribution	15
4.3	Requi	ired Min	imum [	Distribu	itions (RMDs)	15
	(A)	Gener	al Rules	S		15
	(B)	Time a	and Ma	nner of	Distribution	15
		(1)	Requi	ired Be	ginning Date	15
		(2)	Death	ı of Part	ticipant Before Distribution Begins	15
			(a)		se Designated Beneficiary	
			(b)		Spouse Designated Beneficiary	
			(c)	No D	esignated Beneficiary	15
			(d)	Deatl	h of Spouse	16
		(3)			tribution	
	(C)	RMDs			pant's Lifetime	
		(1)	Amou	int of R	MD Each Distribution Calendar Year	16
			(a)	Unifo	orm Life Table	16
			(b)		iger Spouse	
		(2)			Ds Through Year of Participant's Death	
	(D)	RMDs		_	ant's Death	
		(1)	Death		After Distributions Begin	
			(a)	Parti	cipant Survived by Designated Beneficiary	16
				(i)	Participant's Life Expectancy	
				(ii)	Spouse's Life Expectancy	
				(iii)	Non-Spouse's Life Expectancy	
			(b)		esignated Beneficiary	
		(2)	Death		e Date Distributions Begin	
			(2)	Parti	cinant Survived by Decignated Reneficiary	17

		(b)	No Designated Beneficiary	18		
		(c)	Death of Surviving Spouse Before Distributions to Surviving			
			Spouse are Required to Begin	18		
		(d)	5-Year or Life Expectancy Rule; Possible Election			
	(E)		ons			
	( )		nated Beneficiary			
			bution Calendar Year			
		. ,	xpectancy			
			ripant's Account Balance			
		. ,	ired Beginning Date <u>(RBD)</u>			
4.4	Deatl	Death Benefits				
4.5			Severance From Employment			
	(A)		e Emergency			
	(B)		of Rollover Contributions			
	(C)		- <u>Service Distributions</u>			
	(D)	_	oans			
	(E)		th or Adoption Distributions (QBOADs)			
	(F)	Coronavirus-	Related Distributions (CRDs)	<u></u> 23		
	ــــــــــــــــــــــــــــــــــــــ		navirusRelated Distribution Defined			
			egate Dollar Amount			
			ontribution of CRDs			
			cipant Certification			
	(G)		ribution			
4.6	Distributions Under Qualified Domestic Relations Orders (QDROs)					
1.0	(A)		thod of Paymentthod of Payment			
	(B)		lures			
	(C)	•				
	(D)		QDROs			
4.7			gible Rollover Distributions			
1./	(A)		lection			
	(B)		Withholding Notice			
	(C)		ibution or Rollover			
	(D)		Beneficiary Rollover Right			
	(D)	<u>=</u>	in Requirements Not Applicable			
		. ,	Beneficiary			
		. ,	S Not Eligible For Rollover			
	(E)	. ,	nitions			
	(E)		le Rollover Distribution			
		. , .	le Retirement Plan			
			t Rollover			
		(4) Mand	atory Distribution	Z4		
ARTI		PLAN ADMINIST	TRAT <u>ION OR – DUTIES WITH RESPECT TO PARTICIPANTS' ACCO</u>	UNTS		
	25					
5.1						
5.2	Term	/Vacancy		25		

5.3	Powers and Duties25			
	(A) <u>In General</u>			
	(B) Loan Policy Establishment	26		
	(C) QDRO Policy Establishment			
5.4	Without Additional Compensation			
5.5	Authorized Representative			
5.6	Individual Accounts/Records			
5.7	Valuation of Participant's Account			
5.8	Account Administration, Valuation and Expenses			
	(A) Individual Accounts			
	(1) By Contribution Type			
	(2) By Investment Account Type			
	(a) Participant Directed Accounts			
	(b) Segregated Accounts			
	(3) Amount of Account			
	(4) Account Statements			
5.9	Participant Account Charged			
5.10	Participant Direction of Investment			
5.11	Vesting			
5.12	Preservation of Eligible Plan Status			
5.13	Limited Liability			
5.14	Lost Participants			
	(A) Attempt to Locate			
	(B) Failure to Locate			
	(C) Reasonable Expenses			
5.15	Plan Correction			
A DELC	NEW DENERICIARY DEGICNATIONS AND ENDOLLMENT	20		
	CLE VI. BENEFICIARY DESIGNATIONS AND ENROLLMENT			
6.1	Beneficiary Designation			
6.2	No Beneficiary Designation			
6.3	Salary Reduction Agreement			
	(A) General			
	(B) Election Timing			
	(C) Sick, Vacation and Back Pay			
	(D) Modification of Salary Reduction Agreement			
6.4	Personal Data to Plan Administrator			
6.5	Address for Notification			
6.6	Participant or Beneficiary Incapacitated	32		
ARTIC	CLE VII. TRUST PROVISIONS FUNDS	32		
7.1	Establishment of Trust			
7.2	Acceptance/Holding	33		
7.3	Receipt of Contributions			
7.4	Investment of Plan Assets Deferred Compensation Investment Committee			
7.5	Records and Statements Investment Policy Statement			
	(A) The Plan's Objectives			

	(B) The Policies or Procedures	35		
	(C) The Liquidity of Plan Assets			
7.6	Fees and Expenses From FundInvestment Manager			
7.7	Professional Agents Investment Funds			
7.8	Distribution of Cash or PropertyParticipant Direction of Investment			
7.9	Resignation and Removal	35		
7.10	Successor Trustee/Appointment	35		
	Valuation of Trust			
7.12	Participant Direction of Investment	36		
7.13	Invalidity of Any Trust Provision	36		
7.14	Exclusive Benefit	36		
7. <del>15</del> 9	Group Trust Authority	36		
7.16	Documents and Instruments	<u></u> 37		
7.17	More Than One Trustee	<u></u> 37		
7.18	Trustee Duties Limited	<u></u> 37		
<del>7.19</del>	Appointment of Custodian	<u></u> 37		
ARTIC	LE VIII. AMENDMENT, TERMINATION AND TRANSFERS	37		
8.1	Amendment by Employer	37		
	(A) Amendment Purposes Limited	37		
	(B) Method of Amendment	37		
8.2	Termination/Freezing of Plan	38		
8.3	Purchase of Permissive Service Credit			
ARTIC	LE IX. MISCELLANEOUS	38		
9.1	No Assignment or Alienation	38		
9.2	Effect on Other Plans	38		
9.3	Word Usage	38		
9.4	Governing Law	38		
9.5	Employment Not Guaranteed	38		
9.6	Notice, Designation, Election, Consent, and Waiver	39		
9.7	Risk of Loss and Indemnities	39		

#### The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan

#### **BACKGROUND**

Pursuant to Board of Directors' Resolution Number 7683, The Metropolitan Water District of Southern California ("District") established effective November 1, 1976, the Deferred Compensation Plan (the "Plan").

The Plan has been amended from time to time. The Plan was restated by Board of Directors' Minute Item 34625, effective April 1, 1983. Board of Directors' Minute Item 42712 caused the Plan to be codified in Division VI, Chapter 7, Article 8 of The Metropolitan Water District of Southern California Administrative Code ("MWD Code") and named the "Metropolitan Water District of Southern California 1997 Deferred Compensation Plan" effective November 18, 1997. The Plan was restated in its entirety and renamed "The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan" effective August 1, 2018. The Plan was further amended and restated effective January 1, 2020. The Plan was most recently is hereby amended and restated effective January 1, 2024, except as otherwise stated herein.

The Plan was established, and is maintained, as an eligible governmental deferred compensation plan meeting the requirements of Internal Revenue Code (the "Code") § 457(b) and § 457(e)(1)(A) and Treas. Reg. §§ 1.457-3 through 1.457-10.

All contributions made to the Plan, all property and rights purchased with such amounts and all income attributable to such amounts shall be held in trust for the exclusive benefit of Participants, Beneficiaries and Alternate Payees under the Plan in accordance with Code § 457(g). Amounts contributed under the Plan are held and invested, until distributed in accordance with the terms of the Plan.

The purpose of the Plan is to attract and retain certain officers and employees to The Metropolitan Water District of Southern California.

#### ARTICLE I.

#### **DEFINITIONS**

- **1.1** Account. The separate Account(s) established and maintained under the Plan for a Participant's Deferred Compensation, an alternate payee's interest in the Plan under a QDRO, or as needed, for Beneficiaries described in Section 4.3.
- **1.2 Accounting Date.** The last day of the Plan Year. The Plan Administrator will allocate contributions and forfeitures for a particular Plan Year as of the Accounting Date of that Plan Year, and on such other dates, if any, as the Plan Administrator determines, consistent with the Plan's

allocation conditions and other provisions.

- **1.3 Beneficiary.** A person who the Plan or a Participant designates and who is or may become entitled to a Participant's Account upon the Participant's death. If the Plan does not have a valid Beneficiary designation on file, the Participant's Beneficiary will be determined in accordance with Article VI (Beneficiary Designations and Enrollment). A Beneficiary who becomes entitled to a benefit under the Plan remains a Beneficiary under the Plan until the Beneficiary's entire Plan benefit has been fully distributed. A Beneficiary's right to (and the duty to provide to the Beneficiary) information or data concerning the Plan does not arise until the Beneficiary first becomes entitled to receive a benefit under the Plan.
- **1.4 Code.** The Internal Revenue Code of 1986, as amended.
- **1.5 Compensation.** The definition of Compensation is W-2 Wages. W-2 Wages means wages for federal income tax withholding purposes, as defined under Code § 3401(a), plus all other payments to an Employee in the course of the Employer's trade or business, for which the Employer must furnish the Employee a written statement under Code §§ 6041(Information at Source), 6051(Receipts for Employees), and 6052(Returns Regarding Payment of Wages in the Form of Group-Term Life Insurance), but determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or services performed (such as the exception for agricultural labor in Code § 3401(a)(2)).
  - **(A) Compensation Taken Into Account.** For the Plan Year in which an Employee first becomes a Participant, the Plan Administrator will determine the allocation of contributions by taking into account the Employee's Compensation for the entire Plan Year.
  - **(B) Elective Contributions.** Compensation includes elective contributions excludible from the Employee's gross income under Code §§ 125(Cafeteria Plans), 132(f)(4)(De Minimis Fringe Defined), 402(e)(3)(Cash or Deferred Arrangements), 402(h)(1)(B)(Special Rules for Simplified Employee Pensions), 403(b)(Taxability of Beneficiary Under Annuity Purchased by Section 501(c)(3)), 408(p)(Simple Retirement Accounts) or 457.
  - **(C) Post-Severance Compensation.** Post-Severance Compensation is Compensation paid after a Participant's severance from employment with the District as further described in subsection (C)(1) below. Any other amount paid after severance from employment that is not described in subsection5(C)(1) is not Compensation even if payment is made within the time period described below. Post-Severance Compensation does not include severance pay, or payments under a nonqualified unfunded deferred compensation plan unless the payments would have been paid at that time without regard to severance from employment.
    - (1) Timing. Post-Severance Compensation includes the following items of compensation to the extent the Employer pays such amounts by the later of 2½ months after severance from employment or by the end of the Plan Year that

includes the date of such severance from employment:

- **(a) Regular Pay.** The payment of regular Compensation for services during the Participant's regular working hours, or Compensation for services outside the Participant's regular working hours (such as overtime or shift differential), bonuses, or other similar payments, but only if the payment would have been paid to the Participant prior to a severance from employment if the Participant had continued in employment with the Employer.
- **(b) Leave Cash-outs.** Payments for unused accrued bona fide sick, vacation or other leave, but only if the Employee would have been able to use the leave if employment had continued and if Compensation would have included those amounts if they were paid prior to the Participant's severance from employment.
- **(D) Salary Continuation for Disabled Participants.** Compensation paid to a Participant who is permanently and totally disabled (as defined in Code § 22(e)(3)) may be included in Compensation.
- **(E) Differential Wage Payments.** An individual receiving a Differential Wage Payment, as defined by Code § 3401(h)(2), shall be treated as an employee of the employer making the payment and the Differential Wage Payment shall be treated as compensation for purposes of Code § 457(b) and any other Code section that references the definition of compensation under Code § 415, including the definition of Includible Compensation as provided in Section 1.14.
- **1.6 Deferred Compensation.** The amount of Salary Reduction Contributions, Rollover Contributions and Transfers adjusted for allocable net income, gain or loss, in the Participant's Account.
- 1.7 Deferred Compensation Investment Advisory Committee (DCIAC). A Districtestablished advisory The committee which is comprised of the District's Chief Financial Officer, <u>Chief Administrative Officer, and Budget & Treasury and Debt Manager</u>er, the Plan Administrator, a staff member and the Manager of the Human Resources Group's Benefits Unit District's HR Group-Manager; Assistant General Manager, Operations or designee; Treasury & Debt Manager; Assistant General Manager, Chief Financial Officer or Finance Group Manager as designee; General Auditor or designee; and Risk Manager - and five appointed members - one appointed by the Employees' Association of the Metropolitan Water District, AFSCME Local 1902 ("AFSCME Local 1902"); oneappointed by the Management and Professional Employees Association of the Metropolitan Water-District, AFSCME Local 1001 ("MAPA"); one appointed by the Supervisors Association of the Metropolitan Water District of Southern California ("SA"); one appointed by the Association of Confidential Employees ("ACE"); and one, a Plan retiree, appointed by the Committee., and a memberappointed to represent retirees and one representative from each of the District's four bargaining units: AFSCME Local 1902, Management and Professional Association, Supervisor's Association, and Association of Confidential Employees. The DCAC meets quarterly to advise the Plan Administrator-

on plan administrative policies, investments, educational programs, the effect of regulatory and taxlaw changes, and the viability of proposed plan changes based on Participant requests\_The DCIACdecides by an advises only on an affirmative vote by a majority of the committee or by unanimousconsent. Non-voting DCIC attendees include the Human Resource Benefits Manager, the Human-Resources Principal Analyst, and the General Counsel's representative described in Section 7.4.

- **1.8 Effective Date.** The Plan was originally established and effective November 1, 1976. Since then, the Plan has been amended and/ or restated from time to time. The effective date of the most current amendment and restatement of this Plan is January 1, 2024, the effective date of this amendment and restatement of the Plan.
- **1.9 Eligible Employee.** An Employee who is not an Excluded Employee.
- **1.10 Employee.** An individual who provides services for the District as a common law employee of the District.
- **1.11 Employer or District.** The Metropolitan Water District of Southern California.
- **1.12 Excess Deferrals.** Salary Reduction Contributions to the Plan that exceed the Taxable Year maximum limitation of Code §§ 457(b) and (e)(18).
- **1.13 Excluded Employee.** Employees who retired and severed employment and are rehired as annuitants, as defined in California Government Code § 7522.56, to work for a limited duration of time. For purposes of this Plan, Independent Contractors are also treated as Excluded Employees.
- **1.14 Includible Compensation.** For the Employee's Taxable Year, the Employee's total Compensation within the meaning of Code § 415(c)(3) paid to an Employee for services rendered to the Employer. Includible Compensation also includes:
  - (A) Salary Reduction Contributions under this Plan, compensation deferred under any other plan described in Code § 457, and any amount excludible from the Employee's gross income under Code §§ 401(k), 403(b), 125, or 132(f)(4) or any other amount excludible from the Employee's gross income for Federal income tax purposes;
  - **(B)** Payments for unused bona fide sick, vacation, or other leave, if the Employee would have been able to use the leave if employment had continued; and
  - **(C)** Amounts paid prior to severance, or paid within 2 1/2 months after the later of severance from employment or the end of the Plan Year that includes the date of severance.

The Employer will determine Includible Compensation without regard to community property laws.

- **1.15 Independent Contractor.** Any individual who performs service for the District and who the District does not treat as an Employee or a Leased Employee.
- **1.16** Leased Employee. An Employee within the meaning of Code § 414(n).
- **1.17 Normal Retirement Age.** The age is designated by a Participant. A Participant may not designate a Normal Retirement Age (NRA) that is earlier than his or her CalPERS normal retirement

age, which for classic members is age 55, and is age 62 for new members (as defined under PEPRA). Participants shall designate a NRA in the form and manner established by the Plan Administrator.

- **1.18 Participant.** An Employee (other than an Excluded Employee) who becomes a Participant in accordance with the provisions of Article II (Eligibility and Participation).
- **1.19 Plan.** The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan.
- **1.20 Plan Administrator.** The General Manager of the District or other person or entity designated by the District to be responsible for the administration and operation of the Plan.
- **1.21 Plan Year.** The calendar year ending on December 31.
- 1.22 Qualified Birth or Adoption Distribution (QBOAD). A distribution to a Participant if made (a) during the 1-year period beginning on the date on which a child of the Participant is born or on which the legal adoption by the Participant of an eligible adoptee is finalized, (b) with respect to a child born or eligible adoptee the Participant includes the name, age, and the Taxpayer Identification Number of the child or eligible adoptee on the Participant's tax return for the taxable year in which the distribution is made, and (c) only to the extent such distribution from all plans maintained by the Employer to such Participant does not exceed \$5,000, in the aggregate, for each birth or eligible adoptee. An eligible adoptee is any individual (other than a child of the Participant's spouse) who has not attained age 18 or is physically or mentally incapable of self-support.
- **1.22 1.23 Recurrent Employee.** An Employee who is hired for an indefinite period of time on an irregular basis, such as intermittent, emergency or on call.
- **1.23 1.24 Rehired Annuitant.** A rehired Employee who meets the definition of rehired annuitant in California Government Code Section 7522.56.
- **1.24**1.25 **Rollover Contribution.** The amount of cash or property which an eligible retirement plan described in Code § 402(c)(8)(B) distributes to an eligible Employee or Participant in an eligible rollover distribution under Code § 402(c)(4) and which the eligible Employee or Participant transfers directly or indirectly to the Plan. A Rollover Contribution includes net income, gain or loss attributable to the Rollover Contribution. A Rollover Contribution into the Plan excludes after-tax Employee contributions, as adjusted for net income, gain or loss.
- 1.25 Roth Elective Deferrals. A Participant's Salary Reduction Contributions that are includible in the Participant's gross income at the time deferred and have been irrevocably designated as Roth Elective Deferrals by the Participant in his or her Salary Reduction Agreement. A Participant's Roth Elective Deferrals will be separately accounted for, as will gains and losses attributable to those Roth Elective Deferrals. However, forfeitures may not be allocated to such account. The Plan must also maintain a record of a Participant's investment in the contract (i.e., designated Roth contributions that have not been distributed) and the year in which the Participant first made a Roth Elective Deferral.

- **1.26 1.27 Salary Reduction Agreement.** A written agreement between a Participant and the District, authorizing the District to reduce the Participant Compensation and contribute the amount as a Salary Reduction Contribution to the Participant's Account.
- **1.27 1.28 Salary Reduction Contribution.** The Pre-Tax Elective Deferrals and Roth Elective Deferrals the District deducts from a Participant's Compensation and contributes to the Plan pursuant to a Participant's Salary Reduction Agreement.
- **1.29 Service.** Any period of time the Employee is in the employ of the Employer. An Employee terminates Service upon incurring a severance from employment.
  - (A) Qualified Military Service. Service includes any qualified military service the Plan must credit for contributions and benefits in order to satisfy the crediting of Service requirements of Code § 414(u). A Participant whose employment is interrupted by qualified military service under Code § 414(u) or who is on a leave of absence for qualified military service under Code § 414(u) may elect to make additional Salary Reduction Contributions upon resumption of employment with the Employer equal to the maximum Salary Reduction Contributions that the Participant could have elected during that period if the Participant's employment with the Employer had continued (at the same level of Compensation) without the interruption of leave, reduced by the Salary Reduction Contributions, if any, actually made for the Participant during the period of the interruption or leave. This right applies for five years following the resumption of employment (or, if sooner, for a period equal to three times the period of the interruption or leave). The Plan does not provide for matching or non-elective employer contributions as those terms are defined under Code § 457 and the underlying Treasury Regulations. Therefore, the Employer is not required under Code § 414(u) to make any other contributions to such a Participant. The Plan shall apply limitations of Article III (Deferral Contributions and Limitations) to all Salary Reduction Contributions under this paragraph with respect to the year to which the Salary Reduction Contributions relates.
  - **(B) Continuous Service.** Service with the Employer during which the Employee does not incur a severance from employment.
  - (C) Severance from Employment.
    - (1) Employee. An Employee has a severance from employment when the Employee ceases to be an Employee of the Employer. A Participant does not incur a severance from employment if, in connection with a change in employment, the Participant's new employer continues or assumes sponsorship of the Plan or accepts a Transfer of Plan assets as to the Participant.
    - (2) **Deemed Severance.** If a Participant performs service in the uniformed services (as defined in Code § 414(u)(12)(B)) on active duty for a period of more than 30 days, the Participant will be deemed to have a severance from employment solely for

purposes of eligibility for distribution of amounts not subject to Code § 412. However, the Plan will not distribute such a Participant's Account on account of this deemed severance unless the Participant specifically elects to receive a benefit distribution hereunder. If a Participant elects to receive a distribution on account of this deemed severance, then no Salary Reduction Contributions may be made for the Participant during the 6-month period beginning on the date of the distribution. If a Participant would be entitled to a distribution on account of a deemed severance, and a distribution on account of another Plan provision, then the other Plan provision will control and the 6-month suspension will not apply.

- **1.30 State.** One of the 50 states of the United States or the District of Columbia, or a political subdivision of a State, or any agency or instrumentality of a State or its political subdivision. A State does not include the federal government or any agency or instrumentality thereof.
- **1.30 1.31 Taxable Year.** The calendar year or other taxable year of a Participant.
- **1.31 1.32 Temporary Employee.** An Employee who is hired for a specified limited time.
- **1.32 Transfer.** A transfer of assets from one eligible Code § 457(b) plan to another eligible Code § 457(b) plan which is not a Rollover Contribution and which is made in accordance with Section 8.3 (Transfers).
- <u>Trust.</u> The <u>Trust ereated underestablished pursuant to</u> the <u>Trust Agreement between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets in accordance with Code\_§ 457(g).</u>
- **1.33 1.35 Trust Agreement.** The separate agreement entered into by and between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets.
- 1.34 1.36 Trustee. The District or the person, committee(s) or entity, and any successors thereto, named in the Trust Agreement, who is appointed by the District to serve as Trusteehold, administer and accepts such position. The District retains distribute the right of appointment and removal of Trustee(s) and may exercise that right at any time. Use of the defined term, Trustee refers to a discretionary Trustee, appointed in accordance with Section 7.1. However, the District may retain the assistance of a nondiscretionary trustee to carry out some or all or a portion of the Plan assets held in the Trust pursuant to the terms and conditions of the duties of the Trustee or to assist the Trustee in the performance of its powers, duties and responsibilities. See Article VII\_frust Provisions)Agreement.

#### ARTICLE II.

#### **ELIGIBILITY AND PARTICIPATION**

**2.1 Eligibility.** The Plan does not require Eligible Employees to satisfy an age or service requirement in order to participate in the Plan. Each Eligible Employee becomes a Participant in the Plan in accordance with entry date rules described in Section 2.2.

- 2.2 Commencement of Salary Reduction Contributions. Generally, Salary Reduction Contributions for Eligible Employees commence on the first day of the month coinciding with or following the later of the Eligible Employee's date of hire or submission of a Salary Reduction Agreement. Notwithstanding the preceding sentence, Eligible Employees who are Temporary Employees or Recurrent Employees commence Salary Reduction Contributions in accordance with the Minimum Salary Reduction Contribution amounts specified in Section 3.2 as of his or her date of hire. Eligible Employees who are Temporary Employees or Recurrent Employees may change contribute an amount greater than (but not less than) the Minimum Salary Reduction Contribution amount described in Section 3.2 by completing a Salary Reduction Agreement. Changes to such Eligible Employee's Minimum Salary Reduction Contribution amount will take effect in accordance with the general timing rule described above.
- **2.3 Effect of Plan Restatement.** Each Eligible Employee who was a Participant on the day before the Effective Date of the Plan continues as a Participant in the Plan, irrespective of whether he/she satisfies the eligibility conditions in the <u>amended and</u> restated Plan.
- **2.4 Participation Upon Re-Employment.** A Participant who incurs a severance from employment may re-enter the Plan as a Participant as of his or her date of re-employment, so long as the Employee is not rehired in an Excluded Employee classification.

#### ARTICLE III.

#### **DEFERRAL CONTRIBUTIONS AND LIMITATIONS**

#### 3.1. Amount.

- **(A) Contribution Formula.** For each Plan Year, the District will contribute to the Plan, on behalf of each Participant, the Participant's Salary Reduction Contribution and any other type and amount of contributions authorized under the Plan. Effective February 18, 2024. Salary Reduction Contributions may include Roth Elective Deferrals in addition to pre-tax salary deferrals.
- **(B) Return of Contributions.** If any Salary Reduction Contribution (or any other type contributions authorized under the Plan) is made due to a mistake of fact, the Plan Administrator upon written request from the District shall return the contributions (adjusted for net income, gain or loss), within one (1) year after payment of the contribution.
- **(C) Time of Payment of Contribution.** The District shall deposit Salary Reduction Contributions to the Trust within a period that is not longer than is reasonable for the administration of Participant Accounts.
- **3.2. Salary Reduction Contributions.** The Plan permits Salary Reduction Contributions. Participants may designate a dollar or percentage amount by which to reduce his or her Compensation. Such Participant designations shall be made in accordance with a Salary Reduction Agreement.

Temporary Employees and Recurrent Employees are subject to mandatory minimum Salary Reduction Contribution amounts ("Minimum Salary Reduction Contributions"). Effective July 1, 2016, the Minimum Salary Reduction Contribution for Temporary Employees and Recurrent Employees are as follows:

Temporary and Recurrent Employee	Minimum Salary Reduction
Category	Contribution
Classic PERS Members	0.5%
New PERS Members	1.5%
Non PERS Members	7.5%

Provided, however, the Minimum Salary Reduction Contributions for Temporary or Recurrent Employees shall at all times be no less than the difference between 7.5% of the Temporary or Recurrent Employee's Compensation and his or her member contribution to CalPERS, if any. If at any time the Minimum Salary Reduction Contribution required of a Temporary Employee or Recurrent Employee varies from the minimums provided in the table above, the table shall automatically be adjusted.

Temporary Employees and Recurrent Employees may elect a different Salary Reduction Contribution amount, however, such election must be for an amount that is greater than the applicable Minimum Salary Reduction Contribution stated in this Section 3.2.

- **(A) Deferral from Sick, Vacation and Back Pay.** Participants may elect to make Salary Reduction Contributions from accumulated sick pay, accumulated vacation pay or back pay in accordance with the established deduction timing calendar maintained by the District.
- **(B) Application to Leave of Absence and Disability.** Unless a Participant in his or her Salary Reduction Agreement elects otherwise, the Participant's Salary Reduction Agreement shall continue to apply during the Participant's leave of absence or the Participant's disability (as the Plan Administrator shall establish), if the Participant has Compensation other than imputed compensation or disability benefits.
- **(C) Post-Severance Deferrals Limited to Post-Severance Compensation.** Deferrals are permitted from an amount received following severance from employment only if the amount is Post-Severance Compensation.
- **3.3. Normal Limitation.** Except as provided in Sections 3.4 (Age 50 Catch-Up Contribution) and 3.5 (Special Catch-Up Contribution), a Participant's maximum Salary Reduction Contributions (excluding Rollover Contributions and Transfers) under this Plan for a Taxable Year may not exceed the lesser of:
  - **(A)** The applicable dollar amount as specified under Code § 457(e)(15) (or as the Commissioner of the Internal Revenue may prescribe), or
  - **(B)** 100% of the Participant's Includible Compensation for the Taxable Year.

- **3.4. Age 50 Catch-Up Contribution**. A Participant who will have attained age 50 before the close of the Taxable Year is eligible to make age 50 catch-up contributions for that Taxable Year and in subsequent Taxable Years in accordance with, and subject to the limitations of, Code § 414(v). If, for a Taxable Year, a Participant is eligible to make an Age 50 Catch-Up Contribution under this Section 3.4 and a special catchup contribution under Section 3.5 (Special Catch-Up Contribution) such Participant is entitled to the greater of the two catch up amounts determined under Section 3.4 and Section 3.5.
- **3.5. Special Catch-Up Contribution.** A Participant may elect to make a catch-up contribution for one or more of the Participant's last three Taxable Years ending before the Taxable Year in which the Participant attains NRA. In a Taxable Year in which a catch-up contribution is made, the Participant's maximum Salary Reduction Contributions may not exceed the lesser of:
  - (A) Twice the dollar amount under Section 3.3 (Normal Limitation), and
  - (B) The underutilized limitation under Section 3.5(A), as described below
    - (1) Underutilized Limitation. A Participant's underutilized limitation is equal to the sum of: (i) the normal limitation for the Taxable Year, and (ii) the normal limitation for each of the prior Taxable Years of the Participant commencing after 1978 during which the Participant was eligible to participate in the Plan and the Participant's Salary Reduction Contributions were subject to the Normal Limitation or any other Code § 457(b) limit, *less* the amount of Salary Reduction Contributions for each such prior Taxable Year, excluding Age 50 Catch-Up Contributions.
    - (2) Normal Retirement Age (NRA). A Participant may elect his or her applicable NRA for purposes of this Section 3.5. Participants may not elect a NRA that is earlier than age 65 or the age at which Participants have the right to retire and receive benefits under CalPERS without actuarial or other reduction which is age 55 for classic CalPERS members and age 62 for new CalPERS members as defined by PEPRA. Further, participants may not elect a NRA later than age 72.
    - (3) Pre-2002 Coordination. In determining a Participant's underutilized limitation, the Plan Administrator, in accordance with U.S. Treas. Reg. § 1.457-4(c)(3)(iv), must apply the coordination rule in effect under now repealed Code § 457(c)(2). The Plan Administrator also must determine the Normal Limitation for pre-2002 Taxable Years in accordance with Code § 457(b)(2) as then in effect.
- **3.6. Rollover Contributions.** The Plan may accept Rollover Contributions.
  - **(A) Operational Administration.** The District, operationally and on a nondiscriminatory basis, may elect to limit an eligible Employee's or Participant's right to make a Rollover Contribution. Any <a href="Employee or Participant">Employee or Participant</a> (or former <a href="Employee or Participant">Employee or Participant</a>) with the District's written consent and after filing the forms prescribed by the Plan Administrator, may make a Rollover Contribution to the Trust. Before accepting a Rollover Contribution, the Participant (or eligible Employee) may be required to furnish satisfactory evidence

that the proposed transfer is in fact a Rollover Contribution which the applicable tax rules permits an employee to make to an eligible retirement plan. The Plan Administrator, in its sole discretion, may decline to accept a Rollover Contribution of property which could: (1) generate unrelated business taxable income; (2) create difficulty or undue expense in storage, safekeeping or valuation; or (3) create other practical problems for the Trust.

- (B) Separate Accounting. If the District permits Rollover Contributions, the Plan Administrator must account separately for: (1) amounts rolled into this Plan from an eligible retirement plan (other than from another eligible governmental 457(b) plan); and (2) amounts rolled into this Plan from another eligible governmental 457(b) plan. The Plan Administrator for purposes of ordering any subsequent distribution from this Plan may designate a distribution from a Participant's Rollover Contributions as coming first from either of (1) or (2) above if the Participant has both types of Rollover Contribution Accounts.
- (C) Roth Elective Deferrals. The Plan may accept a rollover contribution of Roth Elective Deferrals only if it is a direct rollover from another Plan which permits Roth Elective Deferrals as described in Code § 402A(e)(1) and only to the extent the rollover is permitted under the rules of Code § 402(c). The District, operationally and on a uniform and nondiscriminatory basis, may decide whether to accept any such rollovers. If the Plan chooses to accept rollovers of Roth Elective Deferrals, the amounts must be directly rolled over into this Plan from another plan that is qualified under Code § 401(a), from a Code § 403(b) plan, or from an eligible governmental Code § 457(b) plan. The Plan must account separately for the Rollover Contribution, including Roth Elective Deferrals and the earnings thereon.
- **3.7. Distribution of Excess Deferrals.** In the event that a Participant has Excess Deferrals, the Plan will distribute to the Participant the Excess Deferrals and allocable net income, gain or loss, in accordance with this Section 3.7.
  - **(A) Eligible Governmental 457(b) Plan.** The Plan Administrator will distribute Excess Deferrals from the Plan as soon as is reasonably practicable following the Plan Administrator's determination of the amount of the Excess Deferral.
  - **(B) Individual Limitation.** If a Participant participates in another eligible 457(b) plan maintained by a different employer, and the Participant has Excess Deferrals, the Plan Administrator may, but is not required, to correct the Excess Deferrals by making a corrective distribution from this Plan.
- **3.8. Benefit Accruals for Qualified Military Service.** To the extent required under Code § 414(u), an Employee who returns to employment with the District following a period of qualified military service will receive any contributions, benefits, and service credit required under Code § 414 (u), provided the Employee satisfies all applicable requirements under the Code and regulations. In determining the amount of contributions under Code § 414(u), Plan Compensation will be deemed to be the compensation the Employee would have received during the period while in military service

based on the rate of pay the Employee would have received from the District but for the absence due to military leave. If the compensation the Employee would have received during the leave is not reasonably certain, Plan Compensation will be equal to the Employee's average compensation from the District during the twelve (12) month period immediately preceding the military leave or, if shorter, the Employee's actual period of employment with the District.

For benefit accrual purposes, the Plan treats an individual who dies or becomes disabled (as defined under the terms of the Plan) while performing qualified military service with respect to the District as if the individual had resumed employment in accordance with the individual's reemployment rights under USERRA, on the day preceding death or disability (as the case may be) and terminated employment on the actual date of death or disability.

- 3.9. Roth Elective Deferrals. Effective February 18, 2024, Participants may make Roth Elective Deferrals to the Plan. Unless otherwise indicated, Roth Elective Deferrals shall be treated in the same manner as Elective Deferrals. The District may, in operation, implement deferral election procedures provided such procedures are communicated to Participants and Participants are permitted to modify their elections at least once per Plan Year.
  - (A) Elective Deferrals means a contribution the Employer makes to the Plan pursuant to a Participant's Salary Reduction Agreement and includes Pre-Tax Elective Deferrals and Roth Elective Deferrals.
    - (1) Pre-Tax Elective Deferrals mean a Participant's Salary Reduction Contributions which are not includible in the Participant's gross income at the time deferred and have been irrevocably designated as Pre-Tax Elective Deferrals by the Participant in his or her Salary Reduction Agreement. A Participant's Pre-Tax Elective Deferrals will be separately accounted for, as will gains and losses attributable to those Pre-Tax Elective Deferrals.
    - (2) Roth Elective Deferrals means a Participant's Salary Reduction Contributions that are includible in the Participant's gross income at the time deferred and have been irrevocably designated as Roth Elective Deferrals by the Participant in his or her Salary Reduction Agreement. A Participant's Roth Elective Deferrals will be separately accounted for, as will gains and losses attributable to those Roth Elective Deferrals. However, forfeitures may not be allocated to such account. The Plan must maintain a record of a Participant's Roth Elective Deferrals and the year in which the Participant first made a Roth Elective Deferral.
  - **(B) Ordering Rules for Distributions.** The District may implement an ordering rule for withdrawals from Participant's accounts attributable to Pre-Tax Elective Deferrals and/or Roth Elective Deferrals. Absent an ordering rule established by the District in compliance with applicable law, withdrawals will be made on a pro rata basis from Participant's Pre-Tax Elective Deferrals, and Roth Elective Deferrals.

(C) Loans. The Plan's loan policy or program may be modified to address Roth Elective Deferrals, including but not limited to, whether loans from Participant Roth Elective Deferral Accounts will be permitted.

#### 3.10. In-Plan Roth Conversion.

- (A) In-Plan Roth Conversion Account. Effective February 18XX, 2024, a Participant may elect to transfer some or all of his or her vested Account balance (other than his or her Roth Elective Deferral Account or Roth Rollover Account), and earnings thereon, to an In-Plan Roth Conversion Account without regard to whether the Participant satisfies the requirements for distribution in accordance with Article IV (Time and Method of Payments of Benefits) of the Plan. Any Participant election to make an In-Plan Roth Conversion during a taxable year is irrevocable and cannot be changed after the In-Plan Roth Conversion is completed. Generally, In-Plan Roth Conversion Accounts are subject to the same Plan provisions as Roth Elective Deferral Accounts. However, distribution restrictions that otherwise apply with respect to a specific contribution source will continue to apply if such contribution source is converted to Roth Elective Deferrals. The Plan Administrator will maintain such records as are necessary for the proper reporting of In-Plan Roth Conversions and will administer the In-Plan Roth Conversion Accounts in accordance with the Code, Treasury Regulations, and Plan provisions.
- **(B) Eligibility to Elect In-Plan Roth Rollover.** For purposes of determining eligibility for In-Plan Roth Rollovers, the Plan will treat a Participant's spousal Beneficiary or Participant's alternate payee spouse or former spouse as a Participant. A non-spouse beneficiary is not eligible to make an In-Plan Roth Rollover.
- (C) In-Plan Roth Rollover Procedure. An In-Plan Roth Rollover must be made by the Participant in the form of a Direct In-Plan Roth Rollover. Additionally, an In-Plan Roth Rollover must be accomplished in the form and manner established by the Plan Administrator.
- **(D) Plan Loans.** In-Plan Roth Rollovers may include a Plan loan only if there is no change in the loan's repayment schedule and the loan is not treated as a new loan. Further, Plan loans that include In-Plan Roth Rollover amounts are subject to the Plan's loan policy.
- **(E)** Treatment of In-Plan Roth Rollovers. Notwithstanding anything in the Plan to the contrary, an In-Plan Roth Rollover is not a rollover contribution for purposes of the Plan. The Plan will take into account the amounts attributable to an In-Plan Roth Rollover for purposes of determining whether a Participant's vested Account Balance exceeds \$500 for purposes of the Plan's automatic cash-out distribution rule at Section 4.2(B). Amounts in the Participant's In-Plan Roth Rollover Account may only be withdrawn by a Participant when the Participant is eligible for a distribution under Article IV (Time and Method of Payments of Benefits) of the Plan.

- **(F) Spousal Consent.** A married Plan Participant is not required to obtain spousal consent in connection with an election to make an In-Plan Roth Conversion, even if the Plan would otherwise require the spousal consent for distributions.
- (G) Tax Treatment of In-Plan Roth Conversion. A Participant must include in gross income the taxable amount of an In-Plan Roth Conversion. For this purpose, the taxable amount of an In-Plan Roth Conversion is the fair market value of the distribution reduced by any basis in the converted amounts. If an outstanding loan is rolled over as part of an In-Plan Roth Conversion, the amount includible in gross income includes the balance of the loan. Generally, the taxable amount of an In-Plan Roth Conversion is includible in gross income in the taxable year in which the conversion occurs. However, In-Plan Roth Conversions are not subject to twenty percent (20%) mandatory withholding under Code § 3405(c). Thus, Participants will be required to make arrangements independent of the Plan to ensure the total taxable amount is timely paid.

#### (H)In-Plan Roth Conversion Definitions.

- (1) "Direct In-Plan Roth Rollover" means a direct rollover from a Participant Account (other than Participant's Roth Elective Deferral Account, Roth Rollover Account and In-Plan Roth Rollover Account) to an In-Plan Roth Rollover Account established for the Participant.
- (2) "In-Plan Roth Rollover" means a contribution made from the Participant's Account (other than Participant's Roth Deferral Account or Roth Rollover Account) to the Participant's In-Plan Roth Rollover Account under the Plan in accordance with Code § 402A(c)(4) with respect to distributable amounts under the Plan or § 402A(c)(4)(E) with respect to amounts not otherwise distributable under the Plan.
- (3) "In-Plan Roth Rollover Account" means the subaccount established under the Plan to account for a Participant's In-Plan Roth Rollover.

#### ARTICLE IV.

#### TIME AND METHOD OF PAYMENT OF BENEFITS

- **4.1. Distribution Restrictions.** Except as the Plan provides otherwise in Sections 4.3 (RMDs) and 4.5 (Distributions Prior to Severance From Employment), a Participant may not receive a distribution of the amounts in his or her Account prior to occurrence of one of the following events:
  - (A) The Participant's severance from employment; or
  - **(B)** The Participant's death.
- **4.2. Time and Method of Payment of Account.** Upon a severance from employment the Participant's vested Account shall be distributed as specified in this Section 4.2 (Time and Method of Payment of Account) and in accordance with Code § 401(a)(9) and its underlying Treasury Regulations. In the absence of a Participant election, the Plan Administrator will determine the time

and method of payment applicable to a particular Participant. In no event will distributions commence later than permissible under the required minimum distribution rules described in Section 4.3 (RMDs) and of Code § 401(a)(9) and its underlying Treasury Regulations.

- (A) Participant Election of Time and Method. Participants may elect the timing and method of distribution of his or her vested Account in accordance with this Section 4.2 (Time and Method of Payment of Account). The Plan Administrator must consent to the specific terms of any such Participant election and the Plan Administrator in its sole discretion may withhold consent. Subject to the foregoing conditions, a Participant: (1) may elect to postpone distribution of his or her vested Account to any fixed or determinable date including, but not beyond, the Participant's required beginning date; and (2) may elect a single lump sum or periodic payments (on a monthly, quarterly, semiannual, or annual basis) and/or a partial distribution as the method of payment of his or her vested Account balance provided the payment schedule selected by the Participant complies with Code §§ 457 and 401(a)(9). The Plan Administrator must furnish to the Participant a form for the Participant to elect the time and a method of payment.
- **(B) Automatic Cash-Out Distribution.** Notwithstanding any other distribution election, if the Participant's Account balance is not in excess of \$500 as of the date of distribution, the Plan Administrator may distribute the Participants Account balance in a single lump sum payment following Participant's severance from employment. In determining whether the Participant's Account balance is in excess of \$500, the Plan Administrator may take Rollover Contributions into consideration.
- **4.3. Required Minimum Distributions ("RMDs").** A Participant's Account may not be distributed under a method of payment which, as of the required beginning date, does not satisfy the minimum distribution requirements of Code § 401(a)(9) or which is not consistent with applicable Treasury regulations.
  - **(A) General Rules.** All distributions required under this Section 4.3 will be determined and made in accordance with the Treasury regulations under Code § 401(a)(9), the requirements of which, will take precedence over any inconsistent provisions of the Plan.
  - (B) Time and Manner of Distribution.
    - (1) Required Beginning Date. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's Required Beginning Date as set forth in Section 4.3(E)(5).
    - (2) Death of Participant Before Distribution Begins. If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
      - **(a) Spouse Designated Beneficiary.** If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, then distributions to the surviving spouse must begin by December 31 of the calendar year immediately following

- the Calendar Year in which the Participant dies, or by December 31 of the Calendar Year in which the Participant would have attained age 70½ (or age 72 if Participant attains age 70½ after December 31, 2019), if later.
- **(b) Non-Spouse Designated Beneficiary.** If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, then distributions to the designated Beneficiary will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- **(c) No Designated Beneficiary.** If there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- (d) Death of Spouse. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Section 4.3(B)(2) other than Section 4.3(B)(2)(a) (Spouse Designated Beneficiary), will apply as if the surviving spouse were the Participant. For purposes of this Section 4.3(B) (Time and Manner of Distribution) and Section 4.3(D) (RMDs After Participant's Death), unless this Section 4.3(B)(2)(d) applies, distributions are considered to begin on the Participant's required beginning date. If this Section 4.3(B)(2)(d) applies, distributions are considered to begin on the date distributions are required to begin to the surviving spouse under Section 4.3(B)(2)(a) (Spouse Designated Beneficiary).
- (3) Forms of Distribution. Unless the Participant's interest is distributed in a single sum on or before the required beginning date, as of the first distribution calendar year distributions will be made in accordance with Sections 4.3(C) (RMDs During Participant's Lifetime) and 4.3(D) (RMDs After Participant's Lifetime).
- (C) RMDs During Participant's Lifetime.
  - (1) Amount of RMD Each Distribution Calendar Year. During the Participant's lifetime, the minimum amount that will be distributed for each distribution calendar year is the lesser of:
    - (a) Uniform Life Table. The quotient obtained by dividing the Participant's Account Balance by the number in the Uniform Life Table set forth in U.S. Treas. Reg. § 1.401(a)(9)-9, using the Participant's attained age as of the Participant's birthday in the distribution calendar year; or
    - **(b) Younger Spouse.** If the Participant's sole Designated Beneficiary for the distribution calendar year is the Participant's spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in U.S. Treas. Reg. § 1.401(a)(9)-9, using the

Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the distribution calendar year.

- (2) Lifetime RMDs Through Year of Participant's Death. RMDs will be determined under this Section 4.3(C) (RMDs During Participant's Lifetime) beginning with the first distribution calendar year and up to and including the distribution calendar year that includes the Participant's date of death.
- (D) RMDs After Participant's Death.
  - (1) Death On or After Distributions Begin.
    - (a) Participant Survived by Designated Beneficiary. If the Participant dies on or after the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's Designated Beneficiary, determined as follows:
      - (i) Participant's Life Expectancy. The Participant's remaining life expectancy is calculated using the attained age of the Participant as of the Participant's birthday in the calendar year of death, reduced by one for each subsequent calendar year.
      - (ii) Spouse's Life Expectancy. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining life expectancy of the surviving spouse is calculated for each distribution calendar year after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For distribution calendar years after the year of the surviving spouse's death, the remaining life expectancy of the surviving spouse is calculated using the attained age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one for each subsequent calendar year.
      - (iii) Non-Spouse's Life Expectancy. If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining life expectancy is calculated using the attained age of the Beneficiary as of the Beneficiary's birthday in the calendar year following the calendar year of the Participant's death, reduced by one for each subsequent calendar year.
    - **(b) No Designated Beneficiary.** If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the calendar year after the calendar year of the Participant's death, the minimum amount that will be distributed for each distribution calendar year

after the calendar year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the Participant's remaining life expectancy calculated using the attained age of the Participant as of the Participant's birthday in the calendar year of death, reduced by one for each subsequent calendar year.

#### (2) Death Before Date Distributions Begin.

- (a) Participant Survived by Designated Beneficiary. If the Participant dies before the date distributions begin and there is a designated Beneficiary, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining life expectancy of the Participant's designated Beneficiary, determined as provided in Section 4.3(D)(1) (Death On or After Distributions Begin).
- **(b) No Designated Beneficiary.** If the Participant dies before the date distributions begin and there is no Designated Beneficiary, distribution of the Participant's entire interest will be completed to Participant's estate by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- **(c) Death of Surviving Spouse Before Distributions to Surviving Spouse are Required to Begin.** If the Participant dies before the date distributions begin, the Participant's surviving spouse is the Participant's sole designated Beneficiary, and the surviving spouse dies before distributions are required to begin to the surviving spouse under Section 4.3(B)(2)(a) (Spouse Designated Beneficiary), this Section 4.3(D)(2) will apply as if the surviving spouse were the Participant.
- (d) 5-Year or Life Expectancy Rule; Possible Election. A Designated Beneficiary may elect whether the life expectancy rule under Section 4.3(D)(2)(a) (Participant Survived by Designated Beneficiary) or the 5-year rule under Section 4.3(D)(2)(b) (No Designated Beneficiary) will apply. However, the life expectancy rule is only available in the case of a Designated Beneficiary. The 5-year rule applies as to any Beneficiary who is not a Designated Beneficiary as defined by Code § 401(a)(9). A permitted election under this Section must be made no later than the earlier of September 30 of the calendar year in which distribution would be required to begin under Section 4.3(D)(2)(a) (Participant Survived by Designated Beneficiary), or by September 30 of the calendar year which contains the fifth anniversary of the Participant's (or, if applicable, surviving spouse's) death.

#### (E) RMD Definitions.

- (1) Designated Beneficiary. The individual who is designated as the Beneficiary under the Plan and is the designated beneficiary under Code § 401(a)(9) and U.S. Treas. Reg. 1.401(a)(9)-1, Q&A-4.
- (2) Distribution Calendar Year. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year is the calendar year immediately preceding the calendar year which contains the Participant's required beginning date. For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which the distributions are required to begin under Section 4.3(B)(2). The RMDs for the Participant's first distribution calendar year will be made on or before the Participant's required beginning date. The RMD for other distribution calendar years, including the RMD for the distribution calendar year in which the Participant's required beginning date occurs, will be made on or before December 31 of that distribution calendar year.
- **(3) Life Expectancy.** Life expectancy as computed by use of the Single Life Table in U.S. Treas. Reg. § 1.401(a)(9)-9.
- (4) Participant's Account Balance. The account balance as of the last valuation date in the calendar year immediately preceding the distribution calendar year (valuation calendar year) increased by the amount of any contributions made and allocated to the account balance as of dates in the valuation calendar year after the valuation date and decreased by distributions made in the valuation calendar year after the valuation date. The account balance for the valuation calendar year includes any Rollover Contributions or Transfers to the Plan either in the valuation calendar year or in the distribution calendar year if distributed or transferred in the valuation calendar year.
- (5) Required Beginning Date. A Participant's required beginning date is the April 1 of the calendar year following the later of: (1) the calendar year in which the Participant attains age 70½ (or age 72 if Participant attains age 70½ after December 31, 2019), or (2) the calendar year in which the Participant retires under Code § 401(a)(9). A Rehired Annuitant is treated as a retired Participant for purposes of determining his or her Required Beginning Date.
- **4.4. Death Benefits.** Upon the death of the Participant, the Participant's Account must be paid in accordance with Section 4.2 (Time and Method of Payment of Account). If a Participant dies while performing qualified military service (as defined in Code § 414(u)), the survivors of the Participant are entitled to any additional benefits (other than benefit accruals relating to the period of qualified military service) provided under the Plan as if the Participant had resumed and then terminated employment on account of death. In no event may death benefits commence later that permissible under the required minimum distribution rules described in Section 4.3, and Code § 401(a) and its underlying Treasury Regulations.

- **4.5. Distributions Prior to Severance From Employment.** Notwithstanding the distribution restrictions in Section 4.1 (Distribution Restrictions), the Plan permits the following in-service distributions from a Participant's vested Account balance.
  - (A) Unforeseeable Emergency. In the event of a Participant's unforeseeable emergency, the Plan Administrator may allow a distribution to a Participant who has not incurred a severance from employment (or who has incurred a severance from employment but will not begin to receive payments until some future date and continues to have an Account balance under the Plan). The Plan Administrator will establish a policy for determining whether an unforeseeable emergency exists. An unforeseeable emergency is a severe financial hardship of a resulting from:
    - (1) illness or accident of the Participant, or the Participant's\_spouse or dependent (as defined in Code § 152(a));
    - (2) loss of the Participant's property due to casualty;
    - (3) the need to pay for the funeral expenses of the Participant's spouse or dependent (as defined in Code § 152(a)); or,
    - (4) other similar extraordinary and unforeseeable circumstances arising from events beyond the Participant's control, or which applicable law may define as an unforeseeable emergency.

The Plan Administrator will not allow the Participant to withdraw more than the amount reasonably necessary to satisfy the emergency need, which may include amounts necessary to pay taxes or penalties on the distribution. The Plan Administrator will not allow payment to the extent the Participant\_may relieve the financial hardship by cessation of deferrals under the Plan, through insurance or other reimbursement, or by liquidation of the individual's assets to the extent such liquidation would not cause severe financial hardship.

- **(B) Distribution of Rollover Contributions.** The Plan permits a Participant to request and to receive a distribution from the Participant's Account attributable to Rollover Contributions, plus earnings thereon, before the Participant has a distributable event under Section 4.1 (Distribution Restrictions).
- (C) Age-Based In-Service Distributions. Effective January 1, 2020, the Plan permits a Participant to request and receive distributions from the Participant's Account on or after the date Participant attains age 59½. Prior to January 1, 2020, the Plan permitted Participants to request and receive distributions from the Participant's Account on or after attaining age 70½. Such age-based distributions must be requested by the Participant and may only be made in the frequency, form and manner provided by the Plan Administrator.

- **(D) Participant Loans.** The Plan may establish and maintain a Participant loan program in accordance with the provisions of Section 5.3(BA) (Loan Policy) of the Plan and Code Section 72(p).
- (E) Qualified Birth or Adoption Distributions (QBOADs). The Plan permits a Participant to request and receive QBOADs. QBOADs must be requested by the Participant and may only be made from the Plan in the frequency, form and manner provided by the Plan Administrator. Further, Participants may have an opportunity to repay a QBOAD if the Participant is eligible to make a rollover into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder. This Section 4.5(E) is effective June 1, 2021.
- **(F) Coronavirus-Related Distributions (CRDs).** Participants may request and receive a Coronavirus-Related Distribution (CRD). CRDs must be requested and made in the form and manner provided by the Plan Administrator.
  - **(1)** "Coronavirus-Related Distribution" Defined. A CRD means a distribution from the Plan made between January 1, 2020 and December 31, 2020, to a Participant:
    - (i) who is diagnosed with the virus SARS-CoV-2 or the coronavirus disease 2019 (collectively, "COVID-19") by a test approved by the Centers for Disease Control and Prevention; or
    - (ii) whose spouse or dependent is diagnosed with COVID-19 by a test approved by the Centers for Disease Control and Prevention; or
    - (iii) who experiences adverse financial consequences as a result of a spouse or Participant being quarantined, furloughed or laid off, or having work hours reduced due to COVID-19; or being unable to work due to lack of childcare due to COVID-19 or closing or reducing hours of a business owned or operated by the spouse or Participant, or a member of the Participant's household (one who shares Participant's principal residence) due to COVID-19; or having a reduction in pay due to COVID-19 or having a job offer rescinded or start date for a job delayed due to COVID-19.
  - (2) Aggregate Dollar Limit. The aggregate amount of CRDs to a Participant may not exceed \$100,000 from all plans sponsored by the Employer.
  - (3) Re-Contribution of CRDs. Participants may have an opportunity to repay some or all of the Participant's CRD if completed within 3 years of the distribution, meets the re-contribution requirements in § 2202(a) of the SECURE Act, and the Participant is eligible to make a rollover contribution into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder.
  - **(4) Participant Certification.** The Plan Administrator may rely on a Participant's certification that he or she satisfies the conditions for a CRD.

- **(G) Pro Rata Distribution.** Unless the Plan Administrator provides otherwise, and only in accordance with applicable law, in-service distributions made under this Section 4.7 shall be made from Participant's Account on a pro rata basis across eligible contribution sources and investment options.
- **4.6. Distributions Under Qualified Domestic Relations Orders (QDROs).** The Plan shall recognize and comply with a court order that the Plan Administrator (or designee thereof) determines, in its sole discretion, meets the requirements of Code § 414(p). In order to be approved, any such order must also comply with the provisions of this Section 4.6 and any additional policies or procedures established by the Plan Administrator.
  - (A) Time and Method of Payment. This Plan specifically permits distribution to an alternate payee under a QDRO at any time, notwithstanding any contrary Plan provision and irrespective of whether the Participant has attained his or her earliest retirement age (as defined under Code § 414(p)) under the Plan. A distribution to an alternate payee prior to the Participant's attainment of earliest retirement age is available only if the QDRO specifies distribution at that time or permits an agreement between the Plan and the alternate payee to authorize an earlier distribution. Nothing in this Section 4.6 gives a Participant a right to receive distribution at a time the Plan otherwise does not permit nor authorizes the alternate payee to receive a form of payment the Plan does not permit.
  - **(B) QDRO Procedures.** The Plan Administrator will establish reasonable procedures to determine the qualified status of a domestic relations order. Upon receiving a domestic relations order, the Plan Administrator (or its designee) promptly will notify the Participant and any alternate payee named in the order, in writing, of the receipt of the order and the Plan's procedures for determining the qualified status of the order. Within a reasonable period of time after receiving the domestic relations order, the Plan Administrator must determine the qualified status of the order and must notify the Participant and each alternate payee, in writing, of the Plan Administrator's determination. The Plan Administrator must provide notice under this paragraph by mailing to the individual's address specified in the domestic relations order.
  - (C) Accounting. If any portion of the Participant's Account balance is payable under the domestic relations order during the period the Plan Administrator is making its determination of the qualified status of the domestic relations order, the Plan Administrator must maintain a separate accounting of the amounts payable. If the Plan Administrator determines the order is a QDRO within 18 months of the date amounts first are payable following receipt of the domestic relations order, the Plan Administrator (or its designee) will distribute the payable amounts in accordance with the QDRO. If the Plan Administrator does not make its determination of the qualified status of the order within the 18-month determination period, the Plan Administrator (or its designee) will distribute the payable amounts in the manner the Plan would distribute if the order did not exist and will apply the order prospectively if the Plan Administrator later determines the order is a QDRO.

**(D) Permissible QDROs.** A domestic relations order that otherwise satisfies the requirements for a qualified domestic relations order ("QDRO") will not fail to be a QDRO: (i) solely because the order is issued after, or revises, another domestic relations order or QDRO; or (ii) solely because of the time at which the order is issued, including issuance after or after the Participant's death.

# 4.7. Direct Rollover of Eligible Rollover Distributions.

- (A) Participant Election. A Participant (including for this purpose, a former Employee) may elect, at the time and in the manner the Plan Administrator prescribes, to have any portion of his or her eligible rollover distribution from the Plan paid directly to an eligible retirement plan specified by the Participant in a direct rollover election. For purposes of this election, a "Participant" includes as to their respective interests, a Participant's surviving spouse and the Participant's spouse or former spouse who is an alternate payee under a QDRO.
- **(B) Rollover and Withholding Notice.** At least 30 days and not more than 180 days prior to distribution of an eligible rollover distribution, the Plan Administrator must provide a written notice (including a summary notice as permitted under applicable Treasury regulations) explaining to the distributee the rollover option, the applicability of mandatory twenty percent (-20%) federal withholding to any amount not directly rolled over, and the recipient's right to roll over within 60 days after the date of receipt of the distribution ("rollover notice").
- (C) Default Distribution or Rollover. In the case of a Participant who does not elect timely to roll over or to receive distribution of his or her Account, the Plan Administrator (or its designee) may distribute to the Participant (if Participant's Account balance is \$1,000 or less) or may directly roll over (if Participant's Account balance is greater than \$1,000) the Participant's Account in accordance with the Plan's rollover notice. The Plan's right to roll over Participant Account balance over \$1,000 to an IRA is effective as of the date of the close of the first regular legislative session of the legislative body with the authority to amend the Plan that begins on or after January 1, 2006. For purposes of the automatic rollover of Accounts over \$1,000, the Plan will treat a Participant's Roth Elective Deferral Account and the Participant's other accounts as held under two separate plans for purposes of applying the automatic rollover rules.
- **(D) Non-Spouse Beneficiary Rollover Right.** A non-spouse beneficiary who is a designated beneficiary under Section 4.3(E)(1)(Designated Beneficiary), by a direct trustee-to-trustee transfer ("direct rollover"), may roll over all or any portion of his or her distribution to an individual retirement account the beneficiary establishes for purposes of receiving the distribution. In order to be able to roll over the distribution, the distribution otherwise must satisfy the definition of an eligible rollover distribution.
  - (1) Certain Requirements Not Applicable. Although a non-spouse beneficiary may roll over directly a distribution as provided in this Section 4.7(D), the distribution is

- not subject to the direct rollover requirements of Code § 401(a)(31) (including the automatic rollover provisions of Code § 401(a)(31)(B)), the notice requirements of Code § 402(f) or the mandatory withholding requirements of Code § 3405(c). If a non-spouse beneficiary receives a distribution from the Plan, the distribution is not eligible for a "60-day" rollover.
- (2) Trust Beneficiary. If the Participant's named beneficiary is a trust, the Plan may make a direct rollover to an individual retirement account on behalf of the trust, provided the trust satisfies the requirements to be a designated beneficiary within the meaning of Code  $\S 401(a)(9)(E)$ .
- (3) RMDs Not Eligible For Rollover. A non-spouse beneficiary may not roll over an amount which is an RMD, as determined under applicable U.S. Treasury regulations and other IRS guidance. If the Participant dies before his or her required beginning date and the non-spouse beneficiary rolls over to an IRA the maximum amount eligible for rollover, the beneficiary may elect to use either the 5-year rule or the life expectancy rule, pursuant to U.S. Treas. Reg. § 1.401(a)(9)-3, A-4(c), in determining the RMDs from the IRA that receives the non-spouse beneficiary's distribution.
- **(E) Rollover Definitions.** The following definitions apply to this Section 4.7:
  - (1) Eligible Rollover Distribution. An eligible rollover distribution is any distribution of all or any portion of a Participant's Account, except an eligible rollover distribution does not include: (a) any distribution which is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's Designated Beneficiary, or for a specified period of ten years or more; (b) any Code § 401(a)(9) required minimum distribution; and, (c) any unforeseeable emergency distribution.
  - (2) Eligible Retirement Plan. An eligible retirement plan is an individual retirement account described in Code § 408(a), an individual retirement annuity described in Code § 408(b), an annuity plan described in Code § 403(a), a qualified plan described in Code § 401(a), an annuity contract (or custodial agreement) described in Code § 403(b), or an eligible deferred compensation plan described in Code § 457(b) and maintained by an employer described in Code § 457(e)(1)(A), which accepts the Participant's, the Participant's spouse or alternate payee's eligible rollover distribution. A Participant or beneficiary may elect to roll over directly an eligible rollover distribution to a Roth IRA described in Code § 408A(b). For this purpose, the term "eligible rollover distribution" includes a rollover distribution described in this Section 4.7. Notwithstanding the foregoing, a direct rollover of a distribution from Roth Elective Deferrals shall only be made to an Eligible Retirement Plan which includes Roth Elective Deferrals as described in Code § 408A, and only to the extent the

### rollover is permitted under the rules of Code § 402(c).

- **(3) Direct Rollover.** A direct rollover is a payment by the Plan to the eligible retirement plan specified by the distributee.
- **(4) Mandatory Distribution.** A mandatory distribution is an eligible rollover distribution without the Participant's consent before the Participant attains the later of age 62 or NRA per paragraph 3.5 (Special Catch-Up Contribution). A distribution to a beneficiary is not a mandatory distribution.

### ARTICLE V.

# PLAN ADMINISTRATOR - DUTIES WITH RESPECT TO PARTICIPANTS' ACCOUNTS ADMINISTRATION

- **5.1 Plan Administrator.** The District is the Plan Administrator and has delegated to the General Manager the powers, duties, and responsibilities of the Plan Administrator. The General Manager may delegate, in writing, some or all of the powers, duties and responsibilities of Plan Administrator position as necessary to adequately operate and maintain the Plan.
- **5.2 Term/Vacancy.** Any Plan Administrator designated by the District must accept the Plan Administrator role and responsibilities in writing and will serve until his or her successor is appointed. In case of a vacancy in the position of the Plan Administrator, the District will exercise any and all of the powers, authority, duties and discretion as the Plan Administrator pending the filling of the vacancy.
- **5.3 Powers and Duties.** The Plan Administrator shall have the following powers and duties:
  - (A) In General.
    - (1) To and receive input from Pselect a committee to assist the Plan Administrator with administration of the Plan;
    - (2) To select a secretary for the committee, who need not be a member of the committee;
    - (3)(1) To determine the rights of eligibility of an Employee to participate in the Plan and the value of a Participant's Accounts;
    - (4)(2) To adopt policies, rules and procedures and to create administrative forms necessary for the proper and efficient administration of the Plan provided the rules, procedures and forms are not inconsistent with the terms of the Plan or <a href="MWD Code The-Metropolitan Water District of Southern California Administrative Code">MWD Code The-Metropolitan Water District of Southern California Administrative Code</a>;
    - (5)(3) To construe and enforce the terms of the Plan and the policies, rules and regulations adopted to carry out the terms of the Plan;
    - (6)(4) To direct the distribution of a Participant's Accounts;

- (7)(5) To review and render decisions respecting an administrative claim for (or denial of such a claim for) a benefit under the Plan;
- (8)(6) To furnish the District with information required for tax or other purposes;
- (9)(7) To establish policies regarding distributions from the Plan;
- (10)(8) To establish policies regarding the receipt of Rollover Contributions and automatic rollover distributions;
- (11)(9) To establish a policy regarding the making and the receipt of Transfers;
- (12)(10) To establish a policy regarding Participant or Beneficiary direction of investment;
- (13) To establish an investment policy for the Plan and/or a policy for Participant or Beneficiary directed investments;
- [11] To undertake correction of any Plan failures as necessary to preserve eligible Plan status;
- (12) To act upon instructions from the DCIC to direct the Trustee regarding changes to the Plan's investment options and to execute proxy votes cast by the DCIC;
- (13) To request and receive input from the DCIC, when reasonable and appropriate, regarding non-investment related Plan features; the selection, performance, and retention of the record keeper, investment advisors, service providers, advisors, consultants, and auditor; and Plan amendments;
- (14) To serve as the secretary for the DCIC which includes the following duties: keeping and maintain the DCIC meeting minutes, executing documents on behalf of the DCIC, issuing formal written correspondence of the DCIC; and other such administrative support duties as the DCIC may reasonably delegate to the Plan Administrator; and
- (14)(15) To undertake any other action the Plan Administrator deems reasonable or necessary to administer the Plan.

The Plan Administrator shall have total and complete discretion to interpret and construe the Plan and to determine all questions arising in the administration, interpretation and application of the Plan. Any determination the Plan Administrator makes under the Plan is final and binding upon any affected person. The Plan Administrator shall coordinate with General Counsel concerning any litigation, compromise, or settlement of claims against the Plans in accordance with MWD Code Section 6433.

**(B)** Loan Policy Establishment. The Plan Administrator, in its sole discretion, may establish or amend a loan policy for making Plan loans (if permitted under the Plan) to Participants. If the Plan Administrator adopts a loan policy, the loan policy must be a written document and must include: (1) the identity of the person or positions authorized to administer the participant loan program; (2) the procedure for applying for a loan; (3) the criteria for

approving or denying a loan; (4) the limitations, if any, on the types and amounts of loans available; (5) the procedure for determining a reasonable rate of interest; (6) the types of collateral which may secure the loan; and (7) the events constituting default and the steps the Plan will take to preserve Plan assets in the event of default. A loan policy the Plan Administrator adopts under this Section 5.3(BA) is part of the Plan, except that the Plan Administrator may amend or terminate the policy without regard to Section 9.1 (Amendment By Employer/Sponsor) at any time.

- **(C) QDRO Policy Establishment.** The Plan Administrator will establish QDRO policies and procedures.
- **5.4 Without Additional Compensation.** The Plan Administrator and the members of the DCIAC will serve without additional compensation for services, but the District will pay all reasonable expenses of the Plan Administrator and the DCIAC.
- **5.5 Authorized Representative.** The Plan Administrator may authorize appropriate Human Resources or Finance staff, to sign on the Plan Administrator's behalf any Plan notices, directions, applications, certificates, consents, approvals, waivers, letters or other documents.
- **5.6 Individual Accounts/Records.** The Plan Administrator will maintain a separate Account in the name of each Participant to reflect the value of the Participant's Deferred Compensation under the Plan. The Plan Administrator will maintain records of its activities.
- **5.7 Valuation of Participant's Account.** The value of each Participant's Account consists of his or her accumulated Deferred Compensation, as of the most recent Accounting Date or any later date as the Plan Administrator may determine.
- 5.8 Account Administration, Valuation and Expenses.
  - (A) Individual Accounts. The Plan Administrator, as necessary for the proper administration of the Plan, will maintain, or direct the Trustee to maintain, a separate Account, or multiple Accounts, in the name of each Participant to reflect the Participant's Account balance under the Plan. The Plan Administrator will make its allocations of Earnings, or will request the Trustee to make such allocations, to the Accounts of the Participants as necessary to maintain proper Plan records and in accordance with the applicable: (i) Contribution Types; (ii) allocation conditions; (iii) investment account types; and (iv) Earnings allocation methods. The Plan Administrator may also maintain, or direct the Trustee to maintain, a separate temporary Account for Participant forfeitures which occur during a Plan Year, pending their accrual and allocation in accordance with the Plan terms, or for other special items as the Plan Administrator determines is necessary and appropriate for proper pPlan administration.
    - (1) By Contribution Type. The Plan Administrator will establish Plan Accounts for each Participant as necessary to reflect his or her Accounts attributable to the following Contribution Types and the Earnings attributable thereto: In-Plan Roth Rollovers, Pre-Tax Deferrals, Rollover Contributions, Roth Deferrals, Roth Rollover

<u>Contributions</u>, and Transfers. If additional contribution types are added to the Plan, the Plan Administrator will establish Plan Accounts for each Participant to reflect such new contribution types.

- **(2) By Investment Account Type.** The Plan Administrator will establish separate Accounts for each Participant as necessary to reflect his or her investment account types as described below:
  - (a) Participant-Directed Accounts. Participants may, in accordance with the provisions of the Plan, direct investment of all or a portion of the amounts allocated to an account. Participant-Directed Accounts are Accounts that are established and maintained for a Participant to invest in one or more assets that are not pooled assets held by the Trust. As the Plan Administrator determines, a Participant-Directed Account may provide for a limited number and type of investment options or funds. A Participant may have one or more Participant-Directed Accounts in addition to Segregated Accounts. A Participant-Directed Account is credited and charged with the Earnings. The Plan Administrator will designate how a Participant Account shall be invested in the absence of proper affirmative direction from the Participant. The Plan or Plan Administrator may designate a default fund under the Plan to deposit contributions on behalf of Participant who has not specified an investment choice under the Plan.
  - **(b) Segregated Accounts.** A Segregated Account that is established and maintained for a Participant: (i) to hold a QDRO amount; (ii) to prevent a distortion of Plan Earnings allocations; or (iii) for such other purposes as the Plan Administrator may direct. A Segregated Account receives all income it earns and bears all expense or loss it incurs. The assets of a Segregated Account shall be invested in a manner consistent with the purpose for which the Segregated Account was established.

As of each Valuation Date, the Plan Administrator must reduce a Segregated Account for any forfeiture arising after the Plan Administrator has made all other allocations, changes or adjustments to the Account (excluding Earnings) for the Valuation Period.

(3) Amount of Account. The amount of a Participant's Account, as determined by the Plan Administrator, is equal to the sum of all contributions, Earnings and other additions credited to the Account, less all distributions (including distributions to Beneficiaries and to alternate payees and also including disbursement of Plan loan proceeds), expenses and other charges against the Account as of a Valuation Date or other relevant date. For purposes of a distribution under the Plan, the amount of a Participant's Account balance is determined based upon its value on the Valuation Date immediately preceding or coinciding with the date of the distribution. If any or all Plan investment Accounts are participant-directed, the directing Participant's

Account balance consists of the assets held within the participant-directed Account and the value of the Account is determined based upon the fair market value of such assets.

- (4) Account Statements. As soon as practicable after the Accounting Date of each Plan Year, the Plan Administrator will deliver to each Participant (and to each Beneficiary) a statement reflecting the amount of his or her Account balance in the Trust as of the statement date or most recent Valuation Date. No Participant, except the Plan Administrator/Participant or Trustee/Participant, has the right to inspect the records reflecting the Account of any other Participant.
- **5.9 Participant Account Charged.** The Plan Administrator will charge all distributions made to a Participant or to his or her Beneficiary or transferred under Section 8.3 (Transfers) from his or her Account, against the Account of the Participant when made.
- **5.10** Participant Direction of Investment. Subject to authorization by the Plan's <code>iI</code>nvestment <code>pP</code>olicy Statement and any terms, conditions and/or restrictions established by the Plan Administrator, Participants may elect to direct the investment or re-investment of the assets comprising the Participant's Account. Participants may direct the investment or re-investment of the assets comprising the Participant's Account in any investment options the Plan-Administrator DCIC makes available under the Plan and in accordance with the Plan's <code>iI</code>nvestment <code>pP</code>olicy Statement.

The Plan Administrator will account separately for any Account assets that are invested at the direction of the Participant or invested through a self-directed brokerage account.

The Participant's right to direct investment does not give the Participant any vested interest or secured or preferred position with respect to assets over which Participant has investment responsibility.\_Any assets in the Participant's Account for which the Participant has not made an investment election shall be invested in the Plan's default investment choice in accordance with the Plan's <u>iInvestment</u> <u>pPolicy Statement</u>.

- **5.11 Vesting.** Participants shall be 100% vested in their Accounts at all times.
- **5.12 Preservation of Eligible Plan Status.** The Plan Administrator may elect to sever from this Plan and to treat as a separate 457 plan, the Accounts of any Participants who have Excess Deferrals that the Plan Administrator has not corrected in accordance with Section 3.7 (Disbursement of Excess Deferrals) or in the case of any other Code § 457(b) failure that the District may not otherwise correct, and which failure would result in the Plan ceasing to be an eligible 457(b) plan. In such event, the Plan Administrator will take any necessary or appropriate action consistent with the District's maintenance of separate 457 plans and with preservation of tax-deferred status of this Plan.
- **5.13 Limited Liability.** The District will not be liable to pay plan benefits to a Participant in excess of the value of the Participant's Account as the Plan Administrator determines in accordance with the Plan terms. Neither the District nor the Plan Administrator will be liable for losses arising

from depreciation or shrinkage in the value of any investments acquired under this Plan.

- **5.14** Lost Participants. If the Plan Administrator is unable to locate any Participant or Beneficiary whose Account becomes distributable (a "lost Participant"), the Plan Administrator will apply the provisions of this Section 5.14.
  - (A) Attempt to Locate. The Plan Administrator will attempt to locate a lost Participant or Beneficiary and may use one or more of the following methods: (1) provide a distribution notice to the lost Participant or Beneficiary at his or her last known address by certified or registered mail; (2) use a commercial locator service, the internet or other general search method; (3) use the Social Security Administration or PBGC search program; or (4) use such other methods as the Plan Administrator believes reasonable and prudent.
  - (B) Failure to Locate. The Plan Administrator may declare a forfeiture of a benefit due a Participant or Beneficiary deemed to be lost (i.e., the Participant or Beneficiary cannot be found within 4 years of his or her required beginning date). If after a benefit is forfeited under this Section 5.14(B), the lost Participant or Beneficiary is located, the Plan will restore the forfeited benefit (unadjusted for gains or losses) to such Participant or Beneficiary within a reasonable time. However, if a lost Participant or Beneficiary has not been located by the time the Plan terminates, the forfeiture of such Participant's or Beneficiary's distributable benefit will be irrevocable. Notwithstanding the foregoing, the provisions of this Section 5.14(B) are not the exclusive means for the Plan Administrator to administer the Accounts of lost Participants and Beneficiaries. The Plan Administrator may address forfeitures using any available means provided under applicable law, including escheating the Participant's or Beneficiary's benefit to the state.
  - **(C) Reasonable Expenses.** The Plan Administrator may charge to the Account of a lost Participant the reasonable expenses incurred under this Section 5.14 and which are associated with the lost Participant's Account.
- **5.15 Plan Correction.** The Plan Administrator, in conjunction with other parties as appropriate, may undertake such correction of Plan errors as the Plan Administrator deems necessary, including but not limited to correction to maintain the Plan's status as an eligible 457(b) plan. The Plan Administrator under this Section 5.15 also may undertake Plan correction in accordance with any correction program that the Internal Revenue Service makes applicable to 457 plans.

### ARTICLE VI.

### BENEFICIARY DESIGNATIONS AND ENROLLMENT

**6.1 Beneficiary Designation.** A Participant from time to time may designate, in the form and manner set forth by the Plan Administrator, any person(s) (including a trust or other entity), contingently or successively, to whom the Plan will pay the Participant's Account in the event of death.

However, spousal consent is required in writing with respect to the distribution of death benefits from the Plan if the Participant designates someone other than the Participant's Spouse as a Beneficiary of more than fifty percent (50%) of the Participant's Account balance.

A Spouse's consent to the designation of more than fifty percent (50%) of Participant's Account to a non-spouse beneficiary must be provided at the time and in the form and manner specified by the Plan Administrator.

Subsequent beneficiary designations submitted to the Plan Administrator in good order revoke all designations filed prior to that date by the same Participant.

Participant's marriage or remarriage revokes any prior Beneficiary designations made by the Participant and a new Beneficiary designation form that complies with the consent requirements of this Section 6.1 (Beneficiary Designation) must be submitted.

Further, when designating a trust or other entity as a Beneficiary, the time and form of distribution available to such Beneficiaries may be limited to by the required minimum distribution rules of Code § 401(a)(9) and Section 4.3 (RMDs) of the Plan.

### 6.2 No Beneficiary Designation.

- (A) Effective for Participants who are active Employees of the District on or after August 1, 2018, if a Participant fails to name a Beneficiary in accordance with Section 6.1 (Beneficiary Designation), or if all the Beneficiaries named by a Participant predeceases the Participant, then the Plan Administrator will pay the Participant's remaining Account in accordance with Article IV (Time and Method of Payments of Benefits) in the following order of priority, to:
  - (1) The Participant's surviving spouse; or
  - (2) The Participant's children (including adopted children), in equal shares by right of representation (one share for each surviving child and one share for each child who predeceases the Participant with living descendants); and if none, then to:
  - **(3)** The Participant's estate.
- **(B)** If a Participant has a severance from employment before August 1, 2018, and has a beneficiary designation form dated prior to August 1, 2018, then the Plan Administrator will pay the Participant's remaining Account in accordance with Article IV (Time and Method of Payments of Benefits) to the Participant's estate.
- (C) If the Beneficiary survives the Participant, but dies prior to distribution of the Participant's entire Account, the remaining portion of the Participant's Account will be paid to the Participant's beneficiaries in accordance with, as applicable, Subsection 6.2 (No Beneficiary Designation) subsections (A) or (B) unless: (1) the Participant's Beneficiary designation election provides otherwise; or (2) the Beneficiary has timely and properly designated a Beneficiary under the procedures established by the Plan Administrator. A Beneficiary only may designate a Beneficiary for the Participant's

Account balance remaining at the Beneficiary's death, if the Participant has not previously designated a successive contingent Beneficiary and the Beneficiary's designation otherwise complies with the Plan terms.

## 6.3 Salary Reduction Agreement.

- **(A) General.** A Participant must elect to make Salary Reduction Contributions on a Salary Reduction Agreement form the Plan Administrator provides for this purpose. The Plan Administrator in a Salary Reduction Agreement may impose such other terms and limitations as the Plan Administrator may determine.
- **(B) Election Timing.** A Participant's Salary Reduction Agreement may not take effect earlier than the first day of the calendar month following the date the Participant executes the Salary Reduction Agreement and as to Compensation paid or made available in such calendar month.
- **(C) Sick, Vacation and Back Pay.** Participants may make Salary Reduction Contributions from accumulated sick pay, from accumulated vacation pay or from back pay. A Participant who will incur a severance from employment may execute a Salary Reduction Agreement before such amounts are paid or made available provided: (i) such amounts are paid or made available before the Participant incurs the Severance; and (ii) the Participant is an Employee in that month.
- (D) Modification of Salary Reduction Agreement. A Participant's Salary Reduction Agreement remains in effect until a Participant modifies it or ceases to be eligible to participate in the Plan. However, a Salary Reduction Agreement must stay in effect for at least one (1) month once it has been entered into by a Participant. A Participant may modify his or her Salary Reduction Agreement by executing a new Salary Reduction Agreement. Any modification will become effective no earlier than the beginning of the calendar month commencing after the date the Participant executes the new Salary Reduction Agreement. Filing a new Salary Reduction Agreement will revoke all Salary Reduction Agreements filed prior to that date. The Plan Administrator may restrict the Participant's right to modify his or her Salary Reduction Agreement in any Taxable Year.
- **Personal Data to Plan Administrator.** Each Participant and each Beneficiary of a deceased Participant must furnish to the Plan Administrator such evidence, data or information as the Plan Administrator considers necessary or desirable for the purpose of administering the Plan. The provisions of this Plan are effective for the benefit of each Participant upon the condition precedent that each Participant will furnish promptly full, true and complete evidence, data and information when requested by the Plan Administrator, provided the Plan Administrator advises each Participant of the effect of his or her failure to comply with its request.
- **6.5 Address for Notification.** Each Participant and each Beneficiary of a deceased Participant must file with the Plan Administrator from time to time, in writing, contact information and any changes to such contract information, including but not limited to, his or her electronic and physical

address, phone number and any changes thereto. Any communication, statement or notice addressed to a Participant, or Beneficiary, at his or her last address filed with the Plan Administrator, or as shown on the records of the District, binds the Participant, or Beneficiary, for all purposes of this Plan.

entitled to a Plan distribution is not able to care for his or her affairs because of a mental condition, a physical condition, or by reason of age and the Plan Administrator (or its designee) is directed to make the distribution to the Participant's or Beneficiary's guardian, conservator, trustee, custodian (including under a Uniform Transfers or Gifts to Minors Act) or to his or her attorney-in-fact or to other legal representative upon furnishing evidence of such status satisfactory to the Plan Administrator (or its designee), the Plan Administrator shall make such distribution. The Plan Administrator will not under any circumstance have any liability with respect to payments so made nor any duty to make inquiry as to the competence of any person entitled to receive payments under the Plan.

### ARTICLE VII.

### TRUST PROVISIONS FUNDS

**7.1. Establishment of Trust.** Consistent with Code §-section 457(g) and applicable Treasury regulations, the all Plan's assets are required towill be held and invested in the ‡Trust in accordance with this Plan and the Trust Agreement. The Trust will be established pursuant to a written Trust Agreement that constitutes a valid trust under the laws of the State of California.

The District may enter into a Trust Agreement with the Trustee providing for the investment of the Trust and prescribing the powers, duties, obligations, and functions of the Trustee with respect to the Plan. The Trust Agreement will form a part of this Plan, and any rights and benefits of any Participant or Beneficiary under the Plan whose Accounts are invested in the Trust will be subject to the terms of the Trust Agreement. However, in the case of any conflict between the Plan and the Trust Agreement, the Plan will govern. The Trustee must be one or more banks or other fiduciaries appointed by the District to hold and administer assets of the Plan.

The Trust shall be operated for Trustee must hold the Plan assets in the Trust pursuant to the terms of the Trust Agreement for the exclusive benefit of Participants and their Beneficiaries and te-for defraying reasonable expenses of administering the Plan. Except as provided in Section 3.1, no assets-part of the Plan shall Trust fund will inure to the benefit of the Employer prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries.

The <u>Trust assets shall be held Trustee must hold</u>, managed, administered, valued, invested, reinvested and accounted for <u>by</u> the <u>Trustee Trust assets</u> in accordance with the terms of the Planand the Trust Agreement.

The In its discretion, the Plan Administrator may recommend that the District may further terminate the Trustee, appoint a nondiscretionary tnew Trustees, and authorize the Plan Administrator to carry out or assistinstruct the Trustee with some or all of the duties and

responsibilities delegated to the transfer assets to a successor Trustee or Trustees. A nondiscretionary trustee is authorized to perform the duties delegated to the nondiscretionary trustee or directed to carry out by the District, Trustee, Plan Administrator or Plan Participant, as applicable. The nondiscretionary trustee is liable for any losses caused directly or indirectly from its negligence, misconduct or failure to act in good faith or as a result of any act or omission where it has exerted discretion. An appointment of a nondiscretionary trustee and any agreement documents outlining the duties and responsibilities of a nondiscretionary trustee shall be maintained as part of the Plan's records.

- **7.2.** Acceptance/Holding. The By executing the Trust Agreement, the Trustee accepts the Trust created under the Plan-Trust Agreement and agrees to perform the duties and obligations imposed under the Trust Agreement. The Trustee must hold and invest all Plan assets in the Trust under this Article VIII, all Deferred Compensation the Trust Agreement until paid in accordance with the Plan's terms.
- **7.3. Receipt of Contributions.** The Trustee is accountable to the District for the funds contributed to the Trust.
- 7.4. Investment of Plan AssetsDeferred Compensation Investment Committee. The District grants the Trustee-DCIC full power and authority to select an investment lineup consult and advise the Plan Administrator on the investment lineup for the, manage, and control Plan assets. The DCIC's membership, governance, and powers and duties are set forth in accordance with the Plan's Investment Policy StatementBylaws of the Deferred Compensation Investment Committee. The Trustee shall monitor performance members of the Plan's investment lineupDCIC will serve without additional compensation for services, and as needed, consult and advise but the Plan Administrator to make adjustments to investment lineup. The District grants the Trustee a consultation and advisory role with regard to the investmentwill pay all reasonable expenses of Trust assets, except with respect to a Trust asset under Participant direction of investment, in accordance with Section 7.12-(Participant Direction of Investment). The Trustee is authorized and empowered, but not by way of limitation, to exercise and perform the following powers, rights and duties: DCIC.
  - (A) To the extent permitted by the Plan's Investment Policy Statement, invest any part or all of the Trust in any common or preferred stocks, open-end or closed-end mutual funds, put and call options traded on a national exchange, United States retirement plan bonds, corporate bonds, debentures, convertible debentures, commercial paper, U. S. Treasury bills, U. S. Treasury notes and other direct or indirect obligations of the United States Government or its agencies, , limited partnerships, insurance contracts of any type, and to buy or sell options on common stock on a nationally recognized options exchange with or without holding the underlying common stock, as a prudent person would do under like circumstances. Other investments prudently made or retained by the Trustee of a kind constituting a diversification considered by law suitable for trust investments:

- **(B)** To retain in cash so much of the Trust as it may deem advisable to satisfy liquidity needs of the Plan and to deposit any cash held in the Trust in a bank account at reasonable interest;
- (C) To invest, if the Trustee is a bank or similar financial institution supervised by the United States or by a State, in any type of deposit of the Trustee (or a bank related to the Trustee within the meaning of Code § 414(b)) at a reasonable rate of interest or in a common trust fund as described in Code § 584, or in a collective investment fund, the provisions of which the Trust incorporates by this reference, which the Trustee (or its affiliate, as defined in Code § 1504) maintains exclusively for the collective investment of money contributed by the bank (or its affiliate) in its capacity as Trustee and which conforms to the rules of the Comptroller of the Currency;
- **(D)** To credit and distribute the Trust assets as directed by the Plan Administrator of the Plan. To borrow money, to assume indebtedness, extend mortgages and encumber by mortgage or pledge;
- (E) To compromise, contest, arbitrate or abandon claims and demands, in the Trustee's discretion:
- To have with respect to the Trust all of the rights of an individual owner, including the power to exercise any and all voting rights associated with Trust assets, to give proxies, to participate in any voting trusts, mergers, consolidations or liquidations, to tender shares and to exercise or sell stock subscriptions or conversion rights. Further, the Trustee is authorized to establish a proxy voting balance. All proxies will be voted in accordance with a proxy voting policy:
- **(F)** To hold any securities or other property in the name of the Trustee or its nominee, with depositories or agent depositories or in another form as it may deem best, with or without disclosing the trust relationship;
- **(G)** To retain any funds or property subject to any dispute without liability for the payment of interest, and to decline to make payment or delivery of the funds or property until a court of competent jurisdiction makes a final adjudication;
- (H) To furnish to the District and the Plan Administrator an annual statement of account showing the condition of the Trust and all investments, receipts, disbursements and other transactions effected by the Trustee during the Plan Year covered by the statement and also stating the assets of the Trust held at the end of the Plan Year, which accounts will be conclusive on all persons, including the District and the Plan Administrator, except as to any act or transaction concerning which the District or the Plan Administrator files with the Trustee written exceptions or objections within 90-days after the receipt of the accounts; and to begin, maintain or defend any litigation necessary in connection with the administration of the Trust, except that the Trustee

- will not be obliged or required to do so unless indemnified to the extent authorized by the District.
- (I) To perform any and all other acts in its judgment necessary or appropriate for the proper and advantageous management, investment and distribution of the Trust.
- 7.5. Records and Statements. The records of the Trustee pertaining to the Trust will be open to the inspection of the Plan Administrator and the District at all reasonable times and may be audited from time to time by any person or persons as the District or Plan Administrator may specify in writing. The Trustee will furnish the Plan Administrator whatever information relating to the Trust the Plan Administrator considers necessary. Investment Policy Statement. The DCIC will establish and maintain an Investment Policy Statement (IPS) for the Plan, and will periodically review that policy, as needed. In establishing and maintaining the IPS, the DCIC will review all pertinent Plan and Participant information and data and may consult with any professional advisors as it deems necessary or appropriate, and will consider matters such as the following:
  - (A) the Plan's objectives and the best method to accomplish those objectives, including the range of investment alternatives that should be available to Participants and their Beneficiaries in order to accomplish those objectives;
  - (B) the policies or procedures to be followed by the DCIC in the selection and monitoring of investment alternatives, including the benchmarks and other performance criteria, as well as investment fees and expenses; and
  - (C) the liquidity of Plan assets.
- 7.6. Fees and Expenses From Fund. The Trustee will pay from the Trust all fees and expenses the Trustee reasonably incurs in its administration of the Trust unless the District pays the fees and expenses. An employee Trustee may not receive compensation for performing duties as Trustee. A nonemployee trustee (whether discretionary or nondiscretionary) may receive reasonable annual compensation in an amount agreed upon by the Plan Administrator and Trustee-Investment

  Manager. The Board of Directors may appoint one or more investment managers to manage (including the power to acquire and dispose of) any assets of the Plan. Only a person who is a registered investment advisor under the Investment Advisors Act of 1940, a bank or an insurance company qualified to do business in the State of California may be an investment manager under the Plan. The Trustee may be an investment manager.
- 7.7. Professional Agents. The Trustee may employ and pay from the Trust reasonable compensation to agents, attorneys, accountants and other persons to advise the Trustee as in its opinion may be necessary. The Trustee may delegate any non-fiduciary duty vested in it to any agent, attorney, accountant or other person selected by the Trustee. However, as fiduciary the Trustee may act or refrain from acting on the advice or opinion of any agent, attorney, accountant or other person so selected. Investment Funds. The Trustee will establish and maintain investment funds designated by the DCIC or an investment manager designated by the Board of Directors. The assets of each investment fund will be managed by the Trustee or one or more investment

managers, as designated, except for Participant-directed investments that permit Participants to direct the investment of some or all of the assets in their Account in specific investments in accordance with Section 7.12 (Participant Direction of Investment).

- **7.8. Distribution of Cash or Property.** The Trustee may make distributions under the Plan in cash or property, or partly in each, at its fair market value as determined by the Trustee.
- 7.9. Resignation and Removal. The Trustee may resign at any time by providing the Employer and Plan Administrator its position by giving written notice of resignation at least thirty (30) days prior to the effective date of such resignation unless the District consents in writing to a shorter period. to the District and to the Plan Administrator. The Trustee's notice must specify the effective date of the Trustee's resignation, which date must be at least 30 days following the date of the Trustee's notice, unless the District consents in writing to shorter notice. The Employer District may remove a Trustee at any time, with or without cause, by providing written notice to the Trustee. by giving written notice to the affected party. The District's notice must specify the effective date of removal.
- 7.10. Successor Trustee/Appointment. In the event of Upon the resignation, or the removal, death or incapacity of a Trustee, where no other Trustee continues to serveice, the District must either appoint a successor Trustee or become the successor Trustee to continue the Plan. Upon accepting such appointment, a successor Trustee will have all the powers, rights, and duties conferred upon the preceding Trustee. If it intends to continue the Plan. If two or more persons hold the position of Trustee, in the event of the removal of one such person, during any period the selection of a replacement is pending, or during any period such person is unable to serve for any reason, the remaining person or persons will act as the Trustee. If the District fails to appoint a successor Trustee as of the effective date of the Trustee resignation, or removal, death or incapacity and no other Trustee remains, the District will be treated as the Trustee having appointed itself as Trustee and as having accepted the appointment. filed written acceptance of appointment as successor Trustee with the former Truste. During such period, the Trust continues to be in existence and legally enforceable, and the assets of the Plan shall continue to be protected by the provisions of the Trust.
- 7.11. Valuation of Trust. The Trustee will value the Trust as of each Accounting Date to determine the fair market value of the Trust assets. The Trustee will value the Trust on such other date(s) the Plan Administrator may direct.
- 7.12.7.8. Participant Direction of Investment. Consistent with Article V, the Trustee DCIC may consent in writing to permit Participants in the Plan to direct the investment to their Accounts under the Trust.

The Plan Administrator will advise the Trustee of the portion of the Trust credited to each Participant's Account under the Plan, and subject to such Participant direction.

As a condition of Participant direction, the <u>Trustee DCIC</u> may impose such conditions, limitations and other provisions as the <u>Trustee DCIC</u> may deem appropriate and as are consistent with policies created under the Plan's <u>iInvestment pPolicy Statement</u> and Article V.

The Trustee will report to the <u>DCIC and Plan Administrator</u> the net income, gain or losses incurred by each participant-directed Account separately from the net income, gain or losses incurred by the general Trust during the Plan Year.

**7.13. Invalidity of Any Trust Provision.** If any clause or provision of this Article VII proves to be or is adjudged to be invalid or void for any reason, such void or invalid clause or provision will not affect any of the other provisions of this Article VII and the balance of the Trust provisions will remain operative.

7.14. Exclusive Benefit. The Trustee will hold all the assets of the Trust for the exclusive benefit of the Participants and their Beneficiaries and neither the District nor the Trustee will use or divertany part of the corpus or income of the Trust for purposes other than the exclusive benefit of the Participants and Beneficiaries of the Plan. The District will not have any right to the assets held by the Trustee and the Trust assets will not be subject to the claims of the District's creditors or, except as provided in Section 4.6 (Distributions Under QDROs), of the creditors of any Participant or Beneficiary. No Participant or Beneficiary shall have any right to sell, assign, transfer or otherwise convey his or her Account or any interest in his or her Deferred Compensation. Notwithstanding the foregoing, the Plan Administrator may pay from a Participant's or Beneficiary's Account the amount the Plan Administrator finds is lawfully demanded under a levy issued by the Internal Revenue Service with respect to that Participant or Beneficiary or is sought to be collected by the United States Government under a judgment resulting from an unpaid tax assessment against the Participant or Beneficiary. The Trust created under the Plan is irrevocable and its assets will not inure to the benefit of the District.

7.15-7.9. Group Trust Authority. Notwithstanding any contrary provision in thisSubject to the Plan Administrator's direction, the Trustee may, unless restricted in writing by commingle, for investment or administration purposes, the Plan Administrator, transfer assets (or portion of the assets) of the Plan to a Trust in any common trust fund, collective investment fund, master trust, or group trust that is operated or maintained exclusively for, but only if the commingling and collective investment of monies provided that the funds in the group commingled trust consist exclusively of trust assets held under plans qualified under Code § 401(a), individual retirement accounts that are exempt under Code § 408(e), and eligible governmental plans that meets satisfies the requirements of Code § 457(b). For this purpose, a trust includes a custodial account that is treated as a trust under Code § 401(f) or under Code § 457(g)(3). For purposes of valuation, the value of the interest maintained Revenue Ruling 81-100 (as modified by the Plan in such group trust shall Revenue Ruling 2004-67 and Revenue Ruling 2011-1) to be the fair market value of the portion of the group trust held for Plan, determined in accordance with generally recognized valuation procedures tax exempt.

**7.16 Documents and Instruments.** The Trustee is authorized to execute, acknowledge and deliver all documents of transfer and conveyance, receipts, releases, and any other instruments that the Trustee deems necessary or appropriate to carry outits powers, rights and duties hereunder.

- 7.17 More Than One Trustee. If the Plan has more than one person acting as Trustee, the Trustees may allocate the Trustee responsibilities by mutual agreement. The Trustees may agree to make decisions by a majority vote or may permit any one of the Trustees to make any decision, undertake any action or execute any documents affecting this Trust without the approval of the remaining Trustees. The Trustees may agree to the allocation of responsibilities in a separate trust-agreement or other binding document.
- 7.18 Trustee Duties Limited. The duties and obligations of the Trustee shall be limited to those imposed upon it by this Plan document, the applicable Trust Agreements, applicable law or as subsequently agreed upon by the parties. Responsibility for administrative duties required under the Plan or applicable law not imposed upon or agreed to by the Trustee shall rest with the Plan Administrator and Employer.
- **7.19 Appointment of Custodian.** The Plan Administrator may appoint a Custodian to hold all or any portion of the Plan assets. A Custodian has the powers, rights and responsibilities similar to those of a nondiscretionary trustee unless otherwise provided in a separate agreement.
- 7.20 <u>Modification of Trust Provisions</u>. The Employer may amend the administrative trust or custodial provisions under this Plan (such as provisions relating to investments and the duties of trustees), provided the amended provisions are not in conflict with any other provision of the Plan and do not cause the Plan to fail to be eligible under Code § 457(b). The Employer may document any amendment modifying the trust or custodial provisions under this Plan or other overriding language in an Addendum to the Plan.

### ARTICLE VIII.

### AMENDMENT, TERMINATION AND TRANSFERS

- **8.1. Amendment by Employer.** The Employer has the right, at any time and from time to time, to amend the Plan in whole or in part. Any amendment of the Plan by the Employer shall be expressed in an instrument executed by the Employer on the order of its Board of Directors and filed with the Trustee. Further, the Plan Administrator is also authorized, at any time and from time to time, to amend the Plan in whole or in part, to ensure compliance with applicable laws and regulations and further the administration of the Plan. The Plan Administrator may not amend the Plan to effectuate its termination.
  - **(A) Amendment Purposes Limited.** No Plan amendment may authorize or permit any portion of the assets held under the Plan to be used for or diverted to a purpose other than the exclusive benefit of Participants or Beneficiaries, except to the extent such assets are used to pay taxes or administrative expenses of the Plan. Any Plan amendment also may not cause or permit any portion of the assets held under the Plan to revert to or become property of the District.
  - **(B) Method of Amendment.** The Plan Administrator may amend the Plan in writing and without amendment of the MWD Code.

- (C) Termination/Freezing of Plan. The Districts has the right, at any time, to terminate this Plan or to cease (freeze) further Salary Reduction Contributions to the Plan. Upon freezing Salary Reduction Contributions to the Plan, the provisions of the Plan (other than provisions permitting continued Salary Reduction Contributions) may remain operative until the Plan is terminated. Upon Plan termination, all Deferred Compensation in the Plan shall be distributed to Participants and Beneficiaries as soon as is reasonably practicable following termination. The exercise of such right by the District to terminate the 401(k) or 457(b) Plans does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to MWD Code Section 6102 (Employee Rights).
- **8.2. Purchase of Permissive Service Credit.** A Participant, prior to otherwise incurring a distributable event under Article IV (Time and Method of Payments of Benefits), may request a transfer of all or a portion of his or her Account to a governmental defined benefit plan (under Code § 414(d)) for: (a) the purchase of permissive service credit (under Code § 415(n)(3)(A)) under such plan, or (b) the repayment of contributions and earnings previously refunded with respect to a forfeiture of service credited under the plan (or under another governmental plan within the same State) to which Code § 415 does not apply by reason of Code § 415(k)(3). The Participant must follow the transfer request procedures established by the Plan Administrator.

### ARTICLE IX.

### MISCELLANEOUS

- 9.1. No Assignment or Alienation. A Participant or Beneficiary does not have the right to commute, sell, assign, pledge, transfer or otherwise convey or encumber the right to receive any payments under the Plan or Trust and the Plan Administrator and the Trustee will not recognize any such anticipation, assignment, or alienation. The payments and the rights under this Plan are non-assignable and nontransferable. Furthermore, except as provided in Section 4.6 (Distributions Under QDROs), a Participant's or Beneficiary's interest in the Trust is not subject to attachment, garnishment, levy, execution or other legal or equitable process. Notwithstanding the foregoing, the Plan Administrator may pay from a Participant's or Beneficiary's Account the amount the Plan Administrator finds is lawfully demanded under a levy issued by the Internal Revenue Service with respect to that Participant or Beneficiary or is sought to be collected by the United States

  Government under a judgment resulting from an unpaid tax assessment against the Participant or Beneficiary.
- **9.2. Effect on Other Plans.** This Plan does not affect benefits under any other retirement, pension, or benefit plan or system established for the benefit of the District's Employees, and participation under this Plan does not affect benefits receivable under any such plan or system, except to the extent provided in such plan or system.

- **9.3. Word Usage.** Words used in the masculine will apply to the feminine where applicable, and wherever the context of the Plan dictates, the plural will be read as the singular and the singular as the plural.
- **9.4. Governing Law.** The provisions of this Plan shall be construed, administered, and enforced in accordance with the provisions of applicable Federal law, and, to the extent applicable, the laws of the state of California.
- **9.5 Employment Not Guaranteed.** Nothing contained in this Plan, or any modification or amendment to the Plan, or in the creation of any Account, or the payment of any benefit, gives any Employee, Participant or Beneficiary any right to continue employment, any legal or equitable right against the District, the Plan Administrator, the Trustee, <a href="the DCIC">the DCIC</a>, any other Employee of the District, or any agents thereof except as expressly provided by the Plan.
- 9.6 Notice, Designation, Election, Consent, and Waiver. All notices under the Plan and all Participant or Beneficiary designations, elections, consents or waivers must be in writing and made in the form and manner the Plan Administrator specifies or otherwise approves. To the extent permitted by Treasury regulations or other applicable guidance, any Plan notice, election, consent or waiver may be transmitted telephonically or electronically. Any person entitled to notice under the Plan may waive the notice or shorten the notice period except as otherwise required by the Code or other regulations or pronouncements.
- 9.7 Risk of Loss and Indemnities Indemnification. In accordance with To the extent permitted by law, tThe MWD Code, the the District will indemnify, defend and hold harmless the members of the Board of Directors, the the General Manager, the Plan Administrator, or any Employees who hashave administrative responsibility under the Plan, any employee or formeremployee, board member or former board member of the Districtand the Trustee Deferred Compensation Investment Committee or any member thereof, except to the extent such person has acted with gross negligence or has committed willIful misconduct, shall be indemnified and heldharmless by the District (to the extent not indemnified or held harmless under any liabilityinsurance contract or other indemnification arrangement with respect to the plan) from and againstrespect to any and all-liability, loss, damage or expense to which such person may be subject by reason of resulting from any act or omission made in good faith with respect to, except willful misconduct or gross negligence, in their official capacities in the administration of or management of the assets of the Plan, including attorney, accountant and advisory fees and all otherall expenses reasonably incurred in that person'stheir defense, in the event the District fails to provide such defense after having been requested in writing to do so. The provisions of this Section 9.7 shall not be construed to in any way restrict the privileges and immunities that the District and its-Employees are otherwise entitled to under Government Code Sections 818.8 and 820.2.

### The Metropolitan Water District of Southern California

The amended and restated Plan is hereby adopted, effective

<u>January 1</u> , 202 <u>4</u> .
By:
Katano Kassaine
Assistant General Manager/Chief Financial Officer and
Interim Plan Administrator
Date Signed

# The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan

**January 1, 2024** 

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 457(b) DEFERRED COMPENSATION PLAN

# **ABBREVIATED**

# **TABLE OF CONTENTS**

		<u>Page</u>
BACKGROUND		
ARTICLE I:	DEFINITIONS	1
ARTICLE II:	ELIGIBILITY AND PARTICIPATION	7
ARTICLE III:	DEFERRAL CONTRIBUTIONS AND LIMITATIONS	8
ARTICLE IV:	TIME AND METHOD OF PAYMENT OF BENEFITS	14
ARTICLE V:	PLAN ADMINISTRATION	25
ARTICLE VI:	BENEFICIARY DESIGNATIONS AND ENROLLMENT	30
ARTICLE VII:	TRUST FUNDS	32
ARTICLE VIII:	AMENDMENT, TERMINATION AND TRANSFERS	37
ARTICLE IX:	MISCELLANEOUS	38

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 457(b) DEFERRED COMPENSATION PLAN

# DETAILED TABLE OF CONTENTS

		<u>Page(s)</u>
BACK	GROUND	1
ARTIC	CLE I. DEFINITIONS	1
1.1	Account	
1.2	Accounting Date	
1.3	Beneficiary	
1.4	Code	
1.5	Compensation	
1.0	(A) Compensation Taken Into Account	
	(B) Elective Contributions	
	(C) Post-Severance Compensation	
	(1) Timing	
	(a) Regular Pay	
	(b) Leave Cash-outs	
	(D) Salary Continuation for Disabled Participants	
	(E) Differential Wage Payments	
1.6	Deferred Compensation	
1.7	Deferred Compensation Investment Committee (DCIC)	
1.8	Effective Date	
1.9	Eligible Employee	
1.10	Employee	
1.11	Employer or District	
1.12	Excess Deferrals	
1.13	Excluded Employee	
1.14	Includible Compensation	
1.15	Independent Contractor	
1.16	Leased Employee	4
1.17	Normal Retirement Age	4
1.18	Participant	4
1.19	Plan	4
1.20	Plan Administrator	5
1.21	Plan Year	5
1.22	Qualified Birth or Adoption Distribution (QBOAD)	5
1.23	Recurrent Employee	5
1.24	Rehired Annuitant	5
1.25	Rollover Contribution	
1.26	Roth Elective Deferrals	
1.27	Salary Reduction Agreement	5

1.28	Salary Reduction Contribution						
1.29	Service						
	(A) Qualified Military Service	6					
	(B) Continuous Service	6					
	(C) Severance from Employment	<i>6</i>					
	(1) Employee	<i>6</i>					
	(2) Deemed Severance						
1.30	State						
1.31	Taxable Year	7					
1.32	Temporary Employee						
1.33	Transfer	7					
1.34	Trust						
1.35	Trust Agreement						
1.36	Trustee						
ARTIC	CLE II. ELIGIBILITY AND PARTICIPATION	7					
2.1	Eligibility						
2.2	Commencement of Salary Reduction Contributions	7					
2.3	Effect of Plan Restatement						
2.4	Participation Upon Re-Employment	8					
ARTIC	CLE III. DEFERRAL CONTRIBUTIONS AND LIMITATIONS	8					
3.1	Amount						
	(A) Contribution Formula						
	(B) Return of Contributions	8					
	(C) Time of Payment of Contribution	8					
3.2	Salary Reduction Contributions						
	(A) Deferral from Sick, Vacation and Back Pay						
	(B) Application to Leave of Absence and Disability						
	(C) Post-Severance Deferrals Limited to Post-Severance Compensation						
3.3	Normal Limitation	<u>9</u>					
3.4	Age 50 Catch-Up Contribution						
3.5	Special Catch-Up Contribution	<u>9</u>					
	(1) Underutilized Limitation	10					
	(2) Normal Retirement Age (NRA)	10					
	(3) Pre-2002 Coordination	10					
3.6	Rollover Contributions						
	(A) Operational Administration	10					
	(B) Separate Accounting	10					
	(C) Roth Elective Deferrals						
3.7	Distribution of Excess Deferrals						
	(A) Eligible Governmental 457(b) Plan						
	(B) Individual Limitation						
3.8	Benefit Accruals for Qualified Military Service	11					
3.9	Roth Elective Deferrals						
	(A) Elective Deferrals						

		(1)			e Deferrals				
		(2)	Roth	Elective De	eferrals	12			
	(B)	Ordei	ing Rul	es for Disti	ributions	12			
	(C)	Loans	S			12			
3.10		n Roth	Convers	sion		12			
	(A)								
	(B)								
	(C)				rocedure				
	(D)	Plan Loans							
	(E)	Treatment of In-Plan Roth Conversions							
	(F)	Spousal Consent							
	(G)								
	(H)	In-Pla	n Roth	Conversion	n Definitions	14			
		(1)	Direc	t In-Plan R	oth Rollover	14			
		(2)	In-Pla	n Roth Ro	llover	14			
		(3)	In-Pla	n Roth Ro	llover Account	14			
ARTI(	CLE IV.	TIME A	ND MET	HOD OF P.	AYMENT OF BENEFITS	14			
4.1	Distri	bution l	Restrict	ions		14			
4.2	Time	and Me	thod of	Payment o	f Account	14			
	(A)	Participant Election of Time and Method1							
	(B)	Autor	natic Ca	sh-Out Dis	tribution	15			
4.3	Requi	red Mir	imum I	Distribution	ns (RMDs)	15			
	(A)	Genei	ral Rule	S		15			
	(B)	Time and Manner of Distribution							
		(1)	Requ	Required Beginning Date					
		(2)	Death	ı of Particij	pant Before Distribution Begins	15			
			(a)	Spouse I	Designated Beneficiary	15			
			(b)	Non-Spo	ouse Designated Beneficiary	15			
			(c)	No Desig	gnated Beneficiary	15			
			(d)	Death of	f Spouse	16			
		(3)	Form	s of Distrib	oution	16			
	(C)	RMDs	S During	Participar	ıt's Lifetime	16			
		(1)	Amou	ınt of RMD	Each Distribution Calendar Year	16			
			(a)	Uniform	Life Table	16			
			(b)	Younger	· Spouse	16			
		(2)	Lifeti	me RMDs 7	Гhrough Year of Participant's Death	16			
	(D) RMI		Os After Participant's Death						
		(1)	Death	On or Afte	er Distributions Begin	16			
			(a)	Participa	ant Survived by Designated Beneficiary	16			
				(i) I	Participant's Life Expectancy	17			
				(ii) S	Spouse's Life Expectancy	17			
				(iii) l	Non-Spouse's Life Expectancy	17			
			(b)		gnated Beneficiary				
		(2)	Death	Before Da	nte Distributions Begin	17			
			(a)	Particin:	ant Survived by Designated Beneficiary	17			

			(b)	No Designated Beneficiary	18			
			(c)	Death of Surviving Spouse Before Distributions to Surviving				
				Spouse are Required to Begin	18			
			(d)	5-Year or Life Expectancy Rule; Possible Election				
	(E)	RMD D	efiniti	ons				
		(1)	Desig	nated Beneficiary	18			
		(2)	Distri	bution Calendar Year	18			
		(3)	Life E	xpectancy	19			
		(4)		cipant's Account Balance				
		(5)	Requi	ired Beginning Date (RBD)	19			
4.4	Deatl		-					
4.5	Distr	Distributions Prior to Severance From Employment						
	(A)			e Emergency				
	(B)			of Rollover Contributions				
	(C)			-Service Distributions				
	(D)	_		oans				
	(E)			th or Adoption Distributions (QBOADs)				
	(F)			Related Distributions (CRDs)				
	(- )	(1)		navirus-Related Distribution Defined				
		(2)		egate Dollar Amount				
		(3)		ontribution of CRDs				
		(4)		cipant Certification				
	(G)							
4.6		Distributions Under Qualified Domestic Relations Orders (QDROs)						
1.0	(A)							
	(B)		QDRO Procedures					
	(C)	Accounting						
	(D)	Permissible QDROs						
4.7				gible Rollover Distributions				
11,	(A)			lection				
	(B)			Withholding Notice				
	(C)			bution or Rollover				
	(D)	Non-Spouse Beneficiary Rollover Right						
		(1)		in Requirements Not Applicable				
		(2)		Beneficiary				
		(3)		S Not Eligible For Rollover				
	(E)			nitions				
	(1)	(1)		le Rollover Distribution				
		(2)	U	le Retirement Plan				
		(3)	_	t Rollover				
		(4)		atory Distribution				
		(*)	Hand					
				TRATION				
5.1	_							
5.2								
53	POME	are and Di	ıtiac		20			

	(A) In General	2!				
	(B) Loan Policy Establishment	20				
	(C) QDRO Policy Establishment	20				
5.4	Without Additional Compensation	20				
5.5	Authorized Representative	20				
5.6	Individual Accounts/Records	20				
5.7	Valuation of Participant's Account					
5.8	Account Administration, Valuation and Expenses					
	(A) Individual Accounts					
	(1) By Contribution Type	27				
	(2) By Investment Account Type	27				
	(a) Participant Directed Accounts	27				
	(b) Segregated Accounts	28				
	(3) Amount of Account					
	(4) Account Statements	28				
5.9	Participant Account Charged	28				
5.10	•					
5.11	<u>•</u>					
5.12						
5.13						
5.14	Lost Participants	29				
	(A) Attempt to Locate					
	(B) Failure to Locate					
	(C) Reasonable Expenses					
5.15	Plan Correction	30				
л реги	ICLE VI DENERICIADV DECICNATIONS AND ENDOLLMENT	24				
	ICLE VI. BENEFICIARY DESIGNATIONS AND ENROLLMENT.					
6.1	Beneficiary Designation					
6.2	No Beneficiary Designation					
6.3	Salary Reduction Agreement					
	(A) General					
	(B) Election Timing					
	(C) Sick, Vacation and Back Pay(D) Modification of Salary Reduction Agreement					
<i>(</i> 1	(D) Modification of Salary Reduction Agreement Personal Data to Plan Administrator					
6.4	Address for Notification					
6.5	Participant or Beneficiary Incapacitated					
6.6	Participant of Beneficiary Incapacitated					
ARTI	ICLE VII. TRUST FUNDS	32				
7.1	Establishment of Trust	32				
7.2	Acceptance/Holding	33				
7.3	Receipt of Contributions					
7.4	Deferred Compensation Investment Committee					
7.5	Investment Policy Statement					
	(A) The Plan's Objectives					
	(R) The Policies or Procedures	31				

	(C) The Liquidity of Plan Assets	35
7.6	Investment Manager	35
7.7	Investment Funds	
7.8	Participant Direction of Investment	35
7.9	Group Trust Authority	36
ARTI	ICLE VIII. AMENDMENT, TERMINATION AND TRANSFERS	37
8.1	Amendment by Employer	37
	(A) Amendment Purposes Limited	37
	(B) Method of Amendment	37
8.2	Termination/Freezing of Plan	38
8.3	Purchase of Permissive Service Credit	
ARTI	ICLE IX. MISCELLANEOUS	38
9.1	No Assignment or Alienation	38
9.2	Effect on Other Plans	38
9.3	Word Usage	
9.4	Governing Law	38
9.5	Employment Not Guaranteed	
9.6	Notice, Designation, Election, Consent, and Waiver	39
9.7	Risk of Loss and Indemnities	39

# The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan

### **BACKGROUND**

Pursuant to Board of Directors' Resolution Number 7683, The Metropolitan Water District of Southern California ("District") established effective November 1, 1976, the Deferred Compensation Plan (the "Plan").

The Plan has been amended from time to time. The Plan was restated by Board of Directors' Minute Item 34625, effective April 1, 1983. Board of Directors' Minute Item 42712 caused the Plan to be codified in Division VI, Chapter 7, Article 8 of The Metropolitan Water District of Southern California Administrative Code ("MWD Code") and named the "Metropolitan Water District of Southern California 1997 Deferred Compensation Plan" effective November 18, 1997. The Plan was restated in its entirety and renamed "The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan" effective August 1, 2018. The Plan was further amended and restated effective January 1, 2020. The Plan is hereby amended and restated effective January 1, 2024, except as otherwise stated herein.

The Plan was established, and is maintained, as an eligible governmental deferred compensation plan meeting the requirements of Internal Revenue Code (the "Code") § 457(b) and § 457(e)(1)(A) and Treas. Reg. §§ 1.457-3 through 1.457-10.

All contributions made to the Plan, all property and rights purchased with such amounts and all income attributable to such amounts shall be held in trust for the exclusive benefit of Participants, Beneficiaries and Alternate Payees under the Plan in accordance with Code § 457(g). Amounts contributed under the Plan are held and invested, until distributed in accordance with the terms of the Plan.

The purpose of the Plan is to attract and retain certain officers and employees to The Metropolitan Water District of Southern California.

### ARTICLE I.

### **DEFINITIONS**

- **1.1** Account. The separate Account(s) established and maintained under the Plan for a Participant's Deferred Compensation, an alternate payee's interest in the Plan under a QDRO, or as needed, for Beneficiaries described in Section 4.3.
- **1.2 Accounting Date.** The last day of the Plan Year. The Plan Administrator will allocate contributions and forfeitures for a particular Plan Year as of the Accounting Date of that Plan Year, and on such other dates, if any, as the Plan Administrator determines, consistent with the Plan's

allocation conditions and other provisions.

- **1.3 Beneficiary.** A person who the Plan or a Participant designates and who is or may become entitled to a Participant's Account upon the Participant's death. If the Plan does not have a valid Beneficiary designation on file, the Participant's Beneficiary will be determined in accordance with Article VI (Beneficiary Designations and Enrollment). A Beneficiary who becomes entitled to a benefit under the Plan remains a Beneficiary under the Plan until the Beneficiary's entire Plan benefit has been fully distributed. A Beneficiary's right to (and the duty to provide to the Beneficiary) information or data concerning the Plan does not arise until the Beneficiary first becomes entitled to receive a benefit under the Plan.
- **1.4 Code.** The Internal Revenue Code of 1986, as amended.
- **1.5 Compensation.** The definition of Compensation is W-2 Wages. W-2 Wages means wages for federal income tax withholding purposes, as defined under Code § 3401(a), plus all other payments to an Employee in the course of the Employer's trade or business, for which the Employer must furnish the Employee a written statement under Code §§ 6041(Information at Source), 6051(Receipts for Employees), and 6052(Returns Regarding Payment of Wages in the Form of Group-Term Life Insurance), but determined without regard to any rules that limit the remuneration included in wages based on the nature or location of the employment or services performed (such as the exception for agricultural labor in Code § 3401(a)(2)).
  - **(A) Compensation Taken Into Account.** For the Plan Year in which an Employee first becomes a Participant, the Plan Administrator will determine the allocation of contributions by taking into account the Employee's Compensation for the entire Plan Year.
  - **(B) Elective Contributions.** Compensation includes elective contributions excludible from the Employee's gross income under Code §§ 125(Cafeteria Plans), 132(f)(4)(De Minimis Fringe Defined), 402(e)(3)(Cash or Deferred Arrangements), 402(h)(1)(B)(Special Rules for Simplified Employee Pensions), 403(b)(Taxability of Beneficiary Under Annuity Purchased by Section 501(c)(3)), 408(p)(Simple Retirement Accounts) or 457.
  - **(C) Post-Severance Compensation.** Post-Severance Compensation is Compensation paid after a Participant's severance from employment with the District as further described in subsection (C)(1) below. Any other amount paid after severance from employment that is not described in subsection5(C)(1) is not Compensation even if payment is made within the time period described below. Post-Severance Compensation does not include severance pay, or payments under a nonqualified unfunded deferred compensation plan unless the payments would have been paid at that time without regard to severance from employment.
    - (1) Timing. Post-Severance Compensation includes the following items of compensation to the extent the Employer pays such amounts by the later of 2½ months after severance from employment or by the end of the Plan Year that

includes the date of such severance from employment:

- **(a) Regular Pay.** The payment of regular Compensation for services during the Participant's regular working hours, or Compensation for services outside the Participant's regular working hours (such as overtime or shift differential), bonuses, or other similar payments, but only if the payment would have been paid to the Participant prior to a severance from employment if the Participant had continued in employment with the Employer.
- **(b) Leave Cash-outs.** Payments for unused accrued bona fide sick, vacation or other leave, but only if the Employee would have been able to use the leave if employment had continued and if Compensation would have included those amounts if they were paid prior to the Participant's severance from employment.
- **(D) Salary Continuation for Disabled Participants.** Compensation paid to a Participant who is permanently and totally disabled (as defined in Code § 22(e)(3)) may be included in Compensation.
- **(E) Differential Wage Payments.** An individual receiving a Differential Wage Payment, as defined by Code § 3401(h)(2), shall be treated as an employee of the employer making the payment and the Differential Wage Payment shall be treated as compensation for purposes of Code § 457(b) and any other Code section that references the definition of compensation under Code § 415, including the definition of Includible Compensation as provided in Section 1.14.
- **1.6 Deferred Compensation.** The amount of Salary Reduction Contributions, Rollover Contributions and Transfers adjusted for allocable net income, gain or loss, in the Participant's Account.
- **1.7 Deferred Compensation Investment Committee (DCIC).** The committee described in Section 7.4.
- **1.8 Effective Date.** January 1, 2024, the effective date of this amendment and restatement of the Plan.
- **1.9 Eligible Employee.** An Employee who is not an Excluded Employee.
- **1.10 Employee.** An individual who provides services for the District as a common law employee of the District.
- **1.11 Employer or District.** The Metropolitan Water District of Southern California.
- **1.12 Excess Deferrals.** Salary Reduction Contributions to the Plan that exceed the Taxable Year maximum limitation of Code §§ 457(b) and (e)(18).
- **1.13 Excluded Employee.** Employees who retired and severed employment and are rehired as annuitants, as defined in California Government Code § 7522.56, to work for a limited duration of

time. For purposes of this Plan, Independent Contractors are also treated as Excluded Employees.

- **1.14 Includible Compensation.** For the Employee's Taxable Year, the Employee's total Compensation within the meaning of Code § 415(c)(3) paid to an Employee for services rendered to the Employer. Includible Compensation also includes:
  - (A) Salary Reduction Contributions under this Plan, compensation deferred under any other plan described in Code § 457, and any amount excludible from the Employee's gross income under Code §§ 401(k), 403(b), 125, or 132(f)(4) or any other amount excludible from the Employee's gross income for Federal income tax purposes;
  - **(B)** Payments for unused bona fide sick, vacation, or other leave, if the Employee would have been able to use the leave if employment had continued; and
  - **(C)** Amounts paid prior to severance, or paid within 2 1/2 months after the later of severance from employment or the end of the Plan Year that includes the date of severance.

The Employer will determine Includible Compensation without regard to community property laws.

- **1.15 Independent Contractor.** Any individual who performs service for the District and who the District does not treat as an Employee or a Leased Employee.
- **1.16 Leased Employee.** An Employee within the meaning of Code § 414(n).
- **1.17 Normal Retirement Age.** The age is designated by a Participant. A Participant may not designate a Normal Retirement Age (NRA) that is earlier than his or her CalPERS normal retirement age, which for classic members is age 55, and is age 62 for new members (as defined under PEPRA). Participants shall designate a NRA in the form and manner established by the Plan Administrator.
- **1.18 Participant.** An Employee (other than an Excluded Employee) who becomes a Participant in accordance with the provisions of Article II (Eligibility and Participation).
- **1.19 Plan.** The Metropolitan Water District of Southern California 457(b) Deferred Compensation Plan.
- **1.20 Plan Administrator.** The General Manager of the District or other person or entity designated by the District to be responsible for the administration and operation of the Plan.
- **1.21 Plan Year.** The calendar year ending on December 31.
- **1.22 Qualified Birth or Adoption Distribution (QBOAD).** A distribution to a Participant if made (a) during the 1-year period beginning on the date on which a child of the Participant is born or on which the legal adoption by the Participant of an eligible adoptee is finalized, (b) with respect to a child born or eligible adoptee the Participant includes the name, age, and the Taxpayer Identification Number of the child or eligible adoptee on the Participant's tax return for the taxable year in which the distribution is made, and (c) only to the extent such distribution from all plans maintained by the Employer to such Participant does not exceed \$5,000, in the aggregate, for each

birth or eligible adoptee. An eligible adoptee is any individual (other than a child of the Participant's spouse) who has not attained age 18 or is physically or mentally incapable of self-support.

- **1.23 Recurrent Employee.** An Employee who is hired for an indefinite period of time on an irregular basis, such as intermittent, emergency or on call.
- **1.24 Rehired Annuitant.** A rehired Employee who meets the definition of rehired annuitant in California Government Code Section 7522.56.
- **1.25 Rollover Contribution.** The amount of cash or property which an eligible retirement plan described in Code  $\S$  402(c)(8)(B) distributes to an eligible Employee or Participant in an eligible rollover distribution under Code  $\S$  402(c)(4) and which the eligible Employee or Participant transfers directly or indirectly to the Plan. A Rollover Contribution includes net income, gain or loss attributable to the Rollover Contribution. A Rollover Contribution into the Plan excludes after-tax Employee contributions, as adjusted for net income, gain or loss.
- 1.26 Roth Elective Deferrals. A Participant's Salary Reduction Contributions that are includible in the Participant's gross income at the time deferred and have been irrevocably designated as Roth Elective Deferrals by the Participant in his or her Salary Reduction Agreement. A Participant's Roth Elective Deferrals will be separately accounted for, as will gains and losses attributable to those Roth Elective Deferrals. However, forfeitures may not be allocated to such account. The Plan must also maintain a record of a Participant's investment in the contract (i.e., designated Roth contributions that have not been distributed) and the year in which the Participant first made a Roth Elective Deferral. 1.27 Salary Reduction Agreement. A written agreement between a Participant and the District, authorizing the District to reduce the Participant Compensation and contribute the amount as a Salary Reduction Contribution to the Participant's Account.
- **1.28 Salary Reduction Contribution.** The Pre-Tax Elective Deferrals and Roth Elective Deferrals the District deducts from a Participant's Compensation and contributes to the Plan pursuant to a Participant's Salary Reduction Agreement.
- **1.29 Service.** Any period of time the Employee is in the employ of the Employer. An Employee terminates Service upon incurring a severance from employment.
  - (A) Qualified Military Service. Service includes any qualified military service the Plan must credit for contributions and benefits in order to satisfy the crediting of Service requirements of Code § 414(u). A Participant whose employment is interrupted by qualified military service under Code § 414(u) or who is on a leave of absence for qualified military service under Code § 414(u) may elect to make additional Salary Reduction Contributions upon resumption of employment with the Employer equal to the maximum Salary Reduction Contributions that the Participant could have elected during that period if the Participant's employment with the Employer had continued (at the same level of Compensation) without the interruption of leave, reduced by the Salary Reduction Contributions, if any, actually made for the Participant during the period of

the interruption or leave. This right applies for five years following the resumption of employment (or, if sooner, for a period equal to three times the period of the interruption or leave). The Plan does not provide for matching or non-elective employer contributions as those terms are defined under Code § 457 and the underlying Treasury Regulations. Therefore, the Employer is not required under Code § 414(u) to make any other contributions to such a Participant. The Plan shall apply limitations of Article III (Deferral Contributions and Limitations) to all Salary Reduction Contributions under this paragraph with respect to the year to which the Salary Reduction Contributions relates.

- **(B) Continuous Service.** Service with the Employer during which the Employee does not incur a severance from employment.
- (C) Severance from Employment.
  - (1) Employee. An Employee has a severance from employment when the Employee ceases to be an Employee of the Employer. A Participant does not incur a severance from employment if, in connection with a change in employment, the Participant's new employer continues or assumes sponsorship of the Plan or accepts a Transfer of Plan assets as to the Participant.
  - (2) Deemed Severance. If a Participant performs service in the uniformed services (as defined in Code § 414(u)(12)(B)) on active duty for a period of more than 30 days, the Participant will be deemed to have a severance from employment solely for purposes of eligibility for distribution of amounts not subject to Code § 412. However, the Plan will not distribute such a Participant's Account on account of this deemed severance unless the Participant specifically elects to receive a benefit distribution hereunder. If a Participant elects to receive a distribution on account of this deemed severance, then no Salary Reduction Contributions may be made for the Participant during the 6-month period beginning on the date of the distribution. If a Participant would be entitled to a distribution on account of a deemed severance, and a distribution on account of another Plan provision, then the other Plan provision will control and the 6-month suspension will not apply.
- **1.30 State.** One of the 50 states of the United States or the District of Columbia, or a political subdivision of a State, or any agency or instrumentality of a State or its political subdivision. A State does not include the federal government or any agency or instrumentality thereof.
- **1.31 Taxable Year.** The calendar year or other taxable year of a Participant.
- **1.32 Temporary Employee.** An Employee who is hired for a specified limited time.
- **1.33 Transfer.** A transfer of assets from one eligible Code § 457(b) plan to another eligible Code § 457(b) plan which is not a Rollover Contribution and which is made in accordance with Section 8.3 (Transfers).

- **1.34 Trust.** The trust established pursuant to the Trust Agreement between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets in accordance with Code § 457(g).
- **1.35 Trust Agreement.** The separate agreement entered into by and between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets.
- **1.36 Trustee.** The person(s) or entity, and any successors thereto, named in the Trust Agreement, who is appointed by the District to hold, administer and distribute all or a portion of the Plan assets held in the Trust pursuant to the terms and conditions of the Trust Agreement.

#### ARTICLE II.

#### **ELIGIBILITY AND PARTICIPATION**

- **2.1 Eligibility.** The Plan does not require Eligible Employees to satisfy an age or service requirement in order to participate in the Plan. Each Eligible Employee becomes a Participant in the Plan in accordance with entry date rules described in Section 2.2.
- 2.2 Commencement of Salary Reduction Contributions. Generally, Salary Reduction Contributions for Eligible Employees commence on the first day of the month coinciding with or following the later of the Eligible Employee's date of hire or submission of a Salary Reduction Agreement. Notwithstanding the preceding sentence, Eligible Employees who are Temporary Employees or Recurrent Employees commence Salary Reduction Contributions in accordance with the Minimum Salary Reduction Contribution amounts specified in Section 3.2 as of his or her date of hire. Eligible Employees who are Temporary Employees or Recurrent Employees may change contribute an amount greater than (but not less than) the Minimum Salary Reduction Contribution amount described in Section 3.2 by completing a Salary Reduction Agreement. Changes to such Eligible Employee's Minimum Salary Reduction Contribution amount will take effect in accordance with the general timing rule described above.
- **2.3 Effect of Plan Restatement.** Each Eligible Employee who was a Participant on the day before the Effective Date of the Plan continues as a Participant in the Plan, irrespective of whether he/she satisfies the eligibility conditions in the amended and restated Plan.
- **2.4 Participation Upon Re-Employment.** A Participant who incurs a severance from employment may re-enter the Plan as a Participant as of his or her date of re-employment, so long as the Employee is not rehired in an Excluded Employee classification.

#### ARTICLE III.

#### **DEFERRAL CONTRIBUTIONS AND LIMITATIONS**

#### 3.1. Amount.

**(A) Contribution Formula.** For each Plan Year, the District will contribute to the Plan, on behalf of each Participant, the Participant's Salary Reduction Contribution and any other type and amount of contributions authorized under the Plan. Effective February 18, 2024,

Salary Reduction Contributions may include Roth Elective Deferrals in addition to pre-tax salary deferrals.

- **(B) Return of Contributions.** If any Salary Reduction Contribution (or any other type contributions authorized under the Plan) is made due to a mistake of fact, the Plan Administrator upon written request from the District shall return the contributions (adjusted for net income, gain or loss), within one (1) year after payment of the contribution.
- **(C) Time of Payment of Contribution.** The District shall deposit Salary Reduction Contributions to the Trust within a period that is not longer than is reasonable for the administration of Participant Accounts.
- **3.2. Salary Reduction Contributions.** The Plan permits Salary Reduction Contributions. Participants may designate a dollar or percentage amount by which to reduce his or her Compensation. Such Participant designations shall be made in accordance with a Salary Reduction Agreement.

Temporary Employees and Recurrent Employees are subject to mandatory minimum Salary Reduction Contribution amounts ("Minimum Salary Reduction Contributions"). Effective July 1, 2016, the Minimum Salary Reduction Contribution for Temporary Employees and Recurrent Employees are as follows:

Temporary and Recurrent Employee Category	Minimum Salary Reduction Contribution
Classic PERS Members	0.5%
New PERS Members	1.5%
Non PERS Members	7.5%

Provided, however, the Minimum Salary Reduction Contributions for Temporary or Recurrent Employees shall at all times be no less than the difference between 7.5% of the Temporary or Recurrent Employee's Compensation and his or her member contribution to CalPERS, if any. If at any time the Minimum Salary Reduction Contribution required of a Temporary Employee or Recurrent Employee varies from the minimums provided in the table above, the table shall automatically be adjusted.

Temporary Employees and Recurrent Employees may elect a different Salary Reduction Contribution amount, however, such election must be for an amount that is greater than the applicable Minimum Salary Reduction Contribution stated in this Section 3.2.

**(A) Deferral from Sick, Vacation and Back Pay.** Participants may elect to make Salary Reduction Contributions from accumulated sick pay, accumulated vacation pay or back pay in accordance with the established deduction timing calendar maintained by the District.

- **(B) Application to Leave of Absence and Disability.** Unless a Participant in his or her Salary Reduction Agreement elects otherwise, the Participant's Salary Reduction Agreement shall continue to apply during the Participant's leave of absence or the Participant's disability (as the Plan Administrator shall establish), if the Participant has Compensation other than imputed compensation or disability benefits.
- **(C) Post-Severance Deferrals Limited to Post-Severance Compensation.** Deferrals are permitted from an amount received following severance from employment only if the amount is Post-Severance Compensation.
- **3.3. Normal Limitation.** Except as provided in Sections 3.4 (Age 50 Catch-Up Contribution) and 3.5 (Special Catch-Up Contribution), a Participant's maximum Salary Reduction Contributions (excluding Rollover Contributions and Transfers) under this Plan for a Taxable Year may not exceed the lesser of:
  - **(A)** The applicable dollar amount as specified under Code § 457(e)(15) (or as the Commissioner of the Internal Revenue may prescribe), or
  - **(B)** 100% of the Participant's Includible Compensation for the Taxable Year.
- **3.4. Age 50 Catch-Up Contribution**. A Participant who will have attained age 50 before the close of the Taxable Year is eligible to make age 50 catch-up contributions for that Taxable Year and in subsequent Taxable Years in accordance with, and subject to the limitations of, Code § 414(v). If, for a Taxable Year, a Participant is eligible to make an Age 50 Catch-Up Contribution under this Section 3.4 and a special catchup contribution under Section 3.5 (Special Catch-Up Contribution) such Participant is entitled to the greater of the two catch up amounts determined under Section 3.4 and Section 3.5.
- **3.5. Special Catch-Up Contribution.** A Participant may elect to make a catch-up contribution for one or more of the Participant's last three Taxable Years ending before the Taxable Year in which the Participant attains NRA. In a Taxable Year in which a catch-up contribution is made, the Participant's maximum Salary Reduction Contributions may not exceed the lesser of:
  - (A) Twice the dollar amount under Section 3.3 (Normal Limitation), and
  - **(B)** The underutilized limitation under Section 3.5(A), as described below
    - (1) Underutilized Limitation. A Participant's underutilized limitation is equal to the sum of: (i) the normal limitation for the Taxable Year, and (ii) the normal limitation for each of the prior Taxable Years of the Participant commencing after 1978 during which the Participant was eligible to participate in the Plan and the Participant's Salary Reduction Contributions were subject to the Normal Limitation or any other Code § 457(b) limit, *less* the amount of Salary Reduction Contributions for each such prior Taxable Year, excluding Age 50 Catch-Up Contributions.
    - **(2) Normal Retirement Age (NRA).** A Participant may elect his or her applicable NRA for purposes of this Section 3.5. Participants may not elect a NRA that is earlier than

- age 65 or the age at which Participants have the right to retire and receive benefits under CalPERS without actuarial or other reduction which is age 55 for classic CalPERS members and age 62 for new CalPERS members as defined by PEPRA. Further, participants may not elect a NRA later than age 72.
- (3) Pre-2002 Coordination. In determining a Participant's underutilized limitation, the Plan Administrator, in accordance with U.S. Treas. Reg. § 1.457-4(c)(3)(iv), must apply the coordination rule in effect under now repealed Code § 457(c)(2). The Plan Administrator also must determine the Normal Limitation for pre-2002 Taxable Years in accordance with Code § 457(b)(2) as then in effect.
- **3.6. Rollover Contributions.** The Plan may accept Rollover Contributions.
  - (A) Operational Administration. The District, operationally and on a nondiscriminatory basis, may elect to limit an eligible Employee's or Participant's right to make a Rollover Contribution. Any Employee or Participant (or former Employee or Participant) with the District's written consent and after filing the forms prescribed by the Plan Administrator, may make a Rollover Contribution to the Trust. Before accepting a Rollover Contribution, the Participant (or eligible Employee) may be required to furnish satisfactory evidence that the proposed transfer is in fact a Rollover Contribution which the applicable tax rules permits an employee to make to an eligible retirement plan. The Plan Administrator, in its sole discretion, may decline to accept a Rollover Contribution of property which could: (1) generate unrelated business taxable income; (2) create difficulty or undue expense in storage, safekeeping or valuation; or (3) create other practical problems for the Trust.
  - (B) Separate Accounting. If the District permits Rollover Contributions, the Plan Administrator must account separately for: (1) amounts rolled into this Plan from an eligible retirement plan (other than from another eligible governmental 457(b) plan); and (2) amounts rolled into this Plan from another eligible governmental 457(b) plan. The Plan Administrator for purposes of ordering any subsequent distribution from this Plan may designate a distribution from a Participant's Rollover Contributions as coming first from either of (1) or (2) above if the Participant has both types of Rollover Contribution Accounts.
  - (C) Roth Elective Deferrals. The Plan may accept a rollover contribution of Roth Elective Deferrals only if it is a direct rollover from another Plan which permits Roth Elective Deferrals as described in Code § 402A(e)(1) and only to the extent the rollover is permitted under the rules of Code § 402(c). The District, operationally and on a uniform and nondiscriminatory basis, may decide whether to accept any such rollovers. If the Plan chooses to accept rollovers of Roth Elective Deferrals, the amounts must be directly rolled over into this Plan from another plan that is qualified under Code § 401(a), from a Code § 403(b) plan, or from an eligible governmental Code § 457(b) plan. The Plan must account separately for the Rollover Contribution, including Roth Elective Deferrals and the earnings thereon.

- **3.7. Distribution of Excess Deferrals.** In the event that a Participant has Excess Deferrals, the Plan will distribute to the Participant the Excess Deferrals and allocable net income, gain or loss, in accordance with this Section 3.7.
  - **(A) Eligible Governmental 457(b) Plan.** The Plan Administrator will distribute Excess Deferrals from the Plan as soon as is reasonably practicable following the Plan Administrator's determination of the amount of the Excess Deferral.
  - **(B) Individual Limitation.** If a Participant participates in another eligible 457(b) plan maintained by a different employer, and the Participant has Excess Deferrals, the Plan Administrator may, but is not required, to correct the Excess Deferrals by making a corrective distribution from this Plan.
- **3.8. Benefit Accruals for Qualified Military Service.** To the extent required under Code § 414(u), an Employee who returns to employment with the District following a period of qualified military service will receive any contributions, benefits, and service credit required under Code § 414 (u), provided the Employee satisfies all applicable requirements under the Code and regulations. In determining the amount of contributions under Code § 414(u), Plan Compensation will be deemed to be the compensation the Employee would have received during the period while in military service based on the rate of pay the Employee would have received from the District but for the absence due to military leave. If the compensation the Employee would have received during the leave is not reasonably certain, Plan Compensation will be equal to the Employee's average compensation from the District during the twelve (12) month period immediately preceding the military leave or, if shorter, the Employee's actual period of employment with the District.

For benefit accrual purposes, the Plan treats an individual who dies or becomes disabled (as defined under the terms of the Plan) while performing qualified military service with respect to the District as if the individual had resumed employment in accordance with the individual's reemployment rights under USERRA, on the day preceding death or disability (as the case may be) and terminated employment on the actual date of death or disability.

- **3.9. Roth Elective Deferrals.** Effective February 18, 2024, Participants may make Roth Elective Deferrals to the Plan. Unless otherwise indicated, Roth Elective Deferrals shall be treated in the same manner as Elective Deferrals. The District may, in operation, implement deferral election procedures provided such procedures are communicated to Participants and Participants are permitted to modify their elections at least once per Plan Year.
  - **(A) Elective Deferrals** means a contribution the Employer makes to the Plan pursuant to a Participant's Salary Reduction Agreement and includes Pre-Tax Elective Deferrals and Roth Elective Deferrals.
    - (1) Pre-Tax Elective Deferrals mean a Participant's Salary Reduction Contributions which are not includible in the Participant's gross income at the time deferred and have been irrevocably designated as Pre-Tax Elective Deferrals by the Participant in his or her Salary Reduction Agreement. A Participant's Pre-Tax Elective Deferrals will

- be separately accounted for, as will gains and losses attributable to those Pre-Tax Elective Deferrals.
- (2) Roth Elective Deferrals means a Participant's Salary Reduction Contributions that are includible in the Participant's gross income at the time deferred and have been irrevocably designated as Roth Elective Deferrals by the Participant in his or her Salary Reduction Agreement. A Participant's Roth Elective Deferrals will be separately accounted for, as will gains and losses attributable to those Roth Elective Deferrals. However, forfeitures may not be allocated to such account. The Plan must maintain a record of a Participant's Roth Elective Deferrals and the year in which the Participant first made a Roth Elective Deferral.
- **(B) Ordering Rules for Distributions.** The District may implement an ordering rule for withdrawals from Participant's accounts attributable to Pre-Tax Elective Deferrals and/or Roth Elective Deferrals. Absent an ordering rule established by the District in compliance with applicable law, withdrawals will be made on a pro rata basis from Participant's Pre-Tax Elective Deferrals, and Roth Elective Deferrals.
- **(C) Loans.** The Plan's loan policy or program may be modified to address Roth Elective Deferrals, including but not limited to, whether loans from Participant Roth Elective Deferral Accounts will be permitted.

#### 3.10. In-Plan Roth Conversion.

- (A) In-Plan Roth Conversion Account. Effective February 18, 2024, a Participant may elect to transfer some or all of his or her vested Account balance (other than his or her Roth Elective Deferral Account or Roth Rollover Account), and earnings thereon, to an In-Plan Roth Conversion Account without regard to whether the Participant satisfies the requirements for distribution in accordance with Article IV (Time and Method of Payments of Benefits) of the Plan. Any Participant election to make an In-Plan Roth Conversion during a taxable year is irrevocable and cannot be changed after the In-Plan Roth Conversion is completed. Generally, In-Plan Roth Conversion Accounts are subject to the same Plan provisions as Roth Elective Deferral Accounts. However, distribution restrictions that otherwise apply with respect to a specific contribution source will continue to apply if such contribution source is converted to Roth Elective Deferrals. The Plan Administrator will maintain such records as are necessary for the proper reporting of In-Plan Roth Conversions and will administer the In-Plan Roth Conversion Accounts in accordance with the Code, Treasury Regulations, and Plan provisions.
- **(B) Eligibility to Elect In-Plan Roth Rollover.** For purposes of determining eligibility for In-Plan Roth Rollovers, the Plan will treat a Participant's spousal Beneficiary or Participant's alternate payee spouse or former spouse as a Participant. A non-spouse beneficiary is not eligible to make an In-Plan Roth Rollover.

- **(C) In-Plan Roth Rollover Procedure.** An In-Plan Roth Rollover must be made by the Participant in the form of a Direct In-Plan Roth Rollover. Additionally, an In-Plan Roth Rollover must be accomplished in the form and manner established by the Plan Administrator.
- **(D) Plan Loans.** In-Plan Roth Rollovers may include a Plan loan only if there is no change in the loan's repayment schedule and the loan is not treated as a new loan. Further, Plan loans that include In-Plan Roth Rollover amounts are subject to the Plan's loan policy.
- **(E) Treatment of In-Plan Roth Rollovers.** Notwithstanding anything in the Plan to the contrary, an In-Plan Roth Rollover is not a rollover contribution for purposes of the Plan. The Plan will take into account the amounts attributable to an In-Plan Roth Rollover for purposes of determining whether a Participant's vested Account Balance exceeds \$500 for purposes of the Plan's automatic cash-out distribution rule at Section 4.2(B). Amounts in the Participant's In-Plan Roth Rollover Account may only be withdrawn by a Participant when the Participant is eligible for a distribution under Article IV (Time and Method of Payments of Benefits) of the Plan.
- **(F) Spousal Consent.** A married Plan Participant is not required to obtain spousal consent in connection with an election to make an In-Plan Roth Conversion, even if the Plan would otherwise require the spousal consent for distributions.
- (G) Tax Treatment of In-Plan Roth Conversion. A Participant must include in gross income the taxable amount of an In-Plan Roth Conversion. For this purpose, the taxable amount of an In-Plan Roth Conversion is the fair market value of the distribution reduced by any basis in the converted amounts. If an outstanding loan is rolled over as part of an In-Plan Roth Conversion, the amount includible in gross income includes the balance of the loan. Generally, the taxable amount of an In-Plan Roth Conversion is includible in gross income in the taxable year in which the conversion occurs. However, In-Plan Roth Conversions are not subject to twenty percent (20%) mandatory withholding under Code § 3405(c). Thus, Participants will be required to make arrangements independent of the Plan to ensure the total taxable amount is timely paid.

#### (H) In-Plan Roth Conversion Definitions.

- (1) "Direct In-Plan Roth Rollover" means a direct rollover from a Participant Account (other than Participant's Roth Elective Deferral Account, Roth Rollover Account and In-Plan Roth Rollover Account) to an In-Plan Roth Rollover Account established for the Participant.
- (2) "In-Plan Roth Rollover" means a contribution made from the Participant's Account (other than Participant's Roth Deferral Account or Roth Rollover Account) to the Participant's In-Plan Roth Rollover Account under the Plan in accordance with Code § 402A(c)(4) with respect to distributable amounts under the Plan or

- § 402A(c)(4)(E) with respect to amounts not otherwise distributable under the Plan.
- **(3) "In-Plan Roth Rollover Account"** means the subaccount established under the Plan to account for a Participant's In-Plan Roth Rollover.

#### ARTICLE IV.

#### TIME AND METHOD OF PAYMENT OF BENEFITS

- **4.1. Distribution Restrictions.** Except as the Plan provides otherwise in Sections 4.3 (RMDs) and 4.5 (Distributions Prior to Severance From Employment), a Participant may not receive a distribution of the amounts in his or her Account prior to occurrence of one of the following events:
  - (A) The Participant's severance from employment; or
  - **(B)** The Participant's death.
- **4.2. Time and Method of Payment of Account.** Upon a severance from employment the Participant's vested Account shall be distributed as specified in this Section 4.2 (Time and Method of Payment of Account) and in accordance with Code § 401(a)(9) and its underlying Treasury Regulations. In the absence of a Participant election, the Plan Administrator will determine the time and method of payment applicable to a particular Participant. In no event will distributions commence later than permissible under the required minimum distribution rules described in Section 4.3 (RMDs) and of Code § 401(a)(9) and its underlying Treasury Regulations.
  - (A) Participant Election of Time and Method. Participants may elect the timing and method of distribution of his or her vested Account in accordance with this Section 4.2 (Time and Method of Payment of Account). The Plan Administrator must consent to the specific terms of any such Participant election and the Plan Administrator in its sole discretion may withhold consent. Subject to the foregoing conditions, a Participant: (1) may elect to postpone distribution of his or her vested Account to any fixed or determinable date including, but not beyond, the Participant's required beginning date; and (2) may elect a single lump sum or periodic payments (on a monthly, quarterly, semiannual, or annual basis) and/or a partial distribution as the method of payment of his or her vested Account balance provided the payment schedule selected by the Participant complies with Code §§ 457 and 401(a)(9). The Plan Administrator must furnish to the Participant a form for the Participant to elect the time and a method of payment.
  - **(B) Automatic Cash-Out Distribution.** Notwithstanding any other distribution election, if the Participant's Account balance is not in excess of \$500 as of the date of distribution, the Plan Administrator may distribute the Participants Account balance in a single lump sum payment following Participant's severance from employment. In determining whether the Participant's Account balance is in excess of \$500, the Plan Administrator may take Rollover Contributions into consideration.
- **4.3. Required Minimum Distributions ("RMDs").** A Participant's Account may not be

distributed under a method of payment which, as of the required beginning date, does not satisfy the minimum distribution requirements of Code § 401(a)(9) or which is not consistent with applicable Treasury regulations.

- (A) General Rules. All distributions required under this Section 4.3 will be determined and made in accordance with the Treasury regulations under Code § 401(a)(9), the requirements of which, will take precedence over any inconsistent provisions of the Plan.
- (B) Time and Manner of Distribution.
  - (1) Required Beginning Date. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's Required Beginning Date as set forth in Section 4.3(E)(5).
  - (2) Death of Participant Before Distribution Begins. If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
    - (a) Spouse Designated Beneficiary. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, then distributions to the surviving spouse must begin by December 31 of the calendar year immediately following the Calendar Year in which the Participant dies, or by December 31 of the Calendar Year in which the Participant would have attained age 70½ (or age 72 if Participant attains age 70½ after December 31, 2019), if later.
    - **(b) Non-Spouse Designated Beneficiary.** If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, then distributions to the designated Beneficiary will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
    - **(c) No Designated Beneficiary.** If there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
    - (d) Death of Spouse. If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Section 4.3(B)(2) other than Section 4.3(B)(2)(a) (Spouse Designated Beneficiary), will apply as if the surviving spouse were the Participant. For purposes of this Section 4.3(B) (Time and Manner of Distribution) and Section 4.3(D) (RMDs After Participant's Death), unless this Section 4.3(B)(2)(d) applies, distributions are considered to begin on the Participant's required beginning date. If this Section 4.3(B)(2)(d) applies, distributions are considered to begin on the date distributions are required to begin to the surviving spouse under Section 4.3(B)(2)(a) (Spouse Designated Beneficiary).

- (3) Forms of Distribution. Unless the Participant's interest is distributed in a single sum on or before the required beginning date, as of the first distribution calendar year distributions will be made in accordance with Sections 4.3(C) (RMDs During Participant's Lifetime) and 4.3(D) (RMDs After Participant's Lifetime).
- (C) RMDs During Participant's Lifetime.
  - (1) Amount of RMD Each Distribution Calendar Year. During the Participant's lifetime, the minimum amount that will be distributed for each distribution calendar year is the lesser of:
    - (a) Uniform Life Table. The quotient obtained by dividing the Participant's Account Balance by the number in the Uniform Life Table set forth in U.S. Treas. Reg. § 1.401(a)(9)-9, using the Participant's attained age as of the Participant's birthday in the distribution calendar year; or
    - **(b) Younger Spouse.** If the Participant's sole Designated Beneficiary for the distribution calendar year is the Participant's spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in U.S. Treas. Reg. § 1.401(a)(9)-9, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the distribution calendar year.
  - (2) Lifetime RMDs Through Year of Participant's Death. RMDs will be determined under this Section 4.3(C) (RMDs During Participant's Lifetime) beginning with the first distribution calendar year and up to and including the distribution calendar year that includes the Participant's date of death.
- (D) RMDs After Participant's Death.
  - (1) Death On or After Distributions Begin.
    - **(a) Participant Survived by Designated Beneficiary.** If the Participant dies on or after the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's Designated Beneficiary, determined as follows:
      - (i) Participant's Life Expectancy. The Participant's remaining life expectancy is calculated using the attained age of the Participant as of the Participant's birthday in the calendar year of death, reduced by one for each subsequent calendar year.
      - **(ii) Spouse's Life Expectancy.** If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining life expectancy of the

surviving spouse is calculated for each distribution calendar year after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For distribution calendar years after the year of the surviving spouse's death, the remaining life expectancy of the surviving spouse is calculated using the attained age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one for each subsequent calendar year.

- (iii) Non-Spouse's Life Expectancy. If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining life expectancy is calculated using the attained age of the Beneficiary as of the Beneficiary's birthday in the calendar year following the calendar year of the Participant's death, reduced by one for each subsequent calendar year.
- **(b) No Designated Beneficiary.** If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the calendar year after the calendar year of the Participant's death, the minimum amount that will be distributed for each distribution calendar year after the calendar year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the Participant's remaining life expectancy calculated using the attained age of the Participant as of the Participant's birthday in the calendar year of death, reduced by one for each subsequent calendar year.

#### (2) Death Before Date Distributions Begin.

- (a) Participant Survived by Designated Beneficiary. If the Participant dies before the date distributions begin and there is a designated Beneficiary, the minimum amount that will be distributed for each distribution calendar year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining life expectancy of the Participant's designated Beneficiary, determined as provided in Section 4.3(D)(1) (Death On or After Distributions Begin).
- **(b)** No Designated Beneficiary. If the Participant dies before the date distributions begin and there is no Designated Beneficiary, distribution of the Participant's entire interest will be completed to Participant's estate by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- **(c) Death of Surviving Spouse Before Distributions to Surviving Spouse are Required to Begin.** If the Participant dies before the date distributions begin, the Participant's surviving spouse is the Participant's sole designated Beneficiary, and the surviving spouse dies before distributions are required to

begin to the surviving spouse under Section 4.3(B)(2)(a) (Spouse Designated Beneficiary), this Section 4.3(D)(2) will apply as if the surviving spouse were the Participant.

(d) 5-Year or Life Expectancy Rule; Possible Election. A Designated Beneficiary may elect whether the life expectancy rule under Section 4.3(D)(2)(a) (Participant Survived by Designated Beneficiary) or the 5-year rule under Section 4.3(D)(2)(b) (No Designated Beneficiary) will apply. However, the life expectancy rule is only available in the case of a Designated Beneficiary. The 5-year rule applies as to any Beneficiary who is not a Designated Beneficiary as defined by Code § 401(a)(9). A permitted election under this Section must be made no later than the earlier of September 30 of the calendar year in which distribution would be required to begin under Section 4.3(D)(2)(a) (Participant Survived by Designated Beneficiary), or by September 30 of the calendar year which contains the fifth anniversary of the Participant's (or, if applicable, surviving spouse's) death.

#### (E) RMD Definitions.

- (1) Designated Beneficiary. The individual who is designated as the Beneficiary under the Plan and is the designated beneficiary under Code § 401(a)(9) and U.S. Treas. Reg. 1.401(a)(9)-1, Q&A-4.
- (2) Distribution Calendar Year. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year is the calendar year immediately preceding the calendar year which contains the Participant's required beginning date. For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which the distributions are required to begin under Section 4.3(B)(2). The RMDs for the Participant's first distribution calendar year will be made on or before the Participant's required beginning date. The RMD for other distribution calendar years, including the RMD for the distribution calendar year in which the Participant's required beginning date occurs, will be made on or before December 31 of that distribution calendar year.
- **(3) Life Expectancy.** Life expectancy as computed by use of the Single Life Table in U.S. Treas. Reg. § 1.401(a)(9)-9.
- (4) Participant's Account Balance. The account balance as of the last valuation date in the calendar year immediately preceding the distribution calendar year (valuation calendar year) increased by the amount of any contributions made and allocated to the account balance as of dates in the valuation calendar year after the valuation date and decreased by distributions made in the valuation calendar year after the valuation date. The account balance for the valuation calendar year includes any Rollover Contributions or Transfers to the Plan either in the valuation

- calendar year or in the distribution calendar year if distributed or transferred in the valuation calendar year.
- (5) Required Beginning Date. A Participant's required beginning date is the April 1 of the calendar year following the later of: (1) the calendar year in which the Participant attains age 70½ (or age 72 if Participant attains age 70½ after December 31, 2019), or (2) the calendar year in which the Participant retires under Code § 401(a)(9). A Rehired Annuitant is treated as a retired Participant for purposes of determining his or her Required Beginning Date.
- **4.4. Death Benefits.** Upon the death of the Participant, the Participant's Account must be paid in accordance with Section 4.2 (Time and Method of Payment of Account). If a Participant dies while performing qualified military service (as defined in Code § 414(u)), the survivors of the Participant are entitled to any additional benefits (other than benefit accruals relating to the period of qualified military service) provided under the Plan as if the Participant had resumed and then terminated employment on account of death. In no event may death benefits commence later that permissible under the required minimum distribution rules described in Section 4.3, and Code § 401(a) and its underlying Treasury Regulations.
- **4.5. Distributions Prior to Severance From Employment.** Notwithstanding the distribution restrictions in Section 4.1 (Distribution Restrictions), the Plan permits the following in-service distributions from a Participant's vested Account balance.
  - (A) Unforeseeable Emergency. In the event of a Participant's unforeseeable emergency, the Plan Administrator may allow a distribution to a Participant who has not incurred a severance from employment (or who has incurred a severance from employment but will not begin to receive payments until some future date and continues to have an Account balance under the Plan). The Plan Administrator will establish a policy for determining whether an unforeseeable emergency exists. An unforeseeable emergency is a severe financial hardship of a resulting from:
    - (1) illness or accident of the Participant, or the Participant's spouse or dependent (as defined in Code § 152(a));
    - (2) loss of the Participant's property due to casualty;
    - (3) the need to pay for the funeral expenses of the Participant's spouse or dependent (as defined in Code § 152(a)); or,
    - (4) other similar extraordinary and unforeseeable circumstances arising from events beyond the Participant's control, or which applicable law may define as an unforeseeable emergency.

The Plan Administrator will not allow the Participant to withdraw more than the amount reasonably necessary to satisfy the emergency need, which may include amounts necessary to pay taxes or penalties on the distribution. The Plan Administrator will not allow payment to the extent the Participant may relieve the financial hardship

- by cessation of deferrals under the Plan, through insurance or other reimbursement, or by liquidation of the individual's assets to the extent such liquidation would not cause severe financial hardship.
- **(B) Distribution of Rollover Contributions.** The Plan permits a Participant to request and to receive a distribution from the Participant's Account attributable to Rollover Contributions, plus earnings thereon, before the Participant has a distributable event under Section 4.1 (Distribution Restrictions).
- (C) Age-Based In-Service Distributions. Effective January 1, 2020, the Plan permits a Participant to request and receive distributions from the Participant's Account on or after the date Participant attains age 59½. Prior to January 1, 2020, the Plan permitted Participants to request and receive distributions from the Participant's Account on or after attaining age 70½. Such age-based distributions must be requested by the Participant and may only be made in the frequency, form and manner provided by the Plan Administrator.
- **(D) Participant Loans.** The Plan may establish and maintain a Participant loan program in accordance with the provisions of Section 5.3(B) (Loan Policy) of the Plan and Code Section 72(p).
- **(E) Qualified Birth or Adoption Distributions (QBOADs).** The Plan permits a Participant to request and receive QBOADs. QBOADs must be requested by the Participant and may only be made from the Plan in the frequency, form and manner provided by the Plan Administrator. Further, Participants may have an opportunity to repay a QBOAD if the Participant is eligible to make a rollover into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder. This Section 4.5(E) is effective June 1, 2021.
- **(F) Coronavirus-Related Distributions (CRDs).** Participants may request and receive a Coronavirus-Related Distribution (CRD). CRDs must be requested and made in the form and manner provided by the Plan Administrator.
  - **(1)** "Coronavirus-Related Distribution" Defined. A CRD means a distribution from the Plan made between January 1, 2020 and December 31, 2020, to a Participant:
    - (i) who is diagnosed with the virus SARS-CoV-2 or the coronavirus disease 2019 (collectively, "COVID-19") by a test approved by the Centers for Disease Control and Prevention; or
    - (ii) whose spouse or dependent is diagnosed with COVID-19 by a test approved by the Centers for Disease Control and Prevention; or
    - (iii) who experiences adverse financial consequences as a result of a spouse or Participant being quarantined, furloughed or laid off, or having work hours reduced due to COVID-19; or being unable to work due to lack of childcare due to COVID-19 or closing or reducing hours of a business owned or operated by

the spouse or Participant, or a member of the Participant's household (one who shares Participant's principal residence) due to COVID-19; or having a reduction in pay due to COVID-19 or having a job offer rescinded or start date for a job delayed due to COVID-19.

- **(2) Aggregate Dollar Limit.** The aggregate amount of CRDs to a Participant may not exceed \$100,000 from all plans sponsored by the Employer.
- (3) Re-Contribution of CRDs. Participants may have an opportunity to repay some or all of the Participant's CRD if completed within 3 years of the distribution, meets the re-contribution requirements in § 2202(a) of the SECURE Act, and the Participant is eligible to make a rollover contribution into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder.
- **(4) Participant Certification.** The Plan Administrator may rely on a Participant's certification that he or she satisfies the conditions for a CRD.
- **(G) Pro Rata Distribution.** Unless the Plan Administrator provides otherwise, and only in accordance with applicable law, in-service distributions made under this Section 4.7 shall be made from Participant's Account on a pro rata basis across eligible contribution sources and investment options.
- **4.6. Distributions Under Qualified Domestic Relations Orders (QDROs).** The Plan shall recognize and comply with a court order that the Plan Administrator (or designee thereof) determines, in its sole discretion, meets the requirements of Code § 414(p). In order to be approved, any such order must also comply with the provisions of this Section 4.6 and any additional policies or procedures established by the Plan Administrator.
  - (A) Time and Method of Payment. This Plan specifically permits distribution to an alternate payee under a QDRO at any time, notwithstanding any contrary Plan provision and irrespective of whether the Participant has attained his or her earliest retirement age (as defined under Code § 414(p)) under the Plan. A distribution to an alternate payee prior to the Participant's attainment of earliest retirement age is available only if the QDRO specifies distribution at that time or permits an agreement between the Plan and the alternate payee to authorize an earlier distribution. Nothing in this Section 4.6 gives a Participant a right to receive distribution at a time the Plan otherwise does not permit nor authorizes the alternate payee to receive a form of payment the Plan does not permit.
  - **(B) QDRO Procedures.** The Plan Administrator will establish reasonable procedures to determine the qualified status of a domestic relations order. Upon receiving a domestic relations order, the Plan Administrator (or its designee) promptly will notify the Participant and any alternate payee named in the order, in writing, of the receipt of the order and the Plan's procedures for determining the qualified status of the order. Within a reasonable period of time after receiving the domestic relations order, the Plan

- Administrator must determine the qualified status of the order and must notify the Participant and each alternate payee, in writing, of the Plan Administrator's determination. The Plan Administrator must provide notice under this paragraph by mailing to the individual's address specified in the domestic relations order.
- (C) Accounting. If any portion of the Participant's Account balance is payable under the domestic relations order during the period the Plan Administrator is making its determination of the qualified status of the domestic relations order, the Plan Administrator must maintain a separate accounting of the amounts payable. If the Plan Administrator determines the order is a QDRO within 18 months of the date amounts first are payable following receipt of the domestic relations order, the Plan Administrator (or its designee) will distribute the payable amounts in accordance with the QDRO. If the Plan Administrator does not make its determination of the qualified status of the order within the 18-month determination period, the Plan Administrator (or its designee) will distribute the payable amounts in the manner the Plan would distribute if the order did not exist and will apply the order prospectively if the Plan Administrator later determines the order is a QDRO.
- **(D) Permissible QDROs.** A domestic relations order that otherwise satisfies the requirements for a QDRO will not fail to be a QDRO: (i) solely because the order is issued after, or revises, another domestic relations order or QDRO; or (ii) solely because of the time at which the order is issued, including issuance after or after the Participant's death.

#### 4.7. Direct Rollover of Eligible Rollover Distributions.

- **(A) Participant Election.** A Participant (including for this purpose, a former Employee) may elect, at the time and in the manner the Plan Administrator prescribes, to have any portion of his or her eligible rollover distribution from the Plan paid directly to an eligible retirement plan specified by the Participant in a direct rollover election. For purposes of this election, a "Participant" includes as to their respective interests, a Participant's surviving spouse and the Participant's spouse or former spouse who is an alternate payee under a QDRO.
- **(B) Rollover and Withholding Notice.** At least 30 days and not more than 180 days prior to distribution of an eligible rollover distribution, the Plan Administrator must provide a written notice (including a summary notice as permitted under applicable Treasury regulations) explaining to the distributee the rollover option, the applicability of mandatory twenty percent (20%) federal withholding to any amount not directly rolled over, and the recipient's right to roll over within 60 days after the date of receipt of the distribution ("rollover notice").
- **(C) Default Distribution or Rollover.** In the case of a Participant who does not elect timely to roll over or to receive distribution of his or her Account, the Plan Administrator (or its designee) may distribute to the Participant (if Participant's Account balance is \$1,000 or less) or may directly roll over (if Participant's Account balance is greater than \$1,000) the

Participant's Account in accordance with the Plan's rollover notice. The Plan's right to roll over Participant Account balance over \$1,000 to an IRA is effective as of the date of the close of the first regular legislative session of the legislative body with the authority to amend the Plan that begins on or after January 1, 2006. For purposes of the automatic rollover of Accounts over \$1,000, the Plan will treat a Participant's Roth Elective Deferral Account and the Participant's other accounts as held under two separate plans for purposes of applying the automatic rollover rules.

- **(D) Non-Spouse Beneficiary Rollover Right.** A non-spouse beneficiary who is a designated beneficiary under Section 4.3(E)(1)(Designated Beneficiary), by a direct trustee-to-trustee transfer ("direct rollover"), may roll over all or any portion of his or her distribution to an individual retirement account the beneficiary establishes for purposes of receiving the distribution. In order to be able to roll over the distribution, the distribution otherwise must satisfy the definition of an eligible rollover distribution.
  - (1) Certain Requirements Not Applicable. Although a non-spouse beneficiary may roll over directly a distribution as provided in this Section 4.7(D), the distribution is not subject to the direct rollover requirements of Code § 401(a)(31) (including the automatic rollover provisions of Code § 401(a)(31)(B)), the notice requirements of Code § 402(f) or the mandatory withholding requirements of Code § 3405(c). If a non-spouse beneficiary receives a distribution from the Plan, the distribution is not eligible for a "60-day" rollover.
  - **(2) Trust Beneficiary.** If the Participant's named beneficiary is a trust, the Plan may make a direct rollover to an individual retirement account on behalf of the trust, provided the trust satisfies the requirements to be a designated beneficiary within the meaning of Code § 401(a)(9)(E).
  - (3) RMDs Not Eligible For Rollover. A non-spouse beneficiary may not roll over an amount which is an RMD, as determined under applicable U.S. Treasury regulations and other IRS guidance. If the Participant dies before his or her required beginning date and the non-spouse beneficiary rolls over to an IRA the maximum amount eligible for rollover, the beneficiary may elect to use either the 5-year rule or the life expectancy rule, pursuant to U.S. Treas. Reg. § 1.401(a)(9)-3, A-4(c), in determining the RMDs from the IRA that receives the non-spouse beneficiary's distribution.
- **(E) Rollover Definitions.** The following definitions apply to this Section 4.7:
  - (1) Eligible Rollover Distribution. An eligible rollover distribution is any distribution of all or any portion of a Participant's Account, except an eligible rollover distribution does not include: (a) any distribution which is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's Designated Beneficiary, or for a specified period of ten years or more; (b) any Code § 401(a)(9) required minimum

distribution; and, (c) any unforeseeable emergency distribution.

- (2) Eligible Retirement Plan. An eligible retirement plan is an individual retirement account described in Code § 408(a), an individual retirement annuity described in Code § 408(b), an annuity plan described in Code § 403(a), a qualified plan described in Code § 401(a), an annuity contract (or custodial agreement) described in Code § 403(b), or an eligible deferred compensation plan described in Code § 457(b) and maintained by an employer described in Code § 457(e)(1)(A), which accepts the Participant's, the Participant's spouse or alternate payee's eligible rollover distribution. A Participant or beneficiary may elect to roll over directly an eligible rollover distribution to a Roth IRA described in Code § 408A(b). For this purpose, the term "eligible rollover distribution" includes a rollover distribution described in this Section 4.7. Notwithstanding the foregoing, a direct rollover of a distribution from Roth Elective Deferrals shall only be made to an Eligible Retirement Plan which includes Roth Elective Deferrals as described in Code § 402(e)(1) or to a Roth IRA as described in Code § 408A, and only to the extent the rollover is permitted under the rules of Code § 402(c).
- **(3) Direct Rollover.** A direct rollover is a payment by the Plan to the eligible retirement plan specified by the distributee.
- **(4) Mandatory Distribution.** A mandatory distribution is an eligible rollover distribution without the Participant's consent before the Participant attains the later of age 62 or NRA per paragraph 3.5 (Special Catch-Up Contribution). A distribution to a beneficiary is not a mandatory distribution.

#### ARTICLE V.

#### PLAN ADMINISTRATION

- **5.1 Plan Administrator.** The District is the Plan Administrator and has delegated to the General Manager the powers, duties, and responsibilities of the Plan Administrator. The General Manager may delegate, in writing, some or all of the powers, duties and responsibilities of Plan Administrator position as necessary to adequately operate and maintain the Plan.
- **5.2 Term/Vacancy.** Any Plan Administrator designated by the District must accept the Plan Administrator role and responsibilities in writing and will serve until his or her successor is appointed. In case of a vacancy in the position of the Plan Administrator, the District will exercise any and all of the powers, authority, duties and discretion as the Plan Administrator pending the filling of the vacancy.
- 5.3 Powers and Duties. The Plan Administrator shall have the following powers and duties:(A) In General.
  - (1) To determine the rights of eligibility of an Employee to participate in the Plan and the value of a Participant's Account;

- (2) To adopt policies, rules and procedures and to create administrative forms necessary for the proper and efficient administration of the Plan provided the rules, procedures and forms are not inconsistent with the terms of the Plan or MWD Code;
- (3) To construe and enforce the terms of the Plan and the policies, rules and regulations adopted to carry out the terms of the Plan;
- **(4)** To direct the distribution of a Participant's Account;
- (5) To review and render decisions respecting an administrative claim for (or denial of such a claim for) a benefit under the Plan;
- **(6)** To furnish the District with information required for tax or other purposes;
- (7) To establish policies regarding distributions from the Plan;
- **(8)** To establish policies regarding the receipt of Rollover Contributions and automatic rollover distributions;
- **(9)** To establish a policy regarding the making and the receipt of Transfers;
- **(10)** To establish a policy regarding Participant or Beneficiary direction of investment:
- **(11)** To undertake correction of any Plan failures as necessary to preserve eligible Plan status:
- **(12)** To act upon instructions from the DCIC to direct the Trustee regarding changes to the Plan's investment options and to execute proxy votes cast by the DCIC;
- (13) To request and receive input from the DCIC, when reasonable and appropriate, regarding non-investment related Plan features; the selection, performance, and retention of the record keeper, investment advisors, service providers, advisors, consultants, and auditor; and Plan amendments;
- (14) To serve as the secretary for the DCIC which includes the following duties: keeping and maintain the DCIC meeting minutes, executing documents on behalf of the DCIC, issuing formal written correspondence of the DCIC; and other such administrative support duties as the DCIC may reasonably delegate to the Plan Administrator; and
- (15) To undertake any other action the Plan Administrator deems reasonable or necessary to administer the Plan.

The Plan Administrator shall have total and complete discretion to interpret and construe the Plan and to determine all questions arising in the administration, interpretation and application of the Plan. Any determination the Plan Administrator makes under the Plan is final and binding upon any affected person. The Plan Administrator shall coordinate with General Counsel concerning any litigation, compromise, or settlement of claims against the Plans in accordance with MWD Code Section 6433.

- (B) Loan Policy Establishment. The Plan Administrator, in its sole discretion, may establish or amend a loan policy for making Plan loans (if permitted under the Plan) to Participants. If the Plan Administrator adopts a loan policy, the loan policy must be a written document and must include: (1) the identity of the person or positions authorized to administer the participant loan program; (2) the procedure for applying for a loan; (3) the criteria for approving or denying a loan; (4) the limitations, if any, on the types and amounts of loans available; (5) the procedure for determining a reasonable rate of interest; (6) the types of collateral which may secure the loan; and (7) the events constituting default and the steps the Plan will take to preserve Plan assets in the event of default. A loan policy the Plan Administrator adopts under this Section 5.3(B) is part of the Plan, except that the Plan Administrator may amend or terminate the policy without regard to Section 9.1 (Amendment By Employer/Sponsor) at any time.
- **(C) QDRO Policy Establishment.** The Plan Administrator will establish QDRO policies and procedures.
- **5.4 Without Additional Compensation.** The Plan Administrator will serve without additional compensation for services, but the District will pay all reasonable expenses of the Plan Administrator.
- **5.5 Authorized Representative.** The Plan Administrator may authorize appropriate Human Resources or Finance staff, to sign on the Plan Administrator's behalf any Plan notices, directions, applications, certificates, consents, approvals, waivers, letters or other documents.
- **5.6 Individual Accounts/Records.** The Plan Administrator will maintain a separate Account in the name of each Participant to reflect the value of the Participant's Deferred Compensation under the Plan. The Plan Administrator will maintain records of its activities.
- **5.7 Valuation of Participant's Account.** The value of each Participant's Account consists of his or her accumulated Deferred Compensation, as of the most recent Accounting Date or any later date as the Plan Administrator may determine.
- 5.8 Account Administration, Valuation and Expenses.
  - (A) Individual Accounts. The Plan Administrator, as necessary for the proper administration of the Plan, will maintain, or direct the Trustee to maintain, a separate Account, or multiple Accounts, in the name of each Participant to reflect the Participant's Account balance under the Plan. The Plan Administrator will make its allocations of Earnings, or will request the Trustee to make such allocations, to the Accounts of the Participants as necessary to maintain proper Plan records and in accordance with the applicable: (i) Contribution Types; (ii) allocation conditions; (iii) investment account types; and (iv) Earnings allocation methods. The Plan Administrator may also maintain, or direct the Trustee to maintain, a separate temporary Account for Participant forfeitures which occur during a Plan Year, pending their accrual and allocation in accordance with

the Plan terms, or for other special items as the Plan Administrator determines is necessary and appropriate for proper Plan administration.

- (1) By Contribution Type. The Plan Administrator will establish Plan Accounts for each Participant as necessary to reflect his or her Accounts attributable to the following Contribution Types and the Earnings attributable thereto: In-Plan Roth Rollovers, Pre-Tax Deferrals, Rollover Contributions, Roth Deferrals, Roth Rollover Contributions, and Transfers. If additional contribution types are added to the Plan, the Plan Administrator will establish Plan Accounts for each Participant to reflect such new contribution types.
- **(2)** By Investment Account Type. The Plan Administrator will establish separate Accounts for each Participant as necessary to reflect his or her investment account types as described below:
  - (a) Participant-Directed Accounts. Participants may, in accordance with the provisions of the Plan, direct investment of all or a portion of the amounts allocated to an account. Participant-Directed Accounts are Accounts that are established and maintained for a Participant to invest in one or more assets that are not pooled assets held by the Trust. As the Plan Administrator determines, a Participant-Directed Account may provide for a limited number and type of investment options or funds. A Participant may have one or more Participant-Directed Accounts in addition to Segregated Accounts. A Participant-Directed Account is credited and charged with the Earnings. The Plan Administrator will designate how a Participant Account shall be invested in the absence of proper affirmative direction from the Participant. The Plan or Plan Administrator may designate a default fund under the Plan to deposit contributions on behalf of Participant who has not specified an investment choice under the Plan.
  - **(b) Segregated Accounts.** A Segregated Account that is established and maintained for a Participant: (i) to hold a QDRO amount; (ii) to prevent a distortion of Plan Earnings allocations; or (iii) for such other purposes as the Plan Administrator may direct. A Segregated Account receives all income it earns and bears all expense or loss it incurs. The assets of a Segregated Account shall be invested in a manner consistent with the purpose for which the Segregated Account was established.

As of each Valuation Date, the Plan Administrator must reduce a Segregated Account for any forfeiture arising after the Plan Administrator has made all other allocations, changes or adjustments to the Account (excluding Earnings) for the Valuation Period.

**(3) Amount of Account.** The amount of a Participant's Account, as determined by the Plan Administrator, is equal to the sum of all contributions, Earnings and other additions credited to the Account, less all distributions (including distributions to

Beneficiaries and to alternate payees and also including disbursement of Plan loan proceeds), expenses and other charges against the Account as of a Valuation Date or other relevant date. For purposes of a distribution under the Plan, the amount of a Participant's Account balance is determined based upon its value on the Valuation Date immediately preceding or coinciding with the date of the distribution. If any or all Plan investment Accounts are participant-directed, the directing Participant's Account balance consists of the assets held within the participant-directed Account and the value of the Account is determined based upon the fair market value of such assets.

- (4) Account Statements. As soon as practicable after the Accounting Date of each Plan Year, the Plan Administrator will deliver to each Participant (and to each Beneficiary) a statement reflecting the amount of his or her Account balance in the Trust as of the statement date or most recent Valuation Date. No Participant, except the Plan Administrator/Participant or Trustee/Participant, has the right to inspect the records reflecting the Account of any other Participant.
- **5.9 Participant Account Charged.** The Plan Administrator will charge all distributions made to a Participant or to his or her Beneficiary or transferred under Section 8.3 (Transfers) from his or her Account, against the Account of the Participant when made.
- **5.10 Participant Direction of Investment.** Subject to authorization by the Plan's Investment Policy Statement and any terms, conditions and/or restrictions established by the Plan Administrator, Participants may elect to direct the investment or reinvestment of the assets comprising the Participant's Account. Participants may direct the investment or reinvestment of the assets comprising the Participant's Account in any investment options the DCIC makes available under the Plan and in accordance with the Plan's Investment Policy Statement.

The Plan Administrator will account separately for any Account assets that are invested at the direction of the Participant or invested through a self-directed brokerage account.

The Participant's right to direct investment does not give the Participant any vested interest or secured or preferred position with respect to assets over which Participant has investment responsibility. Any assets in the Participant's Account for which the Participant has not made an investment election shall be invested in the Plan's default investment choice in accordance with the Plan's Investment Policy Statement.

- **5.11 Vesting.** Participants shall be 100% vested in their Accounts at all times.
- **5.12 Preservation of Eligible Plan Status.** The Plan Administrator may elect to sever from this Plan and to treat as a separate 457 plan, the Accounts of any Participants who have Excess Deferrals that the Plan Administrator has not corrected in accordance with Section 3.7 (Disbursement of Excess Deferrals) or in the case of any other Code § 457(b) failure that the District may not otherwise correct, and which failure would result in the Plan ceasing to be an eligible 457(b) plan. In such event, the Plan Administrator will take any necessary or appropriate action consistent with

the District's maintenance of separate 457 plans and with preservation of tax-deferred status of this Plan.

- **5.13 Limited Liability.** The District will not be liable to pay plan benefits to a Participant in excess of the value of the Participant's Account as the Plan Administrator determines in accordance with the Plan terms. Neither the District nor the Plan Administrator will be liable for losses arising from depreciation or shrinkage in the value of any investments acquired under this Plan.
- **5.14 Lost Participants.** If the Plan Administrator is unable to locate any Participant or Beneficiary whose Account becomes distributable (a "lost Participant"), the Plan Administrator will apply the provisions of this Section 5.14.
  - (A) Attempt to Locate. The Plan Administrator will attempt to locate a lost Participant or Beneficiary and may use one or more of the following methods: (1) provide a distribution notice to the lost Participant or Beneficiary at his or her last known address by certified or registered mail; (2) use a commercial locator service, the internet or other general search method; (3) use the Social Security Administration or PBGC search program; or (4) use such other methods as the Plan Administrator believes reasonable and prudent.
  - (B) Failure to Locate. The Plan Administrator may declare a forfeiture of a benefit due a Participant or Beneficiary deemed to be lost (i.e., the Participant or Beneficiary cannot be found within 4 years of his or her required beginning date). If after a benefit is forfeited under this Section 5.14(B), the lost Participant or Beneficiary is located, the Plan will restore the forfeited benefit (unadjusted for gains or losses) to such Participant or Beneficiary within a reasonable time. However, if a lost Participant or Beneficiary has not been located by the time the Plan terminates, the forfeiture of such Participant's or Beneficiary's distributable benefit will be irrevocable. Notwithstanding the foregoing, the provisions of this Section 5.14(B) are not the exclusive means for the Plan Administrator to administer the Accounts of lost Participants and Beneficiaries. The Plan Administrator may address forfeitures using any available means provided under applicable law, including escheating the Participant's or Beneficiary's benefit to the state.
  - **(C) Reasonable Expenses.** The Plan Administrator may charge to the Account of a lost Participant the reasonable expenses incurred under this Section 5.14 and which are associated with the lost Participant's Account.
- **5.15 Plan Correction.** The Plan Administrator, in conjunction with other parties as appropriate, may undertake such correction of Plan errors as the Plan Administrator deems necessary, including but not limited to correction to maintain the Plan's status as an eligible 457(b) plan. The Plan Administrator under this Section 5.15 also may undertake Plan correction in accordance with any correction program that the Internal Revenue Service makes applicable to 457 plans.

#### ARTICLE VI.

#### BENEFICIARY DESIGNATIONS AND ENROLLMENT

**6.1 Beneficiary Designation.** A Participant from time to time may designate, in the form and manner set forth by the Plan Administrator, any person(s) (including a trust or other entity), contingently or successively, to whom the Plan will pay the Participant's Account in the event of death.

However, spousal consent is required in writing with respect to the distribution of death benefits from the Plan if the Participant designates someone other than the Participant's Spouse as a Beneficiary of more than fifty percent (50%) of the Participant's Account balance.

A Spouse's consent to the designation of more than fifty percent (50%) of Participant's Account to a non-spouse beneficiary must be provided at the time and in the form and manner specified by the Plan Administrator.

Subsequent beneficiary designations submitted to the Plan Administrator in good order revoke all designations filed prior to that date by the same Participant.

Participant's marriage or remarriage revokes any prior Beneficiary designations made by the Participant and a new Beneficiary designation form that complies with the consent requirements of this Section 6.1 (Beneficiary Designation) must be submitted.

Further, when designating a trust or other entity as a Beneficiary, the time and form of distribution available to such Beneficiaries may be limited by the required minimum distribution rules of Code § 401(a)(9) and Section 4.3 (RMDs) of the Plan.

#### 6.2 No Beneficiary Designation.

- (A) Effective for Participants who are active Employees of the District on or after August 1, 2018, if a Participant fails to name a Beneficiary in accordance with Section 6.1 (Beneficiary Designation), or if all the Beneficiaries named by a Participant predeceases the Participant, then the Plan Administrator will pay the Participant's remaining Account in accordance with Article IV (Time and Method of Payments of Benefits)in the following order of priority, to:
  - (1) The Participant's surviving spouse; or
  - (2) The Participant's children (including adopted children), in equal shares by right of representation (one share for each surviving child and one share for each child who predeceases the Participant with living descendants); and if none, then to:
  - (3) The Participant's estate.
- **(B)** If a Participant has a severance from employment before August 1, 2018, and has a beneficiary designation form dated prior to August 1, 2018, then the Plan Administrator will pay the Participant's remaining Account in accordance with Article IV (Time and Method of Payments of Benefits) to the Participant's estate.

(C) If the Beneficiary survives the Participant, but dies prior to distribution of the Participant's entire Account, the remaining portion of the Participant's Account will be paid to the Participant's beneficiaries in accordance with, as applicable, Subsection 6.2 (No Beneficiary Designation) subsections (A) or (B) unless: (1) the Participant's Beneficiary designation election provides otherwise; or (2) the Beneficiary has timely and properly designated a Beneficiary under the procedures established by the Plan Administrator. A Beneficiary only may designate a Beneficiary for the Participant's Account balance remaining at the Beneficiary's death, if the Participant has not previously designated a successive contingent Beneficiary and the Beneficiary's designation otherwise complies with the Plan terms.

#### 6.3 Salary Reduction Agreement.

- **(A) General.** A Participant must elect to make Salary Reduction Contributions on a Salary Reduction Agreement form the Plan Administrator provides for this purpose. The Plan Administrator in a Salary Reduction Agreement may impose such other terms and limitations as the Plan Administrator may determine.
- **(B) Election Timing.** A Participant's Salary Reduction Agreement may not take effect earlier than the first day of the calendar month following the date the Participant executes the Salary Reduction Agreement and as to Compensation paid or made available in such calendar month.
- **(C) Sick, Vacation and Back Pay.** Participants may make Salary Reduction Contributions from accumulated sick pay, from accumulated vacation pay or from back pay. A Participant who will incur a severance from employment may execute a Salary Reduction Agreement before such amounts are paid or made available provided: (i) such amounts are paid or made available before the Participant incurs the Severance; and (ii) the Participant is an Employee in that month.
- (D) Modification of Salary Reduction Agreement. A Participant's Salary Reduction Agreement remains in effect until a Participant modifies it or ceases to be eligible to participate in the Plan. However, a Salary Reduction Agreement must stay in effect for at least one (1) month once it has been entered into by a Participant. A Participant may modify his or her Salary Reduction Agreement by executing a new Salary Reduction Agreement. Any modification will become effective no earlier than the beginning of the calendar month commencing after the date the Participant executes the new Salary Reduction Agreement. Filing a new Salary Reduction Agreement will revoke all Salary Reduction Agreements filed prior to that date. The Plan Administrator may restrict the Participant's right to modify his or her Salary Reduction Agreement in any Taxable Year.
- **6.4 Personal Data to Plan Administrator.** Each Participant and each Beneficiary of a deceased Participant must furnish to the Plan Administrator such evidence, data or information as the Plan Administrator considers necessary or desirable for the purpose of administering the Plan. The provisions of this Plan are effective for the benefit of each Participant upon the condition precedent

that each Participant will furnish promptly full, true and complete evidence, data and information when requested by the Plan Administrator, provided the Plan Administrator advises each Participant of the effect of his or her failure to comply with its request.

- **Address for Notification.** Each Participant and each Beneficiary of a deceased Participant must file with the Plan Administrator from time to time, in writing, contact information and any changes to such contract information, including but not limited to, his or her electronic and physical address, phone number and any changes thereto. Any communication, statement or notice addressed to a Participant, or Beneficiary, at his or her last address filed with the Plan Administrator, or as shown on the records of the District, binds the Participant, or Beneficiary, for all purposes of this Plan.
- entitled to a Plan distribution is not able to care for his or her affairs because of a mental condition, a physical condition, or by reason of age and the Plan Administrator (or its designee) is directed to make the distribution to the Participant's or Beneficiary's guardian, conservator, trustee, custodian (including under a Uniform Transfers or Gifts to Minors Act) or to his or her attorney-in-fact or to other legal representative upon furnishing evidence of such status satisfactory to the Plan Administrator (or its designee), the Plan Administrator shall make such distribution. The Plan Administrator will not under any circumstance have any liability with respect to payments so made nor any duty to make inquiry as to the competence of any person entitled to receive payments under the Plan.

#### ARTICLE VII.

#### **TRUST FUNDS**

**7.1. Establishment of Trust.** Consistent with Code section 457(g) and applicable Treasury regulations, all Plan assets will be held and invested in the Trust in accordance with this Plan and the Trust Agreement. The Trust will be established pursuant to a written Trust Agreement that constitutes a valid trust under the laws of the State of California.

The District may enter into a Trust Agreement with the Trustee providing for the investment of the Trust and prescribing the powers, duties, obligations, and functions of the Trustee with respect to the Plan. The Trust Agreement will form a part of this Plan, and any rights and benefits of any Participant or Beneficiary under the Plan whose Accounts are invested in the Trust will be subject to the terms of the Trust Agreement. However, in the case of any conflict between the Plan and the Trust Agreement, the Plan will govern. The Trustee must be one or more banks or other fiduciaries appointed by the District to hold and administer assets of the Plan.

The Trustee must hold the Plan assets in the Trust pursuant to the terms of the Trust Agreement for the exclusive benefit of Participants and their Beneficiaries and for defraying reasonable expenses of administering the Plan. Except as provided in Section 3.1, no part of the Trust fund will inure to the benefit of the Employer prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries.

The Trustee must hold, manage, administer, value, invest, reinvest and account for the Trust assets in accordance with the terms of the Plan and the Trust Agreement.

In its discretion, the Plan Administrator may recommend that the District terminate the Trustee, appoint new Trustee, and authorize the Plan Administrator to instruct the Trustee to transfer assets to a successor Trustee or Trustees.

- **7.2. Acceptance/Holding.** By executing the Trust Agreement, the Trustee accepts the Trust created under the Trust Agreement and agrees to perform the duties and obligations imposed under the Trust Agreement. The Trustee must hold and invest all Plan assets in the Trust under the Trust Agreement until paid in accordance with the Plan's terms.
- **7.3. Receipt of Contributions.** The Trustee is accountable to the District for the funds contributed to the Trust.
- **7.4. Deferred Compensation Investment Committee.** The District grants the DCIC full power and authority to invest, manage, and control Plan assets. The DCIC's membership, governance, and powers and duties are set forth in the Bylaws of the Deferred Compensation Investment Committee. The members of the DCIC will serve without additional compensation for services, but the District will pay all reasonable expenses of the DCIC.
- **7.5. Investment Policy Statement.** The DCIC will establish and maintain an Investment Policy Statement (IPS) for the Plan, and will periodically review that policy, as needed. In establishing and maintaining the IPS, the DCIC will review all pertinent Plan and Participant information and data and may consult with any professional advisors as it deems necessary or appropriate, and will consider matters such as the following:
  - (A) the Plan's objectives and the best method to accomplish those objectives, including the range of investment alternatives that should be available to Participants and their Beneficiaries in order to accomplish those objectives;
  - **(B)** the policies or procedures to be followed by the DCIC in the selection and monitoring of investment alternatives, including the benchmarks and other performance criteria, as well as investment fees and expenses; and
  - **(C)** the liquidity of Plan assets.
- **7.6. Investment Manager.** The Board of Directors may appoint one or more investment managers to manage (including the power to acquire and dispose of) any assets of the Plan. Only a person who is a registered investment advisor under the Investment Advisors Act of 1940, a bank or an insurance company qualified to do business in the State of California may be an investment manager under the Plan. The Trustee may be an investment manager.
- **7.7. Investment Funds.** The Trustee will establish and maintain investment funds designated by the DCIC or an investment manager designated by the Board of Directors. The assets of each investment fund will be managed by the Trustee or one or more investment managers, as designated, except for Participant-directed investments that permit Participants to direct the

investment of some or all of the assets in their Account in specific investments in accordance with Section 7.12 (Participant Direction of Investment).

**7.8. Participant Direction of Investment.** Consistent with Article V, the DCIC may consent in writing to permit Participants in the Plan to direct the investment to their Accounts under the Trust.

The Plan Administrator will advise the Trustee of the portion of the Trust credited to each Participant's Account under the Plan, and subject to such Participant direction.

As a condition of Participant direction, the DCIC may impose such conditions, limitations and other provisions as the DCIC may deem appropriate and as are consistent with policies created under the Plan's Investment Policy Statement and Article V.

The Trustee will report to the DCIC and Plan Administrator the net income, gain or losses incurred by each participant-directed Account separately from the net income, gain or losses incurred by the general Trust during the Plan Year.

**7.9. Group Trust Authority.** Subject to the Plan Administrator's direction, the Trustee may commingle, for investment or administration purposes, the assets (or portion of the assets) of the Trust in any common trust fund, collective investment fund, master trust, or group trust, but only if the commingled trust satisfies the requirements under Revenue Ruling 81-100 (as modified by Revenue Ruling 2004-67 and Revenue Ruling 2011-1) to be tax exempt.

#### ARTICLE VIII.

#### AMENDMENT, TERMINATION AND TRANSFERS

- **8.1. Amendment by Employer.** The Employer has the right, at any time and from time to time, to amend the Plan in whole or in part. Any amendment of the Plan by the Employer shall be expressed in an instrument executed by the Employer on the order of its Board of Directors and filed with the Trustee. Further, the Plan Administrator is also authorized, at any time and from time to time, to amend the Plan in whole or in part, to ensure compliance with applicable laws and regulations and further the administration of the Plan. The Plan Administrator may not amend the Plan to effectuate its termination.
  - **(A) Amendment Purposes Limited.** No Plan amendment may authorize or permit any portion of the assets held under the Plan to be used for or diverted to a purpose other than the exclusive benefit of Participants or Beneficiaries, except to the extent such assets are used to pay taxes or administrative expenses of the Plan. Any Plan amendment also may not cause or permit any portion of the assets held under the Plan to revert to or become property of the District.
  - **(B) Method of Amendment.** The Plan Administrator may amend the Plan in writing and without amendment of the MWD Code.
  - **(C) Termination/Freezing of Plan.** The Districts has the right, at any time, to terminate this Plan or to cease (freeze) further Salary Reduction Contributions to the Plan. Upon

freezing Salary Reduction Contributions to the Plan, the provisions of the Plan (other than provisions permitting continued Salary Reduction Contributions) may remain operative until the Plan is terminated. Upon Plan termination, all Deferred Compensation in the Plan shall be distributed to Participants and Beneficiaries as soon as is reasonably practicable following termination. The exercise of such right by the District to terminate the 401(k) or 457(b) Plans does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to MWD Code Section 6102 (Employee Rights).

**8.2. Purchase of Permissive Service Credit.** A Participant, prior to otherwise incurring a distributable event under Article IV (Time and Method of Payments of Benefits), may request a transfer of all or a portion of his or her Account to a governmental defined benefit plan (under Code § 414(d)) for: (a) the purchase of permissive service credit (under Code § 415(n)(3)(A)) under such plan, or (b) the repayment of contributions and earnings previously refunded with respect to a forfeiture of service credited under the plan (or under another governmental plan within the same State) to which Code § 415 does not apply by reason of Code § 415(k)(3). The Participant must follow the transfer request procedures established by the Plan Administrator.

#### ARTICLE IX.

#### **MISCELLANEOUS**

- 9.1. No Assignment or Alienation. A Participant or Beneficiary does not have the right to commute, sell, assign, pledge, transfer or otherwise convey or encumber the right to receive any payments under the Plan or Trust and the Plan Administrator and the Trustee will not recognize any such anticipation, assignment, or alienation. The payments and the rights under this Plan are non-assignable and nontransferable. Furthermore, except as provided in Section 4.6 (Distributions Under QDROs), a Participant's or Beneficiary's interest in the Trust is not subject to attachment, garnishment, levy, execution or other legal or equitable process. Notwithstanding the foregoing, the Plan Administrator may pay from a Participant's or Beneficiary's Account the amount the Plan Administrator finds is lawfully demanded under a levy issued by the Internal Revenue Service with respect to that Participant or Beneficiary or is sought to be collected by the United States Government under a judgment resulting from an unpaid tax assessment against the Participant or Beneficiary.
- **9.2. Effect on Other Plans.** This Plan does not affect benefits under any other retirement, pension, or benefit plan or system established for the benefit of the District's Employees, and participation under this Plan does not affect benefits receivable under any such plan or system, except to the extent provided in such plan or system.

- **9.3. Word Usage.** Words used in the masculine will apply to the feminine where applicable, and wherever the context of the Plan dictates, the plural will be read as the singular and the singular as the plural.
- **9.4. Governing Law.** The provisions of this Plan shall be construed, administered, and enforced in accordance with the provisions of applicable Federal law, and, to the extent applicable, the laws of the state of California.
- **9.5 Employment Not Guaranteed.** Nothing contained in this Plan, or any modification or amendment to the Plan, or in the creation of any Account, or the payment of any benefit, gives any Employee, Participant or Beneficiary any right to continue employment, any legal or equitable right against the District, the Plan Administrator, the Trustee, the DCIC, any other Employee of the District, or any agents thereof except as expressly provided by the Plan.
- 9.6 Notice, Designation, Election, Consent, and Waiver. All notices under the Plan and all Participant or Beneficiary designations, elections, consents or waivers must be in writing and made in the form and manner the Plan Administrator specifies or otherwise approves. To the extent permitted by Treasury regulations or other applicable guidance, any Plan notice, election, consent or waiver may be transmitted telephonically or electronically. Any person entitled to notice under the Plan may waive the notice or shorten the notice period except as otherwise required by the Code or other regulations or pronouncements.
- **9.7 Indemnification.** In accordance with the MWD Code, the District will indemnify, defend and hold harmless the members of the Board of Directors, the General Manager, the Plan Administrator, or any Employees who have administrative responsibility under the Plan, and the Deferred Compensation Investment Committee or any member thereof, with respect to any liability, loss, damage or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacities in the administration of or management of the assets of the Plan, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.

#### The Metropolitan Water District of Southern California

The amended and restated Plan is hereby adopted, effective January 1, 2024.

By:
Katano Kassaine
Assistant General Manager/Chief Financial Officer and
Interim Plan Administrator
Date Signed

# The Metropolitan Water District of Southern California 401(k) Savings Plan

**January 1, 2024** 

## THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 401(k) SAVINGS PLAN

# ABBREVIATED TABLE OF CONTENTS

BACKGROUND	D	Page 1
SECTION 1:	PLAN DEFINITIONS	Page 1
SECTION 2:	ELIGIBILITY AND PARTICIPATION	Page 10
SECTION 3:	PLAN CONTRIBUTIONS	Page 12
SECTION 4:	SPECIAL RULES AFFECTING GOVERNMENTAL PLANS	Page 17
SECTION 5:	CONTRIBUTION AND BENEFIT LIMITATIONS	Page 18
SECTION 6:	PARTICIPANT VESTING AND FORFEITURES	Page 22
SECTION 7:	PLAN DISTRIBUTIONS	Page 24
SECTION 8:	REQUIRED MINIMUM DISTRIBUTIONS	Page 35
SECTION 9:	SPOUSAL CONSENT RULES	Page 43
SECTION 10:	PLAN ACCOUNTING AND INVESTMENTS	Page 44
SECTION 11:	PLAN ADMINISTRATION AND OPERATION	Page 47
SECTION 12:	TRUST PROVISIONS FUNDS	Page 56
SECTION 13:	PARTICIPANT LOANS	Page 62
SECTION 14:	PLAN AMENDMENT, TERMINATION, MERGERS AND TRANSFERS	Page 65
SECTION 15:	MISCELLANEOUS	Page 69
<u>APPENDIX A</u> :	SECOND AMENDMENT TO THE METROPOLITAN WATER DISTRICT OF SOUTHER	RN CALIFORNIA

SAVINGS PLAN II TRUST AGREEMENT

The Metropolitan Water District of Southern California 401(k) Savings Plan Effective 2024.01.01

**Page** 

## THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 401(k) SAVINGS PLAN

### DETAILED TABLE OF CONTENTS

BACKGROUND......1 **SECTION 1 PLAN DEFINITIONS** 1.1 Account \_\_\_\_\_\_1 Account Balance......2 1.2 1.3 Alternate Payee ......2 1.4 1.5 Cash-Out Distribution......2 1.6 Catch-Up Contributions ......2 1.7 Catch-Up Contribution Limit......2 1.8 Code 2 Collectively Bargained Employee ......2 1.9 Compensation Limit ......2 1.10 1.11 Custodian.....3 -Deferred Compensation <u>InvestmentAdvisory</u> Committee or "DC<u>IAC"......</u>3 1.<mark>1615</mark> Distribution Calendar Year......3 1.<del>22</del>21 Employee .......4 1.<del>23</del>22 Employer or District.......4 1.2423 Employer Contributions .......4 1.<del>26</del>25 ERISA \_\_\_\_\_\_\_4 1.<del>29</del>28 Hardship......4 1.<del>3029</del> Hours of Service .......4 Performance of Duties.....4 (a) (b) Nonperformance of Duties......4 (c) Back Pay Award......4 (d) Related Employers ......5 Maternity/Paternity Leave......5 

The Metropolitan Water District of Southern California 401(k) Savings Plan

Effective January 1, XX, XX-20242

1.3332	Limitation Year	5
1.3433 Matching Contributions		
	Nonresident Alien Employee	
	5 Normal Retirement Age (NRA)	
	6 Participant	
	Part-Time Employee	
1.38	Part-Time Employee	<del></del> €
	<u>8</u> Plan	
1.4039	Plan Administrator	6
1.4140	<u>O</u> Plan Compensation	6
	(a) Determination Period	6
	(b) Partial Period of Participation	7
1.42 <u>41</u>	<u>1</u> Plan Year	7
1.43 <u>42</u>	2 Pre-Tax Deferrals	7
<b>1.443</b>	Qualified Birth or Adoption Distribution (QBOAD)	<u></u> 7
<u>1.4<del>5</del>4</u>	_Qualified Domestic Relations Order (QDRO)	7
<u>1.4<del>6</del>5</u>	Recurrent Employee	7
<u>1.476</u>	Rehired Annuitant	7
<u>1.487</u>	Required Beginning Date	7
	Rollover Contribution	
	<u>9 Roth Deferrals</u>	
	OSalary Deferral Election	
	Salary Deferrals	
	Seasonal Employee	
	Short Plan Year	
	Spouse	
	Total Compensation	
	Trust	
	Trust Agreement	
	Trustee	
<u>1.5<del>9</del>9</u>	Valuation Date	S
	SECTION 2	
	ELIGIBILITY AND PARTICIPATION	4.0
2.1	Eligibility	
2.2	Eligible Employees	
	(a) Excluded Employees	
	(b) Only Employees May Participate in the Plan	
	(c) Ineligible Employee Becomes Eligible Employee	
	(d) Eligible Employee Becomes Ineligible Employee	
2.2	(e) Improper Exclusion of Eligible Participant	
2.3	Eligibility Requirements	
	(a) Service Requirement	
2.4	(b) Entry Date	
2.4	Participation on Effective Date of Plan	11 11

The Metropolitan Water District of Southern California 401(k) Savings Plan

Effective January 1, XX, XX 20242

#### SECTION 3 PLAN CONTRIBUTIONS

<b>3.1</b> Ty	pes of Contributions	12
<b>3.2</b> Sa	ary Deferrals	11
(a	Salary Deferral Election	12
(b	Change in Deferral Election	12
(c)	Catch-Up Contributions	13
	(1) Catch-Up Contribution Limit	13
	(2) Special Treatment of Catch-Up Contributions	13
<b>3.3</b> Ma	tching Contributions	13
(a	Employee Matching Formula	13
(b	Employer Matching Contributions	13
8.4 Ro	llover Contributions	13
<b>8.5</b> Ro	th Deferrals	<u></u> 14
<b>3.6</b> In-	Plan Roth Conversions	
<u>(a</u>	Eligibility to Elect In-Plan Roth Rollover	15
<u>(b</u>		
<u>(c)</u>	Plan Loans	15
<u>(d</u>		
<u>(e</u>		
<u>(f)</u>		
<u>(g</u>		
<u>(h</u>	In-Plan Roth Conversion Definitions	<u></u> 16
	(1) "Direct In-Plan Roth Rollover"	<u></u> 16
	(2) "In-Plan Roth Rollover"	<u></u> 16
	(3) "In-Plan Roth Rollover Account"	16
	SECTION 4	
	SPECIAL RULES AFFECTING GOVERNMENTAL PLANS	
	vernmental Plan	
(a	•	
(b	ERISA Exempt	17
	CDOWN ON F	
	SECTION 5	
11 0	CONTRIBUTION AND BENEFIT LIMITATIONS	4.5
	de § 415 Limitation	
(a)	•	
(b	•	
	(1) This Plan's Code § 415 Limitation	
	(2) Annual Additions Reduction	
(~		
(c)		
	(5) Maximum Permissible Amount	19

The Metropolitan Water District of Southern California 401(k) Savings Plan

Effective January 1, XX, XX 20242

		(6)	Total Compensation	19
			i. Total Compensation Actually Paid or Made Available	20
	(d)	Resto	orative Payments	20
	(e)		ective Provisions	
	<b>(f)</b>	Chan	ge of Limitation Year	2
5.2	Elect	ive Defe	erral Dollar Limit	2
	(a)	Exces	ss Deferrals	2
	(b)	Corre	ection of Excess Deferrals	20
		(1)	Amount of Corrective Distribution	2
		(2)	Allocable Gain or Loss	
		(3)	Taxation of Corrective Distribution	2
		(4)	Coordination of Other Provisions	
		(5)	Correction of Excess Deferrals Under Plans Not Maintained by the Employer	2
			SECTION 6	
. 1	X7		PARTICIPANT VESTING AND FORFEITURES	2
6.1			ontributions	
6.2			Benefits	
	(a)		ing Participant or Beneficiary	
		(1)	Reasonable Diligent Search	
		(2)	Forfeiture of Account of Missing Participant or Beneficiary	
	(la)	(3)	Expenses Attributable to Search for Missing Participant	
<i>(</i> )	(b)		loyer Matching Contributions Attributable to Excess Deferrals	
6.3			Forfeitures	
	(a) (b)		ection of Contributions nent of Plan Expenses	
	(6)	1 dy 11	•	
			SECTION 7 PLAN DISTRIBUTIONS	
7.1	Avail	able Foi	rms of Distribution	2
7.2			ible for Distribution	
7.3	Parti	cipant C	Consent	2
	(a)		untary Cash-Out Threshold	
	(b)	Parti	cipant Notice	2
	(c)	Speci	ial Rules	2
7.4	Direc	t Rollov	/ers	2
	(a)	Defin	iitions	2
		(1)	Eligible Rollover Distribution	2
		(2)	Eligible Retirement Plan	2
		(3)	Direct Rollover	2
	(b)	Direc	t Rollover Notice	2
	(c)		t Rollover by Non-Spouse Beneficiary	
	(d)		t Rollover of Non-Taxable Amounts	
7.5	Distr		Upon Termination of Employment	
	(a)		unt Balance Not Exceeding Cash-Out Threshold	
	(b)		unt Balance Exceeding Cash-Out Threshold	
7.6			Upon Death	
	(a)	Deatl	h After Commencement of Benefits	2

	(b)			Commencement of Benefits	
	(c)		_	ributions	
	(d)			Participant's Beneficiary	
		(1)		Retirement Death Benefit	
		(2)		etirement Death Benefit	
		(3)	Defaul	t Beneficiaries	
			i.	Prior to August 1, 2018	
			ii.	After August 1, 2018	
		<b>(4)</b>		fication of Beneficiaries	
		(5)		of Beneficiary	
		(6)		ce From Spouse	
7.7				ed Employees	
7.8				ns	
	(a)			ributions	
	(b)			tributions	
	(c)			als	
	(d)	-		ervist Distribution	
		(1)	-	ied Reservist Distribution	
		(2)		Duty	
	(e)			ribution	
		(1)	_	arbor Hardship Distribution	
			İ. 	Immediate and Heavy Financial Need	
		(0)	ii.	Determination of Whether Distribution Necessary to Satisfy Need	
		(2)		nt Available for Hardship Distribution	
		(3)		bility to Terminated Employees	
	(0	(4)		onth Suspension of Salary Deferrals	
	(f)			ributions	
	(g)			n or Adoption Distributions (QBOADs)	
	<u>(h)</u>			Related Distributions (CRDs)	
				navirus-Related Distribution" Defined	
		<u>(2)</u>		gate Dollar Limit	
		(3)		ntribution of CRDs	
<b>7</b> 0	<b>C</b>	<u>(4)</u>		ipant Certification	
7.9				n	
7.10				redit Under Defined Benefit Plan	
7.11	Corre	ction of	Quannica	ation Defects	34
				CECTION O	
				SECTION 8	
0 1	Dogui	nad Mir	imum D	REQUIRED MINIMUM DISTRIBUTIONS	24
8.1 8.2				istributions <u>(RMDs)</u> Before Required Distributions Begin	
0.4			-		
	(a)			use as a Designated Beneficiary	
	(b)		-	Designated Beneficiary I Beneficiary	
	(c)		_	iving Spouse	
8.3	(d)			istributions During Participant's Lifetime	
0.3	_				
	(a)			quired Minimum Distributions for Each Distribution Calendar Year	
	(b)	Lilett	me nequ	ired Minimum Distributions Continue Through Year of Participant's Death	50

8.4	Requ	uired Minimum Distributions After Participant's Death	36				
	(a)	Death on or After Date Required Distributions Begin Begin	36				
		(1) Participant Survived by Designated Beneficiary	36				
		(2) No Designated Beneficiary	37				
	(b)	Death Before Date Required Distributions Begin	37				
	(1) Participant Survived by Designated Beneficiary						
		(2) No Designated Beneficiary	37				
		(3) Death of Surviving Spouse Before Distributions to Surviving Spouse					
		are Required to Begin	37				
8.5	Defin	nitions					
	(a)	Designated Beneficiary					
	(b)	Distribution Calendar Year					
	(c)	Life Expectancy					
	(d)	Account Balance					
	(e)	Required Beginning Date <u>(RBD)</u>					
8.6	Speci	ial Rules					
	(a)	Election to Allow Participants or Beneficiaries to Elect 5-Year Rule					
	(b)	Forms of Distribution					
	(c)	Waiver of Required Minimum Distributions					
	(d)	Treatment of Trust Beneficiaries As Designated Beneficiaries					
	(e)	Special Rules Applicable to Trust Beneficiaries					
		(1) Information that Must be Supplied to Plan Administrator	40				
		i. Required Minimum Distribution Before Death Where Spouse is Sole					
		Beneficiary					
		ii. Required Minimum Distribution After Death					
		(2) Relief for Discrepancy					
8.7	Trans	sitional Rule	41				
		SECTION 9					
		SPOUSAL CONSENT RULES					
9.1		ication of Joint and Survivor Annuity Rules					
9.2	Spou	isal Consent	42				
		GROWN AND					
		SECTION 10					
101	Donti	PLAN ACCOUNTING AND INVESTMENTS	42				
10.1		cipant Accounts					
10.2		ation of Accounts					
	(a)	Periodic Valuation					
	(b)	Daily Valuation					
10.3	(c)	Interim Valuationsstments to Participant Accounts					
10.3	,	<u>*</u>					
	(a)	Net Income or Loss					
	(b)	Distributions and Forfeitures from A Participant's Account  Dividends					
	(c)	Contributions and Forfeitures Allocated to A Participant's Account					
	(d)	Directed Accounts					
	(e) (f)	Plan Related Fees and Expenses					
104		e or Unit Accounting					
4 W.T	Julait	C VI VIIICINCUUIICIIE					

10.5	Susper	ise Acc	counts	44					
10.6	Investments Under the Plan Deferred Compensation Investment Committee								
10.7			Options Policy Statement						
			non/Collective Trusts and Collectibles						
10.7	Participant-Directed Investments								
10.8	Investment Manager								
10.9	Investment Funds								
			Directed Investments						
	(a)		s on Participant Investment Direction						
	(b)		re to Direct Investment						
	( )								
			SECTION 11						
			PLAN ADMINISTRATION AND OPERATION						
11.1			strator						
11.2	Role of		Administrator						
	(a)		nation or Removal of Plan Administrator						
			nnification of Plan Administrator and Others						
11.3	Duties,		rs and Responsibilities of the Plan Administrator						
	(a)		ral Duties, Powers and Responsibilities						
	(b)	Deleg	gation of Duties, Powers and Responsibilities	46					
	(c)	Speci	fic Plan Administrator Responsibilities	46					
	(d)		pretation of Plan						
	(e)	Notic	es and Filings	47					
	<b>(f)</b>	Appro	opriate Adjustments	47					
11.4	Immur	nities		48					
11.5	Plan A	dminis	stration Expenses	48					
	(a)	Reaso	onable Plan Administration Expenses	48					
	(b)		Expense Allocation						
	(c)	Exper	nses Related to Administration of Former Employee or Surviving Spouse	49					
11.6	Claims	Proce	dure and Determinations	49					
	(a)	Benef	fit Claims Procedures	49					
	(b)		fit Claims Determinations						
11.7	Qualifi		nestic Relations Orders (QDROs)						
	(a)	In Ge	neral	49					
	(b)	Defin	itions Related to QDROs						
		(1)	QDRO	49					
		(2)	Domestic Relations Order	49					
		(3)	Alternate Payee	50					
	(c)	Recog	gnition as a QDRO	50					
	(d)	Conte	ents of QDRO	50					
	(e)	Impe	rmissible QDRO Provisions	50					
	<b>(f)</b>	Imme	ediate Distribution to Alternate Payee	51					
	(g)	Fee fo	or QDRO Determination	51					
	(h)	Defau	ılt QDRO Procedure	51					
	(i)	Acces	ss to Information	51					
		(1)	Notifications to Participant and Alternate Payee	51					
		(2)	Alternate Payee Representative	51					
		(3)	Evaluation of Domestic Relations Order	51					

		i.	Separate Accounting	51
		ii.	Separate Accounting Until the End of 18-Month Period	
		iii.	Preliminary Review	
		iv.	Notification of Determination	
11.8	Militai	rv Service		
	(a)		ts Under Qualified Military Service	
	(b)		itions	
	(c)		ntributions	
	(-)	op		
			SECTION 12	
12.1	Ectabl	ichment of Tri	TRUST PROVISIONS FUNDS  1st	5.4.
12.1 12.2			1e Trustee	
12.2			ties Regarding Administration of Trust	
			ties Regarding Investment of Plan Assets	
12.3			ne Employer	
			tee	
			Iministrator and Employer	
			sation	
			noval of Trustee/Appointment of Successor	
			rustee	
			red	
			todian	
			t Provisions	
_	_		Annuity Contracts, and Insurance Contracts	
		· · · · · · · · · · · · · · · · · · ·	Trusts	
	GOIIII	ion, donective	114000	<u></u>
			SECTION 13	
			PARTICIPANT LOANS	
13.1	Availa	bility of Partic	ipant Loans	60
13.2	Must b	oe Available in	Reasonably Equivalent Manner	60
13.3	Loan I	Limitations		60
13.4	Limit	on Amount and	d Number of Loans	60
13.5	Partic	ipant Must Be	Creditworthy	61
13.6	Reaso	nable Rate of I	nterest	61
13.7	Adequ	ate Security		61
13.8	Period	lic Repayment		61
13.9	Proced	dures for Loan	Default	62
13.10	Termi	-	loyment	
	(a)	Outstanding	Loan	62
	(b)		/er	
12 11	Amon	dment of Plan	to Fliminate Particinant Loans	62

# **SECTION 14**

The Metropolitan Water District of Southern California 401(k) Savings Plan

14.1	Dlan A	mondn		AMENDMEN 1, 1 ERMINATION, MERGERS AND 1 RANSFERS	63			
14.1				Amend				
	(a)	(1)		idment Purposes Limited				
		(2)		od of Amendment				
	(b)			e of Plan Amendments				
	(D)	(1)		e of Fight Amendments				
		(2)		pactive Effect of PPA, HEART and WRERA Provisions				
		(4)	i.	Hardship Distributions				
			ii.	Direct Rollovers by Non-Spouse Beneficiary				
			iii.	Direct Rollover of Non-Taxable Amounts				
			iv.	Rollovers to Roth IRA				
			<b>v.</b>	Distribution Notice Periods				
			vi.	Content of Notice of A Participant's Right to Defer Receipt of A Distribution				
			vii.	Qualified Domestic Relations of Orders (QDROs)				
			viii.	Penalty-Free Withdrawals for Individuals Called to Active Duty				
			ix.	Benefit Accruals for Participants on Qualified Military Service				
			х.	Differential Pay				
			xi.	Waiver of Required Minimum Distributions				
			xii.	Final Code § 415 Regulations	65			
		(3)	Merge	ed Plans	65			
14.2	Indem	<del>nities</del>			<del>6</del> 5			
<del>14.3</del>	–Plan T							
	(a)	Full and Immediate Vesting						
	(b)		tribution Upon Plan Termination6					
	(c)		issing Participants					
	(d)			nation				
44.40	<u>(e)</u>			<u>ghts</u>				
	Plan Merger or Consolidation							
14. <del>5</del> 4	Transfer of Assets							
	(a)							
	(b) Transfer of Plan to Unrelated Employer							
				SECTION 15				
				MISCELLANEOUS				
15.1	Exclus	ive Ber	nefit		67			
15.2				Contributions				
15.3				Limited				
15.4		-	_					
15.5				for Payments				
15.6				······································				
<b>15.7</b>	No Gu	arantee	Agains	t Loss	68			
<b>15.8</b>	Gende	r and N	lumber.		68			
<b>15.9</b>	Use of	IRS Co	mplianc	ce Programs	68			
<b>15.10</b>	Gover	ning La	w		68			
15.11	Waive	r of No	tice		68			
15.12	Use of	Electro	onic Med	dia	69			
15.13	Severability of Provisions69							

15.14	Binding Effect69
15.15	Headings69
15.16	Uniformity
15.17	Indemnification

## BACKGROUND

The Metropolitan Water District of Southern California (the "District") Board of Directors' Minute Item 35632 established a profit-sharing plan with a cash or deferral arrangement known as The Metropolitan Water District of Southern California Savings Plan ("Savings Plan I") effective July 1, 1985. By Board of Directors' Minute Item 35816, the District also established another profit-sharing plan with a cash or deferral arrangement known as "Savings Plan II" applicable to a separate group of employees effective October 1, 1985.

As a result of Board of Directors' Minute Item 46592, "Savings Plan I" was merged into "Savings Plan II" and renamed The Metropolitan Water District of Southern California Consolidated Savings Plan (the "Plan") effective January 1, 2006, and incorporated into Division VI, Chapter 7, Article 6 of The Metropolitan Water District of Southern California Administrative Code ("MWD Code"). The Plan has been amended and restated from time to time. The Plan was most recently restated and renamed The Metropolitan Water District of Southern California 401(k) Savings Plan effective August 1, 2018. The Plan was further amended and restated effective January 1, 2020. The Plan is hereby amended and restated effective January 1, 2024, except as otherwise stated herein.

The Plan is a tax-qualified governmental defined contribution plan under Internal Revenue Code (the "Code") §§ 401(a) and 414(d) that is a profit sharing plan pursuant to Code § 401(a)(27). The Plan has a grandfathered cash or deferral arrangement under Code § 401(k).

All contributions made to the Plan, all property and rights purchased with such amounts and all income attributable to such amounts shall be held in trust for the exclusive benefit of Participants, Beneficiaries and Alternate Payees under the Plan in accordance with Internal Revenue Code §401(a)(2). Prior to the satisfaction of all liabilities with respect to employees and their beneficiaries under the Plan's Trust, it shall be impossible for any part of the corpus or income of the Trust to be (within the taxable year or thereafter) used for, or diverted to, purposes other than for the exclusive benefit of the District's employees or their beneficiaries.

The purpose of the Plan is to attract and retain certain officers and employees to The Metropolitan Water District of Southern California.

#### **SECTION 1**

### PLAN DEFINITIONS

This Section 1 contains definitions for common terms that are used throughout the Plan. All capitalized terms under the Plan are defined in this Section 1 or in the relevant section of the Plan document where such term is used.

- **1.1 Account.** The separate Account established and maintained for each Participant, Beneficiary or Alternate Payee under the Plan. A Participant Account may have any or all of the following separate sub-Accounts:
  - Pre-Tax Salary Deferral Account
  - Roth Deferral Account
  - Matching Contribution Account

- Rollover Contribution Account
- Roth Rollover Account
- Transfer Account

The Plan Administrator may establish other Account types, as it deems necessary, for the proper administration of the Plan.

- **1.2 Account Balance.** Account Balance shall mean a Participant's balances in all of the Accounts maintained by the Plan on his or her behalf.
- **Alternate Payee.** A person designated as having a right to receive all or a portion of the Participant's benefit pursuant to a QDRO.
- **Beneficiary.** A person, including a trust, designated by the Participant (or in the absence of a proper beneficiary designation, by the terms of the Plan) to receive a benefit under the Plan upon the death of the Participant. See Section 7.6(d) (Determining a Participant's Beneficiary) for the applicable rules for determining a Participant's Beneficiaries under the Plan.
- **1.5** <u>Cash-Out Distribution</u>. A total distribution made to a Participant who incurs a termination from employment.
- **1.6** Catch-Up Contributions. Salary Deferrals that may be made in excess of an otherwise applicable IRS Code § 402(g) limit by a Participant who attains age 50 or over by the end of the applicable Plan Year.
- **1.7** Catch-Up Contribution Limit. The annual limit applicable to Catch-Up Contributions as set forth in Section 3.2(c)(1)(Catch-Up Contributions).
- **1.8 Code.** The Internal Revenue Code of 1986, as amended.
- **1.9** <u>Collectively Bargained Employee</u>. An Employee who is included in a unit of Employees covered by a collective bargaining agreement between the District and Employee representatives and whose retirement benefits are subject to good faith bargaining.
- Year for purposes of determining a Participant's Plan Compensation. For Plan Years beginning on or after January 1, 1994, and before January 1, 2002, the Compensation Limit taken into account for determining benefits provided under the Plan for any Plan Year is \$150,000, as adjusted for increases in cost-of-living in accordance with Code § 401(a)(17)(B). For any Plan Years beginning on or after January 1, 2002, the Compensation Limit is \$200,000, as adjusted for cost-of-living increases in accordance with Code § 401(a)(17)(B). In determining the Compensation Limit for any applicable period (the "Plan Year"), the cost-of-living adjustment in effect at the beginning of the calendar year applies to any Plan Year that begins within such calendar year.

If a Plan Year consists of fewer than 12 months, the Compensation Limit for such period is an amount equal to the otherwise applicable Compensation Limit multiplied by a fraction, the numerator of which is the number of months in the short determination period, and the denominator of which is 12. A Plan Year will not be considered to be less than 12 months merely because compensation is taken into account only for

the period the Employee is a Participant. If Salary Deferrals or Matching Contributions are separately determined on the basis of specified periods within the determination period (e.g., on the basis of payroll periods), no proration of the Compensation Limit is required with respect to such contributions.

- **1.11** <u>Custodian</u>. An organization that has custody of all or any portion of the Plan assets.
- Deferred Compensation InvestmentAdvisory Committee or "DCIAC". An Employer established investmentadvisory The committee which is comprised of the Human Resources Group Manager, Assistant General Manager, Operations (or designee), Treasury & Debt Manager, Finance Group Manager, General Auditor (or designee), Risk Manager District's Chief Financial Officer, Treasurer, Chief Administrative Officer, and Treasury and Debt Manager, the Plan Administrator, a staff member and the Manager of the Human Resources Group's Benefits Unit, and a member appointed to represent retirees and onerepresentative from each of the District's four bargaining units: MAPA/AFSCME Local 1001, AFSCME Local 1902, Supervisor Associations, Affiliate of IBEW Local 11, Unit 76, and Association of Confidential Employees AFSCME Local 1902, Management and Professional Employee's Association, Supervisors' Association, and Association of Confidential Employees. . The DCIC committee meets quarterly and is toresponsible for performing the fiduciary duties related to the management of Plan assets.advise the Plan-Administrator on plan administrative policies, investments, educational programs, the effect of regulatory and tax law changes, and the viability of proposed plan changes based on Participant requests. \_Thecommittee DCIC decides by an advises only on an affirmative vote by a majority of the committee or by unanimous consent. Non-voting DCIC attendees include the Human Resource Benefits Manager, the Human Resources Principal Analyst, and the General Counsel's representative described in Section 10.6.-
- **1.131.12 Defined Contribution Plan.** A plan that provides for individual Accounts for each Participant to which all contributions, forfeitures, income, expenses, gains and/ or losses under the Plan are credited or deducted. A Participant's benefit under a Defined Contribution Plan is based solely on the fair market value of his/her vested Account Balance.
- **1.14**1.13 **Differential Pay.** Certain payments made by the District to an individual while the individual is performing service in the Uniformed Services.
- **1.151.14 Directed Account.** That portion of Participant's Account that is subject to Participant direction of investment and which is maintained by the Plan Administrator as a Directed Account.
- **1.16**1.15 **Distribution Calendar Year.** A calendar year for which a minimum distribution under Code § 409(a)(9) is required.
- **1.17 1.16 Distribution Starting Date.** The date a Participant commences distribution from the Plan. If a Participant commences distribution with respect to a portion of his/her Account Balance, a separate Distribution Starting Date applies to any subsequent distribution.
- **1.18**1.17 **Effective Date.** The Plan was originally established and\_effective July 1, 1985. The Plan has since been amended and restated November 18, 2014. The current Effective Date of the Plan as amended and restated is August 1, 2018. See Section 14.1(bd) (Effective Date of Plan Amendments) for special rules concerning the retroactive effective date of provisions under the Plan designed to comply with the requirements of the Pension Protection Act of 2006 (PPA).

- 1.191.18 Elective Deferral Dollar Limit. The maximum amount of Elective Deferrals a Participant may make for any calendar year.
- **1.20**1.19 **Elective Deferrals.** A Participant's Elective Deferrals is the sum of all Salary Deferrals made by the Participant. Elective Deferrals shall not include any amounts properly distributed as an Excess Amount under Code § 415.
- 1.21 <u>Eligible Employee</u>. An Eligible Employee is an Employee who (i) is not a <u>Seasonal Employee</u>, a <u>Leased Employee excluded under Section 2.2 of the Plan</u> or and <u>Employee</u> excluded by agreement, and (ii) meets the eligibility requirements in Section 2.32 of the Plan.
- **1.221.21 Employee.** An Employee is any individual employed by the District as a common law employee. An independent contractor is not treated as an Employee of the recipient organization for purposes of this Plan or the District.
- **1.23 1.22 Employer or District**. Employer or District means The Metropolitan Water District of Southern California ("District"), a political subdivision.
- **1.241.23 Employer Contributions.** Contributions the Employer makes pursuant to Sections 3.2 (Salary Deferrals) and 3.3 (Matching Contributions).
- **1.25**1.24 Entry Date. The commencement of the first payroll period following the date on which an Eligible Employee satisfies the eligibility requirements described in Section 2 (Eligibility and Participation).
- **1.26**1.25 **ERISA.** The Employee Retirement Income Security Act of 1974, as amended.
- **1.27 1.26 Excess Amount.** Amounts which exceed the Code § 415 Limitation.
- **1.28**1.27 **Excess Deferrals.** Elective Deferrals that exceed the Elective Deferral Dollar Limit.
- **1.29 1.28 Hardship.** A heavy and immediate financial need which meets the requirements of Section 7.8(e) (Hardship Distributions).
- **1.301.29 Hour of Service.** Each Employee of the District will receive credit for each Hour of Service he/she works for purposes of applying the eligibility rules under the Plan. An Employee *will not* receive credit for the same Hour of Service under more than one category listed below:
  - (a) Performance of Duties. Hours of Service include each hour for which an Employee is paid, or entitled to payment, for the performance of duties for the District. These hours will be credited to the Employee for the computation period in which the duties are performed. In the case of Hours of Service to be credited to an Employee in connection with a period of no more than 31 days which extends beyond one computation period, all such Hours of Service may be credited to the first computation period or the second computation period. Hours of Service under this subsection (a) must be credited consistently for all Employees within the same job classifications.
  - **Nonperformance of Duties.** Hours of Service include each hour for which an Employee is paid, or entitled to payment, by the District on account of a period of time during which no duties are performed (irrespective of whether the employment relationship has terminated) due to vacation, holiday, illness, incapacity (including disability), layoff, jury duty, military duty or leave of absence.

The purpose of this subsection (b) is to prevent a break in service occurring due to the leave of absence. Thus, no more than the required 501 hours of service needed to prevent a break in service will be credited under this paragraph.

- **Back Pay Award.** Hours of Service include each hour for which back pay, irrespective of mitigation of damages, is either awarded or agreed to by the District. The same Hours of Service *will not* be credited both under subsection (a) or subsection (b), as the case may be, and under this subsection (c). These hours will be credited to the Employee for the Plan Year to which the award or agreement pertains rather than the Plan Year in which the award, agreement or payment is made.
- (d) Related Employers. Hours of Service will be credited for employment with any Related Employer. A Related Employer includes all members of a controlled group, all commonly controlled entities or affiliated service groups of which the Employer is a part, and any other entity required to be aggregated with the Employer pursuant to regulations under Code §414(o). For purposes of Hours of Service, the Employer and any Related Employers are treated as a single Employer, unless specifically stated otherwise.
- **Maternity/Paternity Leave.** Solely for purposes of determining eligibility, an individual who is absent from work for maternity or paternity reasons *will not* be deemed to have had a break in continuous hours of service by reason of an absence from work for maternity or paternity reasons. For purposes of this paragraph, an absence from work due to maternity or paternity reasons means an absence:
  - by reason of the pregnancy of the individual,
  - by reason of birth of a child of the individual,
  - by reason of the placement of a child with the individual in connection with the adoption of such child by such individual, or
  - for purposes of caring for such child for a period beginning immediately following such birth or placement.
- **1.31**1.30 **Code § 415 Limitation.** The limit on the amount of Annual Additions a Participant may make under the Plan during any Limitation Year.
- 1.321.31 Leased Employee. An individual who performs services for the District pursuant to an agreement between the District and a leasing organization, and who satisfies the definition of a Leased Employee under Code § 414(n).
- **1.33**1.32 **Limitation Year.** The measuring period for determining whether the Plan satisfies the Code § 415 Limitation.
- **1.34 Matching Contributions.** Matching Contributions are contributions made by the District on behalf of a Participant on account of Salary Deferrals made by the Participant under this Plan.

- 1.351.34 Nonresident Alien Employee. An Employee is a nonresident alien if he or she is neither a citizen of the United States nor a resident of the United States for U.S. tax purposes (as defined in Code § 7701(b)), and who does not have any earned income (as defined in Code § 911) for the District that constitutes U.S. source income (within the meaning of Code § 861). If a nonresident alien Employee has U.S. source income, he/she is treated as satisfying this definition if all of his/her U.S. source income from the District is exempt from U.S. income tax under an applicable income tax treaty.
- **1.361.35 Normal Retirement Age (NRA)**. For purposes of in-service distributions, Normal Retirement Age is 59½.
- 1.371.36 Participant. A Participant is an Eligible Employee who has satisfied the conditions for participating under Section 2.3 (Eligibility Requirement) of this Plan. A Participant also includes any Employee (or former Employee) who has an Account Balance under the Plan, including an Account Balance derived from a rollover or transfer from another qualified plan or IRA. A Participant is entitled to share in an allocation of contributions or forfeitures under the Plan for a given year only if the Participant is an Eligible Employee as defined in Section 2.2, and satisfies the allocation conditions set forth in Section 3.3 (Matching Contributions).

An Eligible Employee is treated as a Participant with respect to Salary Deferrals once the Employee has satisfied the eligibility conditions and has reached his or her Entry Date for making such contributions, even if the Eligible Employee chooses not to actually make such contributions to the Plan. An Eligible Employee is treated as a Participant with respect to Matching Contributions once the Eligible Employee has satisfied the eligibility conditions and has reached his or her Entry Date, even if the Employee does not receive a Matching Contribution because of the Employee's failure to make contributions eligible for the Matching Contribution.

- **1.38**1.37 **Part-Time Employee.** A Part-Tim Employee is an Employee who is normally scheduled to work on average 20 to 39 hours per week.
- **1.39**1.38 Plan. Plan means The Metropolitan Water District of Southern California 401(k) Savings Plan.
- **1.40**1.39 **Plan Administrator**. The Plan Administrator is the General Manager or other person or entity designated by the District to be responsible for the administration and operation of the Plan. A Plan Administrator also includes a Qualified Termination Administrator ("QTA") that assumes the responsibilities of Plan Administrator.
- 1.411.40 Plan Compensation. Plan Compensation is Total Compensation except as modified herein. Plan Compensation includes all cash compensation payable by the District during the Plan Year to, or for the benefit of, an Employee in return for services and any deemed compensation under Code § 414(u)(7). Plan Compensation also includes payments to disabled Participants described in Section 1.54(d) if the payments are not the result of a leave donation(s).

Effective as of January 1, 1997, Plan Compensation shall include pre-tax contributions to a cafeteria plan under Code § 125, an eligible deferred compensation plan under § 457, employer pick-up contributions under § 414(h)(2) or elective contributions that are not included in the Employee's gross income as a qualified transportation fringe benefit under 132(f)(4).

Plan Compensation also does not include other fringe benefits such as medical credit, cell phone allowances, or professional development, wellness, and tuition reimbursements.

Nor does Plan Compensation include payments for leave donations (e.g., hours donated by Employees to a Participant) or payments made by Third Party Providers (TPAs). Plan Compensation also excludes all amounts earned with a related employer who is not a participating employer in this Plan.

- (a) <u>Determination period</u>. Plan Compensation is determined based on the Plan Year.
- **Partial period of participation.** Even if an Employee is a Participant for only part of a Plan Year, Plan Compensation shall be determined over the entire Plan Year. In determining whether an Employee is a Participant for purposes of applying this subsection (b), the Employee's status will be determined solely with respect to the contribution type for which the definition of Plan Compensation is being determined. Plan Compensation does not include any amounts earned for any period while an individual is not an Eligible Employee.

In no case may Plan Compensation for any Participant exceed the Compensation Limit.

- **1.42 1.41 Plan Year**. The calendar year.
- **1.43 Pre-Tax Deferrals.** Pre-tax Deferrals are a Participant's Salary Deferrals that are not includible in the Participant's gross income at the time deferred.
- 1.43 Qualified Birth or Adoption Distribution (QBOAD). A distribution to a Participant if made (1) during the 4one-year period beginning on the date on which a child of the Participant is born or on which the legal adoption by the Participant of an eligible adoptee is finalized, (2) with the respect to a child born or eligible adoptee the Participant includes the name, age, and the Taxpayer Identification Number of the child born or eligible adoptee on the Participant's tax return for the taxable year in which the distribution is made, and (3) only to the extent such distribution from all plans maintained by the Employer to such Participant does not exceed \$5,000, in the aggregate, for each child born or eligible adoptee. An eligible adoptee is any individual (other than a child of the Participant's spouse) who has not attained age 18 or is physically or mentally incapable of self-support.
- **1.44 Qualified Domestic Relations Order (QDRO).** A domestic relations order that provides for the payment of all or a portion of the Participant's benefits to an Alternate Payee and satisfies the requirements under Code § 414(p).
- **1.45** Recurrent Employee. Employees hired for an indefinite period of time on an irregular basis, such as intermitted, emergency, or on-call.
- **1.46** Rehired Annuitant. A rehired Employee who meets the definition of retired annuitant in California Government Code § 7522.56.
- **1.47 Required Beginning Date.** The date by which minimum distributions must commence under the Plan.
- **1.48** Rollover Contribution. A contribution made by an Employee to the Plan attributable to an Eligible Rollover Distribution from another qualified plan or IRA.

- 1.49 Roth Deferrals. Deferral contributions that are designated irrevocably by the Participant at the time of the cash or deferred election as a Roth Deferral that is being made in lieu of all or a portion of the Pre-Tax Deferrals the Participant is otherwise eligible to make under the Plan and treated by the Employer as includible in the Participant's income at the time the Participant would have received that amount in cash if the Participant had not made a cash or deferred election.
- **1.50 Salary Deferral Election.** An agreement between a Participant and the District, whereby the Participant elects to have a specific amount withheld from his/her Plan Compensation and the District agrees to contribute such amount into the Plan.
- **1.51** Salary Deferrals. Amounts contributed under the Plan at the election of the Participant, in lieu of cash compensation, which are made pursuant to a Salary Deferral Election or other deferral mechanism. Salary Deferrals may consist of Pre-Tax Deferrals and/or Roth Deferrals. Salary Deferrals shall not include any amounts properly distributed as an Excess Amount under Code § 415. An Employee's Salary Deferrals are treated as employer contributions for all purposes under this Plan, except as otherwise provided under the Code or U.S. Treasury regulations.
- **1.52 Seasonal Employee.** An Employee who normally works on a full-time basis less than five months during any Plan Year.
- **1.53 Short Plan Year.** Any Plan Year that is less than 12 months long, either because of the amendment of the Plan Year, or because the Effective Date of a new Plan is less than 12 months prior to the end of the first Plan Year.
- 1.54 Spouse. Subject to any additional federal tax guidance from the Internal Revenue Service or other agency or court, a Spouse is any individual who is lawfully married to the Participant under a state or foreign jurisdiction, without regard to the location of the Employer or the state where the Participant and Spouse are domiciled. Notwithstanding the previous sentence, a former Spouse of the Participant will be treated as the Spouse or surviving Spouse to the extent provided under a valid QDRO and any current Spouse will not be treated as the Spouse or surviving Spouse with respect to any portion of Participant's benefit covered by a valid QDRO.
- **1.55 Total Compensation.** A Participant's compensation for services with the District, as defined herein:
  - (a) Wages within the meaning of Code § 3401(a) and all other payments of compensation to an Employee by the District (in the course of the District's trade or business) for which the Districtis required to furnish the Employee a written statement under Code §§ 6041(d), 6051(a)(3), and 6052, determined without regard to any rules under Code § 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed.
  - Includes Elective Deferrals (as defined in Section 1.20), elective contributions to a cafeteria plan under Code § 125 or to an eligible deferred compensation plan under Code § 457, employer pick-up contributions under Code § 414(h)(2), and elective contributions that are not includible in the Employee's gross income as a qualified transportation fringe under Code § 132(f)(4).

- (c) Effective for the first Plan Year beginning on or after July 1, 2007, Total Compensation includes compensation paid after an Employee severs employment with the Employer, provided the compensation is paid by the later of 2-½ months after severance from employment with the Employer maintaining the Plan or the end of the Plan Year that includes such date of severance from employment. For this purpose, compensation paid after severance of employment may only be included in Total Compensation to the extent such amounts would have been included as compensation if they were paid prior to the Employee's severance from employment. The following amounts paid after a Participant's severance of employment are included in Total Compensation:
  - (1) Regular Pay. Compensation for services during the Employee's regular working hours, or compensation for services outside the Employee's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments; and
  - (2) <u>Unused Leave Payments</u>. Payment for unused accrued bona fide sick, vacation, or other leave, but only if the Employee would have been able to use the leave if employment had continued.

All other post-severance payments are not included in the definition of Total Compensation, even if such amounts are paid within the time period described in this subsection (c).

- (d) Total Compensation includes post-severance payments to a Participant who is permanently and totally disabled as defined in IRS Code § 22(e)(3) provided that contributions made with respect to amounts treated as compensation under this subsection (d) are non-forfeitable when made.
- **(e)** <u>Differential Pay</u>. Effective for years beginning on or after January 1, 2009, in the case of an individual who receives Differential Pay from the District:
  - (1) such individual will be treated as an Employee of the Employer making the payment, and
  - (2) the Differential Pay shall be treated as wages and will be included in calculating an Employee's Total Compensation under the Plan.

If all Employees performing service in the Uniformed Services are entitled to receive Differential Pay on reasonably equivalent terms and are eligible to make contributions based on the payments on reasonably equivalent terms, the Plan shall not be treated as failing to meet the requirements of any provision described in Code § 414(u)(1)(C)(Treatment of Certain Contributions Made Pursuant to Veteran's Reemployment Rights) by reason of any contribution or benefit based on Differential Pay.

For purposes of this subsection (e), Differential Pay means any payment which is made by an Employer to an individual while the individual is performing service in the Uniformed Services while on active duty for a period of more than 30 days, and represents all or a portion of the wages the individual would have received from the Employer if the individual were performing services for the Employer. In applying the provisions of this subsection (e), Uniformed Services are services as described in Code § 3401(h)(2)(A). The Trust is the separate funding vehicle under the Plan.

**1.56 Trust.** The Ttrust created underestablished pursuant to the Plan Trust Agreement between the District

- and the Trustee to hold, administer and distribute all or a portion of the Plan's assets for the exclusive benefit of Plan Participants, Beneficiaries and Alternate Payees.
- **1.57 Trust Agreement.** The separate agreement entered into by and between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets.
- 1.561.58 Trustee. The District or person, committee(s) or entity that, and any successors thereto, who is appointed by the District to serve as Trusteehold, administer and accepts such position. The District retains the right distribute all or a portion of appointment and removal of Trustee(s) and may exercise that right at any time. Use of the defined term, Trustee refers to a discretionary Trustee, appointed Plan assets held in accordance with Section 12 (the Trust Provisions). However, pursuant to the District may retain the assistance terms and conditions of a nondiscretionary trustee to carry out some or all of the duties of the Trustee or to assist the Trustee in the performance of its powers, duties and responsibilities Trust Agreement.
- **1.57**<u>1.59</u> **Valuation Date**. The date or dates upon which Plan assets are valued. Plan assets are typically valued daily and will be valued at least annually as of the last day of each Plan Year.

#### **SECTION 2**

## **ELIGIBILITY AND PARTICIPATION**

- **Eligibility**. In order to participate in the Plan, an Employee must be an Eligible Employee (as defined in Section 2.2) and must satisfy the eligibility requirements specified in Section 2.3. Once an Eligible Employee satisfies the Plan's eligibility requirements, such Eligible Employee shall become a Plan Participant as of the Entry Date specified in Section 2.3(b). An Employee who meets the eligibility requirements set forth in Section 2.3 but is not an Eligible Employee, will be eligible to participate in the Plan only upon becoming an Eligible Employee. For purposes of determining eligibility to make Salary Deferrals, an Eligible Employee will be deemed to commence participation on a timely basis if he or she is permitted to commence making Salary Deferrals as soon as administratively feasible after satisfying the eligibility conditions under the Plan.
- **Eligible Employees.** Except as specified in subsection 2.2(a) and (b), all Employees of the District are Eligible Employees.
  - **Excluded Employees.** An Employee is not an Eligible Employee and may not participate under the Plan if he or she is classified as a:
    - (1) Rehired Annuitant.
    - (2) Nonresident Alien Employee.
    - (3) Leased Employee.
    - (4) Seasonal Employee working less than 5 months in a Plan Year.
    - (5) Member of the Carpenter's Union Trust.
    - [6] Intern working less than 20 hours a week and less than 5 months in a Plan Year.

- (b) Only Employees May Participate in the Plan. To participate in the Plan, an individual must be an Employee. If an individual is not an Employee (e.g., the individual performs services with the District as an independent contractor) such individual may not participate under the Plan. If an individual who is classified as a non-Employee is later determined by the District or by a court or other government agency to be an Employee of the District, the reclassification of such individual as an Employee will not create retroactive rights to participate in the Plan. For periods prior to the date of such final determination, the reclassified Employee will not have any rights to accrued benefits under the Plan, except as agreed to by the District or mandated by a court or government agency, or as set forth in an amendment adopted by the District.
- (c) Ineligible Employee Becomes Eligible Employee. If an Employee changes status from an ineligible Employee to an Eligible Employee, such Employee will become a Participant immediately on the date he/she changes status to an Eligible Employee, provided the Employee has satisfied the eligibility requirements in Section 2.3 of the Plan and has passed his or her Entry Date that would otherwise have applied had the Employee been an Eligible Employee. If the Employee's original Entry Date (determined as if the Employee was always an Eligible Employee) has not passed as of the date the Employee becomes an Eligible Employee, the Employee will become a Participant upon the occurrence of such Entry Date. This requirement is deemed satisfied with respect to Salary Deferrals if the Employee is permitted to commence making Salary Deferrals under the Plan as soon as administratively feasible after the Employee becomes an Eligible Employee. If an ineligible Employee has not satisfied the eligibility requirement(s) in Section 2.3 at the time such Employee becomes an Eligible Employee, such Employee will become a Participant on the appropriate Entry Date following satisfaction of the eligibility requirements in Section 2.3 of the Plan.
- **Eligible Employee Becomes Ineligible Employee.** If an Employee ceases to qualify as an Eligible Employee (i.e., the Employee changes status from an eligible class to an ineligible class of Employees), such Employee will immediately cease to participate in the Plan. If such Employee should subsequently become an Eligible Employee, he/she will be able to participate in the Plan in accordance with subsection (b) above.
- **Improper Exclusion of Eligible Participant.** If the Plan improperly excludes a Participant who has satisfied the requirements under this Section 2 for participating under the Plan, the District may take **reasonable** action to correct such error, provided such corrective action is consistent with the requirements of the Internal Revenue Service's Employee Plans Compliance Resolution System (EPCRS) program, as amended, or other successor program.

# 2.3 <u>Eligibility Requirements</u>.

- (a) <u>Service Requirement</u>. There is no minimum age requirement to become an Eligible Employee. An Eligible Employee must complete 1,044 continuous Hours of Service hours to participate in the Plan.
- **(b)** Entry Date: An Eligible Employee will enter the Plan and begin Salary Deferral contributions and receive Matching Contributions as of the later of commencement of the first full payroll period

following Eligible Employee's satisfaction of the eligibility requirements specified in Section 2.3(a) or submitting a completed deferral election form in the manner prescribed by the Plan Administrator.

- **Participation on Effective Date of Plan.** An Eligible Employee who has satisfied the Plan's eligibility requirements and reached his/her Entry Date described in Section 2.3 as of the Effective Date of the Plan will be eligible to participate in the Plan as of such Effective Date. If an Eligible Employee has satisfied the minimum eligibility requirements as of the Effective Date in Section 2.3 of the Plan but has not yet reached his/her Entry Date, the Eligible Employee will participate upon reaching his or her Entry Date.
- **Rehired Employees**. If a terminated Employee is subsequently rehired, such Employee will be eligible to participate in the Plan on his/her reemployment date, if the Employee is rehired within a 12-month period and is an Eligible Employee and the Employee had satisfied the Plan's eligibility requirement(s) prior to his/her termination of employment. If a rehired Employee had not satisfied the Plan's eligibility requirement(s) prior to termination of employment, or was rehired more than 12 months after termination, such Employee is eligible to participate in the Plan on the appropriate Entry Date following satisfaction of the eligibility requirements under Section 2.1.

### **SECTION 3**

## **PLAN CONTRIBUTIONS**

- **Types of Contributions.** The Plan provides for Salary Deferrals and Matching Contributions.
- 3.2 Salary Deferrals. Participants may make Salary Deferrals under the Plan. A Participant may elect to change, discontinue and/ or resume Salary Deferrals in accordance with Section 3.2(b). Subject to the conditions and limitations of the Plan, a Participant's election shall remain in effect until any change, suspension, or termination properly elected by the Participant under Section 3.2(b) becomes effective. The amounts to be deferred shall be withheld from the Participant's Plan Compensation and contributed to the Plan on the Participant's behalf. The District will deposit any amounts withheld from a Participant's Plan Compensation as Salary Deferrals into the Participant's Salary Deferral Account under the Plan. Effective January 1, 2024, a Participant may, at the time of his or her Salary Deferral Election or election change, irrevocably designate all or a portion of his or her future Salary Deferrals, including Catch-Up Contributions, as Roth Deferrals instead of Pre-Tax Deferrals.
  - (a) Salary Deferral Election. In order to make Salary Deferrals under the Plan, a Participant must enterinto a Salary Deferral Election which authorizes the District to withhold a specific dollar amount or a specific percentage from the Participant's Plan Compensation. A Participant's election to make a Salary Deferral to the Plan must be made in accordance with the procedures and time requirements established by the Plan Administrator. A Participant's election must irrevocably designate the Salary Deferrals as Pre-Tax Deferrals, Roth Deferrals, or a specific combination of both. A Participant may defer any amount up to one-hundred percent (100%) of Plan Compensation, provided however, the amount deferred never exceeds the Elective Deferral Dollar Limit imposed by Code § 402(g) and the Limit on Annual Additions under Code § 415. In

determining the amount to be withheld from a Participant's Plan Compensation, a Salary Deferral Election may be rounded to the next highest or lowest whole dollar amount. A Salary Deferral Election may only relate to Plan Compensation that is not currently payable or made available at the time the Salary Deferral Election is completed. Further, a Salary Deferral Election may not be effective prior to the later of the date the:

- (1) Employee becomes a Participant, or
- (2) Participant executes the Salary Deferral Election.

In addition, Salary Deferrals made pursuant to a Salary Deferral Election may not be made earlier than the date the Participant performs the services to which such Salary Deferrals relate or the date the compensation subject to such Salary Deferral Election would be currently available to the Participant absent the deferral election (if earlier).

- **Change in Deferral Election.** Participants may enter into new Salary Deferral Elections, modify or terminate existing Salary Deferral Elections within the limits established by the Plan Administrator. Any election to modify, including changing the designation of all or any portion of his or her Salary Deferrals as Pre-Tax Deferrals or Roth Deferrals, or terminate a Salary Deferral Election will take effect the first pay date of the following pay period which does not include the date of the election or modification (For example, for an election made on May 14, 2018, the salary deferrals will commence on the next pay date of June 13, 2018, which is for the following pay period of May 20 June 2).
- (c) <u>Catch-Up Contributions</u>. A Participant who is aged 50 or over before the end of a taxable year may make Catch-Up Contributions, provided such Catch-Up Contributions are in excess of an otherwise applicable limit under the Plan. For this purpose, an otherwise applicable Plan limit is a limit in the Plan that applies to Salary Deferrals without regard to Catch-up Contributions, such as the Code § 415 Limitation, or the Elective Deferral Dollar Limit under Code § 402(g). <u>As with other Salary Deferrals</u>, a Participant may, at the time of the deferral election, designate Catch-Up Contributions as either Pre-Tax Deferrals or Roth Deferrals.
  - (1) <u>Catch-Up Contribution Limit</u>. Catch-up Contributions for a Participant for a taxable year may not exceed the Catch-Up Contribution Limit. The Catch-Up Contribution Limit for the 2018 tax year is \$6,000. For taxable years beginning after 2018, the Catch-Up Contribution Limit will be adjusted for cost-of-living increases under Code § 414(v)(2)(C).
  - **Special Treatment of Catch-Up Contributions.** Catch-up Contributions are not subject to the Elective Deferral Dollar Limit or the Code § 415 Limitation.
- 3.3 <u>Matching Contributions</u>. The District shall have the responsibility for making Matching Contributions under the Plan. On behalf of each Participant, the District shall make a Matching Contribution equal to one-hundred percent (100%) of Participant's contributions to the Plan. The Matching Contributions shall be capped at four and one-half percent (4.5%) of the Participant's Plan Compensation not to exceed the participant's total deferred compensation. Any Matching Contribution made under the Plan will be allocated to Participants' Matching Contribution Account.

- **Employee Matching Formula.** The Matching Contribution formula shall apply to Salary Deferrals made by Participants to the Plan.
- **Matching Contributions.** All Matching Contributions are one-hundred percent (100%) vested immediately.
- Rollover Contributions. An Employee (or former Employee) may make a Rollover Contribution to this Plan from a qualified retirement plan or from an IRA in accordance with administrative procedures established by the Plan Administrator for the acceptance of Rollover Contributions, including Rollover Contributions of Roth amounts, but only if (i) it is a direct rollover from another Roth Deferral Account in an eligible retirement plan described in Code § 402A(e)(1), (ii) it is permitted under Code § 402(c), and (iii) the Plan Administrator of the transferor plan provides the Plan Administrator with information about the rollover required under the Treasury Regulations. A rollover of Roth Deferrals may not be made to this Plan from a Roth IRA. Any Roth Deferrals rolled into this Plan will be held in a separate Roth Rollover Account. A Participant may withdraw amounts from his/her Rollover Contribution Account in accordance with the distribution rules under Section 7 (Plan Distributions). Effective August 1, 2018, Participant withdrawals from his or her Rollover Contribution Account is no longer limited to once per Plan Year. Any amounts received as a Rollover Contribution under this Section 3.4 will not be treated as an Annual Addition for purposes of applying the Code § 415 Limitation.

For purposes of this Section 3.4, a qualified retirement plan is a tax-qualified retirement plan described in Code § 401(a) or Code § 403(a), an annuity contract described in Code § 403(b), or an eligible plan under Code § 457(b), which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state. To qualify as a Rollover Contribution under this Section 3.4, the Rollover Contribution must be transferred directly from the qualified retirement plan or IRA in a Direct Rollover or must be transferred to the Plan by the Employee within sixty (60) days following receipt of the amounts from the qualified plan or IRA.

An Employee may make a Rollover Contribution to the Plan if the Employee is not a Participant with respect to any or all other contributions under the Plan. An Employee who makes a Rollover Contribution to this Plan prior to becoming an Eligible Employee shall be treated as a Participant only with respect to such Rollover Contribution Account, but shall not be treated as a Participant with respect to other contribution sources under the Plan until he/she otherwise satisfies the eligibility conditions under the Plan. To the extent Participant loans are authorized under the Plan, a "limited Participant" under this paragraph may request a Participant loan from the Rollover Contribution Account. A former Employee, however, may only make a Rollover Contribution to the Plan if he or she is a Participant with respect to other contributions under the Plan.

The Plan Administrator may refuse to accept a Rollover Contribution if the Plan Administrator reasonably believes the Rollover Contribution:

- (a) is not being made from a proper plan or IRA;
- (b) is not being made within sixty (60) days from receipt of the amounts from a qualified retirement plan or IRA;

- **(c)** could jeopardize the tax-exempt status of the Plan; or
- **(d)** could create adverse tax consequences for the Plan or the District.

Prior to accepting a Rollover Contribution, the Plan Administrator may require the Employee to provide satisfactory evidence establishing that the Rollover Contribution meets the requirements of this Section 3.4.

The Plan Administrator may apply different conditions for accepting Rollover Contributions from qualified retirement plans and IRAs. For example, the Plan Administrator may decide in its discretion whether to accept a Direct Rollover of a loan note from another qualified plan. Any conditions on Rollover Contributions must be applied uniformly to all Employees under the Plan.

- 3.5 Roth Deferrals. Effective January 1, 2024, a Participant may, at the time of his or her Salary Deferral Election or election change, irrevocably designate all or a portion of his or her future Salary Deferrals, including Catch-Up Contributions, as Roth Deferrals instead of Pre-Tax Deferrals. A Participant's Roth Deferrals will be allocated to and maintained in a separate Account as provided in this Section 3.5.
  - (a) Contributions and withdrawals of Roth Deferrals will be credited and debited to a separate Account maintained for each Participant, and the Plan will maintain a record of the amount of Roth Deferrals in such Account. In addition, gains, losses and other credits or charges must be separately allocated on a reasonable and consistent basis to each Participant's Roth Deferral Account under the Plan.

    However, in no event may Plan forfeitures be allocated to such Account. No contributions other than Roth Deferrals or Rollover Contributions to a Roth Deferral Account will be credited to a Participant's Roth Deferral Account.
  - (b) To the extent required by the Code and the Treasury regulations, reemployed veterans may designate the year for which a Roth Deferral is made. If no such designation is made, the Roth Deferrals will be treated as having been made in the first year of military service for which the veteran could have made the contributions but not earlier than February XX, 2024. This designation must be made solely for purposes of determining the five-taxable-year period under Code § 402A(d)(2).
  - In the absence of a specific designation of Salary Deferrals as Roth Deferrals, Salary Deferrals must be treated as Pre-Tax Deferrals. Unless specifically provided otherwise in the Plan, Roth Deferrals will be treated as Salary Deferrals for all purposes under the Plan, including but not limited to immediate full vesting, the distribution restrictions in Section 7 (Plan Distributions), the Elective Deferral Dollar Limit imposed by Code § 402(g) and the Limit on Annual Additions under Code § 415.
- 3.6 In-Plan Roth Conversions. Effective January 1, 2024, a Participant may elect to transfer some or all of his or her vested Account Balance (other than his or her Roth Deferral Account or Roth Rollover Account), and earnings thereon, to an In-Plan Roth Conversion Account without regard to whether the Participant satisfies the requirements for distribution in accordance with Section 7 (Plan Distributions). Any Participant election to make an In-Plan Roth Conversion during a taxable year is irrevocable and cannot be changed after the In-Plan Roth Conversion is completed. Generally, In-Plan Roth Conversion Accounts are subject to the same Plan provisions as Roth Deferral Accounts. However, distribution restrictions that

otherwise apply with respect to a specific contribution source will continue to apply if such contribution source is converted to Roth Deferrals. For example, if Matching Contributions are converted to Roth Deferrals, such amounts may not be distributed from the Plan upon events not otherwise available for Matching Contributions. The Plan Administrator will maintain such records as are necessary for the proper reporting of In-Plan Roth Conversions and will administer the In-Plan Roth Conversion Accounts in accordance with the Code, Treasury Regulations, and Plan provisions.

- (a) Eligibility to Elect In-Plan Roth Rollover. For purposes of determining eligibility for In-Plan Roth Rollovers, the Plan will treat a Participant's spousal Beneficiary or Participant's alternate payee spouse or former spouse as a Participant. A non-spouse Beneficiary is not eligible to make an In-Plan Roth Rollover.
- (b) In-Plan Roth Rollover Procedure. An In-Plan Roth Rollover must be made by the Participant in the form of a Direct In-Plan Roth Rollover. Additionally, an In-Plan Roth Rollover must be accomplished in the form and manner established by the Plan Administrator.
- (c) Plan Loans. In-Plan Roth Rollovers may include a Plan loan only if there is no change in the loan's repayment schedule and the loan is not treated as a new loan. Further, Plan loans that include In-Plan Roth Rollover amounts are subject to the Plan's loan rules and the Plan's loan policy.
- In-Plan Roth Rollover is not a rollover contribution for purposes of the Plan. The Plan will take into account the amounts attributable to an In-Plan Roth Rollovers for purposes of determining whether a Participant's vested Account Balance exceeds \$500 for purposes of the Plan's involuntary cashout rules. Amounts in the Participant's In-Plan Roth Rollover Account may only be withdrawn by a Participant when the Participant is eligible for a distribution under Section 7 (Plan Distributions).
- (e) Spousal Consent. A married Plan Participant is not required to obtain spousal consent in connection with an election to make an In-Plan Roth Conversion, even if the Plan would otherwise require the spousal consent for distributions.
- Tax Treatment of In-Plan Roth Conversion. A Participant must include in gross income the taxable amount of an In-Plan Roth Conversion. For this purpose, the taxable amount of an In-Plan Roth Conversion is the fair market value of the distribution reduced by any basis in the converted amounts. If an outstanding loan is rolled over as part of an In-Plan Roth Conversion, the amount includible in gross income includes the balance of the loan. Generally, the taxable amount of an In-Plan Roth Conversion is includible in gross income in the taxable year in which the conversion occurs. However, In-Plan Roth Conversions are not subject to the twenty percent (20%) mandatory withholding under Code § 3405(c). Thus, Participants will be required to make arrangements independent of the Plan to ensure the total taxable amount is timely paid.
- **Early Distribution Penalty.** If an amount allocable to the taxable amount of an In-Plan Roth is subsequently distributed within the 5-taxable-year period beginning with the first day of the Participant's taxable year in which the conversion was made, the amount distributed is treated as includible in gross income for purposes of applying the Code § 72(t) early distribution penalty. For

this purpose, the 5--taxable-year period ends on the last day of the Participant's fifth taxable year in the period. This subsection (g) will not apply to the extent the distribution is rolled over to a Roth account in another qualified plan or is rolled over to a Roth IRA. However, the rule under this subsection (g) will apply to any subsequent distributions made from such other Roth account or Roth IRA within the 5-taxable-year period.

# (h) In-Plan Roth Conversion Definitions.

- "Direct In-Plan Roth Rollover" means a direct rollover from a Participant Account (other than Participant's Roth Deferral Account, Roth Rollover Account and In-Plan Roth Rollover Account) to an In-Plan Roth Rollover Account established for the Participant.
- "In-Plan Roth Rollover" means a contribution made from the Participant's Account (other than Participant's Roth Deferral Account or Roth Rollover Account) to the Participant's In-Plan Roth Rollover Account under the Plan in accordance with Code § 402A(c)(4) with respect to distributable amounts under the Plan or § 402A(c)(4)(E) with respect to amounts not otherwise distributable under the Plan.
- (3) "In-Plan Roth Rollover Account" means the subaccount established under the Plan to account for a Participant's In-Plan Roth Rollover.

### **SECTION 4**

#### SPECIAL RULES AFFECTING GOVERNMENTAL PLANS

- **4.1 Governmental Plan.** This Plan is a tax-qualified defined contribution plan under Code § 401(a). In addition, this Plan meets the requirements for establishing and maintaining a governmental plan under Code § 414(d).
  - **Governmental Plan Exemptions.** As a tax-qualified governmental plan, this Plan is exempt from Title I of ERISA and certain qualification rules under Code 401(a), including:
    - (1) The minimum age and service rules under Code § 410(a) and the minimum coverage rules under Code § 410(b).
    - (2) The minimum vesting requirements of Code § 411, including minimum vesting schedules, consent requirements for plan distributions, and the anti-cutback rule under Code § 411(d)(6).
    - (3) The nondiscrimination requirements under Code §§ 401(a)(4), 401(k) and 401(m).
    - (4) The top-heavy rules under Code § 416.
    - (5) The joint and survivor annuity rules under Code §§ 401(a)(11) and 417.
    - The requirements for protecting benefits pursuant to a plan merger or a transfer of plan assets and liabilities, as prescribed by Code § 401(a)(12).
    - (7) The anti-assignment rule under Code § 401(a)(13). However, the Code § 414(p)(11) rules relating to the taxability of benefits distributed pursuant to a QDRO are applicable to

- benefits payable to an alternate payee under the QDRO.
- (8) The commencement of benefit requirements under Code § 401(a)(14).
- (9) The protections under Code  $\S 401(a)(19)$ .
- **ERISA Exempt.** Adoption by the District of provisions similar to the requirements applicable to plans covered under Title I of ERISA or to otherwise inapplicable Code requirements will not affect the Plan's status as a governmental plan. In the event such provisions are adopted, the Plan shall remain exempt from ERISA and certain Code requirements as a governmental plan.

#### **SECTION 5**

#### CONTRIBUTION AND BENEFIT LIMITATIONS

## 5.1 <u>Code § 415 Limitation.</u>

- (a) No Other Plan Participation. If the Participant does not participate in, and has never participated in another qualified retirement plan, a welfare benefit fund (as defined under Code § 419(e)), an individual medical account (as defined under Code § 415(l)(2)), or a SEP (as defined under Code § 408(k)) maintained by the District which provides an Annual Addition as defined in subsection (c)(1), then the amount of Annual Additions which may be credited to the Participant's Account for any Limitation Year will not exceed the lesser of the Maximum Permissible Amount or any other limitation contained in this Plan.
  - If an Employer Contribution that would otherwise be contributed or allocated to a Participant's Account will cause that Participant's Annual Additions for the Limitation Year to exceed the Maximum Permissible Amount, the amount to be contributed or allocated to such Participant will be reduced so that the Annual Additions allocated to such Participant's Account for the Limitation Year will equal the Maximum Permissible Amount. However, if a contribution or allocation is made to a Participant's Account in an amount that exceeds the Maximum Permissible Amount, such excess Annual Additions may be corrected pursuant to the correction procedures outlined under the IRS' Employee Plans Compliance Resolution System (EPCRS).
- **Participation in Another Plan.** This subsection 5.1(b) applies if, in addition to this Plan, the Participant receives an Annual Addition during any Limitation Year from another Defined Contribution Plan, a welfare benefit fund (as defined under Code § 419(e)), an individual medical account (as defined under Code § 415(l)(2)), or a SEP (as defined under Code § 408(k)) maintained by the District.
  - This Plan's Code § 415 Limitation. The Annual Additions that may be credited to a Participant's Account under this Plan for any Limitation Year will not exceed the Maximum Permissible Amount (defined in subsection 5.1(c)(6) below) reduced by the Annual Additions credited to a Participant's Account under any other Defined Contribution Plan, welfare benefit fund, individual medical account, or SEP maintained by the District for the same Limitation Year.

- Annual Additions Reduction. If the Annual Additions with respect to the Participant under any other Defined Contribution Plan, welfare benefit fund, individual medical account, or SEP maintained by the District are less than the Maximum Permissible Amount and the Annual Additions that would otherwise be contributed or allocated to the Participant's Account under this Plan would exceed the Code § 415 Limitation for the Limitation Year, the amount contributed or allocated will be reduced so that the Annual Additions under all such Plans and funds for the Limitation Year will equal the Maximum Permissible Amount. However, if a contribution or allocation is made to a Participant's Account in an amount that exceeds the Maximum Permissible Amount, such excess Annual Additions may be corrected pursuant to the correction procedures outlined under the IRS' Employee Plans Compliance Resolution System (EPCRS).
- (3) No Annual Additions Permitted. If the Annual Additions with respect to the Participant under such other Defined Contribution Plan(s), welfare benefit fund(s), individual medical account(s), or SEP(s) in the aggregate are equal to or greater than the Maximum Permissible Amount, no amount will be contributed or allocated to the Participant's Account under this Plan for the Limitation Year. However, if a contribution or allocation is made to a Participant's Account in an amount that exceeds the Maximum Permissible Amount, such excess Annual Additions may be corrected pursuant to the correction procedures outlined under the IRS' Employee Plans Compliance Resolution System (EPCRS).

## (c) § 415 Definitions.

- (1) <u>Annual Additions</u>. The amounts credited to a Participant's Account for the Limitation Year that are taken into account in applying the Code § 415 Limitation, including:
  - i Employer Contributions, including Matching Contributions and Salary Deferrals; and
  - ii Forfeitures.

An Annual Addition is credited to a Participant's Account for a particular Limitation Year if such amount is allocated to the Participant's Account as of any date within that Limitation Year. An Annual Addition *will not* be deemed credited to a Participant's Account for a particular Limitation Year unless such amount is actually contributed to the Plan no later than 30 days after the time prescribed by law for filing the District's income tax return (including extensions) for the taxable year with or within which the Limitation Year ends. In the case of After-Tax Employee Contributions, such amount shall not be deemed credited to a Participant's Account for a particular Limitation Year unless the contributions are actually contributed to the Plan no later than 30 days after the close of that Limitation Year.

- (2) <u>Defined Contribution Dollar Limitation</u>. \$40,000, as adjusted under Code § 415(d).
- (3) <u>Excess Amount</u>. The excess of the Participant's Annual Additions for the Limitation Year over the Maximum Permissible Amount.
- (4) <u>Limitation Year</u>. The Plan Year is the Limitation Year.

- (5) <u>Maximum Permissible Amount</u>. For Limitation Years beginning on or after January 1, 2002, the Maximum Permissible Amount is the lesser of:
  - i. the Defined Contribution Dollar Limitation, or
  - ii. 100 percent of the Participant's Total Compensation for the Limitation Year.

The Total Compensation limitation referred to in (ii) shall not apply to any contribution for medical benefits (within the meaning of Code  $\S$  401(h) or  $\S$  419A(f)(2)) which is otherwise treated as an Annual Addition.

If a short Limitation Year is created because of an amendment changing the Limitation Year to a different 12-consecutive month period, the Maximum Permissible Amount *will not* exceed the Defined Contribution Dollar Limitation multiplied by the following fraction:

## Number of months in the short Limitation Year 12

If a short Limitation Year is created because the Plan has an *initial* Plan Year that is less than 12 months, no proration of the Defined Contribution Dollar Limitation is required.

- **Total Compensation.** The amount of compensation as defined under Section 1.54, subject to the following:
  - i. <u>Total Compensation Actually Paid or Made Available</u>. For purposes of applying the limitations of this Section 5.1, Total Compensation for a Limitation Year is the Total Compensation actually paid or made available to an Employee during such Limitation Year. The Employer may include in Total Compensation for a Limitation Year amounts earned but not paid in the Limitation Year because of the timing of pay periods and pay days, but only if:
    - (A) the amounts are paid during the first few weeks of the next Limitation Year,
    - **(B)** such amounts are included on a uniform and consistent basis with respect to all similarly-situated employees, and
    - (C) no amounts are included in Total Compensation in more than one Limitation Year.
- (d) Restorative Payments. Restorative payments are not considered Annual Additions for any Limitation Year. For this purpose, restorative payments are payments made to restore losses to the Plan resulting from actions (or a failure to act) by a fiduciary or third-party administrator for which there is a reasonable risk of liability under applicable federal or state law, where Participants who are similarly situated are treated similarly with respect to the payments.
- **Corrective Provisions.** The District may use reasonable correction methods to the extent permitted by the IRS or under an IRS correction program.
- **Change of Limitation Year.** Where there is a change of Limitation Year, a "short" Limitation Year exists for the period beginning with the first day of the Limitation Year and ending on the day

before the change in Limitation Year is effective. For this purpose, if the Plan is terminated effective as of a date other than the last day of the Limitation Year, the Plan is treated as if it were amended to change its Limitation Year.

**Elective Deferral Dollar Limit.** The Elective Deferral Dollar Limit under this Section 5.2 applies with respect to a Participant's Salary Deferrals. Under this Elective Deferral Dollar Limit, a Participant may not make Elective Deferrals under this Plan during any calendar year in an amount that exceeds the Elective Deferral Dollar Limit in effect for each Plan Year.

The Elective Deferral Dollar Limit is \$18,500 for the 2018. For taxable years beginning after 2018, the Elective Deferral Dollar Limit will be adjusted for cost-of-living increases under Code § 402(g)(4).

If a Participant is aged 50 by the end of the taxable year, the Elective Deferral Dollar Limit is increased by the Catch-Up Contribution Limit described in Section 3.2(c).

- **Excess Deferrals.** Excess Deferrals are Elective Deferrals made during the Participant's taxable year that exceed the Elective Deferral Dollar Limit for such year.
- (b) Correction of Excess Deferrals. If a Participant makes Elective Deferrals in excess of the Elective Deferral Dollar Limit ("Excess Deferrals") under this Plan, such Excess Deferrals (plus allocable income or loss) shall be distributed to the Participant. A distribution of Excess Deferrals may be made at any time (subject to the correction provisions under EPCRS or subsequent guidance). If the corrective distribution of Excess Deferrals is made by April 15 of the calendar year following the year the Excess Deferrals are made to the Plan, such amounts will be taxable in the year of deferral, but not in the year of distribution. If a corrective distribution of Excess Deferrals is made after April 15 of the following calendar year, such amounts will be taxable in both the year of deferral and the year of distribution. Unless the Plan Administrator determines otherwise, if the Participant made both Pre-Tax Deferrals and Roth Deferrals during the same Plan Year, the distribution of Excess Deferrals will be made first solely from the Participant's Pre-Tax Salary Deferral Account until all Pre-Tax Deferrals, plus any allocable gains and minus any allocable losses, for the Plan Year, have been distributed, and only then from the Participant's Roth Deferral Account.
  - (1) Amount of Corrective Distribution. The amount to be distributed from this Plan as a correction of Excess Deferrals equals the amount of Elective Deferrals the Participant contributes during the taxable year to this Plan and any other plan maintained by the District in excess of the Elective Deferral Dollar Limit, reduced by any corrective distribution of Excess Deferrals the Participant receives during the calendar year from this Plan or other plan(s) maintained by the District.
  - Allocable Gain or Loss. A corrective distribution of Excess Deferrals must include any allocable gain or loss for the taxable year in which the Excess Deferrals are contributed to the Plan. The gain or loss allocable to Excess Deferrals may be determined in any reasonable manner, provided the manner used to determine allocable gain or loss is applied consistently for all Participants and in a manner that is reasonably reflective of the method used by the Plan for allocating income to Participants' Accounts. A corrective distribution of

Excess Deferrals will not include any income or loss allocable to the period between the end of the taxable year and the date of distribution.

- Taxation of Corrective Distribution. If a corrective distribution of Excess Deferrals is made by April 15 of the following calendar year, amounts attributable to the Excess Deferrals will be includible in the Participant's gross income in the taxable year in which such amounts are deferred under the Plan and amounts attributable to income or loss on the Excess Deferrals will be includible in gross income in the year of distribution.
  - If a corrective distribution of Excess Deferrals is made after April 15, the amount of the corrective distribution attributable to Excess Deferrals will be includible in the Participant's gross income in both the taxable year in which such amounts are deferred under the Plan and the taxable year in which such amounts are distributed.
- (4) <u>Coordination with Other Provisions</u>. A corrective distribution of Excess Deferrals made by April 15 of the following calendar year may be made without consent of the Participant or the Participant's Spouse, and without regard to any distribution restrictions applicable under Section 7 (Plan Distributions). A corrective distribution of Excess Deferrals made by the appropriate April 15 also is not treated as a distribution for purposes of applying the required minimum distribution rules under Section 8 (Required Minimum Distributions).
- Correction of Excess Deferrals Under Plans Not Maintained by the Employer. The correction provisions under this subsection 5.2(b) apply only if a Participant makes Excess Deferrals under this Plan (or under this Plan and other plans maintained by the District). However, if a Participant has Excess Deferrals for a calendar year on account of making Elective Deferrals to a plan of an unrelated employer, the Participant may assign to this Plan any portion of his/her Elective Deferrals made under all plans during the calendar year to the extent such Elective Deferrals exceed the Elective Deferral Dollar Limit. The Participant must notify the Plan Administrator in writing on or before March 1 of the following calendar year of the amount of the Excess Deferrals to be assigned to this Plan.

Upon receipt of a timely notification, the Excess Deferrals assigned to this Plan will be distributed (along with any allocable income or loss) to the Participant in accordance with the corrective distribution provisions under this subsection 5.2(b). A Participant is deemed to notify the Plan Administrator of Excess Deferrals to the extent such Excess Deferrals arise only under this Plan and any other plan maintained by the District.

## **SECTION 6**

### PARTICIPANT VESTING AND FORFEITURES

- **6.1 Vesting of Contributions.** Participants are one-hundred percent (100%) vested in his or her Account at all times.
- **Forfeiture of Benefits.** Notwithstanding immediate and full vesting, upon the occurrence of any of the events described below or at any such time as the Plan Administrator determines, a Participant Account

may be forfeited. The Plan Administrator has the responsibility to determine the amount of a Participant's forfeiture. Until an amount is forfeited pursuant to this Section 6.2 (Forfeiture of Benefits), a Participant's entire Account must remain in the Plan and continue to share in gains and losses of the Trust.

- (a) Missing Participant or Beneficiary. If a Participant or Beneficiary cannot be located within a reasonable period following a reasonable diligent search, the missing Participant's or Beneficiary's Account may be forfeited, as provided in subsection (2) below. The District will be deemed to have performed a reasonable diligent search if it performs the actions described in subsection (1) below. In determining whether a reasonable period has elapsed following a reasonable diligent search, the District or Plan Administrator may follow any applicable guidance provided under statute, regulation, or other IRS or DOL guidance of general applicability. However, the District or Plan Administrator will be deemed to have waited a reasonable period following a reasonable diligent search if the District or Plan Administrator waits at least 6 months following the completion of the actions described in subsection (1) below.
  - **Reasonable Diligent Search.** The District or Plan Administrator will be deemed to have performed a reasonable diligent search if it performs the following actions:
    - i. Send a certified letter to the Participant's or Beneficiary's last known address.
    - ii. Check related records of the District (e.g., health plan records) to determine if a more current address exists for the Participant or Beneficiary.
    - iii. If the Participant cannot be located, the District or Plan Administrator may attempts to identify and contact any individual that the Participant has designated as a Beneficiary under the Plan for –updated information concerning the location of the missing Participant.
    - iv. Utilize the Social Security Administration (SSA) letter-forwarding service for locating lost participants. Additional information regarding the SSA letter forwarding program can be located at www.ssa.gov.In addition to the search methods discussed above, the District or Plan Administrator may use other search methods, including the use of Internet search tools, commercial locator services, and credit reporting agencies to locate the missing Participant.
  - (2) Forfeiture of Account of Missing Participant or Beneficiary. A Participant or Beneficiary deemed to be missing (i.e., cannot be found within 4 years of his or her required beginning date), the Plan administrator may forfeit the distributable amount of such missing Participant's or Beneficiary's vested Account. If, after an amount is forfeited under this subsection (2), the missing Participant or Beneficiary is located, the Plan will restore the forfeited amount (unadjusted for gains or losses) to such Participant or Beneficiary within a reasonable time. However, if a missing Participant or Beneficiary has not been located by the time the Plan terminates, the forfeiture of such Participant's or Beneficiary's distributable amount will be irrevocable. Notwithstanding the foregoing, the Plan Administrator may address forfeitures using any available means provided under

- applicable law, including escheating the Participant's or Beneficiary's Account to the state.
- (3) Expenses Attributable to Search for Missing Participant. Reasonable expenses attendant to locating a missing Participant may be charged to such Participant's Account, provided that the amount of such expenses is reasonable. The Plan Administrator may take into account the size of a Participant's Account in relation to the cost of the search when deciding how extensive a search is required before declaring such Participant as missing under subsection (a).
- **Employer Matching Contribution Attributable to Excess Deferrals.** If a Participant receives a distribution of Excess Deferrals, the portion of his/her Matching Contribution Account which is attributable to such distributed amounts will be forfeited. A forfeiture of Matching Contributions under this subsection 6.2(b) occurs in the Plan Year in which the Participant receives the distribution of Excess Deferrals.
- **Allocation of Forfeitures**. Forfeitures may be used in the Plan Year in which the forfeitures occur or in the Plan Year following the Plan Year in which the forfeitures occur. In applying the forfeiture provisions under the Plan, if there are any unused forfeitures as of the end of the Plan Year, any remaining forfeiture will be used in the immediately following Plan Year.
  - Reduction of Contributions. The District may, in its discretion, use such forfeitures to reduce Matching Contributions. The District may adjust its contribution deposits in any manner, provided the total Matching Contributions made for the Plan Year properly take into account the forfeitures that are to be used to reduce such contributions from that Plan Year. If contributions are allocated over multiple allocation periods, the District may reduce its contribution for any allocation periods within the Plan Year in which the forfeitures are to be allocated so that the total amount allocated for the Plan Year is proper. If the Plan provides for Matching Contribution and the District elects not to make a Matching Contribution for the Plan Year, any forfeitures will be allocated to eligible Participants as an additional Matching Contribution.
  - (b) Payment of Plan Expenses. The District may use forfeitures to pay Plan expenses for the Plan Year in which the forfeitures would otherwise be applied. If any forfeitures remain after the payment of Plan expenses under this subsection 6.3(b), the remaining forfeitures will be allocated to reduce Matching Contributions to the Plan. This subsection 6.3(b) only applies to the extent Plan expenses are paid by the Plan. Nothing herein affects the ability of the District to pay Plan expenses. In determining the Plan expenses that may be offset by Plan forfeitures, the District may use any reasonable method to determine the Plan expenses attributable to a particular year.

### **SECTION 7**

### PLAN DISTRIBUTIONS

A Participant may receive a distribution of his/her vested Account Balance at the time and in the manner provided under this Section 7. Upon reaching the Required Beginning Date, a Participant must begin receiving distributions under the Plan in accordance with the provisions of Section 8 (Required Minimum Distributions).

- 7.1 Available Forms of Distribution. Upon termination of employment, a Participant under the Plan may take a distribution of his/her entire vested Account Balance in a single lump sum, periodic payments (on a monthly, quarterly, semiannual, or annual basis), or partial distribution. Further, the Plan Administrator may, in its discretion, permit a Participant to take distributions of less than entire vested Account Balance provided all Participants have an opportunity to take such partial distributions upon termination of employment. In addition, the Plan Administrator may, in its discretion, permit Participants to take single lump sum, periodic payments (on a monthly, quarterly, semiannual, or annual basis), or partial distribution to the extent necessary to satisfy the required minimum distribution rules under Code § 401(a)(9). (See Section 8 Required Minimum Distributions).
- 7.2 Amount Eligible for Distribution. For purposes of determining the amount a Participant or Beneficiary may receive as a distribution from the Plan, a Participant's Account Balance is determined as of the Valuation Date immediately preceding the date the Participant or Beneficiary receives his/her distribution from the Plan. For this purpose, the Account Balance must be increased for any contributions allocated to the Participant's Account since the most recent Valuation Date and must be reduced for any distributions made from the Participant's Account since the most recent Valuation Date. A Participant or Beneficiary does not share in any allocation of gains or losses attributable to the period between the most recent Valuation Date and the date of the distribution, unless provided otherwise under uniform funding and valuation procedures established by the Plan Administrator.
- **Participant Consent**. To the extent a Participant's entire vested Account Balance exceeds the \$500 Involuntary Cash-Out threshold per this Section 7.3(a), the Participant must consent to any distribution of such Account Balance prior to his/her Required Beginning Date except as set forth in Section 8 (Required Minimum Distributions)
  - Involuntary Cash-Out Threshold. For purposes of determining whether a distribution is subject to the Participant consent requirements as described in this Section 7.3, the Involuntary Cash-Out threshold is \$500. For purposes of determining whether a Participant's vested Account Balance exceeds the Involuntary Cash-Out threshold, the value of the Participant's vested Account Balance includes Rollover Contributions (and earnings allocable thereto) within the meaning of Code §§ 402(c), 403(a)(4), 403(b)(8), 408(d)(3)(A)(ii), and 457(e)(16) is taken into account.
  - **Participant Notice.** If a distribution is subject to Participant consent, the Participant must consent in writing to the distribution within a reasonable period prior to the Distribution Starting Date. For this purpose, any consent made within the 180-day period ending on the Distribution Starting Date will be deemed to be made within a reasonable period. If the distribution is subject to spousal consent, the Participant's Spouse also must consent to the distribution in accordance with Section 9.2 (Spousal Consent).

Prior to receiving a distribution from the Plan, a Participant must be notified of his/her right to defer any distribution from the Plan. Effective for Plan Years beginning on or after January 1, 2007, the Participant notice must include a description of the consequences of a Participant's decision not to defer the receipt of a distribution. The notice must be provided no less than 30 days and no more than 180 days prior to the Participant's Distribution Starting Date. However, distribution may

commence less than 30 days after the notice is given, if the Participant is clearly informed of his/or her right to take 30 days after receiving the notice to decide whether or not to elect a distribution (and, if applicable, a particular distribution option), and the Participant, after receiving the notice, affirmatively elects to receive the distribution prior to the expiration of the 30-day minimum period. The notice requirements described in this paragraph may be satisfied by providing a summary of the required information, so long as the conditions described in applicable regulations for the provision of such a summary are satisfied, and the full notice is also provided (without regard to the 180-day period described in this subsection (b)).

- **Special Rules.** The consent rules under this Section 7.3 apply to distributions made after the Participant's termination of employment and to distributions made prior to the Participant's termination of employment. However, the consent of the Participant shall not be required to the extent that a distribution is required to satisfy the required minimum distribution rules under Section 8 (Required Minimum Distributions) or to satisfy the requirements of Code § 415, as described in Section 5.2 (Elective Deferral Dollar Limit). A Participant also will not be required to consent to a corrective distribution of Excess Deferrals.
- 7.4 <u>Direct Rollovers</u>. A Participant (other than a Participant subject to an Involuntary Cash-Out) may elect, at the time and the manner prescribed by the Plan Administrator, to have all or any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan (a "Direct Rollover"). For purposes of this Section 7.4, a Participant includes a Participant or former Participant. In addition, this Section applies to any distribution from the Plan made to a Participant's surviving Spouse or to a Participant's Spouse or former Spouse who is the Alternate Payee under a QDRO. For distributions made on or after January 1, 2007, this Section 7.4 also applies to distributions made to a Participant's non-Spouse beneficiary, as set forth in subsection (c) below.

# (a) Definitions.

- (1) <u>Eligible Rollover Distribution</u>. An Eligible Rollover Distribution is any distribution of all or any portion of a Participant's Account Balance, except an Eligible Rollover Distribution does not include:
  - i. any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's Beneficiary, or for a specified period of ten years or more;
  - ii. any distribution to the extent such distribution is a required minimum distribution under Code § 401(a)(9), as described under Section 8 (Required Minimum Distributions);
  - iii. any Hardship distribution, as described in Section 7.8(e) (Hardship Distribution);
  - iv. the portion of any distribution that is not includible in gross income; or,
  - v. any distribution if it is reasonably expected (at the time of the distribution) that the total

amount the Participant will receive as a distribution during the calendar year will total less than \$200. Any distribution from a Participant's Roth Deferral Account is not taken into account in determining whether distributions from a Participant's other Accounts are reasonably expected to total less than \$200 during a year. Further, the Plan will not provide for a Direct Rollover for distributions from a Participant's Roth Deferral Account if it is reasonably expected (at the time of distribution) that the total amount the Participant will receive as a distribution during the calendar year will total less than \$200; or,

<u>vi.v.</u> a distribution made to satisfy the requirements of Code § 415 (as described in Section 5.2) or a distribution to correct Excess Deferrals.

- **Eligible Retirement Plan.** For purposes of applying the Direct Rollover provisions under this Section 7.4, an Eligible Retirement Plan is:
  - i. a qualified plan described in Code § 401(a);
  - ii. an individual retirement account described in Code § 408(a);
  - iii. an individual retirement annuity described in Code § 408(b);
  - iv. an annuity plan described in Code § 403(a);
  - v. an annuity contract described in Code § 403(b); or,
  - vi. an eligible plan under Code § 457(b) which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this Plan.

If any portion of an Eligible Rollover Distribution is attributable to payments or distributions from a Roth Deferral Account, an Eligible Retirement Plan with respect to such portion will include only another Roth Deferral Account under an applicable retirement plan described in Code § 402A(e)(1) of the individual from whose Account the payments or distributions were made, or to a Roth IRA described in Code § 408A of such individual, and only to the extent the rollover is permitted under the rules of Code § 402(c).

The definition of Eligible Retirement Plan also applies in the case of a distribution to a surviving Spouse, or to a Spouse or former Spouse who is the Alternate Payee under a QDRO.

- (3) <u>Direct Rollover</u>. A Direct Rollover is a payment made directly from the Plan to the Eligible Retirement Plan specified by the Participant. The Plan Administrator may develop reasonable procedures for accommodating Direct Rollover requests.
- **(b)** <u>Direct Rollover Notice</u>. A Participant entitled to an Eligible Rollover Distribution must receive a written explanation of his/her right to a Direct Rollover, the tax consequences of not making a

Direct Rollover, and, if applicable, any available special income tax elections. The notice must be provided within 30–180 days prior to the Participant's Distribution Starting Date in the same manner as described in Section 7.3(b) (Participant Notice). The Direct Rollover notice must be provided to all Participants, unless the total amount the Participant will receive as a distribution during the calendar year is expected to be less than \$200.

- Direct Rollover by Non-Spouse Beneficiary. Effective for Plan Years beginning after December 31, 2009, the Plan permits a non-Spouse beneficiary (as defined in Code § 401(a)(9)(E)) to make a Direct Rollover of an Eligible Rollover Distribution to an individual retirement account under Code § 408(a) or an individual retirement annuity under Code § 408(b) that is established on behalf of the designated beneficiary and that will be treated as an inherited IRA pursuant to the provisions of Code § 402(c)(11). A non-Spouse rollover made after December 31, 2009 will be subject to the direct rollover requirements under Code § 401(a)(31), the rollover notice requirements under Code § 402(f) or the mandatory withholding requirements under Code § 3405(c).
- (d) <u>Direct Rollover of Non-Taxable Amounts</u>. Notwithstanding any other provision of the Plan, effective for taxable years beginning on or after January 1, 2007, an Eligible Rollover Distribution may include the portion of any distribution that is not includible in gross income. For this purpose, an Eligible Retirement Plan includes a Defined Contribution or defined benefit plan qualified under Code § 401(a) and a tax-sheltered annuity plan under Code § 403(b), provided the rollover is accomplished through a direct rollover and the recipient Eligible Retirement Plan separately accounts for any amounts attributable to the rollover of any nontaxable distribution and earnings thereon.
- **7.5 Distribution Upon Termination of Employment.** Subject to the required minimum distribution provisions under Section 8, a Participant who terminates employment for any reason (other than death) is entitled to receive a distribution of his/her vested Account Balance in accordance with this Section 7.5.
  - (a) Account Balance Not Exceeding Cash-Out Threshold. If a Participant's vested Account Balance does not exceed \$500 at the time of distribution, the only distribution option available under the Plan is an immediate lump sum option.
  - **Account Balance Exceeding Cash-Out Threshold.** If a Participant's vested Account Balance exceeds \$500 at the time of distribution, the Participant may elect to receive a distribution of his/her vested Account Balance at any time after the Participant terminates employment, but prior to his or her required beginning date.
- **7.6 Distribution Upon Death.** Subject to the Required Minimum Distribution rules in Section 8, a Participant's vested Account Balance will be distributed to the Participant's Beneficiary(ies) in accordance with this Section 7.6.
  - (a) <u>Death After Commencement of Benefits</u>. If a Participant begins receiving a distribution of his/her benefits under the Plan, and subsequently dies prior to receiving the full value of his/her vested Account Balance, the remaining benefit will continue to be paid to the Participant's

Beneficiary(ies) in accordance with the form of payment that has already commenced. If a Participant commences distribution prior to death only with respect to a portion of his/her Account Balance, then the rules in subsection (b) apply to the rest of the Account Balance.

- **(b)** <u>Death Before Commencement of Benefits</u>. If a Participant dies before commencing distribution of his or her benefits under the Plan, such benefit will be paid to the Participant's Beneficiary(ies) in a single lump sum as soon as administratively feasible following Participant's death.
- (c) <u>Timing of Distributions</u>. In no event will any death benefit be paid later than the end of the fifth year following the year of the Participant's death or in any other manner that is inconsistent with the Required Minimum Distribution rules under Section 8 (Required Minimum Distributions).
- **Determining a Participant's Beneficiary.** The determination of a Participant's Beneficiary(ies) to receive any death benefits under the Plan will be based on the Participant's Beneficiary designation under the Plan. If a Participant does not designate a Beneficiary to receive the death benefits under the Plan, distribution will be made to the default Beneficiaries, as set forth in subsection (3) below.
  - (1) Post-Retirement Death Benefit. If a Participant dies after commencing distribution of benefits under the Plan (but prior to receiving a distribution of his/her entire vested Account Balance under the Plan), the Beneficiary of any post-retirement death benefit is determined in accordance with the Beneficiary selected under the distribution option in effect prior to death.
  - (2) <u>Pre-Retirement Death Benefit</u>. If a Participant dies before commencing distribution of his/her benefits under the Plan, the Participant's Beneficiary(ies) will be treated as the sole Beneficiary(ies).
  - (3) <u>Default Beneficiaries</u>.
    - i. Prior to August 1, 2018. If a Participant dies prior to August 1, 2018 or fails to effectively update his or her Beneficiary designation form subsequent to August 1, 2018, or if the Beneficiary named by a Participant predeceases the Participant, then the Plan Administrator will pay the Participant's remaining Account balance to the Participant's estate.
    - **ii. After August 1, 2018.** Effective for individuals who become Participants in the Plan on or after August 1, 2018 or file a beneficiary designation form subsequent to such date:
      - (A) If a Participant fails to name a Beneficiary or the Beneficiary named by a Participant predeceases the Participant, then the Plan Administrator will pay the Participant's remaining Account balance in the following order of priority, to the:
        - **I.** Participant's surviving spouse; or if no surviving spouse, to the
        - II. Participant's children (including adopted children), in equal shares

by right of representation (one share for each surviving child and one share for each child who predeceases the Participant with living descendants); and if none, then to

- **III.** Participant's estate.
- (B) Where a Participant names more than one Beneficiary and one or more of them, but not all of them, predecease Participant, then the benefits that would have been paid to the deceased Beneficiaries shall be distributed to Participant's next of kin in accordance with subsection (ii)(A)(I through III).
- [4] Identification of Beneficiaries. The Plan Administrator may request proof of the Participant's death and may require the Beneficiary to provide evidence of his/her right to receive a distribution from the Plan as designated in accordance with the rules and procedures developed by the Plan Administrator. The Plan Administrator's determination of the Participant's death and of the right of a Beneficiary to receive payment under the Plan shall be conclusive. If a distribution is to be made to a minor or incompetent Beneficiary, payments may be made to the person's legal guardian, conservator recognized under state law, or custodian in accordance with the Uniform Gifts to Minors Act or similar law as permitted under the laws of the state where the Beneficiary resides. The Plan Administrator will not be liable for any payments made in accordance with this subsection (3) and will not be required to make any inquiries with respect to the competence of any person entitled to benefits under the Plan.
- (5) <u>Death of Beneficiary</u>. If a Beneficiary does not predecease the Participant but dies before distribution of the death benefit is made to the Beneficiary, the death benefit will be paid to the Beneficiary's designated Beneficiary. If the Beneficiary did not provide the Plan with a valid Beneficiary designation, the death benefit will be paid to the Beneficiary's estate.
- 6) <u>Divorce From Spouse</u>. If a Participant designates his/her Spouse as Beneficiary and subsequent to such Beneficiary designation, the Participant and Spouse are subsequently divorced, the designation of the Spouse as Beneficiary under the Plan is automatically rescinded unless specifically provided otherwise under a divorce decree or QDRO, or unless the Participant enters into a new Beneficiary designation naming the prior Spouse as Beneficiary. In addition, the provisions under this subsection will not apply if the Participant has entered into a Beneficiary designation that specifically overrides the provisions of this subsection 7.6(d)(6).

For periods prior to the date this Plan is executed by the District this subsection also applies to situations where the Participant and Spouse are legally separated.

- **7.7 Distribution to Disabled Employees.** A Participant who terminates employment on account of becoming Disabled may receive a distribution of his/her Account Balance in the same manner as a regular distribution upon termination.
- **7.8 In-Service Distributions.** The Plan permits the following in-service distributions.

- **Rollover Contributions.** A Participant may make in-service distributions from Rollover Contributions.
- **Matching Contributions.** Employer Matching Contributions may be withdrawn upon the occurrence of a Hardship or attainment of age 59½.
- (c) <u>Salary Deferrals</u>. Salary Deferrals (including any earnings on such amounts) generally may not be distributed prior to the Participant's severance from employment, death, or disability. However, Salary Deferrals are available for in-service distributions upon attainment of age 59½, upon a Hardship or upon meeting the requirements for a Qualified Reservist Distribution, as defined under subsection (d) below.
- **Qualified Reservist Distribution.** Effective September 11, 2001, the distribution provisions applicable to Salary Deferrals include a Qualified Reservist Distribution. If a Participant takes a Qualified Reservist Distribution, such distributions will not be subject to the ten percent (10%) penalty tax under Code § 72(t).
  - **Qualified Reservist Distribution.** For purposes of this subsection (d), a Qualified Reservist Distribution means any distribution to an individual if:
    - i. such distribution is from amounts attributable to elective deferrals described in Code § 402(g)(3)(A) or (C) or Code § 501(c)(18)(D)(iii),
    - ii. such individual was (by reason of being a member of a reserve component (as defined in § 101 of Title 37 of the United States Code)) ordered or called to active duty for a period in excess of 179 days or for an indefinite period, and
    - iii. such distribution is made during the period beginning on the date of such order or call and ending at the close of the active duty period.
  - **Active Duty.** A Qualified Reservist Distribution will only be available for individuals who are ordered or called into active duty after September 11, 2001.
- **Hardship Distribution.** The Plan allows in-service distribution at the request of a Participant or primary Beneficiary upon the occurrence of a Hardship event. A Hardship distribution under the Plan must meet the requirements of a safe harbor Hardship as described under subsection (1) below.
  - (1) <u>Safe Harbor Hardship Distribution</u>. To qualify for a safe harbor Hardship distribution, a Participant must demonstrate an immediate and heavy financial need, as described in subsection (i), and the distribution must be necessary to satisfy such need, as described in subsection (ii).
    - i. <u>Immediate and Heavy Financial Need</u>. To be considered an immediate and heavy financial need, the Hardship distribution must be made to satisfy one of the following financial needs:
      - (A) to pay expenses incurred or necessary for medical care (as described in

- Code § 213(d)) of the Participant, the Participant's Spouse or, dependents (determined without regard to whether the expenses exceed seven and one-half percent (7.5%) of adjusted gross income);
- **(B)** for the purchase (excluding mortgage payments) of a principal residence for the Participant;
- (C) for payment of tuition and related educational fees (including room and board but not including the repayment of student loans) for the next 12 months of post-secondary education for the Participant, the Participant's Spouse, children or dependents;
- **(D)** to prevent the eviction of the Participant from, or a foreclosure on the mortgage of, the Participant's principal residence;
- to pay funeral or burial expenses for the Participant's deceased parent, Spouse, child or₁ dependent;
- (F) effective January 1, 2020, to pay expenses to repair damage to the Participant's principal residence that would qualify for a casualty deduction under Code § 165 (determined without regard to whether the loss exceeds ten percent (10%) of the participant's adjusted gross income, and effective January 1, 2020, without regard to Code § 165(h)(5);
- to pay expenses and losses (including loss of income) incurred by the participant on account of a Federal Emergency Management Agency ("FEMA") -declared disaster, provided that the participant's principle residence or principle place of employment at the time of the disaster was located in an area designated by FEMA for individual assistance with respect to the disaster.
- **(H)** for any other event that the IRS recognizes as a safe harbor Hardship distribution event under ruling, notice or other guidance of general applicability.

The payment of funeral or burial expenses under subsection (E) above and the payment of expenses to repair damage to a principal residence under subsection (F) above only apply to Plan Years beginning on or after January 1, 2006. For purposes of determining eligibility of a Hardship distribution under this subsection (i), a dependent is determined under Code § 152. However, for taxable years beginning on or after January 1, 2005, the determination of dependent for purposes of tuition and education fees under subsection (C) above will be made without regard to Code §§ 152(b)(1), (b)(2), and (d)(1)(B) and the determination of dependent for purposes of funeral or burial expenses under subsection (E) above will be made without

regard to Code § 152(d)(1)(B).

- **ii. Determination of Whether Distribution Necessary to Satisfy Need.** A distribution will be considered as necessary to satisfy an immediate and heavy financial need of the Participant under the IRS's general standard test if:
  - (A) the distribution is not in excess of the amount of the immediate and heavy financial need, including amounts necessary to pay any federal, state or local income taxes or penalties reasonably anticipated to result from the distribution;
  - **(B)** the Participant obtained all <del>currently</del> available distributions under the plan and all other plans of deferred compensation maintained by the District, whether qualified or non-qualified; and
  - (C) effective for Hardship distributions made on or after January 1, 2020, the Participant must represent in writing (including via electronic medium) that he or she has insufficient cash or other liquid assets to satisfy the need. The Plan Administrator may rely on such representation absent actual knowledge to the contrary.
- (2) <u>Amount Available for Hardship Distribution</u>. A Participant may receive a Hardship distribution of Deferrals (including Roth Deferrals), Matching Employer Contributions and Rollover Contributions. Provided, however, distributions made from Salary Deferrals, when added to other Hardship distributions, does not exceed the total account balance which includes salary deferrals, Employer matching contributions, rollovers, and interest earned.
- (3) <u>Availability to Terminated Employees</u>. A Participant may take such a Hardship distribution after termination of employment to the extent no other distribution is available from the Plan.
- (4) <u>Six-Month Suspension of Salary Deferrals</u>. Effective for Hardship distributions made on or after January 1, 2020, the Plan will no longer suspend a Participant's Salary Deferrals after a Hardship distribution.
- (f) <u>Age 59½ Distributions</u>. A Participant may take in-service distributions upon attaining age 59½ in accordance with the terms and conditions established by the Plan Administrator. Effective January 1, 2019, in-service distributions at age 59½ are no longer limited to one per Plan Year.
- Qualified Birth or Adoption Distributions (QBOADs). The Plan permits a Participant to request and receive QBOADs. QBOADs must be requested by the Participant and may only be made from the Plan in the frequency, form and manner provided by the Plan Administrator. Further, Participants may have an opportunity to repay a QBOAD if the Participant is eligible to make a rollover into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder. This Section 7.8(g) is effective June 1, 2021.

- (h) Coronavirus-Related Distributions (CRDs). Participants may request and receive a Coronavirus-Related Distribution (CRD). CRDs must be requested and made in the form and manner provided by the Plan Administrator.
  - (1) "Coronavirus-Related Distribution" Defined. A CRD means a distribution from the Plan made between January 1, 2020 and December 31, 2020, to a Participant:
    - i who is diagnosed with the virus SARS-CoV-2 or the coronavirus disease 2019 (collectively, "COVID-19") by a test approved by the Centers for Disease Control and Prevention; or
    - ii whose spouse or dependent is diagnosed with COVID-19 by a test approved by the Centers for Disease Control and Prevention; or
    - being quarantined, furloughed or laid off, or having work hours reduced due to COVID-19; or being unable to work due to lack of childcare due to COVID-19 or closing or reducing hours of a business owned or operated by the spouse or Participant, or a member of the Participant's household (one who shares Participant's principal residence) due to COVID-19; or having a reduction in pay due to COVID-19 or having a job offer rescinded or start date for a job delayed due to COVID-19.
  - (2) Aggregate Dollar Limit. The aggregate amount of CRDs to a Participant may not exceed \$100,000 from all plans sponsored by the Employer.
  - (3) Re-Contribution of CRDs. Participants may have an opportunity to repay some or all of the Participant's CRD if completed within 3 years of the distribution, meets the recontribution requirements in Section 2202(a) of the SECURE Act, and the Participant is eligible to make a rollover contribution into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder.
  - (4) Participant Certification. The Plan Administrator may rely on a Participant's certification that he or she satisfies the conditions for a CRD.
- 7.9 Sources of Distribution. Unless otherwise indicated in the Plan, distributions will be made on a pro rata basis from all Accounts from which a distribution is permitted. Alternatively, to the extent permissible by law, the Plan Administrator may permit Participants to direct the Plan Administrator as to which Account the distribution is to be made as long as the directions provided by the Participant are from a source which such distributions are permitted. Regardless of a Participant's direction as to the source of any distribution, the tax effect of such a distribution will be governed by Code § 72 and the regulations thereunder. Further, Roth Deferrals and In-Plan Roth Rollovers that are distributed from the Plan within five years of the Participant's first designated Roth Deferral to the Plan will be subject to tax on the earnings portion of the distribution from the Roth Deferral Account or In-Plan Roth Rollover Account.
- **7.10 Purchase of Service Credit Under Defined Benefit Plan**. Notwithstanding any provision of the Plan to the contrary, the Plan may permit a Participant whose Distribution Starting Date has not yet occurred to

transfer as a direct plan-to-plan transfer, any portion of the Participant's Account to any other plan that is qualified under Code § 401(a) for the purpose of purchasing service credits under such other plan. Prior to permitting such transfer, however, the Plan shall reasonably determine all of the following:

- (a) that such other plan shall accept such a transfer;
- (b) that such transferred amounts are subject to withdrawal and distribution restrictions so that the Participant may not withdraw such transferred amounts prior to retirement, death, disability or severance from employment;
- (c) that such transferred amounts are held for the benefit of the Participant as elective contributions on a fully vested and non-forfeitable basis; and,
- (d) that applicable interest is credited to such transfer amounts.

The Plan Administrator shall establish a written policy for transfers for service credit purchases.

7.11 Correction of Qualification Defects. Nothing in this Section 7 precludes the Plan Administrator from making a distribution to a Participant to correct a qualification defect consistent with the correction procedures under IRS voluntary compliance programs. Thus, for example, if an Employee is permitted to enter the Plan prior to his/her proper Entry Date under Section 2.3 and the Plan Administrator determines that a corrective distribution is a proper means of correcting the operational violation, nothing in this Section 7 would prevent the Plan from making such corrective distribution. Any such distribution must be made in accordance with the correction procedures applicable under IRS voluntary correction programs under IRS Rev. Proc. 2013-12 (or successive guidance).

## **SECTION 8**

# **REQUIRED MINIMUM DISTRIBUTIONS**

- 8.1 Required Minimum Distributions (RMDs). The provisions of this Section 8 apply to calendar years beginning on or after January 1, 2003. A Participant's entire interest under the Plan will be distributed, or begin to be distributed, to the Participant no later than the Participant's Required Beginning Date. All distributions required under this Section 8 will be determined and made in accordance with the regulations under Code § 401(a)(9) and the minimum distribution incidental benefit requirement of Code § 401(a)(9)(G). For purposes of applying the required minimum distribution rules under this Section 8, any distribution made in a form other than a lump sum must be made over one of the following periods (or a combination thereof):
  - (a) the life of the Participant;
  - **(b)** the life of the Participant and a Designated Beneficiary;
  - (c) a period certain not extending beyond the life expectancy of the Participant; or
  - (d) a period certain not extending beyond the joint and last survivor life expectancy of the Participant and a Designated Beneficiary.

- **8.2 Death of Participant Before Required Distributions Begin.** If the Participant dies before required distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
  - Surviving Spouse as a Designated Beneficiary. If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, the surviving Spouse must take a distribution(s) under the five-year rule described in Section 8.6(a) below or under the life expectancy method. Distributions to the surviving Spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age 70½ (or age 72 if Participant attains age 70½ after December 31, 2019), if later. If such surviving Spouse does not make an election and a distribution must commence, the Plan will make distributions under the life expectancy method.
  - **Non-Spousal Designated Beneficiary.** If the Participant's surviving Spouse is not the Participant's sole Designated Beneficiary but the Plan segregates each Designated Beneficiary's Account, the Designated Beneficiary may elect to take distributions under the five-year rule described in Section 8.6(a) below or under the life expectancy method. If the non-Spousal Designated Beneficiary(ies) do not elect to commence distributions by December 31 of the calendar year immediately following the calendar year in which the Participant dies, distributions will be made under the life expectancy method.
  - (c) <u>No Designated Beneficiary</u>. If there is no Designated Beneficiary as of the date of the Participant's death who remains a Beneficiary as of September 30 of the year immediately following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
  - **Death of Surviving Spouse.** If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary and the surviving Spouse dies after the Participant but before distributions to the surviving Spouse begin, this Section 8.2 (other than subsection (a)) will apply as if the surviving Spouse were the Participant.
    - For distributions under this subsection (d), distributions are considered to begin on the date distributions are required to begin to the surviving Spouse under subsection (a) above.
- **Required Minimum Distributions During Participant's Lifetime.** During the Participant's lifetime, the minimum amount that must be distributed for each Distribution Calendar Year is described in this Section 8.3.
  - (a) Amount of Required Minimum Distribution for Each Distribution Calendar Year. During the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:
    - the quotient obtained by dividing the Participant's Account Balance by the distribution period set forth in the Uniform Lifetime Table found in Treas. Reg. § 1.401(a)(9)-9, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or

- if the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's Spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in Treas. Reg. § 1.401(a)(9)-9, Q&A-3, using the Participant's and Spouse's attained ages as of the Participant's and Spouse's birthdays in the Distribution Calendar Year.
- (b) <u>Lifetime Required Minimum Distributions Continue Through Year of Participant's Death.</u>
  Required Minimum Distributions will be determined under this subsection (b) beginning with the first Distribution Calendar Year and continuing up to, and including, the Distribution Calendar Year that includes the Participant's date of death.

# 8.4 Required Minimum Distributions After Participant's Death.

- (a) <u>Death On or After Date Required Distributions Begin</u>. Where the Participant's death occurs on or after his Required Beginning Date and Required Minimum Distributions have already begun, the minimum distributions made to Beneficiaries after the Participant's death must be completed at least as rapidly as under the minimum distribution period in effect at Participant's death.
  - (1) Participant Survived by Designated Beneficiary. If the Participant dies on or after the date required distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's Designated Beneficiary, determined as follows:
    - i. The Participant's remaining life expectancy is calculated in accordance with the Single Life Table found in Treas. Reg. § 1.401(a)(9)-9, Q&A-1, using the age of the Participant in the year of death, reduced by one for each subsequent year.
    - ii. If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, the remaining life expectancy of the surviving Spouse is calculated using the Single Life Table found in Treas. Reg. § 1.401(a)(9)-9, Q&A-1, for each Distribution Calendar Year after the year of the Participant's death using the surviving Spouse's age as of the Spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving Spouse's death, the remaining life expectancy of the surviving Spouse is calculated using the age of the surviving Spouse as of the Spouse's birthday in the calendar year of the Spouse's death, reduced by one for each subsequent calendar year.
    - iii. If the Participant's surviving Spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining life expectancy is calculated under the Single Life Table using the age of the Designated Beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
  - (2) <u>No Designated Beneficiary</u>. If the participant dies on or after the date required distributions

begin and there is no Designated Beneficiary as of the Participant's date of death who remains a Designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the Participant's remaining life expectancy under the Single Life Table calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.

- **(b)** Death Before Date Required Distributions Begin. Where the Participant's death occurs on or after his Required Beginning Date but his or her Required Minimum Distributions have not yet begun, the minimum distributions made to the Participant's Beneficiaries must be made in accordance with the rules in this Section 8.4(b).
  - (1) Participant Survived by Designated Beneficiary. If the Participant dies before the date required distributions begin, and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining life expectancy of the Participant's Designated Beneficiary, determined as provided in subsection 8.4 (a).
  - (2) No Designated Beneficiary. If the Participant dies before the date distributions begin and there is no Designated Beneficiary as of the date of death of the Participant who remains a Designated Beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
  - Death of Surviving Spouse Before Distributions to Surviving Spouse are Required to Begin. If the Participant dies before the date distributions begin, and the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, and the surviving Spouse dies before distributions are required to begin to the surviving Spouse under Section 8.2(a), then this subsection 8.4(b) will apply as if the surviving Spouse were the Participant.

# 8.5 <u>Definitions</u>.

- (a) <u>Designated Beneficiary</u>. A Beneficiary designated by the Participant (or the Plan), whose life expectancy may be taken into account to calculate minimum distributions, pursuant to Code § 401(a)(9) and Treas. Reg. § 1.401(a)(9)-4.
- (b) <u>Distribution Calendar Year</u>. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year that contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin pursuant to Section 8.2. The Required Minimum Distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Beginning Date. The Required Minimum

Distribution for other Distribution Calendar Years, including the Required Minimum Distribution for the Distribution Calendar Year in which the Participant's Required Beginning Date occurs, will be made on or before December 31 of that Distribution Calendar Year.

- **Life Expectancy.** For purposes of determining a Participant's Required Minimum Distribution amount, life expectancy is computed using one of the following tables, as appropriate: (1) Single Life Table, (2) Uniform Life Table, or (3) Joint and Last Survivor Table found in Treas. Reg. § 1.401(a)(9)-9.
- Account Balance. For purposes of determining a Participant's Required Minimum Distribution, the Participant's Account Balance is determined based on the Account Balance as of the last Valuation Date in the calendar year immediately preceding the Distribution Calendar Year (the "valuation calendar year") increased by the amount of any contributions or forfeitures allocated to the Account Balance as of dates in the calendar year after the Valuation Date and decreased by distributions made in the calendar year after the Valuation Date. The Account Balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the Distribution Calendar Year if distributed or transferred in the valuation calendaryear.
- **Required Beginning Date (RBD).** A Participant's Required Beginning Date under the Plan is April 1 that follows the end of the calendar year in which the later of the following two events occurs:
  - the Participant attains age 70½ (or age 72 if a Participant attains age 70½ after December 31, 2019), or
  - (2) the Participant terminates employment. For purposes of the Required Minimum Distribution, a Rehired Annuitant is considered terminated from employment.

A Participant may begin in-service distributions prior to his/her Required Beginning Date only to the extent authorized under Section 7.8 (In-Service Distributions).

## 8.6 **Special Rules**.

- **Election to Allow Participants or Beneficiaries to Elect 5-Year Rule.** A Participant or Designated Beneficiary may elect whether the life expectancy rule under Section 8.2 above or the five—year rule under subsection 8.6(a), no later than the earlier of September 30 of the calendar year in which distribution would be required to begin under Section 8.2 or by September 30 of the calendar year which contains the fifth anniversary of the Participant's (or, if applicable, surviving Spouse's) death.
- **(b)** Forms of Distribution. Unless the Participant's interest is distributed in a lump sum on or before the Required Beginning Date, as of the first Distribution Calendar Year distributions will be made in accordance with Sections 8.2 and 8.4.
- (c) <u>Waiver of Required Minimum Distributions</u>. For calendar year 2009, the Required Minimum

Distribution rules will not apply. In applying the provisions of this Section 8 for the 2009 Distribution Calendar Year,

- the Required Beginning Date with respect to any individual shall be determined without regard to this subsection (c) for purposes of applying this paragraph for Distribution Calendar Years after 2009, and
- required distributions to a beneficiary upon the death of the Participant shall be determined without regard to calendar year 2009.

A Participant or beneficiary who would have been required to receive a Required Minimum Distribution for the 2009 Distribution Calendar Year but for the enactment of Code § 401(a)(9)(H) ("2009 RMD"), may elect whether or not to receive the 2009 RMD (or any portion of such distribution). A distribution of the 2009 RMD or a series of substantially equal distributions (that include the 2009 RMDs) made at least annually and expected to last for the life (or life expectancy) of the participant, the joint lives (or joint life expectancy) of the participant and the participant's designated beneficiary, or for a period of at least 10 years, will be treated as an Eligible Rollover Distribution. However, if all or any portion of a distribution during 2009 is treated as an Eligible Rollover Distribution but would not be so treated if the Required Minimum Distribution requirements under this Section 8 had applied during 2009, such distribution shall not be treated as an Eligible Rollover Distribution for purposes of Code §§ 401(a)(31), 402(f) or 3405(c). (See Notice 2009-82 for transitional rules that apply for purposes of applying the rollover rules to the distribution of 2009 RMDs.).

- (d) Treatment of Trust Beneficiaries As Designated Beneficiaries. If a trust is properly named as a Beneficiary under the Plan, the beneficiaries of the trust will be treated as the Designated Beneficiaries of the Participant solely for purposes of determining the distribution period under this Section 8 with respect to the trust's interests in the Participant's vested Account Balance. The beneficiaries of a trust will be treated as Designated Beneficiaries for this purpose only if, during any period during which required minimum distributions are being determined by treating the beneficiaries of the trust as Designated Beneficiaries, the following requirements are met:
  - (1) the trust is a valid trust under state law, or would be but for the fact there is no corpus;
  - (2) the trust is irrevocable or will, by its terms, become irrevocable upon the death of the Participant;
  - (3) the beneficiaries of the trust who are beneficiaries with respect to the trust's interests in the Participant's vested Account Balance are identifiable from the trust instrument; and
  - (4) the Plan Administrator receives the documentation described in subsection (e)(1) below.

If the foregoing requirements are satisfied and the Plan Administrator receives such additional information as it may request, the Plan Administrator may treat such beneficiaries of the trust as Designated Beneficiaries.

(e) Special Rules Applicable to Trust Beneficiaries.

# (1) <u>Information That Must Be Supplied to Plan Administrator.</u>

- i. Required Minimum Distribution Before Death Where Spouse Is Sole

  Beneficiary. If a Participant designates a trust as the beneficiary of his/her entire benefit and the Participant's Spouse is the sole beneficiary of the trust, the Participant must provide the information under (A) or (B) below to satisfy the information requirements under subsection 8.6(d)(4) above.
  - (A) The Participant must provide to the Plan Administrator a copy of the trust instrument and agree that if the trust instrument is amended at any time in the future, the Participant will, within a reasonable time, provide to the Plan Administrator a copy of each such amendment; or
  - **(B)** The Participant must:
    - (I) provide to the Plan Administrator a list of all of the beneficiaries of the trust (including contingent and remaindermen beneficiaries with a description of the conditions on their entitlement sufficient to establish that the Spouse is the sole beneficiary) for purposes of Code § 401(a)(9);
    - (II) certify that, to the best of the Participant's knowledge, the list under subsection (B)(I) above is correct and complete and that the requirements of subsection 8.6(d) above are satisfied;
    - (III) agree that, if the trust instrument is amended at any time in the future, the Participant will, within a reasonable time, provide to the Plan Administrator corrected certifications to the extent that the amendment changes any information previously certified; and
    - **(IV)** agree to provide a copy of the trust instrument to the Plan Administrator upon demand.
- ii. Required Minimum Distribution After Death. In order to satisfy the documentation requirement of subsection 8.6(d)(4) above for required minimum distributions after the death of the Participant (or Spouse in a case to which U.S. Treas. Reg. § 1.401(a)(9)-3, Q&A-5 applies), the trustee of the trust must satisfy the requirements of subsection (A) or (B) by October 31 of the calendar year immediately following the calendar year in which the Participant died.
  - **(A)** The trustee of the trust must:
    - (I) provide the Plan Administrator with a final list of all beneficiaries of the trust (including contingent and remaindermen beneficiaries with a description of the conditions on their entitlement) as of September 30 of the calendar year following the calendar year of the Participant's death;

- (II) certify that, to the best of the trustee's knowledge, the list in subsection (B)(I) is correct and complete and that the requirements of subsection 8.6(d) above are satisfied; and
- (III) agree to provide a copy of the trust instrument to the Plan Administrator upon demand.
- (B) The trustee of the trust must provide the Plan Administrator with a copy of the actual trust document for the trust that is named as a beneficiary of the Participant under the Plan as of the Participant's date of death.
- Relief for Discrepancy. If required minimum distributions are determined based on the information provided to the Plan Administrator in certifications or trust instruments described in subsection (1) above, the Plan will not fail to satisfy Code § 401(a)(9) merely because the actual terms of the trust instrument are inconsistent with the information in those certifications or trust instruments previously provided to the Plan Administrator, provided the Plan Administrator reasonably relied on the information provided and the required minimum distributions for calendar years after the calendar year in which the discrepancy is discovered are determined based on the actual terms of the trust instrument.
- **8.7 Transitional Rule.** Notwithstanding the other requirements of this Section 8, distribution on behalf of any Employee may be made in accordance with all of the following requirements (regardless of when such distribution commences):
  - (a) The distribution by the Plan is one that would not have disqualified the Plan under Code § 401(a)(9) as in effect prior to amendment by the Deficit Reduction Act of 1984.
  - **(b)** The distribution is in accordance with a method of distribution designated by the Participant whose interest in the Plan is being distributed or, if the Participant is deceased, by a Beneficiary of such Participant.
  - (c) Such designation was in writing, was signed by the Participant or the beneficiary, and was made before January 1, 1984.
  - **(d)** The Participant had accrued a benefit under the Plan as of December 31,1983.
  - (e) The method of distribution designated by the Participant or the beneficiary specifies the time at which distribution will commence, the period over which distributions will be made, and in the case of any distribution upon the Participant's death, the beneficiaries of the Participant listed in order of priority.

A distribution upon death will not be covered by this transitional rule unless the information in the designation contains the required information described above with respect to the distributions to be made upon the death of the Participant.

For any distribution which commences before January 1, 1984, but continues after December 31, 1983, the Participant, or the Beneficiary, to whom such distribution is being made, will be presumed to have

designated the method of distribution under which the distribution is being made if the method of distribution was specified in writing and the distribution satisfies the requirements in subsections (a) - (e) above.

If a designation is revoked any subsequent distribution must satisfy the requirements of Code § 401(a)(9) and the proposed regulations thereunder. If a designation is revoked subsequent to the date distributions are required to begin, the Plan must distribute by the end of the calendar year following the calendar year in which the revocation occurs the total amount not yet distributed which would have been required to have been distributed to satisfy Code § 401(a)(9) and the proposed regulations thereunder, but for the TEFRA § 242(b)(2) election. For calendar years beginning after December 31, 1988, such distributions must meet the minimum distribution incidental benefit requirements. Any changes in the designation will be considered to be a revocation of the designation. However, the mere substitution or addition of another Beneficiary (one not named in the designation) under the designation will not be considered to be a revocation of the designation, so long as such substitution or addition does not alter the period over which distributions are to be made under the designation, directly or indirectly (for example, by altering the relevant measuring life). In the case in which an amount is transferred or rolled over from one plan to another plan, the rules in Treas. Reg. § 1.401(a)(9)-8, Q&A-14 and Q&A-15 shall apply.

## **SECTION 9**

## **SPOUSAL CONSENT RULES**

- **9.1** Application of Joint and Survivor Annuity Rules. As a governmental plan, the Qualified Joint and Survivor Annuity rules under Code §§ 401(a)(11) and 417 do not apply to the Plan. Notwithstanding the foregoing, the District requires spousal consent for certain designations of Plan benefits.
- **Spousal Consent**. Spousal consent is only required with respect to the distribution of death benefits from the Plan if the Participant designates someone other than the Participant's Spouse as a Beneficiary of more than fifty percent (50%) of the Participant's Account Balance.

A Spousal consent to the Beneficiary designation of more than fifty percent (50%) of Participant's Account to a non-spouse beneficiary must be provided at the time and in the form and manner specified by the Plan Administrator.

<u>Subsequent beneficiary designations submitted to the Plan Administrator in good order revoke all designations filed prior to that date by the same Participant.</u>

Participant's marriage or remarriage revokes any prior Beneficiary designations made by the Participant and a new Beneficiary designation form that complies with the consent requirements of this Section 9.2 (Spousal Consent) must be submitted.

Further, when designating a trust or other entity as a Beneficiary, the time and form of distribution available to such Beneficiaries may be limited by the required minimum distribution rules of Code § 401(a)(9) and Section 8 (RMDs) of the Plan.

#### **SECTION 10**

## PLAN ACCOUNTING AND INVESTMENTS

- **Participant Accounts.** The Plan Administrator will establish and maintain a separate Account for each Participant to reflect the Participant's entire interest under the Plan. Each Participant Account established by the Plan Administrator may include one or more of the following separate sub-Accounts:
  - Pre-Tax Salary Deferral Account
  - Matching Contribution Account
  - Rollover Contribution Account
  - Transfer Account

The Plan Administrator may establish other Accounts, as it deems necessary, for the proper administration of the Plan.

- **Valuation of Accounts.** A Participant's portion of the Trust assets is determined as of each Valuation Date under the Plan. The value of a Participant's Account consists of the fair market value of the Participant's share of the Trust assets. The Trustee must value Plan assets at least annually. The Trustee's determination of the value of Trust assets shall be final and conclusive.
  - **Periodic Valuation.** The Plan may elect to value assets on a periodic basis more frequent than annually including daily valuation. The Trustee and the Plan Administrator may adopt reasonable procedures for performing such valuations.
  - (b) <u>Daily Valuation</u>. For any period in which the Plan Administrator uses daily valuation, the Plan Administrator may adopt reasonable procedures for performing such valuations. Unless otherwise set forth in the written procedures, a daily valued Plan will have its assets valued at the end of each business day during which the New York Stock Exchange is open. The Plan Administrator has authority to interpret the provisions of this Plan in the context of a daily valuation procedure. This includes, but is not limited to, the determination of the value of the Participant's Account for purposes of Participant loans, distribution and consent rights, and corrective distributions.
  - **(c) Interim Valuations.** The Plan Administrator may request the Trustee to perform interim valuations.
- **10.3** Adjustments to Participant Accounts. Unless the Plan Administrator adopts other reasonable administrative procedures, as of each Valuation Date under the Plan, each Participant's Account is adjusted in the following manner:
  - (a) <u>Net Income or Loss</u>. A Participant's Account will be adjusted accordingly for any net income or loss from Participant's selected investment vehicles.
  - **(b)** <u>Distributions and Forfeitures from A Participant's Account</u>. A Participant's Account will be reduced by any distributions, forfeitures and other reductions from the Account since the previous Valuation Date.

- **(c) <u>Dividends.</u>** The Account will be credited with any interest, dividends, distributions received, or credits paid by the Trust for the benefit of the Participant.
- (d) <u>Contributions and Forfeitures Allocated to A Participant's Account</u>. A Participant's Account will be credited with any contribution, forfeiture or other additions allocated to the Participant since the previous Valuation Date.
- (e) <u>Directed Accounts</u>.
  - (1) Net income or loss attributable to the investments made by a Directed Account is allocated to such account in a manner that reasonably reflects the investment experience of such Directed Account.
  - Assets held in a Directed Account may be segregated from other investments held in Trust, and when this occurs, allocation of net income or loss shall not result in a Participant being entitled to distribution from the ancount that exceeds the value of the ancount as of the date of distribution.
- **Plan Related Fees and Expenses**. A Participant's Account will be reduced by any and all reasonable expenses and fees related to plan administration and investment charges allocable to his or her Account.
- **10.4 Share or Unit Accounting.** The Plan's investment procedures may provide for share or unit accounting to reflect the value of Accounts, if such method is appropriate for the investments allocable to such Accounts.
- **10.5 Suspense Accounts.** The Plan's investment procedures also may provide for special valuation procedures for suspense accounts that are properly established under the Plan.
- 10.6 <u>Investments Under the PlanDeferred Compensation Investment Committee</u>. The <u>DCICPlan</u>
  Administrator or other authorized person(s) responsible for the investment of Plan assets or the selection
  of investment options to be made available under the Plan is authorized has the full power and authority to
  invest, manage, and control Plan assets in any prudent i. The DCIC's membership, governance, and powers
  and duties are set forth in the Bylaws of the Deferred Compensation Investment option or select any
  prudent investment options as part of the Plan's investment lineupCommittee</u>, as approved by the Board.
- 10.7 Investment OptionsPolicy Statement. The DCICPlan Administrator shall select will establish and determine whichmaintain an iInvestment options are suitablePolicy Statement (IPS) for the Plan, and prudent based on the advicewill periodically review theat policy, as needed. at least annually. In establishing and guidance of anmaintaining the Plan's IPSiadvisor and the DCAC, the DCIC will review all pertinent Plan and are consistentParticipant information and data and may consult with the Plan's Investment Policy Statementany professional advisors as it deems necessary or appropriate, and will consider matters such as the following:
  - (a) <u>Common/Collective Trusts and Collectibles</u>. Plan assets may also be invested in a common/collective trust fund, or in a group trust fund that satisfies the requirements of IRS-Revenue Ruling 81–100 (as modified by Rev. Rul. 2004-67 and Rev. Rul. 2011-1). All of the terms and provisions of any such common/collective trust fund or group trust into which Plan assets are

invested are incorporated by reference into the provisions of the Trust for this Plan. No portion of any voluntary, tax deductible Employee contributions being held under the Plan (or any earnings thereon) may be invested in life insurance contracts or, as with any Participant directed investment, in tangible personal property characterized by the IRS as a collectible. the Plan's objectives and the best method to accomplish those objectives, including the range of investment alternatives that should be available to Participants and their Beneficiaries in order to accomplish those objectives;

- the policies or procedures to be followed by the DCIC in the selection and monitoring of investment alternatives, including the benchmarks and other performance criteria, as well as investment fees and expenses; and
- (a)(c) the liquidity of Plan assets.
- 10.8 Investment Manager. The Board of Directors may appoint one or more investment managers to manage (including the power to acquire and dispose of) any assets of the Plan. Only a person who is a registered investment advisor under the Investment Advisors Act of 1940, a bank or an insurance company qualified to do business in the State of California may be an investment manager under the Plan. The Trustee may be an investment manager.
- 10.9 Investment Funds. The Trustee will establish and maintain investment funds designated by the DCIC or an investment manager designated by the Board of Directors. The assets of each investment fund will be managed by the Trustee or by one or more investment managers, as designated, except for participant-directed investments that permit Participants to direct the investment of some or all of the assets in their Account in specific investments.
- Participant-Directed Investments. The Plan permits Participant direction of investments. Each Participant shall have the exclusive right, in accordance with the provisions of the Plan, to direct the investment of all or a portion of the amounts allocated to the separate sub-Accounts of the Participant under the Plan. This Section 10.710 also applies to any Beneficiary or Alternate Payee eligible to direct investments under the Plan.
  - Limits on Participant Investment Direction. The Plan's separate investment procedures may limit Participant direction of investment to specific types of contributions or with respect to specific investment options. If Participant investment direction is limited to specific investment options, it shall be the responsibility of the <a href="DCIC District or Plan Administrator">DCIC District or Plan Administrator</a> to select the investment options available under the Plan for Participant investment. In no case may Participants direct that investments be made in collectibles, including but not limited to U.S. Government or State-issued gold and silver coins. In addition, investment directions that result in the following will not be followed:
    - (1) result in a prohibited transaction;
    - (2) cause the assets of the Plan to be maintained outside the jurisdiction of the U.S. courts;
    - (3) jeopardize the Plan's tax qualification;

- **(4)** be contrary to the Plan's governing documents;
- cause the assets to be invested in collectibles within the meaning of Code § 408(m);
- **(6)** generate unrelated business taxable income; or
- (7) result (or could result) in a loss exceeding the value of the Participant's Account.
- **(b)** Failure to Direct Investment. The Plan Administrator DCIC will designate how a Participant Account shall be invested in the absence of proper affirmative direction from the Participant. The Plan or Plan Administrator DCIC may designate a default fund under the Plan to deposit contributions to the Trust on behalf of a Participant who has not specified an investment choice under the Plan.

<del>(b)</del> -

## **SECTION 11**

## PLAN ADMINISTRATION AND OPERATION

- **Plan Administrator.** The District is the Plan Administrator and has delegated to the General Manager the powers, duties, and responsibilities of the Plan Administrator. The General Manager may delegate, in writing, some or all of the powers, duties and responsibilities of the Plan Administrator position as necessary to adequately operate and maintain the Plan.
- **11.2 Role of Plan Administrator.** Any Plan Administrator designated by the District must accept the Plan Administrator role and responsibilities in writing and will serve until his or her successor is appointed. In case of a vacancy in the position of the Plan Administrator, the District will exercise any and all of the powers, authority, duties and discretion as the Plan Administrator pending the filling of the vacancy.
  - (a) Resignation or Removal of Plan Administrator. A Plan Administrator may resign by delivering notice of resignation to the Employer. The Employer may remove a Plan Administrator by delivering notice of removal. If a Plan Administrator resigns or is removed and no new Plan Administrator is designated, the District becomes the Plan Administrator.
  - Indemnification of Plan Administrator. The District will indemnify, defend and hold harmless members of the District's Board of Directors, the General Manager, or any Employees who have administrative responsibility under the 401(k) or 457(b) Plans, and the Deferred Compensation Investment Committee or any member thereof, with respect to any liability, loss, damage or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacities in the administration of the 401(k) or 457(b) Plans, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.
- 11.3 <u>Duties, Powers and Responsibilities of the Plan Administrator.</u>

- (a) <u>General Duties, Powers and Responsibilities</u>. The Plan Administrator shall have the responsibility and authority to administer the Plan and shall do so for the exclusive benefit of the Plan Participants and Beneficiaries, and in accordance with the terms of the Plan. <u>The Plan Administrator shall have the responsibility and authority to appoint or remove any investment manager</u>. The Plan Administrator shall have the authority to appoint or remove any Employee who has administrative responsibilities under the Plan.
- **Delegation of Duties. Powers and Responsibilities.** The Plan Administrator may delegate its duties, powers or responsibilities to one or more persons. Such delegation must be in writing and accepted by the person or persons receiving the delegation. However, the Employer must agree to any such delegation in writing when made by a delegate of the Plan Administrator. The Plan Administrator has designated the Human Resources Group Manager to carry out the powers, duties and responsibilities of the Plan Administrator.
- **(c)** Specific Plan Administrator Responsibilities. The Plan Administrator has the general responsibility to control and manage the operation of the Plan. More specifically, this responsibility includes, but is not limited to, the following:
  - (1) To communicate with the Trustee and other responsible persons with respect to the crediting of Plan contributions, the disbursement of Plan distributions and other relevant matters;
  - (2) To develop separate procedures, if necessary, consistent with the terms of the Plan to assist in the administration of the Plan, including the adoption of a separate or modified loan policy, procedures for direction of investment by Participants, procedures for determining whether domestic relations orders are QDROs, and procedures for the determination of investment earnings to be allocated to Participants' Accounts;
  - (3) To maintain all records necessary for tax and other administration purposes;
  - (4) To furnish and to file all appropriate notices, reports and other information to Participants, Beneficiaries, the Employer, the Trustee and government agencies (as necessary);
  - (5) To provide information relating to Plan Participants and Beneficiaries;
  - (6) To retain the services of other persons, including investment managers, attorneys, consultants, advisers and others, to assist in the administration of the Plan;
  - (7) To review and decide on <u>administrative</u> claims for benefits under the Plan;
  - (8) To correct any defect or error in the operation of the Plan; and
  - To interpret and enforce the provisions of the Plan, including those related to Plan eligibility, vesting and benefits; to interpret and apply provisions of the Plan in a uniform, nondiscriminatory manner:
  - (10) To act upon instructions from the DCIC to direct the Trustee regarding changes to the Plan's

<u>investment options and to execute proxy votes cast by the DCIC:</u>

- (11) To request and receive input from the DCIC, when reasonable and appropriate, regarding non-investment related Plan features; selection, performance, and retention of the record keeper, investment advisors, service providers, advisors, consultants, and auditor; and Plan amendments; and
- (12) To serve as the secretary for the DCIC which includes the following duties: keep and maintain the DCIC meeting minutes, execute documents on behalf of the DCIC, issue formal written correspondence of the DCIC; and provide other administrative support as the DCIC may reasonably delegate to the Plan Administrator.

The Plan Administrator shall coordinate with General Counsel concerning any litigation, compromise, or settlement of claims against the Plans in accordance with MWD Code Section 6433-

- **Interpretation of Plan.** The Employer shall have the responsibility and authority, as an express grant of discretionary authority, where necessary, to interpret and resolve ambiguities in this Plan document, provided such interpretation is consistent with the rules of Code § 401(a), and to make discretionary decisions regarding the interpretation of the Plan's terms, and Participant's rights and status under the Plan including, but not limited to, who is eligible to participate under the Plan, and the benefit rights of a Participant or Beneficiary.
- **(e)** Notices and Filings. The Plan Administrator may modify any notice period required by the Plan or designate any office to serve as the recipient of any form or notice that has to be filed under the Plan.
- (f) Appropriate Adjustments. Any action taken by the Plan Administrator with respect to 1) the rights or benefits under the Plan or 2) any Participant or Beneficiary shall be revocable by the Plan Administrator as to payments, distributions or deliveries not theretofore made hereunder pursuant to such action. Appropriate adjustments may be made in future payments or distributions to a Participant or Beneficiary.
- **Immunities.** The provisions of this Section 11.4 shall not be construed to in any way restrict the privileges and immunities that the Employer and its Employees are otherwise entitled to under Government Code Sections 818.8 and 820.2. Furthermore,
  - (a) Neither the District, its Board of Directors, nor its Employees shall be liable: 1) for any action taken or not taken with respect to the Plan including, but not limited to, any action allocated by the foregoing to another person or entity, except for its or their own gross negligence or willful misconduct; or 2) upon any contract, agreement or other instrument made or executed for administration of the Plan. With regards to the District's responsibility and authority to interpret the Plan, unless an interpretation or decision is determined to be arbitrary and capricious, the General Manager Plan Administrator will not be held liable for any interpretation of the Plan terms or decision regarding the application of a Plan provision.

(b) The District, its Board of Directors, each Employee thereof, specifically including the General Manager, and any other persons to whom the General Manager delegates or assigns any duty with respect to the Plan, may rely and shall be fully protected in acting in good faith upon: 1) the advice of counsel, who may be counsel for the District; 2) the records of the District; 3) the opinion, certificate, valuation, report, recommendation, or determination of the Trustee, an investment advisor, the DCIC Deferred Compensation Advisory Committee or any member thereof, or the District's General Auditor; or, 4) any certificate, statement or other representation made by or any information furnished by a Participant, Beneficiary, Alternate Payee, Trustee, DCIC Deferred Compensation Advisory Committee or any member thereof, or investment advisor, concerning any fact required to be determined under any of the provisions of the Plan.

# 11.5 <u>Plan Administration Expenses</u>.

- (a) Reasonable Plan Administration Expenses. All reasonable expenses related to plan administration will be paid from Plan assets, except to the extent the expenses are paid (or reimbursed) by the Employer. For this purpose, Plan expenses include, but are not limited to, all reasonable costs, charges and expenses incurred by the Trustee, DCIC, Plan Sponsor or Plan Administrator in connection with the administration of the Plan and Trust. If liquid assets of the Trust are insufficient to cover the fees of the Trustee or the Plan Administrator, then Trust assets shall be liquidated to the extent necessary for such fees. In the event any part of the Trust becomes subject to tax, all taxes incurred will be paid from the Trust.
- (b) Plan Expense Allocation. The Plan Administrator will allocate plan expenses among the accounts of Plan Participants. The Plan Administrator has authority to allocate these expenses either proportionally based on the value of the Account Balances or pro rata based on the number of Participants in the Plan. The Plan Administrator will determine the proper method for allocating expenses in accordance with such reasonable nondiscriminatory rules as the Plan Administrator deems appropriate under the circumstances. Unless the Plan Administrator decides otherwise, the following expenses will be allocated to the Participant's Account relative to which the expense is incurred: distribution expenses, including those relating to lump sums, installments, QDROs, hardship, in-service and required minimum distributions; loan expenses; participant direction expenses, including brokerage fees; and benefit calculations.
- **Expenses Related to Administration of Former Employee or Surviving Spouse.** If the Plan is making distributions to a former Employee or surviving Spouse, the Plan may charge reasonable Plan administrative expenses to the Account of that former Employee or surviving Spouse, but only if the administrative expenses are on a pro rata basis. Under the pro rata basis, the expenses are based on the amount in each account of a former Employee or surviving Spouse receiving benefits from the Plan. The Plan Administrator may use another reasonable basis for charging the expenses.

# 11.6 <u>Claims Procedure and Determinations.</u>

**Benefit Claims Procedures.** The Plan Administrator is authorized to conduct an examination of the relevant facts to determine the merits of a Participant's or Beneficiary's claim for Plan benefits.

The Plan Administrator may establish procedures for administering benefit claims. Such benefit claims procedures should provide claimants with a reasonable opportunity to have a full and fair review of a denied claim. Any claims procedure will incorporate the guidelines under this Section 11.6. To the extent any of the time periods specified in this Section 11.6 are amended by law, the time frames specified herein shall automatically be changed in accordance with such law or regulation.

(b) <u>Benefit Claims Determinations</u>. The Plan Administrator, in case of disputes, may make findings of fact and determinations with respect to any benefit claims made under the Plan. Subject to the provisions of this Section 11.6, such determinations and findings shall be final and conclusive, to the extent permitted by law, as to all interested persons for all purposes of the Plan. The Employer shall instruct the Plan Administrator and Trustee as to its final determination and shall furnish the Plan Administrator and Trustee with any further information reasonably required to administer the Plan consistent with the final determination.

# 11.7 Qualified Domestic Relations Orders (QDROs).

**In General.** Upon receipt of an order which appears to be a QDRO, the Plan Administrator will notify the Participant involved and each Alternate Payee under the order. The Plan Administrator will determine whether the order is a QDRO and will notify each affected individual of such determination. The Plan Administrator may use the default QDRO procedures set forth in subsection 11.7(h) below or may develop separate QDRO procedures for administering any QDROs submitted under the Plan.

# (b) <u>Definitions Related to ODROs</u>.

- QDRO. A QDRO is a domestic relations order that creates or recognizes the existence of an Alternate Payee's right to receive, or assigns to an Alternate Payee the right to receive, all or a portion of the benefits payable with respect to a Participant under the Plan. (See Code § 414(p).) The QDRO must contain certain information and meet other requirements described in this Section 11.7.
- **Domestic Relations Order.** A domestic relations order is a judgment, decree, or order (including the approval of a property settlement) that is made pursuant to state domestic relations law (including community property law).
- (3) <u>Alternate Payee</u>. An Alternate Payee must be a Spouse, former Spouse, child, or other dependent of a Participant.
- **Recognition as a QDRO.** To be a QDRO, an order must be a domestic relations order that relates to the provision of child support, alimony payments, or marital property rights for the benefit of an Alternate Payee. The Plan Administrator is not required to determine whether the court or agency issuing the domestic relations order had jurisdiction to issue an order, whether state law is correctly applied in the order, whether service was properly made on the parties, or whether an individual identified in an order as an Alternate Payee is a proper Alternate Payee under state law.

Effective April 6, 2007, a domestic relations order otherwise meeting the requirements to be a QDRO shall not fail to be treated as a QDRO solely because:

- (1) the order is issued after, or revises, another domestic relations order or QDRO; or
- (2) of the time at which the order is issued, including orders issued after the death of the Participant.

Any QDRO described in this Section 11.7 shall be subject to the same requirements and protections which apply to QDROs under Code  $\S$  414(p)(7).

- **(d) Contents of ODRO.** A QDRO must contain the following information:
  - (1) the name and last known mailing address of the Participant and each Alternate Payee;
  - (2) the name of each plan to which the order applies;
  - (3) the dollar amount or percentage (or the method of determining the amount or percentage) of the benefit to be paid to the Alternate Payee; and
  - (4) the number of payments or time period to which the order applies.

# (e) <u>Impermissible QDRO Provisions</u>.

- (1) The order must not require the Plan to provide an Alternate Payee or Participant with any type or form of benefit, or any option, not otherwise provided under the Plan;
- (2) The order must not require the Plan to provide for increased benefits (determined on the basis of actuarial value);
- (3) The order must not require the Plan to pay benefits to an Alternate Payee that are required to be paid to another Alternate Payee under another order previously determined to be a QDRO; and
- (4) The order must not require the Plan to pay benefits to an Alternate Payee in the form of a Qualified Jointand Survivor Annuity for the lives of the Alternate Payee and his or her subsequent Spouse.
- **Immediate Distribution to Alternate Payee.** Even if a Participant is not eligible to receive an immediate distribution from the Plan, an Alternate Payee may receive a QDRO benefit immediately in a lump sum, provided such distribution is consistent with the QDRO provisions.
- **(g) Fee for QDRO Determination.** The Plan Administrator shall condition the making of a QDRO determination on the payment of a fee by the Participant or an Alternate Payee either directly or as a charge against the Participant's Account.
- **(h) <u>Default QDRO Procedure.</u>** If the Plan Administrator chooses this default QDRO procedure or if the Plan Administrator does not establish a separate QDRO procedure, this subsection 11.7(h) will

- apply as the procedure the Plan Administrator will use to determine whether a domestic relations order is a QDRO. This default QDRO procedure incorporates the requirements set forth below.
- (i) Access to Information. The Plan Administrator will provide access to Plan and Participant benefit information sufficient for a prospective Alternate Payee to prepare a QDRO. Such information might include the summary plan description, other relevant plan documents, and a statement of the Participant's benefit entitlements. The disclosure of this information is conditioned on the prospective Alternate Payee providing to the Plan Administrator information sufficient to reasonably establish that the disclosure request is being made in connection with a domestic relations order.
  - (1) Notifications to Participant and Alternate Payee. The Plan Administrator will promptly notify the affected Participant and each Alternate Payee named in the domestic relations order of the receipt of the order. The Plan Administrator will send the notification to the address included in the domestic relations order. Along with the notification, the Plan Administrator will provide a copy of the Plan's procedures for determining whether a domestic relations order is a QDRO.
  - (2) <u>Alternate Payee Representative</u>. The prospective Alternate Payee may designate a representative to receive copies of notices and Plan information that are sent to the Alternate Payee with respect to the domestic relations order.
  - (3) Evaluation of Domestic Relations Order. Within a reasonable period of time, the Plan Administrator will evaluate the domestic relations order to determine whether it is a QDRO. A reasonable period will depend on the specific circumstances. The domestic relations order must contain the information described in subsection 11.7(d). If the order is only deficient in a minor respect, the Plan Administrator may supplement information in the order from information within the Plan Administrator's control or through communication with the prospective Alternate Payee.
    - **Separate Accounting.** Upon receipt of a domestic relations order, the Plan Administrator will separately account for and preserve the amounts that would be payable to an Alternate Payee until a determination is made with respect to the status of the order. During the period in which the status of the order is being determined, the Plan Administrator will take whatever steps are necessary to ensure that amounts that would be payable to the Alternate Payee, if the order were a QDRO, are not distributed to the Participant or any other person. The separate accounting requirement may be satisfied, at the Plan Administrator's discretion, by a segregation of the assets that are subject to separate accounting.
    - **Separate Accounting Until the End of 18-Month Period.** The Plan Administrator will continue to separately account for amounts that are payable under the QDRO until the end of an 18-month period. The 18-month period will begin on the first date following the Plan's receipt of the order upon which a payment would be

required to be made to an Alternate Payee under the order. If, within the 18-month period, the Plan Administrator determines that the order is a QDRO, the Plan Administrator must pay the Alternate Payee in accordance with the terms of the QDRO. If, however, the Plan Administrator determines within the 18-month period that the order is not a QDRO, or, if the status of the order is not resolved by the end of the 18-month period, the Plan Administrator may pay out the amounts otherwise payable under the order to the person or persons who would have been entitled to such amounts if there had been no order. If the order is later determined to be a QDRO, the order will apply only prospectively; that is, the Alternate Payee will be entitled only to amounts payable under the order after the subsequent determination.

- **Preliminary Review.** The Plan Administrator will perform a preliminary review of the domestic relations order to determine if it is a QDRO. If this preliminary review indicates the order is deficient in some manner, the Plan Administrator will allow the parties to attempt to correct any deficiency before issuing a final decision on the domestic relations order. The ability to correct is limited to a reasonable period of time.
- **Notification of Determination.** The Plan Administrator will notify in writing the Participant and each Alternate Payee of the Plan Administrator's decision as to whether a domestic relations order is a QDRO. In the case of a determination that an order is not a QDRO, the written notice will contain the following information:
  - (A) references to the Plan provisions on which the Plan Administrator based its decision;
  - (B) an explanation of any time limits that apply to rights available to the parties under the Plan (such as the duration of any protective actions the Plan Administrator will take); and
  - (C) a description of any additional material, information, or modifications necessary for the order to be a QDRO and an explanation of why such material, information, or modifications are necessary.
  - Administrator will act in accordance with the terms of the QDRO as if it were a part of the Plan. Except as designated otherwise under this subsection 11.7(h)(4)(v), an Alternate Payee will be considered a Beneficiary under the Plan and be afforded the same rights as a Beneficiary. The Plan Administrator will provide any appropriate disclosure information relating to the Plan to the Alternate Payee. In determining the rights of an Alternate Payee, the following rules apply.
  - **(E) Loans.** An Alternate Payee is not permitted to take a loan from the Plan.

- (F) Death Benefits. If an Alternate Payee dies prior to receiving the entire amount designated under the QDRO, such benefits will be paid in accordance with Section 7.7, treating the Alternate Payee as the Beneficiary. If the Alternate Payee dies without a designated Beneficiary, the benefits will be paid to the Alternate Payee's estate. Any death benefit will be paid in a single sum as soon as administratively feasible after the Alternate Payee's death.
- **(G)** <u>Direction of Investments</u>. An Alternate Payee has the right to direct the investment of the portion of the Participant's benefit that is segregated for the Alternate Payee's benefit pursuant to a QDRO in the same manner as the Participant.
- Military Service. To the extent required under Code § 414(u), an Employee who returns to employment with the Employer following a period of qualified military service will receive any contributions, benefits and service credit required under Code § 414(u), provided the Employee satisfies all applicable requirements under the Code and regulations. In determining the amount of contributions under Code § 414(u), Plan Compensation will be deemed to be the compensation the Employee would have received during the period while in military service based on the rate of pay the Employee would have received from the Employer but for the absence due to military leave. If the compensation the Employee would have received during the leave is not reasonably certain, Plan Compensation will be equal to the Employee's average compensation from the Employer during the twelve (12) month period immediately preceding the military leave or, if shorter, the Employee's actual period of employment with the Employer.
  - **Death Benefits Under Qualified Military Service.** In the case of a Participant who dies while performing qualified military service (as defined in Code § 414(u)), the survivors of the Participant are entitled to any additional benefits (other than benefit accruals relating to the period of qualified military service) provided under the Plan as though the Participant resumed and then terminated employment on account of death. This provision is effective with respect to deaths occurring on or after January 1, 2007.
  - (b) Plan Distributions. Notwithstanding the provisions of Section 1.54(c) regarding the treatment of Differential Pay, an individual shall be treated as having been severed from employment during any period the individual is performing service in the Uniformed Services for purposes of receiving a Plan distribution under Code § 401(k)(2)(B)(i)(I). If an individual elects to receive a distribution while on military leave, the individual may not make Salary Deferrals under the Plan during the 6-month period beginning on the date of the distribution.
  - (c) <u>Make-Up Contributions</u>. A Participant who is reemployed following a qualified military leave shall have the right to make up any Salary Deferrals to which he/she would have been entitled but for the fact the Participant was on qualified military leave. The Employer will also make any Matching Contributions the Participant would have earned during the period of qualified military leave had the Participant remained employed during such period. The Employer will only be required to make Matching Contributions if the reemployed Participant makes up the underlying contributions

that were eligible for the Matching Contributions.

In determining the amount of Make-Up Contributions a Participant may make under this subsection (d), a Participant will be treated as earning Plan Compensation during the period the Participant was on qualified military leave equal to:

- (1) the rate of pay the Participant would have received from the Employer during such period had the Participant not been on qualified military leave, or
- (2) if the Plan Compensation the Participant would have received during such period was not reasonably certain, the Participant's average Plan Compensation during the 12-month period immediately preceding the qualified military leave (or the entire period of employment, if shorter).

If the Employer is required under this subsection (d) to make Employer Contributions for a reemployed Participant, the Employer must make such Employer Contributions not later than 90 days after the date of reemployment or the date the Employer Contributions are otherwise due for the year in which the military service was performed. For Salary Deferrals and After-Tax Employee Contributions, a Participant who is reemployed following a qualified military leave may make up such contributions during the period beginning on the date of reemployment and ending on the earlier of the date that is three times the length of the military service period or 5 years from the date of reemployment. Any required Matching Contributions must be made in the same manner as other Matching Contribution under the Plan following the Participant's contribution of the amounts eligible for the Matching Contributions.

Any make up contributions under this subsection 11.8(c) are subject to Code § 415 Limitation under Section 5.2 (Elective Deferral Dollar Limit) for the year for which the make-up contribution would have been made had the Participant not been on qualified military leave.

## **SECTION 12**

## TRUST PROVISIONS FUNDS

**12.1 Establishment of Trust.** In conjunction with the establishment of this Plan, a domestic Trust consisting of such sums as shall from time to time be paid to the Trust under the Plan and such earnings, income and appreciation as may accrue thereon will be established pursuant to a written Trust Agreement.

The District may enter into a Trust Agreement with the Trustee providing for the investment of the Trust and prescribing the powers, duties, obligations, and functions of the Trustee with respect to the Plan. The Trust Agreement will form a part of this Plan, and any rights and benefits of any Participant or Beneficiary under the Plan whose Accounts are invested in the Trust will be subject to the terms of the Trust Agreement. However, in the case of any conflict between the Plan and the Trust Agreement, the Plan will govern. The Trustee must be one or more banks or other fiduciaries appointed by the District to hold and administer assets of the Plan.

The Trust shall be operated solelyee must hold the Plan assets received by it in <u>Trust pursuant to</u> the interest terms of the <u>Trust Agreement for the exclusive benefit of</u> Participants and their Beneficiaries and for the exclusive purpose of providing benefits to Participants and their Beneficiaries and defraying the reasonable

expenses of administering the Plan. Except as provided in Section 15.2, no assets of the Plan shall inure to the benefit of the Employer <u>prior to the satisfaction of all liabilities with respect to Participants and their</u>
Beneficiaries.

The Trust-shall be heldee must hold, managed, administered, valued, invested, reinvested and accounted for by the Trustee assets in accordance with the terms of the Plan and the Trust Agreement.

The In its discretion, the Plan Administrator may recommend that the District may further terminate the Trustee, appoint a nondiscretionarynew to to carry out or assistinstruct the Trustee with some or all of the duties and responsibilities delegated transfer assets to the a successor Trustee or Trustees. A nondiscretionary trustee is authorized to perform the duties delegated to the nondiscretionary trustee or directed to carry out by the District, Trustee, Plan Administrator or Plan-Participant, as applicable. The nondiscretionary trustee is liable for any losses caused directly or indirectly from its negligence, misconduct or failure to act in good faith or as a result of any act or omission where it has exerted discretion. An appointment of a nondiscretionary trustee and any agreement documents outlining the duties and responsibilities of a nondiscretionary trustee shall be maintained as part of the Plan's records.

12.2 Responsibilities of the Trustee. In addition to the powers, rights and responsibilities enumerated under this Section 12, the Trustee has all powers necessary to carry out its duties in a prudent manner. The Trustee's powers, rights and responsibilities may be modified, supplemented or limited by the District, by the Plan's Investment Policy Statement, or by any other binding document entered into between the Trustee and the District.

# (a) Responsibilities Regarding Administration of Trust.

- (1) The Trustee shall discharge the duties and responsibilities under this Plan solely in the interest of Participants and their Beneficiaries in the following manner:
  - i. for the exclusive purpose of providing benefits to Participants and their Beneficiaries and defraying reasonable expenses of administering the Plan;
  - ii. with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent person acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims; and
  - iii. by diversifying the available investments under the Plan so as to minimize the risk of large losses, unless under the circumstances it is clearly prudent not to do so.
- The Trustee will receive all contributions, earnings and other amounts made to and under the terms of the Plan. The Trustee is not liable for the manner in which such amounts are deposited or the allocation between Participant's Accounts, to the extent the Trustee follows the written direction of the Plan Administrator, the District, Participant or Beneficiary, as applicable.
- (3) The Trustee will make distributions from the Trust in accordance with the written directions Plan document.

- (4) The Trustee may employ agents, attorneys accountants and other third parties to provide counsel on behalf of the Plan, where the Trustee deems advisable. The Trustee may reimburse such persons from the Trust for reasonable expenses and compensation incurred as a result of such employment. The Trustee shall not be liable for the actions of such persons, provided the Trustee acted prudently in the employment and retention of such persons. In addition, the Trustee will not be liable for any actions taken as a result of good faith reliance on the advice of such persons.
- (5) The Trustee shall keep full and accurate accounts of all receipts, investments, disbursements and other transactions hereunder, including such specific records as may be agreed upon inwriting between the District and the Trustee. All such accounts, books and records shall be open to inspection and audit at all reasonable times by any authorized representative of the District or the Plan Administrator. A Participant may examine only those individual account records pertaining directly to him or her.
- (6) Except as provided in Section 15.2, at no time prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries under the Plan shall any part of the corpus or income of the Fund beused for, or diverted to, purposes other than for the exclusive benefit of Participants or their Beneficiaries, or for defraying reasonable expenses of administering the Plan.
- (b) Responsibilities Regarding Investment of Plan Assets.
  - (1) The Trustee shall be responsible for holding the assets of the Trust in accordance with the provisions of this Plan.
  - (2) The Trustee may invest, manage and control the Plan assets in a manner that is consistent with the Plan's funding policy and the Plan's Investment Policy Statement.
    - The Trustee may invest in any investment, as authorized under this subsection 12.3(b), which the Trustee deems advisable and prudent, and where applicable, subject to the proper written direction of the Plan Administrator, the District, a properly appointed investment manager, or a Plan Participant.
    - The Trustee is not liable for the investment of Plan assets to the extent the Trustee is following the proper direction of the Plan Administrator, the Employer, a Participant, an investment manager, or other person or persons duly appointed by the Employer to provide investment direction.
    - In addition, the Trustee does not guarantee the Trust in any manner against investment loss or depreciation in asset value if not the result of an error or omission made by the Trustee.
  - (3) The Trustee may hold any securities or other property in the name of the Trustee or in the name of the Trustee's nominee, and may hold any investments in bearer form, provided the books and records of the Trustee at all times show such investment to be part of the Trust.

- (4) The Trustee may retain such portion of the Plan assets in cash or cash balances as the Trustee may, from time to time, deem to be in the best interests of the Plan, without liability for interest thereon.
- (5) The Trustee may collect and receive any and all moneys and other property due the Plan and to settle, compromise, or submit to arbitration any claims, debts, or damages with respect to the Plan, and to commence or defend on behalf of the Plan any lawsuit, or other legal or administrative proceedings.
- (6) The Trustee <u>DCIC</u> may pay expenses out of Plan assets as necessary to administer the Trust and as authorized under the Plan.
- (7) The Trustee is authorized to execute, acknowledge and deliver all documents of transferand conveyance, receipts, releases, and any other instruments that the Trustee deemsnecessary or appropriate to carry outits powers, rights and duties hereunder.
- (8) The Trustee is authorized to enter into a transfer agreement with the <u>t</u>Trustee of another qualified retirement plan and to accept a transfer of assets from such retirement plan on behalf of any Employee of the Employer. The Trustee is also authorized to transfer some or all of a Participant's vested Account Balance to another qualified retirement plan on behalf of such Participant.
- (9) If the Employer maintains more than one Plan, the assets of such Plans may be commingled for investment purposes. The Trustee must separately account for the assets of each Plan.

  A commingling of assets does not cause the Trusts maintained with respect to the Employer's Plans to be treated as a single Trust, except as provided in a separate document authorized in the first paragraph of this Section 12.3.
- (10) The Trustee is authorized to invest Plan assets in a common/collective trust fund, or in a group trust fund that satisfies the requirements of IRS Revenue Ruling 81-100, as clarified by Revenue Ruling 2004-67. All of the terms and provisions of any such common/collective trust fund or group trust into which Plan assets are invested are incorporated by reference into the provisions of the Trust for this Plan. The assets in a group trust may be pooled with the assets of a custodial account under Code § 403(b)(7), a retirement income account under Code § 403(b)(9), and Code § 401(a)(24) governmental plans without affecting the tax status of the group trust, subject to the requirements under Rev. Rul. 2011-1 (as modified by Notice 2012-6).
- 12.3 Responsibilities of the Employer. The Employer will provide to the Trustee written notification of the appointment of any person or persons as Plan Administrator or investment manager and the names, titles and authorities of any individuals who are authorized to act on behalf of such persons. The Trustee shall be entitled to rely upon such information until it receives written notice of a change in such appointments or authorizations.

The Employer may authorize the Trustee to enter into a merger agreement with the trustee of another planto effect such merger or consolidation. A merger agreement entered into by the Trustee is not part of this Plan and does not affect the assets transferred to this Plan from another plan.

- 12.4 More Than One Trustee. If the Plan has more than one person acting as Trustee, the Trustees may allocate the Trustee responsibilities by mutual agreement. The Trustees may agree to make decisions by a majority vote or may permit any one of the Trustees to make any decision, undertake any action or execute any documents affecting this Trust without the approval of the remaining Trustees. The Trustees may agree to the allocation of responsibilities in a separate trust agreement or other binding document.
- 12.5 <u>Annual Valuation</u>. The Plan assets will be valued at least on an annual basis. The Employer may designate more frequent Valuation Dates in a written policy. The Trustee and Plan Administrator may agree to value the Trust on a more frequent basis, and/or to perform an interim valuation of the Trust.
- 12.6 Reporting to Plan Administrator and Employer. Within a reasonable time after the end of each Plan Year or within a reasonable time after its removal or resignation, the Trustee shall file with the Plan Administrator a written account of the administration of the Trust showing all transactions effected by the Trustee from the last preceding accounting to the end of such Plan Year or date of removal or resignation. The accounting will-include a statement of cash receipts, disbursements and other transactions effected by the Trustee since the date of its last accounting, and such further information as the Trustee and/or Employer deems appropriate. The Trustee shall have a reasonable time following its receipt of a written disapproval from the Employer to provide the Employer with a written explanation of the terms in question. If the Employer again disapproves of the accounting, the Trustee may file its accounting with a court of competent jurisdiction for audit and adjudication.
  - Reasonable Compensation. The Trustee may be paid reasonable compensation in an amount agreed upon by the Plan Administrator and Trustee. The Trustee may also be reimbursed for any reasonable expenses or fees incurred in its function as Trustee. An individual Trustee who is already receiving pay as an Employee of the Employer shall not receive additional compensation for services as Trustee. The Plan will pay the reasonable compensation and expenses incurred by the Trustee, unless the Employer pays such compensation and expenses. Any compensation or expense paid directly by the Employer to the Trustee is not an Employer Contribution to the Plan. A nonemployee trustee (whether discretionary or nondiscretionary) may be paid reasonable annual compensation in an amount agreed upon by the Plan Administrator and Trustee.
  - Resignation and Removal of Trustee/Appointment of Successor. The Trustee may resign at any time by providing Employer and the Plan Administrator written notice of resignation at least thirty (30) days prior to the effective date of such resignation, unless the District Employer consents in writing to a shorter period. The Trustee's notice must specify the effective date of the Trustee's resignation. Employer may remove the Trustee at any time, with or without cause, by providing written notice to the Trustee. The District's notice must specify the effective date of removal., which shall not be less than thirty (30) days after delivery therof, unless such notice shall be waived.

Upon the resignation, removal, death or incapacity of a Trustee, where no other Trustee continues to serve, the District Employer must either either appoint a successor Trustee or become the successor Trustee to continue the Plan. Upon accepting such appointment, a successor successor Trustee will have all the powers, rights and duties conferred upon the preceding Trustee. If two or more persons hold the position of Trustee, in the event of the removal of one such person, during any period the selection of a replacement is pending, or during any period such person is unable to serve for any reason, the remaining person or persons will act as the Trustee. If the District Employer fails to appoint a successor Trustee as of the effective date of the Trustee resignation, removal, death or incapacity and no other Trustee remains, the District will be treated as the Trustee having appointed itself as Trustee and as having accepted the ed the appointment. During such period, the Trust continues to be in existence and legally enforceable, and the assets of the Plan shall continue to be protected by the provisions of the Trust.

- <u>Indemnification of Trustee</u>. Except to the extent that it is judicially determined that the Trustee has acted with gross negligence negligently or has committed willful misconduct, the Employer shall indemnify the Trustee (whether or not the Trustee has resigned or been removed) against any liabilities, losses, damages, and expenses, including attorney, accountant, and other advisory fees, and all other expenses reasonably incurred in their defense as a result of:
  - (a) any action of the Trustee taken in good faith in accordance with any information, instruction, direction, or opinion given to the Trustee by the Employer, the Plan Administrator, investment manager, or legal counsel of the Employer, or any person or entity appointed by any of them and authorized to give any information, instruction, direction, or opinion to the Trustee;
  - (b) the failure of the Employer, the Plan Administrator, investment manager, or any person or entity appointed by any of them to make timely disclosure to the Trustee of information which any of them or any appointee knows or should know if it acted in a reasonably prudent manner; or
  - (c) any breach of fiduciary duty by the Employer, the Plan Administrator, investment manager, or any person or entity appointed by any of them, other than such a breach which is caused by any failure of the Trustee to perform its duties under this Trust.
- 12.10 <u>Liability of Trustee Duties Limited</u>. The duties and obligations of the Trustee shall be limited to those imposed upon it by this Plan document, the Trust Agreement, applicable law or as subsequently agreed upon by the parties. Responsibility for administrative duties required under the Plan or applicable law not imposed upon or agreed to by the Trustee shall rest with the Plan Administrator and the Employer.
- **12.11** Appointment of Custodian. The Plan Administrator may appoint a Custodian to hold all or any portion of the Plan assets. A Custodian has the powers, rights and responsibilities similar to those of a nondiscretionary trustee unless otherwise provided in a separate agreement.
- 12.12 <u>Modification of Trust Provisions</u>. The Employer may amend the administrative trust or custodial provisions under this Plan (such as provisions relating to investments and the duties of trustees), provided the amended provisions are not in conflict with any other provision of the Plan and do not cause the Plan to fail to qualify under Code § 401(a). The Employer may document any amendment modifying the trust or custodial provisions under this Plan or other overriding language in an Addendum to the Plan.

12.13 12.2 Custodial Accounts, Annuity Contracts, and Insurance Contracts. As provided under Code § 401(f), a custodial account, an annuity contract or a contract issued by an Insurer is treated as a qualified trust under the Plan if (i) the custodial account or contract would, except for the fact that it is not a trust, constitute a qualified trust under Code § 401(a) and (ii) in the case of a custodial account the assets thereof are held by a bank (as defined in Code § 408(n)) or another person who demonstrates to the IRS that the manner in which the assets are held are consistent with the requirements of Code § 401(a). Common/Collective Trusts and Collectibles.

Plan Subject to the DCIC's or the Plan Administrator's direction, the Trustee may commingle, for investment or administration purposes the assets may also be invested (or portion of the assets) of the Trust in any common/collective trust fund, collective investment fund, master trust or in a group trust fund that, but only if the commingled trust satisfies the requirements of under IRS Revenue Ruling 81-100 (as modified by Rev. Rul. 2004-67 and Rev. Rul. 2011-1) to be tax exempt. All of the terms and provisions of any such common/collective trust fund or group trust into which Plan assets are invested are incorporated by reference into the provisions of the Trust for this Plan. No portion of any voluntary, tax deductible Employee contributions being held under the Plan (or any earnings thereon) may be invested in life insurance contracts or, as with any Participant directed investment, in tangible personal property characterized by the IRS as a collectible.

## **SECTION 13**

## PARTICIPANT LOANS

**13.1 Availability of Participant Loans.** In accordance with Code § 72(p), the District may permit Participants to take loans from their vested Account Balance under the Plan.

To receive a Participant loan, a Participant must sign a promissory note along with a pledge or assignment of the portion of the Account Balance used for security on the loan. The loan will be evidenced by a legally enforceable agreement which specifies the amount and term of the loan, and the repayment schedule.

A loan policy shall be established to determine the terms, conditions and restrictions applicable to such loans as well as the time and manner in which loans may be made available. The loan policy will ensure that all loans made under this Plan meet the requirements of Code § 72(p) and its underlying regulations.

- **Must be Available in Reasonably Equivalent Manner**. Participant loans must be made available to Participants in a reasonably equivalent manner. The loan policy may limit the availability of Participant loans to specified events.
- **13.3 Loan Limitations.** A Participant loan may not be made to the extent such loan (when added to the outstanding balance of all other loans made to the Participant) exceeds the lesser of:
  - (a) Fifty-thousand dollars (\$50,000) (reduced by the excess, if any, of the Participant's highest outstanding balance of loans from the Planduring the one-year period ending on the day before the date on which such loan is made, over the Participant's outstanding balance of loans from the Plan as of the date such loan is made) or
  - (b) One-half (½) of the Participant's vested Account Balance, determined as of the Valuation Date coinciding with or immediately preceding such loan, adjusted for any contributions or distributions made since such Valuation Date.

Notwithstanding the above described loan limitation, a Participant may take a loan equal to the greater of \$10,000 or fifty percent (50%) of the Participant's vested Account Balance. However, if a Participant takes a loan in excess of (50%) of the Participant's vested Account Balance, such loan is still subject to the adequate security requirements under Section 13.7 (Adequate Security).

In applying the limitations under this Section 13.3 (Loan Limitations), all plans maintained by the District, including an eligible governmental 457(b) plan, are aggregated and treated as a single plan. In addition, any assignment or pledge of any portion of the Participant's interest in the Plan and any loan, pledge, or assignment with respect to any insurance contract purchased under the Plan will be treated as loan under this Section.

In applying the Loan Limitations described in this Section 13.3, the Participant's Roth Deferral Account and In-Plan Roth Rollover Account will be aggregated with all of the Participant's other Accounts from which loans are permissible.

- **Limit on Amount and Number of Loans.** A Participant may not receive a Participant loan of less than \$1,000. Participant may not have more than two Participant loans outstanding at any time. Notwithstanding the foregoing sentences, the Plan Administrator may amend the minimum loan amount and maximum number of outstanding loans by amending the loan policy.
- 13.5 Participant Must Be Creditworthy. The Plan Administrator may refuse to make a loan to any Participant who is determined to not be creditworthy. For this purpose, a Participant is not creditworthy if, based on the facts and circumstances, it is reasonable to believe that the Participant will not repay the loan. A Participant who has defaulted on a previous loan from the Plan (or on a loan from another plan sponsored by the District) and has not repaid such loan (with accrued interest) will be treated as not creditworthy until such time as the Participant repays the defaulted loan (with accrued interest).
- **Reasonable Rate of Interest.** All Participant loans will be charged a commercially reasonable rate of interest. The interest rate assumptions will be periodically reviewed to ensure the interest rate charged on Participant loans is reasonable.
  - If a Participant is in military service while he/she has an outstanding Participant loan, the applicable interest charged on such loan during the period while the Participant is in military service will not exceed six percent (6%) per year provided the Participant provides written notice and a copy of his/her call-up or extension orders to the Plan Administrator within 180 days following the Participant's termination or release from military service. For this purpose, military service is as defined in the Soldier's and Sailor's Civil Relief Act of 1940 as modified by the Service members Civil Relief Act of 2003. The Participant may voluntarily waive this six percent (6%) interest limitation and the Plan Administrator may petition the court to retain the original interest rate if the ability to repay is not affected by the Participant's activation to military duty.
- Adequate Security. All Participant loans must be adequately secured. Generally, the Participant's vested Account Balance shall be used as security for a Participant loan provided the outstanding balance of all Participant loans made to such Participant does not exceed: -1) fifty percent (50%) of the Participant's vested Account Balance, or 2) the Participant's vested Account Balance, where the Participant seeks a loan

equal to the greater of \$10,000 or fifty percent (50%) of the Participant's vested Account Balance, in either case, as determined immediately after the origination of each loan. The Plan's loan policy shall prescribe rules for defining and obtaining adequate security for loans under the Plan.

- 13.8 Periodic Repayment. A Participant loan must provide for level amortization with payments to be made not less frequently than quarterly. A Participant loan must be repaid in full within a period not exceeding five (5) years from the date the Participant receives the loan from the Plan, unless the loan is for the purchase of the Participant's principal residence, in which case the loan must be repaid within fifteen (15) years or such other period that is commensurate with the repayment period permitted by commercial lenders for similar loans, permissible by the Code § 72(p) and set out in the Plan's loan policy. To the extent a loan is made from a Participant's Roth Deferral Account, the loan repayment requirement must be satisfied separately with respect to that portion of the loan and the portion of the loan from the Participant's other Accounts. The Plan's loan policy shall establish the method loans shall be repaid. The loan policy shall address loan repayments during leave of absence periods in a manner that is consistent with the applicable tax rules.
- 13.9 Procedures for Loan Default. A Participant will be considered to be in default with respect to a loan if any scheduled repayment with respect to such loan is not made by the end of the calendar quarter following the calendar quarter in which the missed payment was due. Once a loan is in default, the total outstanding balance of the loan will be deemed distributed and reported on an IRS Form 1099-R for the calendar year in which the Participant defaulted on the loan. In accordance with the Code, the deemed distribution of a defaulted loan will not be designated as a qualified distribution from a Roth Deferral Account or In-Plan Roth Rollover Account.

If a Participant defaults on a Participant loan, the Plan may not offset the Participant's Account Balance until the Participant is otherwise entitled to an immediate distribution of the portion of the Account Balance which will be offset and such amount being offset is available as security on the loan, pursuant to Section 13.7 (Adequate Security). For this purpose, a loan default is treated as an immediate distribution event to the extent the law does not prohibit an actual distribution of the type of contributions which would be offset as a result of the loan default). The Participant may repay the outstanding balance of a defaulted loan (including accrued interest through the date of repayment) at any time.

Pending the offset of a Participant's Account Balance following a defaulted loan, the following rules apply to the amount in default.

- (a) Interest continues to accrue on the amount in default until the time of the loan offset or, if earlier, the date the loan repayments are made current or the amount is satisfied with other collateral.
- **(b)** A subsequent offset of the amount in default is not reported as a taxable distribution, except to the extent the taxable portion of the default amount was not previously reported by the Plan as a taxable distribution.
- (c) The post-default accrued interest included in the loan offset is not reported as a taxable distribution at the time of the offset.

## 13.10 <u>Termination of Employment</u>.

- (a) <u>Outstanding Loan</u>. The loan policy may provide Participant's with an opportunity to continue to repay the loan after termination of employment. The terms and conditions of repayment upon termination of employment shall be established by the Plan Administrator.
- **Direct Rollover.** Upon termination of employment, a Participant may request a Direct Rollover of the loan note (provided the distribution is an Eligible Rollover Distribution as defined in Section 7.4) to another qualified plan which agrees to accept a Direct Rollover of the loan note. A Participant may not engage in a Direct Rollover of a loan to the extent the Participant has already received a deemed distribution with respect to such loan.
- **Amendment of Plan to Eliminate Participant Loans.** The Plan may be amended at any time to eliminate Participant loans on a prospective basis. However, the elimination of a Participant loan feature may not result in the acceleration of payment of any existing Participant loans, unless the terms of the Participant loan permit such acceleration.

#### **SECTION 14**

## PLAN AMENDMENT, TERMINATION, MERGERS AND TRANSFERS

## 14.1 Plan Amendment.

- **Authority to Amend.** The Employer has the right, at any time and from time to time, to amend the Plan in whole or in part. Any amendment of the Plan by the Employer shall be expressed in an instrument executed by the Employer on the order of its Board of Directors and filed with the Trustee. Further, the Plan Administrator is also authorized, at any time and from time to time, to amend the Plan in whole or in part, to ensure compliance with applicable laws and regulations and further the administration of the Plan. The Plan Administrator may not amend the Plan to effectuate its termination.
  - (1) Amendment Purposes Limited. No Plan amendment may authorize or permit any portion of the assets held under the Plan to be used for or diverted to a purpose other than the exclusive benefit of Participants or Beneficiaries, except to the extent such assets are used to pay taxes or administrative expenses of the Plan. An amendment also may not cause or permit any portion of the assets held under the Plan to revert to or become property of the District.
  - **Method of Amendment**. The Plan Administrator may amend the Plan in writing and without amendment of the MWD Code.
- **Effective Date of Plan Amendments.** If the Plan is restated or amended, such restatement or amendment is generally effective as of the Effective Date of the restatement or amendment except where the context indicates a reference to an earlier Effective Date. The District may designate special effective dates for individual provisions under the Plan.

- required to comply with IRS guidance or law), the provisions of the Plan as amended generally override the provisions of any prior Plan. However, if the provisions of the Plan as amended are different from the provisions of the District's prior plan and, after the retroactive Effective Date of the Plan as amended, the District operates in compliance with the provisions of the prior plan, the provisions of such prior plan are incorporated into the Plan for purposes of determining whether the District operated the Plan in compliance with its terms, provided operation in compliance with the terms of the prior plan do not violate any qualification requirements under the Code, regulations, or other IRS guidance.
- **(2)** Retroactive Effect of PPA, HEART and WRERA Provisions. This Plan is designed to comply with the Code, regulations, and general guidance applicable to qualified retirement plans, including the provisions of the Pension Protection Act of 2006 (PPA), the Heroes Earnings Assistance And Relief Tax Act Of 2008 (HEART Act), and the Worker, Retiree, and Employer Recovery Act of 2008 (WRERA). If this Plan is being restated or amended to comply with the provisions of PPA, HEART and/or WRERA, the Plan contains special effective dates that apply with respect to such provisions. If the Plan is being restated within the remedial amendment period for retroactive compliance with the PPA, HEART and WRERA provisions, the special effective dates for such provisions (as described below) will apply, even if such special effective dates precede the Effective Date of this restatement. Thus, if the Plan is being restated or amended to comply with PPA, HEART and/or WRERA, and the Effective Date of this restatement or amendment is later than the special effective date applicable to any of the PPA, HEART or WRERA provisions described below, such special effective dates will apply and any prior plan being replaced by this Plan will be considered to have been timely amended for the PPA, HEART and WRERA provisions.

The following provisions contain special effective dates for purposes of complying with the requirements of PPA, HEART and WRERA:

- **i. Hardship Distributions**. Section 7.8(e) of this Plan allows Hardship distributions to be determined with respect to primary Beneficiaries.
- **ii. Direct Rollovers By Non-Spouse Beneficiaries.** The provisions allowing for direct rollovers by non- Spouse beneficiaries are effective for distributions made on or after January 1, 2007.
- **Direct Rollover of Non-Taxable Amounts.** Effective for taxable years beginning on or after January 1, 2007, the definition of Eligible Rollover Distribution to include the portion of a distribution that is not includible in gross income.
- **iv.** Rollovers to Roth IRA. For distributions occurring on or after January 1, 2008, Participants or Beneficiaries are permitted to rollover a qualified Eligible Rollover Distribution to a Roth IRA.
- v. <u>Distribution Notice Periods</u>. Effective for Plan Years beginning on or after

- January 1, 2007, the period for providing the Code § 402(f) rollover notice and the period for providing the notice regarding the right to defer receipt of a distribution is increased to 180 days.
- vi. Content of Notice of A Participant's Right to Defer Receipt of A Distribution.

  Effective for Plan Years beginning on or after January 1, 2007, notice relating to a

  Participant's right to defer receipt of a distribution must include a description of the
  consequences of a Participant's decision not to defer the receipt of a distribution.
- **vii.** Qualified Domestic Relations Orders (QDROs). The Plan expands the definition of a QDRO effective April 6, 2007 to include modified orders and orders issued after the Participant's death.
- **Penalty-Free Withdrawals for Individuals Called to Active Duty.** Effective September 11, 2001, Qualified Reservist Distribution may be taken from elective deferrals.
  - ix. Benefit Accruals for Participants on Qualified Military Service. The HEART Act provisions addressing Participants on qualified military leave are effective for Plan Years beginning on or after January 1, 2007.
  - x. <u>Differential Pay.</u> Effective for years beginning on or after January 1, 2009, Section 1.54(c) of this Plan permits the Employer to include Differential Pay as Total Compensation under the Plan.
- **xi. Waiver of Required Minimum Distributions.** Waiver of the Required Minimum Distribution rules for calendar year 2009 as prescribed under WRERA is available.
- **xii. Final Code § 415 Regulations.** Provisions required by the final Code § 415 regulations are effective for Limitation Years beginning on or after July 1,2007.
- (3) Merged Plans. Except for retroactive application of the provisions under this subsection 14.1(d), if one or more qualified retirement plans have been merged into this Plan, the provisions of the merging plan(s) will remain in full force and effect until the Effective Date of the Plan merger(s).
- 14.2 Indemnities. To the extent permitted by law, any employee or former employee, board member or former board member of the District shall be indemnified and held harmless (to the extent not indemnified or held harmless under any liability insurance contract or other indemnification arrangement with respect to the plan) from and against any and all liability to which such person may be subject by reason of any act or omission made in good faith with respect to the administration of the Plan, including all expenses reasonably incurred in that person's defense in the event the District fails to provide such defense after having been requested in writing to do so. The provisions of this Section 14.29.7 shall not be construed to in any way restrict the privileges and immunities that the District and its Employees are otherwise entitled to under Government Code Sections 818.8 and 820.2.

- **14.314.2 Plan Termination**. The District may terminate (partial or full termination) this Plan at any time by Board resolution.
  - **Full and Immediate Vesting.** Upon a full or partial termination of the Plan all amounts credited to an affected Participant's Account become one-hundred percent (100%) vested. The Plan Administrator has discretion to determine whether a partial termination has occurred.
  - (b) <u>Distribution Upon Plan Termination</u>. Upon the termination of the Plan, the Plan Administrator shall direct the distribution of Plan assets to Participants in accordance with the provisions under Section 7 (Plan Distribution). For purposes of applying the provisions of this subsection 14.2(b), distribution may be delayed until the District receives a favorable determination letter from the IRS, or an equivalent, as to the qualified status of the Plan upon termination, provided the determination letter request is made within a reasonable period following the termination of the Plan. Until all Plan assets have been distributed from the Plan, the District must amend the Plan in order to comply with current laws and regulations and may take any other actions necessary to retain the qualified status of the Plan.
  - (c) <u>Missing Participants</u>. Upon termination of the Plan, if any Participant cannot be located after a reasonable diligent search as defined in Section 6 (Participant Vesting and Forfeitures), the Plan Administrator may make a direct rollover to an IRA selected by the Plan Administrator. For this purpose, the Plan Administrator will adopt procedures for making automatic rollovers. An Automatic Rollover under this subsection 14.2(c) may be made on behalf of any missing Participant, regardless of the value of his/her vested Account Balance under the Plan.
  - (d) Partial Termination. In determining whether a Plan has experienced a partial termination as described under Code § 411(d)(3), the Plan Administrator will apply the principals set forth under IRS Revenue Ruling 2007-43.
  - (e) Employee Rights. The exercise of such right by the District to terminate the 401(k) or 457(b)

    Plans does not preclude employees or their representatives from either consulting or meeting and conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to MWD Code Section 6102 (Employee Rights).
- Plan Merger or Consolidation. In the event the Plan is merged or consolidated with another plan, each Participant must be entitled to a benefit immediately after such merger or consolidation that is at least equal to the benefit the Participant was entitled to immediately before such merger or consolidation (had the Plan terminated). If the District amends the Plan from one type of Defined Contribution Plan (e.g., a Profit Sharing Plan) into another type of Defined Contribution Plan (e.g., a Money Purchase Plan) will not result in a partial termination or any other event that would require full vesting of some or all Plan Participants.
- **14.5**14.4 **Transfer of Assets.** The Plan may accept a transfer of assets from another qualified retirement plan on behalf of any Employee, even if such Employee is not eligible to receive other contributions under the Plan. If a transfer of assets is made on behalf of an Employee prior to the Employee's becoming a

Participant, the Employee shall be treated as a Participant for all purposes with respect to such transferred amount. Any assets transferred to this Plan from another plan must be accompanied by written instructions designating the name of each Employee for whose benefit such amounts are being transferred, the current value of such assets, and the sources from which such amounts are derived. The Plan Administrator will deposit any transferred assets in the appropriate Participant's Transfer Account. The Transfer Account will contain any sub-Accounts necessary to separately track the sources of the transferred assets. Each sub-Account will be treated in the same manner as the corresponding Plan Account.

7-11

Any amounts transferred (and any gains attributable to such transferred amounts) continue to be subject to the distribution restrictions applicable to plan assets under the transferor plan. Such amounts may not be distributed for reasons other than death, disability, attainment of Normal Retirement Age, attainment of age  $59\frac{1}{2}$ , or termination of employment, regardless of any distribution provisions under this Plan that would otherwise permit a distribution prior to such events.

The Plan Administrator may refuse to accept a transfer of assets if the Plan Administrator reasonably believes the transfer: (1) is not being made from a proper qualified plan; (2) could jeopardize the tax-exempt status of the Plan; or (3) could create adverse tax consequences for the Plan or the District. Prior to accepting a transfer of assets, the Plan Administrator may require evidence documenting that the transfer of assets meets the requirements of this Section 14.

- **Trustee's Right to Refuse Transfer.** If the assets to be transferred to the Plan under this Section 14.4 are not susceptible to proper valuation and identification or are of such a nature that their valuation is incompatible with other Plan assets, the Trustee may refuse to accept the transfer of all or any specific asset, or may condition acceptance of the assets on the sale or disposition of any specific asset.
- **Transfer of Plan to Unrelated Employer.** The Employer may not transfer sponsorship of the Plan to an unrelated employer if the transfer is not in connection with a transfer of business assets or operations from the Employer to the unrelated employer.

#### **SECTION 15**

#### **MISCELLANEOUS**

- **15.1 Exclusive Benefit.** Plan assets will not be used for, or diverted to, a purpose other than the exclusive benefit of Participants or Beneficiaries.
- 15.2 Return of Employer Contributions. Upon written request by the Employer, the Trustee must return any Employer Contributions provided that the circumstances and the time frames described below are satisfied. Any Employer Contributions made because of a mistake of fact must be returned to the Employer within one year of the contribution. The Trustee may request the Employer to provide additional information to ensure the amounts may be properly returned. Any amounts returned shall not include earnings, but must be reduced by any losses.
- **15.3 Participants' Rights Limited.** The adoption of this Plan by the District does not give any Participant,

Beneficiary, or Employee a right to continued employment with the District and does not affect the District's right to discharge an Employee or Participant at any time. This Plan also does not create any legal or equitable rights in favor of any Participant, Beneficiary, or Employee against the District, Plan Administrator, DCIC, or Trustee. Unless the context indicates otherwise, any amendment to this Plan is not applicable to determine the benefits accrued, or the extent to which such benefits are vested, by a Participant or former Employee whose employment terminated before the effective date of such amendment, except where application of such amendment to the terminated Participant or former Employee is required by statute, regulation or other guidance of general applicability. Where the provisions of the Plan are ambiguous as to the application of an amendment to a terminated Participant or former Employee, the Plan Administrator has the authority to make a final determination on the proper interpretation of the Plan.

#### 15.4 Alienation.

- (a) Except as otherwise provided in subsection (c), no benefit which shall be payable out of the Trust to any person (including a Participant or Beneficiary) shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, or charge, and any attempt to anticipate, alienate, sell, transfer, assign, pledge, encumber, or charge the same shall be void; and no such benefit shall in any manner be liable for, or subject to, the debts, contracts, liabilities, engagements, or torts of any such person, nor shall it be subject to attachment or legal process for or against such person, and the same shall not be recognized by the Trustee, except to such extent as may be required by law.
- (b) In the event a Participant's benefits are garnished or attached by order of any court, the District may bring an action for a declaratory judgment in a court of competent jurisdiction to determine the proper recipient of the benefits to be paid by the Plan. During the pendency of said action, any benefits that become payable shall be paid into the court as they become payable, to be distributed by the court to the recipient it deems proper at the close of said action. Any costs attributable to these actions may be charged to the Participant Account.
- (c) This Section shall not apply to the creation, assignment or recognition of a right to any benefit payable pursuant to a domestic relations order, unless such order is determined to be a QDRO by the Plan Administrator, as defined in Code § 414(p). The Trustee shall be entitled to make distributions to an alternate payee pursuant to a QDRO in accordance with such an Order, without regard to the age or employment status of the Participant.
- 15.5 Receipt and Release for Payments. Any payment to any Participant, Participant's legal representative, Beneficiary, or to any guardian or committee appointed for such Participant or Beneficiary in accordance with the provisions of the Plan, shall, to the extent thereof, be in full satisfaction of all claims hereunder against the Trustee and the District, either of whom may require such Participant, legal representative, Beneficiary, guardian or committee, as a condition precedent to such payment, to execute a receipt and release thereof in such form as shall be determined by the Trustee or District.
- **15.6 Determination**. The District, in case of disputes, may make determinations and findings of fact with

respect to any matter arising in connection with the administration of the Plan. Such determinations and findings shall be final and conclusive, to the extent permitted by the Plan and by law, as to all interested persons for all purposes of the Plan. The District shall instruct the Plan Administrator and Trustee as to any final determination made hereunder with information as reasonably required to administer the Plan in accordance with the final determination.

- 15.7 No Guarantee Against Loss. The District does not guarantee the Participants' Accounts or investments or any part thereof against loss or depreciation. All persons having any interest in the Accounts or investments shall look solely to such sources for payment with respect to such interest. No action by the District shall be considered to be either an endorsement or guarantee of any investment option, nor shall it be considered to attest the financial soundness or the suitability of any investment for the purpose of meeting future obligations provided for in the Plan.
- **15.8 Gender and Number.** Wherever any words are used herein in the masculine, feminine or neuter gender, they shall be construed as though they were also used in another gender in all cases where they would so apply, and whenever any words are used herein in the singular or plural form, they shall be construed as though they were also used in the other form in all cases where they would so apply.
- **15.9** <u>Use of IRS Compliance Programs</u>. Nothing in this Plan document should be construed to limit the District's ability to correct any tax-qualification error using whatever corrective actions or principals permitted under the IRS's Employee Plans Compliance Resolution System.
- **15.10 Governing Law.** The provisions of this Plan shall be construed, administered, and enforced in accordance with the provisions of applicable Federal Law and, to the extent applicable, the laws of the state of California.
- **15.11 Waiver of Notice.** Any person entitled to a notice under the Plan may waive the right to receive such notice, to the extent such a waiver is not prohibited by law, regulation or other pronouncement.
- 15.12 <u>Use of Electronic Media</u>. The District, Plan Administrator, Trustee and any other designated individual responsible for providing applicable notices or disclosures under the Plan, and any Participant or beneficiary making an election under the Plan may use telephonic or electronic media to satisfy any notice requirements required by this Plan. Any use of electronic medium under the Plan must comply with the requirements outlined in Treas. Reg. § 1.401(a)-21 or other general guidance concerning the use of telephonic or electronic media. The Plan Administrator also may use telephonic or electronic media to conduct plan transactions such as enrolling participants, making (and changing) salary reduction elections, electing (and changing) investment allocations, applying for Plan loans, and other transactions, to the extent permissible under regulations (or other generally applicable guidance).
- **15.13 Severability of Provisions.** In the event that any provision of this Plan shall be held to be illegal, invalid or unenforceable for any reason, the remaining provisions under the Plan shall be construed as if the illegal, invalid or unenforceable provisions had never been included in the Plan.
- **15.14 Binding Effect.** The Plan, and all actions and decisions made thereunder, shall be binding upon all applicable parties, and their heirs, executors, administrators, successors and assigns.

- **15.15 Headings**. The headings and subheadings of this Savings Plan have been inserted for convenience of reference and are to be ignored in any construction of the provisions hereof.
- **15.16 Uniformity.** All provisions of this Plan shall be interpreted and applied in a uniform, nondiscriminatory manner.
- of Directors, the General Manager, the Plan Administrator, the DCIC or any member thereof, or any Employees who have administrative responsibility under the Plan with respect to any liability, loss, damage, or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacities in the administration of or management of the assets of the Plan, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.

The amended and restated Plan is hereby adopted, effective <u>December 31, XX XX, 20232</u> January 1, 2024.

Camorma
By:
Katano Kassaine Diane Pitman
Group Manager - Human Resources Assistant General
Manager/Chief Financial Officer and   ✓Interim Plan
<u>Administrator</u>
Date Signed

The Metropolitan Water District of Southern

California

# The Metropolitan Water District of Southern California 401(k) Savings Plan

**January 1, 2024** 

# THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 401(k) SAVINGS PLAN

# ABBREVIATED TABLE OF CONTENTS

BACKGROUNI	)	Page 1
SECTION 1:	PLAN DEFINITIONS	Page 1
SECTION 2:	ELIGIBILITY AND PARTICIPATION	Page 10
SECTION 3:	PLAN CONTRIBUTIONS	Page 12
SECTION 4:	SPECIAL RULES AFFECTING GOVERNMENTAL PLANS	Page 17
SECTION 5:	CONTRIBUTION AND BENEFIT LIMITATIONS	Page 18
SECTION 6:	PARTICIPANT VESTING AND FORFEITURES	Page 22
SECTION 7:	PLAN DISTRIBUTIONS	Page 24
SECTION 8:	REQUIRED MINIMUM DISTRIBUTIONS	Page 35
SECTION 9:	SPOUSAL CONSENT RULES	Page 43
SECTION 10:	PLAN ACCOUNTING AND INVESTMENTS	Page 44
SECTION 11:	PLAN ADMINISTRATION AND OPERATION	Page 47
SECTION 12:	TRUST FUNDS	Page 56
SECTION 13:	PARTICIPANT LOANS	Page 62
SECTION 14:	PLAN AMENDMENT, TERMINATION, MERGERS AND TRANSFERS	Page 65
SECTION 15:	MISCELLANEOUS	Page 69

**Page** 

## THE METROPOLITAN WATER DISTRICT OF SOUTHERN CALIFORNIA 401(k) SAVINGS PLAN

### DETAILED TABLE OF CONTENTS

BACKGROUND......1 **SECTION 1 PLAN DEFINITIONS** 1.1 Account \_\_\_\_\_\_1 Account Balance......2 1.2 1.3 Alternate Payee ......2 1.4 1.5 Cash-Out Distribution......2 1.6 Catch-Up Contributions ......2 1.7 Catch-Up Contribution Limit......2 1.8 1.9 Compensation Limit \_\_\_\_\_\_2 1.10 1.11 1.12 Differential Pay......3 1.13 1.14 1.15 1.16 1.17 1.18 1.19 Eligible Employee......4 1.20 1.21 Employee ......4 1.22 Employer or District......4 1.23 Employer Contributions ......4 1.24 Entry Date \_\_\_\_\_4 1.25 ERISA 4 1.26 Excess Deferrals ......4 1.27 1.28 Hardship......4 1.29 Hours of Service ......4 Performance of Duties ......4 (a) (b) Nonperformance of Duties......4 (c) Back Pay Award......4 (d) Related Employers ......5 Maternity/Paternity Leave......5 1.30 Code § 415 Limitation......5 1.31 Limitation Year ......5 1.32

The Metropolitan Water District of Southern California 401(k) Savings Plan

Effective January 1, 2024

1.33	Matching Contributions	5
1.34	Nonresident Alien Employee	
1.35	Normal Retirement Age (NRA)	5
1.36	Participant	6
1.37	Part-Time Employee	6
1.38	Plan	6
1.39	Plan Administrator	6
1.40	Plan Compensation	6
	(a) Determination Period	6
	(b) Partial Period of Participation	7
1.41	Plan Year	7
1.42	Pre-Tax Deferrals	7
1.43	Qualified Birth or Adoption Distribution (QBOAD)	7
1.44	Qualified Domestic Relations Order (QDRO)	
1.45	Recurrent Employee	
1.46	Rehired Annuitant	7
1.47	Required Beginning Date	7
1.48	Rollover Contribution	
1.49	Roth Deferrals	
1.50	Salary Deferral Election	
1.51	Salary Deferrals	8
1.52	Seasonal Employee	8
1.53	Short Plan Year	8
1.54	Spouse	8
1.55	Total Compensation	8
1.56	Trust	9
1.57	Trust Agreement	
1.58	Trustee	
1.59	Valuation Date	
	SECTION 2	
	ELIGIBILITY AND PARTICIPATION	
2.1	Eligibility	
2.2	Eligible Employees	
	(a) Excluded Employees	
	(b) Only Employees May Participate in the Plan	
	(c) Ineligible Employee Becomes Eligible Employee	
	(d) Eligible Employee Becomes Ineligible Employee	
	(e) Improper Exclusion of Eligible Participant	
2.3	Eligibility Requirements	
	(a) Service Requirement	
	(b) Entry Date	
2.4	Participation on Effective Date of Plan	
2.5	Rehired Employees	11

#### SECTION 3 PLAN CONTRIBUTIONS

3.1	Type	Types of Contributions					
3.2	Salary Deferrals						
	(a)	Salary Deferral Election	12				
	(b)	Change in Deferral Election					
	(c)	Catch-Up Contributions	13				
		(1) Catch-Up Contribution Limit	13				
		(2) Special Treatment of Catch-Up Contributions	13				
3.3	Matcl	hing Contributions					
	(a)	Employee Matching Formula	13				
	(b)	Employer Matching Contributions	13				
3.4	Rollo	ver Contributions	13				
3.5	Roth	Deferrals	14				
3.6	In-Pla	an Roth Conversions	15				
	(a)	Eligibility to Elect In-Plan Roth Rollover	15				
	(b)	In-Plan Roth Rollover Procedure	15				
	(c)	Plan Loans	15				
	(d)	Treatment of In-Plan Roth Rollovers	16				
	(e)	Spousal Consent	16				
	(f)	Tax Treatment of In-Plan Roth Conversion	16				
	(g)	Early Distribution Penalty					
	(h)	In-Plan Roth Conversion Definitions					
		(1) "Direct In-Plan Roth Rollover"					
		(2) "In-Plan Roth Rollover"					
		(3) "In-Plan Roth Rollover Account"	16				
		CECTION 4					
		SECTION 4 SPECIAL RULES AFFECTING GOVERNMENTAL PLANS					
4.1	Gove	rnmental Plan					
	(a)	Governmental Plan Exemptions					
	(b)	ERISA Exempt					
	( )						
		SECTION 5 CONTRIBUTION AND BENEFIT LIMITATIONS					
5.1	Code	§ 415 Limitation	15				
J. 1	(a)	No Other Plan Participation					
	(b)	Participation in Another Plan					
	(6)	(1) This Plan's Code § 415 Limitation					
		(2) Annual Additions Reduction					
		(3) No Annual Additions Permitted					
	(c)	§ 415 Definitions					
	(0)	(1) Annual Additions					
		(2) Defined Contribution Dollar Limitation					
		(3) Excess Amount					
		(4) Limitation Year					
		(5) Maximum Permissible Amount					
		(6) Total Compensation					
		i. Total Compensation Actually Paid or Made Available					
	(d)	Restorative Payments					

	(e)	Corre	ctive Provisions	20			
	(f) Change of Limitation Year						
5.2	Electi	ve Defer	rral Dollar Limit	20			
	(a)	Exces	s Deferrals	20			
	(b)	Corre	ction of Excess Deferrals	20			
		(1)	Amount of Corrective Distribution	21			
		(2)	Allocable Gain or Loss				
		(3)	Taxation of Corrective Distribution				
		(4)	Coordination of Other Provisions				
		(5)	Correction of Excess Deferrals Under Plans Not Maintained by the Employer	22			
			SECTION 6				
6.1	Vocti	ng of Cor	PARTICIPANT VESTING AND FORFEITURES  ntributions	22			
6.2			Benefits				
0.2	(a)		ng Participant or Beneficiary				
	(a)	(1)	Reasonable Diligent Search				
		(2)	Forfeiture of Account of Missing Participant or Beneficiary				
		(3)	Expenses Attributable to Search for Missing Participant				
	(b)		oyer Matching Contributions Attributable to Excess Deferrals	23			
6.3			Forfeitures				
0.0	(a)						
	(b)		ent of Plan Expenses				
			SECTION 7				
			PLAN DISTRIBUTIONS				
7.1			ms of Distribution				
7.2			ble for Distribution				
7.3			onsent				
	(a)		untary Cash-Out Threshold				
	(b)		ripant Notice				
	(c)	-	al Rules				
7.4			ers				
	(a)		itions				
		•	Eligible Rollover DistributionEligible Retirement Plan				
		(2) (3)	Direct Rollover				
	(b)		t Rollover Notice				
	(c)		t Rollover by Non-Spouse Beneficiary				
	(d)		t Rollover of Non-Taxable Amounts				
7.5			Jpon Termination of Employment				
7.0	(a)		int Balance Not Exceeding Cash-Out Threshold				
	(b)		int Balance Exceeding Cash-Out Threshold				
7.6			Jpon Death				
	(a)		After Commencement of Benefits				
	(b)		Before Commencement of Benefits				
	(c)		ng of Distributions				
	(4)		mining a Particinant's Reneficiary	28			

		(1)	Post-Retirement Death Benefit	28				
		(2)	Pre-Retirement Death Benefit					
		(3)	Default Beneficiaries	28				
			i. Prior to August 1, 2018					
			ii. After August 1, 2018					
		(4)	Identification of Beneficiaries					
		(5)	Death of Beneficiary					
		(6)	Divorce From Spouse					
7.7	Distri	bution	to Disabled Employees	30				
7.8	In-Se	rvice Di	stributions	30				
	(a)	Rollo	ver Contributions	30				
	(b)	Matcl	hing Contributions	30				
	(c)	Salar	y Deferrals	30				
	(d)	Quali	fied Reservist Distribution	30				
		(1)	Qualified Reservist Distribution	30				
		(2)	Active Duty	30				
	(e)	Hard	ship Distribution	30				
		(1)	Safe Harbor Hardship Distribution	31				
			i. Immediate and Heavy Financial Need					
			ii. Determination of Whether Distribution Necessary to Satisfy Need					
		(2)	Amount Available for Hardship Distribution					
		(3)	Availability to Terminated Employees					
		(4)	Six-Month Suspension of Salary Deferrals					
	<b>(f)</b>		9½ Distributions					
	(g)	Quali	fied Birth or Adoption Distributions (QBOADs)	33				
	(h)	Coro	navirus-Related Distributions (CRDs)					
		(1)	"Coronavirus-Related Distribution" Defined					
		(2)	Aggregate Dollar Limit					
		(3)	Re-Contribution of CRDs	33				
		(4)	Participant Certification	33				
7.9			stributionstribution					
7.10	Purch	nase of S	Service Credit Under Defined Benefit Plan	34				
7.11	Corre	ection of	Qualification Defects	34				
			gramvov o					
			SECTION 8 REQUIRED MINIMUM DISTRIBUTIONS					
8.1	Reau	ired Mir	nimum Distributions (RMDs)	34				
8.2		Death of Participant Before Required Distributions Begin						
	(a)		viving Spouse as a Designated Beneficiary					
	(b)		Spousal Designated Beneficiary					
	(c)		esignated Beneficiary					
	(d)		of Surviving Spouse					
8.3			nimum Distributions During Participant's Lifetime					
	(a) Amount of Required Minimum Distribution for Each Distribution Calendar Year							
	(b) Lifetime Required Minimum Distributions Continue Through Year of Participant's Death							
8.4			nimum Distributions After Participant's Death					
	(a)		n on After Date Required Distributions Begin					
		(1)	Participant Survived by Designated Beneficiary					
		. ,						

		(2) No Designated Beneficiary	37
	(b)	Death Before Date Required Distributions Begin	37
		(1) Participant Survived by Designated Beneficiary	37
		(2) No Designated Beneficiary	37
		(3) Death of Surviving Spouse Before Distributions to Surviving Spouse	
		are Required to Begin	37
8.5	Defin	nitions	38
	(a)	Designated Beneficiary	38
	(b)	Distribution Calendar Year	
	(c)	Life Expectancy	38
	(d)	Account Balance	38
	(e)	Required Beginning Date (RBD)	38
8.6	Speci	ial Rules	39
	(a)	Election to Allow Participants or Beneficiaries to Elect 5-Year Rule	39
	(b)	Forms of Distribution	39
	(c)	Waiver of Required Minimum Distributions	39
	(d)	Treatment of Trust Beneficiaries As Designated Beneficiaries	39
	(e)	Special Rules Applicable to Trust Beneficiaries	40
		(1) Information that Must be Supplied to Plan Administrator	40
		i. Required Minimum Distribution Before Death Where Spouse is Sole	
		Beneficiary	40
		ii. Required Minimum Distribution After Death	41
		(2) Relief for Discrepancy	41
8.7	Trans	sitional Rule	41
		SECTION 9	
		SPOUSAL CONSENT RULES	
9.1		ication of Joint and Survivor Annuity Rules	
9.2	Spou	sal Consent	42
		SECTION 10	
		PLAN ACCOUNTING AND INVESTMENTS	
10.1		cipant Accounts	
10.2		ation of Accounts	
	(a)	Periodic Valuation	
	(b)	Daily Valuation	
	(c)	Interim Valuations	
10.3		stments to Participant Accounts	
	(a)	Net Income or Loss	
	(b)	Distributions and Forfeitures from A Participant's Account	
	(c)	Dividends	
	(d)	Contributions and Forfeitures Allocated to A Participant's Account	
	(e)	Directed Accounts	
	<b>(f)</b>	Plan Related Fees and Expenses	
10.4		e or Unit Accounting	
10.5		ense Accounts	
10.6		rred Compensation Investment Committee	
107	Inves	stment Policy Statement	4.4

10.7	Participant-Directed Investments4						
10.8	Investment Manager						
10.9	Investment Funds						
10.10	Participant-Directed Investments						
	(a)	Limits	s on Par	ticipant Investment Direction	45		
	(b)	Failur	e to Dir	ect Investment	45		
				SECTION 11			
11.1	Dlan A	dminic	trator	PLAN ADMINISTRATION AND OPERATION	16		
11.1				trator			
11.2	(a)			or Removal of Plan Administrator			
11.3		_		Responsibilities of the Plan Administrator			
11.3	(a)			es, Powers and Responsibilities			
	(a) (b)			f Duties, Powers and Responsibilities			
	(b) (c)			Administrator Responsibilities			
	(d)			n of Plan			
				Filings			
	(e) (f)			Adjustments			
11.4				Aujustinents			
11.5				Expenses			
11.5	(a)			Plan Administration Expenses			
	(a) (b)			e Allocation			
	(c)		-	ated to Administration of Former Employee or Surviving Spouse			
11.6				d Determinationsd			
11.0	(a) Benefit Claims Procedures						
11.7	(b) Benefit Claims Determinations						
11./	(a) In General						
				elated to QDROs			
	(b)			))			
		(1) (2)	-	estic Relations Order			
		(3)		nate Payee			
	(a)			as a QDRO			
	(d)	_	,	· ·			
	(d) (e)	Impo	rmiccihl	QDROle QDRO Provisions	50 50		
	(f)	-		istribution to Alternate Payee			
	(r) (g)			Determination			
	(b)		-	O Procedure			
	(i)		-	ormation			
	(1)	(1)		ications to Participant and Alternate Payee			
		(2)		nate Payee Representative			
		(3)		nate rayee Representative			
		(3)	i.	Separate Accounting			
			i. ii.	Separate Accounting Until the End of 18-Month Period			
			iii.	Preliminary Review			
			iv.	Notification of Determination			
11.8	Milita	rv Servi		Notification of Determination			
-		J • •					

	(a)		fits Under Qualified Military Service				
	(b)		utions				
	(c)	Make-Up Co	ontributions	54			
			SECTION 12				
			TRUST FUNDS				
12.1	Estab	lishment of Tr	ust	54			
12.2			e Trusts54				
			SECTION 13				
			PARTICIPANT LOANS				
13.1		•	cipant Loans				
13.2			n Reasonably Equivalent Manner				
13.3							
13.4			nd Number of Loans				
13.5			creditworthy				
13.6			Interest				
13.7 13.8	_	-					
13.8		1 0	t				
			n Defaultoloyment				
13.10	(a)		g Loan				
	(a) (b)		yer				
13.11			to Eliminate Participant Loans				
			SECTION 14				
		PLA	N AMENDMENT, TERMINATION, MERGERS AND TRANSFERS				
14.1	Plan A			63			
	(a)						
		(1) Ame	endment Purposes Limited	63			
		<b>(2)</b> Met	hod of Amendment	63			
	(b)	Effective Da	te of Plan Amendments	63			
			oactive Effective Date				
		<b>(2)</b> Retr	oactive Effect of PPA, HEART and WRERA Provisions				
		i.	Hardship Distributions				
			Direct Rollovers by Non-Spouse Beneficiary				
		iii.	Direct Rollover of Non-Taxable Amounts				
		iv.	Rollovers to Roth IRA				
		v.	Distribution Notice Periods				
		vi.	Content of Notice of A Participant's Right to Defer Receipt of A Distribution				
		vii.	Qualified Domestic Relations of Orders				
		viii.					
		ix.	Benefit Accruals for Participants on Qualified Military Service				
		х.	Differential Pay				
		xi.	Waiver of Required Minimum Distributions				
		Xii.	Final Code § 415 Regulationsged Plans				
14.2	Dlan 7	` '	ged Plans				
17.4	(a)		mediate Vesting				
	(u)	i an and mil	11041460 4 004116				

	(b) Distribution Upon Plan Termination	65
	(c) Missing Participants	
	(d) Partial Termination	
	(e) Employee Rights	
14.3	Plan Merger or Consolidation	
14.4	Transfer of Assets	
	(a) Trustee's Right to Refuse Transfer	
	(b) Transfer of Plan to Unrelated Employer	
	SECTION 15	
	MISCELLANEOUS	
15.1	Exclusive Benefit	
<b>15.2</b>	Return of Employer Contributions	
<b>15.3</b>	Participants' Rights Limited	
<b>15.4</b>	Alienation	
15.5	Receipt and Release for Payments	
<b>15.6</b>	Determination	68
<b>15.7</b>	No Guarantee Against Loss	68
<b>15.8</b>	Gender and Number	68
15.9	Use of IRS Compliance Programs	68
<b>15.10</b>	Governing Law	68
15.11	Waiver of Notice	68
15.12	Use of Electronic Media	69
<b>15.13</b>	Severability of Provisions	69
<b>15.14</b>	Binding Effect	69
15.15	Headings	69
15.16	Uniformity	69
1517	Indomnification	60

#### BACKGROUND

The Metropolitan Water District of Southern California (the "District") Board of Directors' Minute Item 35632 established a profit-sharing plan with a cash or deferral arrangement known as The Metropolitan Water District of Southern California Savings Plan ("Savings Plan I") effective July 1, 1985. By Board of Directors' Minute Item 35816, the District also established another profit-sharing plan with a cash or deferral arrangement known as "Savings Plan II" applicable to a separate group of employees effective October 1, 1985.

As a result of Board of Directors' Minute Item 46592, "Savings Plan I" was merged into "Savings Plan II" and renamed The Metropolitan Water District of Southern California Consolidated Savings Plan (the "Plan") effective January 1, 2006, and incorporated into Division VI, Chapter 7, Article 6 of The Metropolitan Water District of Southern California Administrative Code ("MWD Code"). The Plan has been amended and restated from time to time. The Plan was most recently restated and renamed The Metropolitan Water District of Southern California 401(k) Savings Plan effective August 1, 2018. The Plan was further amended and restated effective January 1, 2020. The Plan is hereby amended and restated effective January 1, 2024, except as otherwise stated herein.

The Plan is a tax-qualified governmental defined contribution plan under Internal Revenue Code (the "Code") §§ 401(a) and 414(d) that is a profit sharing plan pursuant to Code § 401(a)(27). The Plan has a grandfathered cash or deferral arrangement under Code § 401(k).

All contributions made to the Plan, all property and rights purchased with such amounts and all income attributable to such amounts shall be held in trust for the exclusive benefit of Participants, Beneficiaries and Alternate Payees under the Plan in accordance with Internal Revenue Code §401(a)(2). Prior to the satisfaction of all liabilities with respect to employees and their beneficiaries under the Plan's Trust, it shall be impossible for any part of the corpus or income of the Trust to be (within the taxable year or thereafter) used for, or diverted to, purposes other than for the exclusive benefit of the District's employees or their beneficiaries.

The purpose of the Plan is to attract and retain certain officers and employees to The Metropolitan Water District of Southern California.

#### **SECTION 1**

#### PLAN DEFINITIONS

This Section 1 contains definitions for common terms that are used throughout the Plan. All capitalized terms under the Plan are defined in this Section 1 or in the relevant section of the Plan document where such term is used.

- **1.1 Account.** The separate Account established and maintained for each Participant, Beneficiary or Alternate Payee under the Plan. A Participant Account may have any or all of the following separate sub-Accounts:
  - Pre-Tax Salary Deferral Account
  - Roth Deferral Account
  - Matching Contribution Account

- Rollover Contribution Account
- Roth Rollover Account
- Transfer Account

The Plan Administrator may establish other Account types, as it deems necessary, for the proper administration of the Plan.

- **1.2 Account Balance.** Account Balance shall mean a Participant's balances in all of the Accounts maintained by the Plan on his or her behalf.
- **1.3 Alternate Payee.** A person designated as having a right to receive all or a portion of the Participant's benefit pursuant to a QDRO.
- **Beneficiary.** A person, including a trust, designated by the Participant (or in the absence of a proper beneficiary designation, by the terms of the Plan) to receive a benefit under the Plan upon the death of the Participant. See Section 7.6(d) (Determining a Participant's Beneficiary) for the applicable rules for determining a Participant's Beneficiaries under the Plan.
- **1.5** <u>Cash-Out Distribution</u>. A total distribution made to a Participant who incurs a termination from employment.
- **1.6** <u>Catch-Up Contributions</u>. Salary Deferrals that may be made in excess of an otherwise applicable IRS Code § 402(g) limit by a Participant who attains age 50 or over by the end of the applicable Plan Year.
- **1.7** Catch-Up Contribution Limit. The annual limit applicable to Catch-Up Contributions as set forth in Section 3.2(c)(1)(Catch-Up Contributions).
- **1.8 Code**. The Internal Revenue Code of 1986, as amended.
- **1.9** <u>Collectively Bargained Employee</u>. An Employee who is included in a unit of Employees covered by a collective bargaining agreement between the District and Employee representatives and whose retirement benefits are subject to good faith bargaining.
- Year for purposes of determining a Participant's Plan Compensation. For Plan Years beginning on or after January 1, 1994, and before January 1, 2002, the Compensation Limit taken into account for determining benefits provided under the Plan for any Plan Year is \$150,000, as adjusted for increases in cost-of-living in accordance with Code § 401(a)(17)(B). For any Plan Years beginning on or after January 1, 2002, the Compensation Limit is \$200,000, as adjusted for cost-of-living increases in accordance with Code § 401(a)(17)(B). In determining the Compensation Limit for any applicable period (the "Plan Year"), the cost-of-living adjustment in effect at the beginning of the calendar year applies to any Plan Year that begins within such calendar year.

If a Plan Year consists of fewer than 12 months, the Compensation Limit for such period is an amount equal to the otherwise applicable Compensation Limit multiplied by a fraction, the numerator of which is the number of months in the short determination period, and the denominator of which is 12. A Plan Year will not be considered to be less than 12 months merely because compensation is taken into account only for

the period the Employee is a Participant. If Salary Deferrals or Matching Contributions are separately determined on the basis of specified periods within the determination period (e.g., on the basis of payroll periods), no proration of the Compensation Limit is required with respect to such contributions.

- **1.11 Deferred Compensation Investment Committee or DCIC.** The committee described in Section 10.6.
- 1.12 <u>Defined Contribution Plan</u>. A plan that provides for individual Accounts for each Participant to which all contributions, forfeitures, income, expenses, gains and/ or losses under the Plan are credited or deducted. A Participant's benefit under a Defined Contribution Plan is based solely on the fair market value of his/her vested Account Balance.
- **1.13** <u>Differential Pay</u>. Certain payments made by the District to an individual while the individual is performing service in the Uniformed Services.
- **1.14 Directed Account.** That portion of Participant's Account that is subject to Participant direction of investment and which is maintained by the Plan Administrator as a Directed Account.
- **1.15 Distribution Calendar Year.** A calendar year for which a minimum distribution under Code § 409(a)(9) is required.
- **1.16 Distribution Starting Date.** The date a Participant commences distribution from the Plan. If a Participant commences distribution with respect to a portion of his/her Account Balance, a separate Distribution Starting Date applies to any subsequent distribution.
- **1.17** Effective Date. The Plan was originally established and effective July 1, 1985. The Plan has since been amended and restated November 18, 2014. The current Effective Date of the Plan as amended and restated is August 1, 2018. See Section 14.1(bd) (Effective Date of Plan Amendments) for special rules concerning the retroactive effective date of provisions under the Plan designed to comply with the requirements of the Pension Protection Act of 2006 (PPA).
- **1.18** Elective Deferral Dollar Limit. The maximum amount of Elective Deferrals a Participant may make for any calendar year.
- **1.19** Elective Deferrals. A Participant's Elective Deferrals is the sum of all Salary Deferrals made by the Participant. Elective Deferrals shall not include any amounts properly distributed as an Excess Amount under Code § 415.
- **1.20** Eligible Employee. An Eligible Employee is an Employee who (i) is not excluded under Section 2.2 of the Plan or excluded by agreement, and (ii) meets the eligibility requirements in Section 2.3 of the Plan.
- **1.21 Employee**. An Employee is any individual employed by the District as a common law employee. An independent contractor is not treated as an Employee of the recipient organization for purposes of this Plan or the District.
- **1.22 Employer or District**. Employer or District means The Metropolitan Water District of Southern California ("District"), a political subdivision.
- **1.23 Employer Contributions.** Contributions the Employer makes pursuant to Sections 3.2 (Salary Deferrals) and 3.3 (Matching Contributions).

- **1.24 Entry Date.** The commencement of the first payroll period following the date on which an Eligible Employee satisfies the eligibility requirements described in Section 2 (Eligibility and Participation).
- **1.25 ERISA.** The Employee Retirement Income Security Act of 1974, as amended.
- **1.26 Excess Amount**. Amounts which exceed the Code § 415 Limitation.
- **1.27 Excess Deferrals.** Elective Deferrals that exceed the Elective Deferral Dollar Limit.
- **1.28 Hardship.** A heavy and immediate financial need which meets the requirements of Section 7.8(e) (Hardship Distributions).
- **1.29 Hour of Service.** Each Employee of the District will receive credit for each Hour of Service he/she works for purposes of applying the eligibility rules under the Plan. An Employee *will not* receive credit for the same Hour of Service under more than one category listed below:
  - (a) Performance of Duties. Hours of Service include each hour for which an Employee is paid, or entitled to payment, for the performance of duties for the District. These hours will be credited to the Employee for the computation period in which the duties are performed. In the case of Hours of Service to be credited to an Employee in connection with a period of no more than 31 days which extends beyond one computation period, all such Hours of Service may be credited to the first computation period or the second computation period. Hours of Service under this subsection (a) must be credited consistently for all Employees within the same job classifications.
  - (b) Nonperformance of Duties. Hours of Service include each hour for which an Employee is paid, or entitled to payment, by the District on account of a period of time during which no duties are performed (irrespective of whether the employment relationship has terminated) due to vacation, holiday, illness, incapacity (including disability), layoff, jury duty, military duty or leave of absence. The purpose of this subsection (b) is to prevent a break in service occurring due to the leave of absence. Thus, no more than the required 501 hours of service needed to prevent a break in service will be credited under this paragraph.
  - **Back Pay Award.** Hours of Service include each hour for which back pay, irrespective of mitigation of damages, is either awarded or agreed to by the District. The same Hours of Service *will not* be credited both under subsection (a) or subsection (b), as the case may be, and under this subsection (c). These hours will be credited to the Employee for the Plan Year to which the award or agreement pertains rather than the Plan Year in which the award, agreement or payment is made.
  - (d) Related Employers. Hours of Service will be credited for employment with any Related Employer. A Related Employer includes all members of a controlled group, all commonly controlled entities or affiliated service groups of which the Employer is a part, and any other entity required to be aggregated with the Employer pursuant to regulations under Code §414(o). For purposes of Hours of Service, the Employer and any Related Employers are treated as a single Employer, unless specifically stated otherwise.
  - (e) <u>Maternity/Paternity Leave</u>. Solely for purposes of determining eligibility, an individual who is

absent from work for maternity or paternity reasons *will not* be deemed to have had a break in continuous hours of service by reason of an absence from work for maternity or paternity reasons. For purposes of this paragraph, an absence from work due to maternity or paternity reasons means an absence:

- by reason of the pregnancy of the individual,
- by reason of birth of a child of the individual,
- by reason of the placement of a child with the individual in connection with the adoption of such child by such individual, or
- for purposes of caring for such child for a period beginning immediately following such birth or placement.
- **1.30** <u>Code § 415 Limitation.</u> The limit on the amount of Annual Additions a Participant may make under the Plan during any Limitation Year.
- **Leased Employee**. An individual who performs services for the District pursuant to an agreement between the District and a leasing organization, and who satisfies the definition of a Leased Employee under Code § 414(n).
- **1.32** <u>Limitation Year</u>. The measuring period for determining whether the Plan satisfies the Code § 415 Limitation.
- **1.33 Matching Contributions.** Matching Contributions are contributions made by the District on behalf of a Participant on account of Salary Deferrals made by the Participant under this Plan.
- 1.34 Nonresident Alien Employee. An Employee is a nonresident alien if he or she is neither a citizen of the United States nor a resident of the United States for U.S. tax purposes (as defined in Code § 7701(b)), and who does not have any earned income (as defined in Code § 911) for the District that constitutes U.S. source income (within the meaning of Code § 861). If a nonresident alien Employee has U.S. source income, he/she is treated as satisfying this definition if all of his/her U.S. source income from the District is exempt from U.S. income tax under an applicable income tax treaty.
- **1.35 Normal Retirement Age (NRA)**. For purposes of in-service distributions, Normal Retirement Age is 59½.
- 1.36 Participant. A Participant is an Eligible Employee who has satisfied the conditions for participating under Section 2.3 (Eligibility Requirement) of this Plan. A Participant also includes any Employee (or former Employee) who has an Account Balance under the Plan, including an Account Balance derived from a rollover or transfer from another qualified plan or IRA. A Participant is entitled to share in an allocation of contributions or forfeitures under the Plan for a given year only if the Participant is an Eligible Employee as defined in Section 2.2, and satisfies the allocation conditions set forth in Section 3.3 (Matching Contributions).

An Eligible Employee is treated as a Participant with respect to Salary Deferrals once the Employee has satisfied the eligibility conditions and has reached his or her Entry Date for making such contributions, even if the Eligible Employee chooses not to actually make such contributions to the Plan. An Eligible

Employee is treated as a Participant with respect to Matching Contributions once the Eligible Employee has satisfied the eligibility conditions and has reached his or her Entry Date, even if the Employee does not receive a Matching Contribution because of the Employee's failure to make contributions eligible for the Matching Contribution.

- **1.37 Part-Time Employee.** A Part-Tim Employee is an Employee who is normally scheduled to work on average 20 to 39 hours per week.
- **1.38 Plan.** Plan means The Metropolitan Water District of Southern California 401(k) Savings Plan.
- **1.39 Plan Administrator.** The Plan Administrator is the General Manager or other person or entity designated by the District to be responsible for the administration and operation of the Plan. A Plan Administrator also includes a Qualified Termination Administrator ("QTA") that assumes the responsibilities of Plan Administrator.
- **Plan Compensation.** Plan Compensation is Total Compensation except as modified herein. Plan Compensation includes all cash compensation payable by the District during the Plan Year to, or for the benefit of, an Employee in return for services and any deemed compensation under Code § 414(u)(7). Plan Compensation also includes payments to disabled Participants described in Section 1.54(d) if the payments are not the result of a leave donation(s).

Effective as of January 1, 1997, Plan Compensation shall include pre-tax contributions to a cafeteria plan under Code § 125, an eligible deferred compensation plan under § 457, employer pick-up contributions under § 414(h)(2) or elective contributions that are not included in the Employee's gross income as a qualified transportation fringe benefit under 132(f)(4).

Plan Compensation also does not include other fringe benefits such as medical credit, cell phone allowances, or professional development, wellness, and tuition reimbursements.

Nor does Plan Compensation include payments for leave donations (e.g., hours donated by Employees to a Participant) or payments made by Third Party Providers (TPAs). Plan Compensation also excludes all amounts earned with a related employer who is not a participating employer in this Plan.

- **Determination period.** Plan Compensation is determined based on the Plan Year.
- **Partial period of participation.** Even if an Employee is a Participant for only part of a Plan Year, Plan Compensation shall be determined over the entire Plan Year. In determining whether an Employee is a Participant for purposes of applying this subsection (b), the Employee's status will be determined solely with respect to the contribution type for which the definition of Plan Compensation is being determined. Plan Compensation does not include any amounts earned for any period while an individual is not an Eligible Employee.

In no case may Plan Compensation for any Participant exceed the Compensation Limit.

- **1.41 Plan Year.** The calendar year.
- **1.42 Pre-Tax Deferrals.** Pre-tax Deferrals are a Participant's Salary Deferrals that are not includible in the Participant's gross income at the time deferred.

- 1.43 Qualified Birth or Adoption Distribution (QBOAD). A distribution to a Participant if made (1) during the one-year period beginning on the date on which a child of the Participant is born or on which the legal adoption by the Participant of an eligible adoptee is finalized, (2) with the respect to a child born or eligible adoptee the Participant includes the name, age, and the Taxpayer Identification Number of the child born or eligible adoptee on the Participant's tax return for the taxable year in which the distribution is made, and (3) only to the extent such distribution from all plans maintained by the Employer to such Participant does not exceed \$5,000, in the aggregate, for each child born or eligible adoptee. An eligible adoptee is any individual (other than a child of the Participant's spouse) who has not attained age 18 or is physically or mentally incapable of self-support.
- **Qualified Domestic Relations Order (QDRO).** A domestic relations order that provides for the payment of all or a portion of the Participant's benefits to an Alternate Payee and satisfies the requirements under Code § 414(p).
- **1.45 Recurrent Employee.** Employees hired for an indefinite period of time on an irregular basis, such as intermitted, emergency, or on-call.
- **1.46** Rehired Annuitant. A rehired Employee who meets the definition of retired annuitant in California Government Code § 7522.56.
- **1.47 Required Beginning Date.** The date by which minimum distributions must commence under the Plan.
- **1.48 Rollover Contribution**. A contribution made by an Employee to the Plan attributable to an Eligible Rollover Distribution from another qualified plan or IRA.
- **Roth Deferrals.** Deferral contributions that are designated irrevocably by the Participant at the time of the cash or deferred election as a Roth Deferral that is being made in lieu of all or a portion of the Pre-Tax Deferrals the Participant is otherwise eligible to make under the Plan and treated by the Employer as includible in the Participant's income at the time the Participant would have received that amount in cash if the Participant had not made a cash or deferred election.
- **1.50** Salary Deferral Election. An agreement between a Participant and the District, whereby the Participant elects to have a specific amount withheld from his/her Plan Compensation and the District agrees to contribute such amount into the Plan.
- 1.51 Salary Deferrals. Amounts contributed under the Plan at the election of the Participant, in lieu of cash compensation, which are made pursuant to a Salary Deferral Election or other deferral mechanism. Salary Deferrals may consist of Pre-Tax Deferrals and/or Roth Deferrals. Salary Deferrals shall not include any amounts properly distributed as an Excess Amount under Code § 415. An Employee's Salary Deferrals are treated as employer contributions for all purposes under this Plan, except as otherwise provided under the Code or U.S. Treasury regulations.
- **1.52 Seasonal Employee.** An Employee who normally works on a full-time basis less than five months during any Plan Year.

- **1.53 Short Plan Year.** Any Plan Year that is less than 12 months long, either because of the amendment of the Plan Year, or because the Effective Date of a new Plan is less than 12 months prior to the end of the first Plan Year.
- 1.54 Spouse. Subject to any additional federal tax guidance from the Internal Revenue Service or other agency or court, a Spouse is any individual who is lawfully married to the Participant under a state or foreign jurisdiction, without regard to the location of the Employer or the state where the Participant and Spouse are domiciled. Notwithstanding the previous sentence, a former Spouse of the Participant will be treated as the Spouse or surviving Spouse to the extent provided under a valid QDRO and any current Spouse will not be treated as the Spouse or surviving Spouse with respect to any portion of Participant's benefit covered by a valid QDRO.
- **1.55 Total Compensation.** A Participant's compensation for services with the District, as defined herein:
  - (a) Wages within the meaning of Code § 3401(a) and all other payments of compensation to an Employee by the District (in the course of the District's trade or business) for which the Districtis required to furnish the Employee a written statement under Code §§ 6041(d), 6051(a)(3), and 6052, determined without regard to any rules under Code § 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed.
  - (b) Includes Elective Deferrals (as defined in Section 1.20), elective contributions to a cafeteria plan under Code § 125 or to an eligible deferred compensation plan under Code § 457, employer pick-up contributions under Code § 414(h)(2), and elective contributions that are not includible in the Employee's gross income as a qualified transportation fringe under Code § 132(f)(4).
  - (c) Effective for the first Plan Year beginning on or after July 1, 2007, Total Compensation includes compensation paid after an Employee severs employment with the Employer, provided the compensation is paid by the later of 2½ months after severance from employment with the Employer maintaining the Plan or the end of the Plan Year that includes such date of severance from employment. For this purpose, compensation paid after severance of employment may only be included in Total Compensation to the extent such amounts would have been included as compensation if they were paid prior to the Employee's severance from employment. The following amounts paid after a Participant's severance of employment are included in Total Compensation:
    - (1) Regular Pay. Compensation for services during the Employee's regular working hours, or compensation for services outside the Employee's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments; and
    - (2) <u>Unused Leave Payments</u>. Payment for unused accrued bona fide sick, vacation, or other leave, but only if the Employee would have been able to use the leave if employment had continued.

All other post-severance payments are not included in the definition of Total Compensation, even if such amounts are paid within the time period described in this subsection (c).

- (d) Total Compensation includes post-severance payments to a Participant who is permanently and totally disabled as defined in IRS Code § 22(e)(3) provided that contributions made with respect to amounts treated as compensation under this subsection (d) are nonforfeitable when made.
- **Differential Pay.** Effective for years beginning on or after January 1, 2009, in the case of an individual who receives Differential Pay from the District:
  - (1) such individual will be treated as an Employee of the Employer making the payment, and
  - the Differential Pay shall be treated as wages and will be included in calculating an Employee's Total Compensation under the Plan.

If all Employees performing service in the Uniformed Services are entitled to receive Differential Pay on reasonably equivalent terms and are eligible to make contributions based on the payments on reasonably equivalent terms, the Plan shall not be treated as failing to meet the requirements of any provision described in Code § 414(u)(1)(C)(Treatment of Certain Contributions Made Pursuant to Veteran's Reemployment Rights) by reason of any contribution or benefit based on Differential Pay.

For purposes of this subsection (e), Differential Pay means any payment which is made by an Employer to an individual while the individual is performing service in the Uniformed Services while on active duty for a period of more than 30 days, and represents all or a portion of the wages the individual would have received from the Employer if the individual were performing services for the Employer. In applying the provisions of this subsection (e), Uniformed Services are services as described in Code § 3401(h)(2)(A). The Trust is the separate funding vehicle under the Plan.

- **1.56 Trust**. The trust established pursuant to the Trust Agreement between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets.
- **1.57 Trust Agreement.** The separate agreement entered into by and between the District and the Trustee to hold, administer and distribute all or a portion of the Plan's assets.
- **1.58 Trustee.** The person(s) or entity, and any successors thereto, who is appointed by the District to hold, administer and distribute all or a portion of the Plan assets held in the Trust pursuant to the terms and conditions of the Trust Agreement.
- **1.59 Valuation Date.** The date or dates upon which Plan assets are valued. Plan assets are typically valued daily and will be valued at least annually as of the last day of each Plan Year.

#### **SECTION 2**

#### **ELIGIBILITY AND PARTICIPATION**

**Eligibility.** In order to participate in the Plan, an Employee must be an Eligible Employee (as defined in Section 2.2) and must satisfy the eligibility requirements specified in Section 2.3. Once an Eligible Employee satisfies the Plan's eligibility requirements, such Eligible Employee shall become a Plan Participant as of the Entry Date specified in Section 2.3(b). An Employee who meets the eligibility requirements set forth in Section 2.3 but is not an Eligible Employee, will be eligible to participate in the Plan only upon becoming an Eligible Employee. For purposes of determining eligibility to make Salary

Deferrals, an Eligible Employee will be deemed to commence participation on a timely basis if he or she is permitted to commence making Salary Deferrals as soon as administratively feasible after satisfying the eligibility conditions under the Plan.

- **Eligible Employees.** Except as specified in subsection 2.2(a) and (b), all Employees of the District are Eligible Employees.
  - **Excluded Employees.** An Employee is not an Eligible Employee and may not participate under the Plan if he or she is classified as a:
    - (1) Rehired Annuitant.
    - (2) Nonresident Alien Employee.
    - (3) Leased Employee.
    - (4) Seasonal Employee working less than 5 months in a Plan Year.
    - **(5)** Member of the Carpenter's Union Trust.
    - (6) Intern working less than 20 hours a week and less than 5 months in a Plan Year.
  - (b) Only Employees May Participate in the Plan. To participate in the Plan, an individual must be an Employee. If an individual is not an Employee (e.g., the individual performs services with the District as an independent contractor) such individual may not participate under the Plan. If an individual who is classified as a non-Employee is later determined by the District or by a court or other government agency to be an Employee of the District, the reclassification of such individual as an Employee will not create retroactive rights to participate in the Plan. For periods prior to the date of such final determination, the reclassified Employee will not have any rights to accrued benefits under the Plan, except as agreed to by the District or mandated by a court or government agency, or as set forth in an amendment adopted by the District.
  - (c) Ineligible Employee Becomes Eligible Employee. If an Employee changes status from an ineligible Employee to an Eligible Employee, such Employee will become a Participant immediately on the date he/she changes status to an Eligible Employee, provided the Employee has satisfied the eligibility requirements in Section 2.3 of the Plan and has passed his or her Entry Date that would otherwise have applied had the Employee been an Eligible Employee. If the Employee's original Entry Date (determined as if the Employee was always an Eligible Employee) has not passed as of the date the Employee becomes an Eligible Employee, the Employee will become a Participant upon the occurrence of such Entry Date. This requirement is deemed satisfied with respect to Salary Deferrals if the Employee is permitted to commence making Salary Deferrals under the Plan as soon as administratively feasible after the Employee becomes an Eligible Employee. If an ineligible Employee has not satisfied the eligibility requirement(s) in Section 2.3 at the time such Employee becomes an Eligible Employee, such Employee will become a Participant on the appropriate Entry Date following satisfaction of the eligibility requirements in Section 2.3 of the Plan.

- **Eligible Employee Becomes Ineligible Employee.** If an Employee ceases to qualify as an Eligible Employee (i.e., the Employee changes status from an eligible class to an ineligible class of Employees), such Employee will immediately cease to participate in the Plan. If such Employee should subsequently become an Eligible Employee, he/she will be able to participate in the Plan in accordance with subsection (b) above.
- **Improper Exclusion of Eligible Participant.** If the Plan improperly excludes a Participant who has satisfied the requirements under this Section 2 for participating under the Plan, the District may take **reasonable** action to correct such error, provided such corrective action is consistent with the requirements of the Internal Revenue Service's Employee Plans Compliance Resolution System (EPCRS) program, as amended, or other successor program.

#### 2.3 <u>Eligibility Requirements</u>.

- **Service Requirement.** There is no minimum age requirement to become an Eligible Employee. An Eligible Employee must complete 1,044 continuous Hours of Service to participate in the Plan.
- **Entry Date:** An Eligible Employee will enter the Plan and begin Salary Deferral contributions and receive Matching Contributions as of the later of commencement of the first full payroll period following Eligible Employee's satisfaction of the eligibility requirements specified in Section 2.3(a) or submitting a completed deferral election form in the manner prescribed by the Plan Administrator.
- **Participation on Effective Date of Plan.** An Eligible Employee who has satisfied the Plan's eligibility requirements and reached his/her Entry Date described in Section 2.3 as of the Effective Date of the Plan will be eligible to participate in the Plan as of such Effective Date. If an Eligible Employee has satisfied the minimum eligibility requirements as of the Effective Date in Section 2.3 of the Plan but has not yet reached his/her Entry Date, the Eligible Employee will participate upon reaching his or her Entry Date.
- 2.5 Rehired Employees. If a terminated Employee is subsequently rehired, such Employee will be eligible to participate in the Plan on his/her reemployment date, if the Employee is rehired within a 12-month period and is an Eligible Employee and the Employee had satisfied the Plan's eligibility requirement(s) prior to his/her termination of employment. If a rehired Employee had not satisfied the Plan's eligibility requirement(s) prior to termination of employment, or was rehired more than 12 months after termination, such Employee is eligible to participate in the Plan on the appropriate Entry Date following satisfaction of the eligibility requirements under Section 2.1.

#### **SECTION 3**

#### PLAN CONTRIBUTIONS

- **Types of Contributions.** The Plan provides for Salary Deferrals and Matching Contributions.
- **Salary Deferrals.** Participants may make Salary Deferrals under the Plan. A Participant may elect to change, discontinue and/or resume Salary Deferrals in accordance with Section 3.2(b). Subject to the

conditions and limitations of the Plan, a Participant's election shall remain in effect until any change, suspension, or termination properly elected by the Participant under Section 3.2(b) becomes effective. The amounts to be deferred shall be withheld from the Participant's Plan Compensation and contributed to the Plan on the Participant's behalf. The District will deposit any amounts withheld from a Participant's Plan Compensation as Salary Deferrals into the Participant's Salary Deferral Account under the Plan. Effective January 1, 2024, a Participant may, at the time of his or her Salary Deferral Election or election change, irrevocably designate all or a portion of his or her future Salary Deferrals, including Catch-Up Contributions, as Roth Deferrals instead of Pre-Tax Deferrals.

- (a) Salary Deferral Election. In order to make Salary Deferrals under the Plan, a Participant must enter into a Salary Deferral Election which authorizes the District to withhold a specific dollar amount or a specific percentage from the Participant's Plan Compensation. A Participant's election to make a Salary Deferral to the Plan must be made in accordance with the procedures and time requirements established by the Plan Administrator. A Participant's election must irrevocably designate the Salary Deferrals as Pre-Tax Deferrals, Roth Deferrals, or a specific combination of both. A Participant may defer any amount up to one-hundred percent (100%) of Plan Compensation, provided however, the amount deferred never exceeds the Elective Deferral Dollar Limit imposed by Code § 402(g) and the Limit on Annual Additions under Code § 415. In determining the amount to be withheld from a Participant's Plan Compensation, a Salary Deferral Election may be rounded to the next highest or lowest whole dollar amount. A Salary Deferral Election may only relate to Plan Compensation that is not currently payable or made available at the time the Salary Deferral Election is completed. Further, a Salary Deferral Election may not be effective prior to the later of the date the:
  - (1) Employee becomes a Participant, or
  - (2) Participant executes the Salary Deferral Election.

In addition, Salary Deferrals made pursuant to a Salary Deferral Election may not be made earlier than the date the Participant performs the services to which such Salary Deferrals relate or the date the compensation subject to such Salary Deferral Election would be currently available to the Participant absent the deferral election (if earlier).

- **Change in Deferral Election.** Participants may enter into new Salary Deferral Elections, modify or terminate existing Salary Deferral Elections within the limits established by the Plan Administrator. Any election to modify, including changing the designation of all or any portion of his or her Salary Deferrals as Pre-Tax Deferrals or Roth Deferrals, or terminate a Salary Deferral Election will take effect the first pay date of the following pay period which does not include the date of the election or modification (For example, for an election made on May 14, 2018, the salary deferrals will commence on the next pay date of June 13, 2018, which is for the following pay period of May 20 June 2).
- **Catch-Up Contributions.** A Participant who is aged 50 or over before the end of a taxable year may make Catch-Up Contributions, provided such Catch-Up Contributions are in excess of an

otherwise applicable limit under the Plan. For this purpose, an otherwise applicable Plan limit is a limit in the Plan that applies to Salary Deferrals without regard to Catch-up Contributions, such as the Code § 415 Limitation, or the Elective Deferral Dollar Limit under Code § 402(g). As with other Salary Deferrals, a Participant may, at the time of the deferral election, designate Catch-Up Contributions as either Pre-Tax Deferrals or Roth Deferrals.

- (1) <u>Catch-Up Contribution Limit</u>. Catch-up Contributions for a Participant for a taxable year may not exceed the Catch-Up Contribution Limit. The Catch-Up Contribution Limit for the 2018 tax year is \$6,000. For taxable years beginning after 2018, the Catch-Up Contribution Limit will be adjusted for cost-of-living increases under Code § 414(v)(2)(C).
- **Special Treatment of Catch-Up Contributions.** Catch-up Contributions are not subject to the Elective Deferral Dollar Limit or the Code § 415 Limitation.
- 3.3 <u>Matching Contributions</u>. The District shall have the responsibility for making Matching Contributions under the Plan. On behalf of each Participant, the District shall make a Matching Contribution equal to one-hundred percent (100%) of Participant's contributions to the Plan. The Matching Contributions shall be capped at four and one-half percent (4.5%) of the Participant's Plan Compensation not to exceed the participant's total deferred compensation. Any Matching Contribution made under the Plan will be allocated to Participants' Matching Contribution Account.
  - **Employee Matching Formula.** The Matching Contribution formula shall apply to Salary Deferrals made by Participants to the Plan.
  - **Matching Contributions.** All Matching Contributions are one-hundred percent (100%) vested immediately.
- Rollover Contributions. An Employee (or former Employee) may make a Rollover Contribution to this Plan from a qualified retirement plan or from an IRA in accordance with administrative procedures established by the Plan Administrator for the acceptance of Rollover Contributions, including Rollover Contributions of Roth amounts, but only if (i) it is a direct rollover from another Roth Deferral Account in an eligible retirement plan described in Code § 402A(e)(1), (ii) it is permitted under Code § 402(c), and (iii) the Plan Administrator of the transferor plan provides the Plan Administrator with information about the rollover required under the Treasury Regulations. A rollover of Roth Deferrals may not be made to this Plan from a Roth IRA. Any Roth Deferrals rolled into this Plan will be held in a separate Roth Rollover Account. A Participant may withdraw amounts from his/her Rollover Contribution Account in accordance with the distribution rules under Section 7 (Plan Distributions). Effective August 1, 2018, Participant withdrawals from his or her Rollover Contribution Account is no longer limited to once per Plan Year. Any amounts received as a Rollover Contribution under this Section 3.4 will not be treated as an Annual Addition for purposes of applying the Code § 415 Limitation.

For purposes of this Section 3.4, a qualified retirement plan is a tax-qualified retirement plan described in Code § 401(a) or Code § 403(a), an annuity contract described in Code § 403(b), or an eligible plan under Code § 457(b), which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state. To qualify as a Rollover Contribution under this

Section 3.4, the Rollover Contribution must be transferred directly from the qualified retirement plan or IRA in a Direct Rollover or must be transferred to the Plan by the Employee within sixty (60) days following receipt of the amounts from the qualified plan or IRA.

An Employee may make a Rollover Contribution to the Plan if the Employee is not a Participant with respect to any or all other contributions under the Plan. An Employee who makes a Rollover Contribution to this Plan prior to becoming an Eligible Employee shall be treated as a Participant only with respect to such Rollover Contribution Account, but shall not be treated as a Participant with respect to other contribution sources under the Plan until he/she otherwise satisfies the eligibility conditions under the Plan. To the extent Participant loans are authorized under the Plan, a "limited Participant" under this paragraph may request a Participant loan from the Rollover Contribution Account. A former Employee, however, may only make a Rollover Contribution to the Plan if he or she is a Participant with respect to other contributions under the Plan.

The Plan Administrator may refuse to accept a Rollover Contribution if the Plan Administrator reasonably believes the Rollover Contribution:

- (a) is not being made from a proper plan or IRA;
- (b) is not being made within sixty (60) days from receipt of the amounts from a qualified retirement plan or IRA;
- **(c)** could jeopardize the tax-exempt status of the Plan; or
- **(d)** could create adverse tax consequences for the Plan or the District.

Prior to accepting a Rollover Contribution, the Plan Administrator may require the Employee to provide satisfactory evidence establishing that the Rollover Contribution meets the requirements of this Section 3.4.

The Plan Administrator may apply different conditions for accepting Rollover Contributions from qualified retirement plans and IRAs. For example, the Plan Administrator may decide in its discretion whether to accept a Direct Rollover of a loan note from another qualified plan. Any conditions on Rollover Contributions must be applied uniformly to all Employees under the Plan.

- **Roth Deferrals.** Effective January 1, 2024, a Participant may, at the time of his or her Salary Deferral Election or election change, irrevocably designate all or a portion of his or her future Salary Deferrals, including Catch-Up Contributions, as Roth Deferrals instead of Pre-Tax Deferrals. A Participant's Roth Deferrals will be allocated to and maintained in a separate Account as provided in this Section 3.5.
  - (a) Contributions and withdrawals of Roth Deferrals will be credited and debited to a separate Account maintained for each Participant, and the Plan will maintain a record of the amount of Roth Deferrals in such Account. In addition, gains, losses and other credits or charges must be separately allocated on a reasonable and consistent basis to each Participant's Roth Deferral Account under the Plan. However, in no event may Plan forfeitures be allocated to such Account. No contributions other than Roth Deferrals or Rollover Contributions to a Roth Deferral Account will be credited to a Participant's Roth Deferral Account.

- (b) To the extent required by the Code and the Treasury regulations, reemployed veterans may designate the year for which a Roth Deferral is made. If no such designation is made, the Roth Deferrals will be treated as having been made in the first year of military service for which the veteran could have made the contributions but not earlier than February XX, 2024. This designation must be made solely for purposes of determining the five-taxable-year period under Code § 402A(d)(2).
- (c) In the absence of a specific designation of Salary Deferrals as Roth Deferrals, Salary Deferrals must be treated as Pre-Tax Deferrals. Unless specifically provided otherwise in the Plan, Roth Deferrals will be treated as Salary Deferrals for all purposes under the Plan, including but not limited to immediate full vesting, the distribution restrictions in Section 7 (Plan Distributions), the Elective Deferral Dollar Limit imposed by Code § 402(g) and the Limit on Annual Additions under Code § 415.
- 3.6 In-Plan Roth Conversions. Effective January 1, 2024, a Participant may elect to transfer some or all of his or her vested Account Balance (other than his or her Roth Deferral Account or Roth Rollover Account), and earnings thereon, to an In-Plan Roth Conversion Account without regard to whether the Participant satisfies the requirements for distribution in accordance with Section 7 (Plan Distributions). Any Participant election to make an In-Plan Roth Conversion during a taxable year is irrevocable and cannot be changed after the In-Plan Roth Conversion is completed. Generally, In-Plan Roth Conversion Accounts are subject to the same Plan provisions as Roth Deferral Accounts. However, distribution restrictions that otherwise apply with respect to a specific contribution source will continue to apply if such contribution source is converted to Roth Deferrals. For example, if Matching Contributions are converted to Roth Deferrals, such amounts may not be distributed from the Plan upon events not otherwise available for Matching Contributions. The Plan Administrator will maintain such records as are necessary for the proper reporting of In-Plan Roth Conversions and will administer the In-Plan Roth Conversion Accounts in accordance with the Code, Treasury Regulations, and Plan provisions.
  - **Eligibility to Elect In-Plan Roth Rollover**. For purposes of determining eligibility for In-Plan Roth Rollovers, the Plan will treat a Participant's spousal Beneficiary or Participant's alternate payee spouse or former spouse as a Participant. A non-spouse Beneficiary is not eligible to make an In-Plan Roth Rollover.
  - **In-Plan Roth Rollover Procedure.** An In-Plan Roth Rollover must be made by the Participant in the form of a Direct In-Plan Roth Rollover. Additionally, an In-Plan Roth Rollover must be accomplished in the form and manner established by the Plan Administrator.
  - (c) <u>Plan Loans</u>. In-Plan Roth Rollovers may include a Plan loan only if there is no change in the loan's repayment schedule and the loan is not treated as a new loan. Further, Plan loans that include In-Plan Roth Rollover amounts are subject to the Plan's loan rules and the Plan's loan policy.
  - (d) <u>Treatment of In-Plan Roth Rollovers</u>. Notwithstanding anything in the Plan to the contrary, an In-Plan Roth Rollover is not a rollover contribution for purposes of the Plan. The Plan will take into account the amounts attributable to an In-Plan Roth Rollovers for purposes of determining whether

- a Participant's vested Account Balance exceeds \$500 for purposes of the Plan's involuntary cashout rules. Amounts in the Participant's In-Plan Roth Rollover Account may only be withdrawn by a Participant when the Participant is eligible for a distribution under Section 7 (Plan Distributions).
- **Spousal Consent.** A married Plan Participant is not required to obtain spousal consent in connection with an election to make an In-Plan Roth Conversion, even if the Plan would otherwise require the spousal consent for distributions.
- Tax Treatment of In-Plan Roth Conversion. A Participant must include in gross income the taxable amount of an In-Plan Roth Conversion. For this purpose, the taxable amount of an In-Plan Roth Conversion is the fair market value of the distribution reduced by any basis in the converted amounts. If an outstanding loan is rolled over as part of an In-Plan Roth Conversion, the amount includible in gross income includes the balance of the loan. Generally, the taxable amount of an In-Plan Roth Conversion is includible in gross income in the taxable year in which the conversion occurs. However, In-Plan Roth Conversions are not subject to the twenty percent (20%) mandatory withholding under Code § 3405(c). Thus, Participants will be required to make arrangements independent of the Plan to ensure the total taxable amount is timely paid.
- **Early Distribution Penalty.** If an amount allocable to the taxable amount of an In-Plan Roth is subsequently distributed within the 5-taxable-year period beginning with the first day of the Participant's taxable year in which the conversion was made, the amount distributed is treated as includible in gross income for purposes of applying the Code § 72(t) early distribution penalty. For this purpose, the 5-taxable-year period ends on the last day of the Participant's fifth taxable year in the period. This subsection (g) will not apply to the extent the distribution is rolled over to a Roth account in another qualified plan or is rolled over to a Roth IRA. However, the rule under this subsection (g) will apply to any subsequent distributions made from such other Roth account or Roth IRA within the 5-taxable-year period.

#### (h) <u>In-Plan Roth Conversion Definitions</u>.

- (1) "Direct In-Plan Roth Rollover" means a direct rollover from a Participant Account (other than Participant's Roth Deferral Account, Roth Rollover Account and In-Plan Roth Rollover Account) to an In-Plan Roth Rollover Account established for the Participant.
- (2) "In-Plan Roth Rollover" means a contribution made from the Participant's Account (other than Participant's Roth Deferral Account or Roth Rollover Account) to the Participant's In-Plan Roth Rollover Account under the Plan in accordance with Code § 402A(c)(4) with respect to distributable amounts under the Plan or § 402A(c)(4)(E) with respect to amounts not otherwise distributable under the Plan.
- **(3) "In-Plan Roth Rollover Account"** means the subaccount established under the Plan to account for a Participant's In-Plan Roth Rollover.

#### **SECTION 4**

#### SPECIAL RULES AFFECTING GOVERNMENTAL PLANS

- **4.1 Governmental Plan.** This Plan is a tax-qualified defined contribution plan under Code § 401(a). In addition, this Plan meets the requirements for establishing and maintaining a governmental plan under Code § 414(d).
  - **Governmental Plan Exemptions.** As a tax-qualified governmental plan, this Plan is exempt from Title I of ERISA and certain qualification rules under Code 401(a), including:
    - The minimum age and service rules under Code § 410(a) and the minimum coverage rules under Code § 410(b).
    - (2) The minimum vesting requirements of Code § 411, including minimum vesting schedules, consent requirements for plan distributions, and the anti-cutback rule under Code § 411(d)(6).
    - (3) The nondiscrimination requirements under Code §§ 401(a)(4), 401(k) and 401(m).
    - (4) The top-heavy rules under Code § 416.
    - (5) The joint and survivor annuity rules under Code §§ 401(a)(11) and 417.
    - (6) The requirements for protecting benefits pursuant to a plan merger or a transfer of plan assets and liabilities, as prescribed by Code § 401(a)(12).
    - (7) The anti-assignment rule under Code § 401(a)(13). However, the Code § 414(p)(11) rules relating to the taxability of benefits distributed pursuant to a QDRO are applicable to benefits payable to an alternate payee under the QDRO.
    - (8) The commencement of benefit requirements under Code § 401(a)(14).
    - (9) The protections under Code  $\S$  401(a)(19).
  - **ERISA Exempt.** Adoption by the District of provisions similar to the requirements applicable to plans covered under Title I of ERISA or to otherwise inapplicable Code requirements will not affect the Plan's status as a governmental plan. In the event such provisions are adopted, the Plan shall remain exempt from ERISA and certain Code requirements as a governmental plan.

#### **SECTION 5**

#### CONTRIBUTION AND BENEFIT LIMITATIONS

#### 5.1 <u>Code § 415 Limitation.</u>

(a) No Other Plan Participation. If the Participant does not participate in, and has never participated in another qualified retirement plan, a welfare benefit fund (as defined under Code § 419(e)), an individual medical account (as defined under Code § 415(l)(2)), or a SEP (as defined under Code § 408(k)) maintained by the District which provides an Annual Addition as defined in subsection (c)(1), then the amount of Annual Additions which may be credited to the Participant's

Account for any Limitation Year *will not* exceed the lesser of the Maximum Permissible Amount or any other limitation contained in this Plan.

If an Employer Contribution that would otherwise be contributed or allocated to a Participant's Account will cause that Participant's Annual Additions for the Limitation Year to exceed the Maximum Permissible Amount, the amount to be contributed or allocated to such Participant will be reduced so that the Annual Additions allocated to such Participant's Account for the Limitation Year will equal the Maximum Permissible Amount. However, if a contribution or allocation is made to a Participant's Account in an amount that exceeds the Maximum Permissible Amount, such excess Annual Additions may be corrected pursuant to the correction procedures outlined under the IRS' Employee Plans Compliance Resolution System (EPCRS).

- **Participation in Another Plan.** This subsection 5.1(b) applies if, in addition to this Plan, the Participant receives an Annual Addition during any Limitation Year from another Defined Contribution Plan, a welfare benefit fund (as defined under Code § 419(e)), an individual medical account (as defined under Code § 415(l)(2)), or a SEP (as defined under Code § 408(k)) maintained by the District.
  - This Plan's Code § 415 Limitation. The Annual Additions that may be credited to a Participant's Account under this Plan for any Limitation Year will not exceed the Maximum Permissible Amount (defined in subsection 5.1(c)(6) below) reduced by the Annual Additions credited to a Participant's Account under any other Defined Contribution Plan, welfare benefit fund, individual medical account, or SEP maintained by the District for the same Limitation Year.
  - Annual Additions Reduction. If the Annual Additions with respect to the Participant under any other Defined Contribution Plan, welfare benefit fund, individual medical account, or SEP maintained by the District are less than the Maximum Permissible Amount and the Annual Additions that would otherwise be contributed or allocated to the Participant's Account under this Plan would exceed the Code § 415 Limitation for the Limitation Year, the amount contributed or allocated will be reduced so that the Annual Additions under all such Plans and funds for the Limitation Year will equal the Maximum Permissible Amount. However, if a contribution or allocation is made to a Participant's Account in an amount that exceeds the Maximum Permissible Amount, such excess Annual Additions may be corrected pursuant to the correction procedures outlined under the IRS' Employee Plans Compliance Resolution System (EPCRS).
  - (3) No Annual Additions Permitted. If the Annual Additions with respect to the Participant under such other Defined Contribution Plan(s), welfare benefit fund(s), individual medical account(s), or SEP(s) in the aggregate are equal to or greater than the Maximum Permissible Amount, no amount will be contributed or allocated to the Participant's Account under this Plan for the Limitation Year. However, if a contribution or allocation is made to a Participant's Account in an amount that exceeds the Maximum Permissible Amount, such

excess Annual Additions may be corrected pursuant to the correction procedures outlined under the IRS' Employee Plans Compliance Resolution System (EPCRS).

# (c) § 415 Definitions.

- (1) <u>Annual Additions</u>. The amounts credited to a Participant's Account for the Limitation Year that are taken into account in applying the Code § 415 Limitation, including:
  - i Employer Contributions, including Matching Contributions and Salary Deferrals; and
  - ii Forfeitures.

An Annual Addition is credited to a Participant's Account for a particular Limitation Year if such amount is allocated to the Participant's Account as of any date within that Limitation Year. An Annual Addition *will not* be deemed credited to a Participant's Account for a particular Limitation Year unless such amount is actually contributed to the Plan no later than 30 days after the time prescribed by law for filing the District's income tax return (including extensions) for the taxable year with or within which the Limitation Year ends. In the case of After-Tax Employee Contributions, such amount shall not be deemed credited to a Participant's Account for a particular Limitation Year unless the contributions are actually contributed to the Plan no later than 30 days after the close of that Limitation Year.

- (2) <u>Defined Contribution Dollar Limitation</u>. \$40,000, as adjusted under Code § 415(d).
- (3) <u>Excess Amount</u>. The excess of the Participant's Annual Additions for the Limitation Year over the Maximum Permissible Amount.
- **Limitation Year.** The Plan Year is the Limitation Year.
- (5) <u>Maximum Permissible Amount</u>. For Limitation Years beginning on or after January 1, 2002, the Maximum Permissible Amount is the lesser of:
  - i. the Defined Contribution Dollar Limitation, or
  - ii. 100 percent of the Participant's Total Compensation for the Limitation Year.

The Total Compensation limitation referred to in (ii) shall not apply to any contribution for medical benefits (within the meaning of Code  $\S$  401(h) or  $\S$  419A(f)(2)) which is otherwise treated as an Annual Addition.

If a short Limitation Year is created because of an amendment changing the Limitation Year to a different 12-consecutive month period, the Maximum Permissible Amount *will not* exceed the Defined Contribution Dollar Limitation multiplied by the following fraction:

### Number of months in the short Limitation Year 12

If a short Limitation Year is created because the Plan has an *initial* Plan Year that is less than 12 months, no proration of the Defined Contribution Dollar Limitation is required.

**Total Compensation.** The amount of compensation as defined under Section 1.54, subject

to the following:

- i. <u>Total Compensation Actually Paid or Made Available</u>. For purposes of applying the limitations of this Section 5.1, Total Compensation for a Limitation Year is the Total Compensation actually paid or made available to an Employee during such Limitation Year. The Employer may include in Total Compensation for a Limitation Year amounts earned but not paid in the Limitation Year because of the timing of pay periods and pay days, but only if:
  - (A) the amounts are paid during the first few weeks of the next Limitation Year,
  - (B) such amounts are included on a uniform and consistent basis with respect to all similarly-situated employees, and
  - (C) no amounts are included in Total Compensation in more than one Limitation Year.
- **Restorative Payments.** Restorative payments are not considered Annual Additions for any Limitation Year. For this purpose, restorative payments are payments made to restore losses to the Plan resulting from actions (or a failure to act) by a fiduciary or third-party administrator for which there is a reasonable risk of liability under applicable federal or state law, where Participants who are similarly situated are treated similarly with respect to the payments.
- **Corrective Provisions.** The District may use reasonable correction methods to the extent permitted by the IRS or under an IRS correction program.
- **Change of Limitation Year**. Where there is a change of Limitation Year, a "short" Limitation Year exists for the period beginning with the first day of the Limitation Year and ending on the day before the change in Limitation Year is effective. For this purpose, if the Plan is terminated effective as of a date other than the last day of the Limitation Year, the Plan is treated as if it were amended to change its Limitation Year.
- **Elective Deferral Dollar Limit.** The Elective Deferral Dollar Limit under this Section 5.2 applies with respect to a Participant's Salary Deferrals. Under this Elective Deferral Dollar Limit, a Participant may not make Elective Deferrals under this Plan during any calendar year in an amount that exceeds the Elective Deferral Dollar Limit in effect for each Plan Year.

The Elective Deferral Dollar Limit is \$18,500 for the 2018. For taxable years beginning after 2018, the Elective Deferral Dollar Limit will be adjusted for cost-of-living increases under Code § 402(g)(4).

If a Participant is aged 50 by the end of the taxable year, the Elective Deferral Dollar Limit is increased by the Catch-Up Contribution Limit described in Section 3.2(c).

- (a) <u>Excess Deferrals</u>. Excess Deferrals are Elective Deferrals made during the Participant's taxable year that exceed the Elective Deferral Dollar Limit for such year.
- **(b)** <u>Correction of Excess Deferrals</u>. If a Participant makes Elective Deferrals in excess of the Elective Deferral Dollar Limit ("Excess Deferrals") under this Plan, such Excess Deferrals (plus allocable

income or loss) shall be distributed to the Participant. A distribution of Excess Deferrals may be made at any time (subject to the correction provisions under EPCRS or subsequent guidance). If the corrective distribution of Excess Deferrals is made by April 15 of the calendar year following the year the Excess Deferrals are made to the Plan, such amounts will be taxable in the year of deferral, but not in the year of distribution. If a corrective distribution of Excess Deferrals is made after April 15 of the following calendar year, such amounts will be taxable in both the year of deferral and the year of distribution. Unless the Plan Administrator determines otherwise, if the Participant made both Pre-Tax Deferrals and Roth Deferrals during the same Plan Year, the distribution of Excess Deferrals will be made first solely from the Participant's Pre-Tax Salary Deferral Account until all Pre-Tax Deferrals, plus any allocable gains and minus any allocable losses, for the Plan Year, have been distributed, and only then from the Participant's Roth Deferral Account.

- (1) Amount of Corrective Distribution. The amount to be distributed from this Plan as a correction of Excess Deferrals equals the amount of Elective Deferrals the Participant contributes during the taxable year to this Plan and any other plan maintained by the District in excess of the Elective Deferral Dollar Limit, reduced by any corrective distribution of Excess Deferrals the Participant receives during the calendar year from this Plan or other plan(s) maintained by the District.
- Allocable Gain or Loss. A corrective distribution of Excess Deferrals must include any allocable gain or loss for the taxable year in which the Excess Deferrals are contributed to the Plan. The gain or loss allocable to Excess Deferrals may be determined in any reasonable manner, provided the manner used to determine allocable gain or loss is applied consistently for all Participants and in a manner that is reasonably reflective of the method used by the Plan for allocating income to Participants' Accounts. A corrective distribution of Excess Deferrals will not include any income or loss allocable to the period between the end of the taxable year and the date of distribution.
- (3) Taxation of Corrective Distribution. If a corrective distribution of Excess Deferrals is made by April 15 of the following calendar year, amounts attributable to the Excess Deferrals will be includible in the Participant's gross income in the taxable year in which such amounts are deferred under the Plan and amounts attributable to income or loss on the Excess Deferrals will be includible in gross income in the year of distribution.
  - If a corrective distribution of Excess Deferrals is made after April 15, the amount of the corrective distribution attributable to Excess Deferrals will be includible in the Participant's gross income in both the taxable year in which such amounts are deferred under the Plan and the taxable year in which such amounts are distributed.
- (4) <u>Coordination with Other Provisions</u>. A corrective distribution of Excess Deferrals made by April 15 of the following calendar year may be made without consent of the Participant or the Participant's Spouse, and without regard to any distribution restrictions applicable under Section 7 (Plan Distributions). A corrective distribution of Excess Deferrals made by

the appropriate April 15 also is not treated as a distribution for purposes of applying the required minimum distribution rules under Section 8 (Required Minimum Distributions).

Correction of Excess Deferrals Under Plans Not Maintained by the Employer. The correction provisions under this subsection 5.2(b) apply only if a Participant makes Excess Deferrals under this Plan (or under this Plan and other plans maintained by the District). However, if a Participant has Excess Deferrals for a calendar year on account of making Elective Deferrals to a plan of an unrelated employer, the Participant may assign to this Plan any portion of his/her Elective Deferrals made under all plans during the calendar year to the extent such Elective Deferrals exceed the Elective Deferral Dollar Limit. The Participant must notify the Plan Administrator in writing on or before March 1 of the following calendar year of the amount of the Excess Deferrals to be assigned to this Plan.

Upon receipt of a timely notification, the Excess Deferrals assigned to this Plan will be distributed (along with any allocable income or loss) to the Participant in accordance with the corrective distribution provisions under this subsection 5.2(b). A Participant is deemed to notify the Plan Administrator of Excess Deferrals to the extent such Excess Deferrals arise only under this Plan and any other plan maintained by the District.

### **SECTION 6**

### PARTICIPANT VESTING AND FORFEITURES

- **Vesting of Contributions.** Participants are one-hundred percent (100%) vested in his or her Account at all times.
- **Forfeiture of Benefits.** Notwithstanding immediate and full vesting, upon the occurrence of any of the events described below or at any such time as the Plan Administrator determines, a Participant Account may be forfeited. The Plan Administrator has the responsibility to determine the amount of a Participant's forfeiture. Until an amount is forfeited pursuant to this Section 6.2 (Forfeiture of Benefits), a Participant's entire Account must remain in the Plan and continue to share in gains and losses of the Trust.
  - (a) Missing Participant or Beneficiary. If a Participant or Beneficiary cannot be located within a reasonable period following a reasonable diligent search, the missing Participant's or Beneficiary's Account may be forfeited, as provided in subsection (2) below. The District will be deemed to have performed a reasonable diligent search if it performs the actions described in subsection (1) below. In determining whether a reasonable period has elapsed following a reasonable diligent search, the District or Plan Administrator may follow any applicable guidance provided under statute, regulation, or other IRS or DOL guidance of general applicability. However, the District or Plan Administrator will be deemed to have waited a reasonable period following a reasonable diligent search if the District or Plan Administrator waits at least 6 months following the completion of the actions described in subsection (1) below.
    - **Reasonable Diligent Search.** The District or Plan Administrator will be deemed to have performed a reasonable diligent search if it performs the following actions:

- i. Send a certified letter to the Participant's or Beneficiary's last known address.
- ii. Check related records of the District (e.g., health plan records) to determine if a more current address exists for the Participant or Beneficiary.
- **iii.** If the Participant cannot be located, the District or Plan Administrator may attempts to identify and contact any individual that the Participant has designated as a Beneficiary under the Plan for updated information concerning the location of the missing Participant.
- iv. Utilize the Social Security Administration (SSA) letter-forwarding service for locating lost participants. Additional information regarding the SSA letter forwarding program can be located at www.ssa.gov.In addition to the search methods discussed above, the District or Plan Administrator may use other search methods, including the use of Internet search tools, commercial locator services, and credit reporting agencies to locate the missing Participant.
- deemed to be missing (i.e., cannot be found within 4 years of his or her required beginning date), the Plan administrator may forfeit the distributable amount of such missing Participant's or Beneficiary's vested Account. If, after an amount is forfeited under this subsection (2), the missing Participant or Beneficiary is located, the Plan will restore the forfeited amount (unadjusted for gains or losses) to such Participant or Beneficiary within a reasonable time. However, if a missing Participant or Beneficiary has not been located by the time the Plan terminates, the forfeiture of such Participant's or Beneficiary's distributable amount will be irrevocable. Notwithstanding the foregoing, the Plan Administrator may address forfeitures using any available means provided under applicable law, including escheating the Participant's or Beneficiary's Account to the state.
- (3) Expenses Attributable to Search for Missing Participant. Reasonable expenses attendant to locating a missing Participant may be charged to such Participant's Account, provided that the amount of such expenses is reasonable. The Plan Administrator may take into account the size of a Participant's Account in relation to the cost of the search when deciding how extensive a search is required before declaring such Participant as missing under subsection (a).
- **Employer Matching Contribution Attributable to Excess Deferrals.** If a Participant receives a distribution of Excess Deferrals, the portion of his/her Matching Contribution Account which is attributable to such distributed amounts will be forfeited. A forfeiture of Matching Contributions under this subsection 6.2(b) occurs in the Plan Year in which the Participant receives the distribution of Excess Deferrals.
- **Allocation of Forfeitures.** Forfeitures may be used in the Plan Year in which the forfeitures occur or in the Plan Year following the Plan Year in which the forfeitures occur. In applying the forfeiture provisions under the Plan, if there are any unused forfeitures as of the end of the Plan Year, any remaining forfeiture

will be used in the immediately following Plan Year.

- **Reduction of Contributions.** The District may, in its discretion, use such forfeitures to reduce Matching Contributions. The District may adjust its contribution deposits in any manner, provided the total Matching Contributions made for the Plan Year properly take into account the forfeitures that are to be used to reduce such contributions from that Plan Year. If contributions are allocated over multiple allocation periods, the District may reduce its contribution for any allocation periods within the Plan Year in which the forfeitures are to be allocated so that the total amount allocated for the Plan Year is proper. If the Plan provides for Matching Contribution and the District elects not to make a Matching Contribution for the Plan Year, any forfeitures will be allocated to eligible Participants as an additional Matching Contribution.
- **Payment of Plan Expenses.** The District may use forfeitures to pay Plan expenses for the Plan Year in which the forfeitures would otherwise be applied. If any forfeitures remain after the payment of Plan expenses under this subsection 6.3(b), the remaining forfeitures will be allocated to reduce Matching Contributions to the Plan. This subsection 6.3(b) only applies to the extent Plan expenses are paid by the Plan. Nothing herein affects the ability of the District to pay Plan expenses. In determining the Plan expenses that may be offset by Plan forfeitures, the District may use any reasonable method to determine the Plan expenses attributable to a particular year.

### **SECTION 7**

#### PLAN DISTRIBUTIONS

A Participant may receive a distribution of his/her vested Account Balance at the time and in the manner provided under this Section 7. Upon reaching the Required Beginning Date, a Participant must begin receiving distributions under the Plan in accordance with the provisions of Section 8 (Required Minimum Distributions).

- 7.1 Available Forms of Distribution. Upon termination of employment, a Participant under the Plan may take a distribution of his/her entire vested Account Balance in a single lump sum, periodic payments (on a monthly, quarterly, semiannual, or annual basis), or partial distribution. Further, the Plan Administrator may, in its discretion, permit a Participant to take distributions of less than entire vested Account Balance provided all Participants have an opportunity to take such partial distributions upon termination of employment. In addition, the Plan Administrator may, in its discretion, permit Participants to take single lump sum, periodic payments (on a monthly, quarterly, semiannual, or annual basis), or partial distribution to the extent necessary to satisfy the required minimum distribution rules under Code § 401(a)(9). (See Section 8 Required Minimum Distributions).
- Amount Eligible for Distribution. For purposes of determining the amount a Participant or Beneficiary may receive as a distribution from the Plan, a Participant's Account Balance is determined as of the Valuation Date immediately preceding the date the Participant or Beneficiary receives his/her distribution from the Plan. For this purpose, the Account Balance must be increased for any contributions allocated to the Participant's Account since the most recent Valuation Date and must be reduced for any distributions made from the Participant's Account since the most recent Valuation Date. A Participant or Beneficiary does not share in any allocation of gains or losses attributable to the period between the most recent

Valuation Date and the date of the distribution, unless provided otherwise under uniform funding and valuation procedures established by the Plan Administrator.

- **Participant Consent.** To the extent a Participant's entire vested Account Balance exceeds the \$500 Involuntary Cash-Out threshold per this Section 7.3(a), the Participant must consent to any distribution of such Account Balance prior to his/her Required Beginning Date except as set forth in Section 8 (Required Minimum Distributions)
  - **Involuntary Cash-Out Threshold.** For purposes of determining whether a distribution is subject to the Participant consent requirements as described in this Section 7.3, the Involuntary Cash-Out threshold is \$500. For purposes of determining whether a Participant's vested Account Balance exceeds the Involuntary Cash-Out threshold, the value of the Participant's vested Account Balance includes Rollover Contributions (and earnings allocable thereto) within the meaning of Code §§ 402(c), 403(a)(4), 403(b)(8), 408(d)(3)(A)(ii), and 457(e)(16) is taken into account.
  - **Participant Notice.** If a distribution is subject to Participant consent, the Participant must consent in writing to the distribution within a reasonable period prior to the Distribution Starting Date. For this purpose, any consent made within the 180-day period ending on the Distribution Starting Date will be deemed to be made within a reasonable period. If the distribution is subject to spousal consent, the Participant's Spouse also must consent to the distribution in accordance with Section 9.2 (Spousal Consent).

Prior to receiving a distribution from the Plan, a Participant must be notified of his/her right to defer any distribution from the Plan. Effective for Plan Years beginning on or after January 1, 2007, the Participant notice must include a description of the consequences of a Participant's decision not to defer the receipt of a distribution. The notice must be provided no less than 30 days and no more than 180 days prior to the Participant's Distribution Starting Date. However, distribution may commence less than 30 days after the notice is given, if the Participant is clearly informed of his/ or her right to take 30 days after receiving the notice to decide whether or not to elect a distribution (and, if applicable, a particular distribution option), and the Participant, after receiving the notice, affirmatively elects to receive the distribution prior to the expiration of the 30-day minimum period. The notice requirements described in this paragraph may be satisfied by providing a summary of the required information, so long as the conditions described in applicable regulations for the provision of such a summary are satisfied, and the full notice is also provided (without regard to the 180-day period described in this subsection (b)).

(c) Special Rules. The consent rules under this Section 7.3 apply to distributions made after the Participant's termination of employment and to distributions made prior to the Participant's termination of employment. However, the consent of the Participant shall not be required to the extent that a distribution is required to satisfy the required minimum distribution rules under Section 8 (Required Minimum Distributions) or to satisfy the requirements of Code § 415, as described in Section 5.2 (Elective Deferral Dollar Limit). A Participant also will not be required to consent to a corrective distribution of Excess Deferrals.

7.4 <u>Direct Rollovers</u>. A Participant (other than a Participant subject to an Involuntary Cash-Out) may elect, at the time and the manner prescribed by the Plan Administrator, to have all or any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan (a "Direct Rollover"). For purposes of this Section 7.4, a Participant includes a Participant or former Participant. In addition, this Section applies to any distribution from the Plan made to a Participant's surviving Spouse or to a Participant's Spouse or former Spouse who is the Alternate Payee under a QDRO. For distributions made on or after January 1, 2007, this Section 7.4 also applies to distributions made to a Participant's non-Spouse beneficiary, as set forth in subsection (c) below.

# (a) Definitions.

- (1) <u>Eligible Rollover Distribution</u>. An Eligible Rollover Distribution is any distribution of all or any portion of a Participant's Account Balance, except an Eligible Rollover Distribution does not include:
  - i. any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Participant or the joint lives (or joint life expectancies) of the Participant and the Participant's Beneficiary, or for a specified period of ten years or more;
  - ii. any distribution to the extent such distribution is a required minimum distribution under Code § 401(a)(9), as described under Section 8 (Required Minimum Distributions);
  - iii. any Hardship distribution, as described in Section 7.8(e) (Hardship Distribution);
  - iv. the portion of any distribution that is not includible in gross income; or,
  - v. a distribution made to satisfy the requirements of Code § 415 (as described in Section 5.2) or a distribution to correct Excess Deferrals.
- **Eligible Retirement Plan**. For purposes of applying the Direct Rollover provisions under this Section 7.4, an Eligible Retirement Plan is:
  - i. a qualified plan described in Code § 401(a);
  - ii. an individual retirement account described in Code § 408(a);
  - iii. an individual retirement annuity described in Code § 408(b);
  - iv. an annuity plan described in Code § 403(a);
  - v. an annuity contract described in Code § 403(b); or,
  - vi. an eligible plan under Code § 457(b) which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this Plan.

If any portion of an Eligible Rollover Distribution is attributable to payments or distributions from a Roth Deferral Account, an Eligible Retirement Plan with respect to such portion will include only another Roth Deferral Account under an applicable retirement plan described in Code § 402A(e)(1) of the individual from whose Account the payments or distributions were made, or to a Roth IRA described in Code § 408A of such individual, and only to the extent the rollover is permitted under the rules of Code § 402(c).

The definition of Eligible Retirement Plan also applies in the case of a distribution to a surviving Spouse, or to a Spouse or former Spouse who is the Alternate Payee under a QDRO.

- (3) <u>Direct Rollover</u>. A Direct Rollover is a payment made directly from the Plan to the Eligible Retirement Plan specified by the Participant. The Plan Administrator may develop reasonable procedures for accommodating Direct Rollover requests.
- **Direct Rollover Notice.** A Participant entitled to an Eligible Rollover Distribution must receive a written explanation of his/her right to a Direct Rollover, the tax consequences of not making a Direct Rollover, and, if applicable, any available special income tax elections. The notice must be provided within 30–180 days prior to the Participant's Distribution Starting Date in the same manner as described in Section 7.3(b) (Participant Notice). The Direct Rollover notice must be provided to all Participants.
- Direct Rollover by Non-Spouse Beneficiary. Effective for Plan Years beginning after December 31, 2009, the Plan permits a non-Spouse beneficiary (as defined in Code § 401(a)(9)(E)) to make a Direct Rollover of an Eligible Rollover Distribution to an individual retirement account under Code § 408(a) or an individual retirement annuity under Code § 408(b) that is established on behalf of the designated beneficiary and that will be treated as an inherited IRA pursuant to the provisions of Code § 402(c)(11). A non-Spouse rollover made after December 31, 2009 will be subject to the direct rollover requirements under Code § 401(a)(31), the rollover notice requirements under Code § 402(f) or the mandatory withholding requirements under Code § 3405(c).
- **Direct Rollover of Non-Taxable Amounts.** Notwithstanding any other provision of the Plan, effective for taxable years beginning on or after January 1, 2007, an Eligible Rollover Distribution may include the portion of any distribution that is not includible in gross income. For this purpose, an Eligible Retirement Plan includes a Defined Contribution or defined benefit plan qualified under Code § 401(a) and a tax-sheltered annuity plan under Code § 403(b), provided the rollover is accomplished through a direct rollover and the recipient Eligible Retirement Plan separately accounts for any amounts attributable to the rollover of any nontaxable distribution and earnings thereon.
- **7.5 Distribution Upon Termination of Employment.** Subject to the required minimum distribution provisions under Section 8, a Participant who terminates employment for any reason (other than death) is entitled to receive a distribution of his/her vested Account Balance in accordance with this Section 7.5.

- (a) Account Balance Not Exceeding Cash-Out Threshold. If a Participant's vested Account Balance does not exceed \$500 at the time of distribution, the only distribution option available under the Plan is an immediate lump sum option.
- **Account Balance Exceeding Cash-Out Threshold.** If a Participant's vested Account Balance exceeds \$500 at the time of distribution, the Participant may elect to receive a distribution of his/her vested Account Balance at any time after the Participant terminates employment, but prior to his or her required beginning date.
- **7.6 Distribution Upon Death.** Subject to the Required Minimum Distribution rules in Section 8, a Participant's vested Account Balance will be distributed to the Participant's Beneficiary(ies) in accordance with this Section 7.6.
  - **Death After Commencement of Benefits**. If a Participant begins receiving a distribution of his/her benefits under the Plan, and subsequently dies prior to receiving the full value of his/her vested Account Balance, the remaining benefit will continue to be paid to the Participant's Beneficiary(ies) in accordance with the form of payment that has already commenced. If a Participant commences distribution prior to death only with respect to a portion of his/her Account Balance, then the rules in subsection (b) apply to the rest of the Account Balance.
  - **(b)** Death Before Commencement of Benefits. If a Participant dies before commencing distribution of his or her benefits under the Plan, such benefit will be paid to the Participant's Beneficiary(ies) in a single lump sum as soon as administratively feasible following Participant's death.
  - (c) <u>Timing of Distributions</u>. In no event will any death benefit be paid later than the end of the fifth year following the year of the Participant's death or in any other manner that is inconsistent with the Required Minimum Distribution rules under Section 8 (Required Minimum Distributions).
  - **Determining a Participant's Beneficiary.** The determination of a Participant's Beneficiary(ies) to receive any death benefits under the Plan will be based on the Participant's Beneficiary designation under the Plan. If a Participant does not designate a Beneficiary to receive the death benefits under the Plan, distribution will be made to the default Beneficiaries, as set forth in subsection (3) below.
    - (1) Post-Retirement Death Benefit. If a Participant dies after commencing distribution of benefits under the Plan (but prior to receiving a distribution of his/her entire vested Account Balance under the Plan), the Beneficiary of any post-retirement death benefit is determined in accordance with the Beneficiary selected under the distribution option in effect prior to death.
    - (2) <u>Pre-Retirement Death Benefit</u>. If a Participant dies before commencing distribution of his/her benefits under the Plan, the Participant's Beneficiary(ies) will be treated as the sole Beneficiary(ies).
    - (3) <u>Default Beneficiaries</u>.
      - i. <u>Prior to August 1, 2018</u>. If a Participant dies prior to August 1, 2018 or fails to effectively update his or her Beneficiary designation form subsequent to August 1,

2018, or if the Beneficiary named by a Participant predeceases the Participant, then the Plan Administrator will pay the Participant's remaining Account balance to the Participant's estate.

- ii. <u>After August 1, 2018</u>. Effective for individuals who become Participants in the Plan on or after August 1, 2018 or file a beneficiary designation form subsequent to such date:
  - (A) If a Participant fails to name a Beneficiary or the Beneficiary named by a Participant predeceases the Participant, then the Plan Administrator will pay the Participant's remaining Account balance in the following order of priority, to the:
    - **I.** Participant's surviving spouse; or if no surviving spouse, to the
    - II. Participant's children (including adopted children), in equal shares by right of representation (one share for each surviving child and one share for each child who predeceases the Participant with living descendants); and if none, then to
    - **III.** Participant's estate.
  - (B) Where a Participant names more than one Beneficiary and one or more of them, but not all of them, predecease Participant, then the benefits that would have been paid to the deceased Beneficiaries shall be distributed to Participant's next of kin in accordance with subsection (ii)(A)(I through III).
- [4] Identification of Beneficiaries. The Plan Administrator may request proof of the Participant's death and may require the Beneficiary to provide evidence of his/her right to receive a distribution from the Plan as designated in accordance with the rules and procedures developed by the Plan Administrator. The Plan Administrator's determination of the Participant's death and of the right of a Beneficiary to receive payment under the Plan shall be conclusive. If a distribution is to be made to a minor or incompetent Beneficiary, payments may be made to the person's legal guardian, conservator recognized under state law, or custodian in accordance with the Uniform Gifts to Minors Act or similar law as permitted under the laws of the state where the Beneficiary resides. The Plan Administrator will not be liable for any payments made in accordance with this subsection (3) and will not be required to make any inquiries with respect to the competence of any person entitled to benefits under the Plan.
- (5) <u>Death of Beneficiary</u>. If a Beneficiary does not predecease the Participant but dies before distribution of the death benefit is made to the Beneficiary, the death benefit will be paid to the Beneficiary's designated Beneficiary. If the Beneficiary did not provide the Plan with a valid Beneficiary designation, the death benefit will be paid to the Beneficiary's estate.

- 6) Divorce From Spouse. If a Participant designates his/her Spouse as Beneficiary and subsequent to such Beneficiary designation, the Participant and Spouse are subsequently divorced, the designation of the Spouse as Beneficiary under the Plan is automatically rescinded unless specifically provided otherwise under a divorce decree or QDRO, or unless the Participant enters into a new Beneficiary designation naming the prior Spouse as Beneficiary. In addition, the provisions under this subsection will not apply if the Participant has entered into a Beneficiary designation that specifically overrides the provisions of this subsection 7.6(d)(6).
  - For periods prior to the date this Plan is executed by the District this subsection also applies to situations where the Participant and Spouse are legally separated.
- 7.7 <u>Distribution to Disabled Employees</u>. A Participant who terminates employment on account of becoming Disabled may receive a distribution of his/her Account Balance in the same manner as a regular distribution upon termination.
- **7.8 In-Service Distributions.** The Plan permits the following in-service distributions.
  - **Rollover Contributions.** A Participant may make in-service distributions from Rollover Contributions.
  - **Matching Contributions.** Employer Matching Contributions may be withdrawn upon the occurrence of a Hardship or attainment of age 59½.
  - (c) <u>Salary Deferrals</u>. Salary Deferrals (including any earnings on such amounts) generally may not be distributed prior to the Participant's severance from employment, death, or disability. However, Salary Deferrals are available for in-service distributions upon attainment of age 59½, upon a Hardship or upon meeting the requirements for a Qualified Reservist Distribution, as defined under subsection (d) below.
  - **Qualified Reservist Distribution.** Effective September 11, 2001, the distribution provisions applicable to Salary Deferrals include a Qualified Reservist Distribution. If a Participant takes a Qualified Reservist Distribution, such distribution *will not* be subject to the ten percent (10%) penalty tax under Code § 72(t).
    - **Qualified Reservist Distribution.** For purposes of this subsection (d), a Qualified Reservist Distribution means any distribution to an individual if:
      - i. such distribution is from amounts attributable to elective deferrals described in Code § 402(g)(3)(A) or (C) or Code § 501(c)(18)(D)(iii),
      - ii. such individual was (by reason of being a member of a reserve component (as defined in § 101 of Title 37 of the United States Code)) ordered or called to active duty for a period in excess of 179 days or for an indefinite period, and
      - **iii.** such distribution is made during the period beginning on the date of such order or call and ending at the close of the active duty period.

- **Active Duty.** A Qualified Reservist Distribution will only be available for individuals who are ordered or called into active duty after September 11, 2001.
- **Hardship Distribution.** The Plan allows in-service distribution at the request of a Participant or primary Beneficiary upon the occurrence of a Hardship event. A Hardship distribution under the Plan must meet the requirements of a safe harbor Hardship as described under subsection (1) below.
  - (1) <u>Safe Harbor Hardship Distribution</u>. To qualify for a safe harbor Hardship distribution, a Participant must demonstrate an immediate and heavy financial need, as described in subsection (i), and the distribution must be necessary to satisfy such need, as described in subsection (ii).
    - i. <u>Immediate and Heavy Financial Need</u>. To be considered an immediate and heavy financial need, the Hardship distribution must be made to satisfy one of the following financial needs:
      - (A) to pay expenses incurred or necessary for medical care (as described in Code § 213(d)) of the Participant, the Participant's Spouse or, dependents (determined without regard to whether the expenses exceed seven and one-half percent (7.5%) of adjusted gross income);
      - **(B)** for the purchase (excluding mortgage payments) of a principal residence for the Participant;
      - (C) for payment of tuition and related educational fees (including room and board but not including the repayment of student loans) for the next 12 months of post-secondary education for the Participant, the Participant's Spouse, children or dependents;
      - **(D)** to prevent the eviction of the Participant from, or a foreclosure on the mortgage of, the Participant's principal residence;
      - **(E)** to pay funeral or burial expenses for the Participant's deceased parent, Spouse, child or, dependent;
      - effective January 1, 2020, to pay expenses to repair damage to the Participant's principal residence that would qualify for a casualty deduction under Code § 165 (determined without regard to whether the loss exceeds ten percent (10%) of the participant's adjusted gross income, and effective January 1, 2020, without regard to Code § 165(h)(5);
      - (G) to pay expenses and losses (including loss of income) incurred by the participant on account of a Federal Emergency Management Agency ("FEMA") declared disaster, provided that the participant's principle residence or principle place of employment at the time of the disaster was located in an area designated by FEMA for individual assistance with respect

to the disaster.

- (H) for any other event that the IRS recognizes as a safe harbor Hardship distribution event under ruling, notice or other guidance of general applicability.
- ii. The payment of funeral or burial expenses under subsection (E) above and the payment of expenses to repair damage to a principal residence under subsection (F) above only apply to Plan Years beginning on or after January 1, 2006. For purposes of determining eligibility of a Hardship distribution under this subsection (i), a dependent is determined under Code § 152. However, for taxable years beginning on or after January 1, 2005, the determination of dependent for purposes of tuition and education fees under subsection (C) above will be made without regard to Code §§ 152(b)(1), (b)(2), and (d)(1)(B) and the determination of dependent for purposes of funeral or burial expenses under subsection (E) above will be made without regard to Code § 152(d)(1)(B). Determination of Whether Distribution Necessary to Satisfy Need. A distribution will be considered as necessary to satisfy an immediate and heavy financial need of the Participant under the IRS's general standard test if:
  - (A) the distribution is not in excess of the amount of the immediate and heavy financial need, including amounts necessary to pay any federal, state or local income taxes or penalties reasonably anticipated to result from the distribution;
  - **(B)** the Participant obtained all available distributions under the plan and all other plans of deferred compensation maintained by the District, whether qualified or non-qualified; and
  - (C) effective for Hardship distributions made on or after January 1, 2020, the Participant must represent in writing (including via electronic medium) that he or she has insufficient cash or other liquid assets to satisfy the need. The Plan Administrator may rely on such representation absent actual knowledge to the contrary.
- (2) Amount Available for Hardship Distribution. A Participant may receive a Hardship distribution of Deferrals (including Roth Deferrals), Matching Employer Contributions and Rollover Contributions. Provided, however, distributions made from Salary Deferrals, when added to other Hardship distributions, does not exceed the total account balance which includes salary deferrals, Employer matching contributions, rollovers, and interest earned.
- (3) Availability to Terminated Employees. A Participant may take such a Hardship distribution after termination of employment to the extent no other distribution is available from the Plan.
- (4) <u>Six-Month Suspension of Salary Deferrals</u>. Effective for Hardship distributions made on

or after January 1, 2020, the Plan will no longer suspend a Participant's Salary Deferrals after a Hardship distribution.

- (f) <u>Age 59½ Distributions</u>. A Participant may take in-service distributions upon attaining age 59½ in accordance with the terms and conditions established by the Plan Administrator. Effective January 1, 2019, in-service distributions at age 59½ are no longer limited to one per Plan Year.
- **Qualified Birth or Adoption Distributions (QBOADs).** The Plan permits a Participant to request and receive QBOADs. QBOADs must be requested by the Participant and may only be made from the Plan in the frequency, form and manner provided by the Plan Administrator. Further, Participants may have an opportunity to repay a QBOAD if the Participant is eligible to make a rollover into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder. This Section 7.8(g) is effective June 1, 2021.
- **Coronavirus-Related Distributions (CRDs).** Participants may request and receive a Coronavirus-Related Distribution (CRD). CRDs must be requested and made in the form and manner provided by the Plan Administrator.
  - (1) <u>"Coronavirus-Related Distribution" Defined</u>. A CRD means a distribution from the Plan made between January 1, 2020 and December 31, 2020, to a Participant:
    - i who is diagnosed with the virus SARS-CoV-2 or the coronavirus disease 2019 (collectively, "COVID-19") by a test approved by the Centers for Disease Control and Prevention; or
    - ii whose spouse or dependent is diagnosed with COVID-19 by a test approved by the Centers for Disease Control and Prevention; or
    - iii who experiences adverse financial consequences as a result of a spouse or Participant being quarantined, furloughed or laid off, or having work hours reduced due to COVID-19; or being unable to work due to lack of childcare due to COVID-19 or closing or reducing hours of a business owned or operated by the spouse or Participant, or a member of the Participant's household (one who shares Participant's principal residence) due to COVID-19; or having a reduction in pay due to COVID-19 or having a job offer rescinded or start date for a job delayed due to COVID-19.
  - **Aggregate Dollar Limit.** The aggregate amount of CRDs to a Participant may not exceed \$100,000 from all plans sponsored by the Employer.
  - (3) Re-Contribution of CRDs. Participants may have an opportunity to repay some or all of the Participant's CRD if completed within 3 years of the distribution, meets the recontribution requirements in Section 2202(a) of the SECURE Act, and the Participant is eligible to make a rollover contribution into the Plan under Section 3.4 (Rollover Contributions) and the policies and procedures established thereunder.

- **Participant Certification.** The Plan Administrator may rely on a Participant's certification that he or she satisfies the conditions for a CRD.
- 50 Sources of Distribution. Unless otherwise indicated in the Plan, distributions will be made on a pro rata basis from all Accounts from which a distribution is permitted. Alternatively, to the extent permissible by law, the Plan Administrator may permit Participants to direct the Plan Administrator as to which Account the distribution is to be made as long as the directions provided by the Participant are from a source which such distributions are permitted. Regardless of a Participant's direction as to the source of any distribution, the tax effect of such a distribution will be governed by Code § 72 and the regulations thereunder. Further, Roth Deferrals and In-Plan Roth Rollovers that are distributed from the Plan within five years of the Participant's first designated Roth Deferral to the Plan will be subject to tax on the earnings portion of the distribution from the Roth Deferral Account or In-Plan Roth Rollover Account.
- **7.10** Purchase of Service Credit Under Defined Benefit Plan. Notwithstanding any provision of the Plan to the contrary, the Plan may permit a Participant whose Distribution Starting Date has not yet occurred to transfer as a direct plan-to-plan transfer, any portion of the Participant's Account to any other plan that is qualified under Code § 401(a) for the purpose of purchasing service credits under such other plan. Prior to permitting such transfer, however, the Plan shall reasonably determine all of the following:
  - (a) that such other plan shall accept such a transfer;
  - (b) that such transferred amounts are subject to withdrawal and distribution restrictions so that the Participant may not withdraw such transferred amounts prior to retirement, death, disability or severance from employment;
  - (c) that such transferred amounts are held for the benefit of the Participant as elective contributions on a fully vested and non-forfeitable basis; and,
  - (d) that applicable interest is credited to such transfer amounts.

The Plan Administrator shall establish a written policy for transfers for service credit purchases.

7.11 <u>Correction of Qualification Defects</u>. Nothing in this Section 7 precludes the Plan Administrator from making a distribution to a Participant to correct a qualification defect consistent with the correction procedures under IRS voluntary compliance programs. Thus, for example, if an Employee is permitted to enter the Plan prior to his/her proper Entry Date under Section 2.3 and the Plan Administrator determines that a corrective distribution is a proper means of correcting the operational violation, nothing in this Section 7 would prevent the Plan from making such corrective distribution. Any such distribution must be made in accordance with the correction procedures applicable under IRS voluntary correction programs under IRS Rev. Proc. 2013-12 (or successive guidance).

#### **SECTION 8**

### REQUIRED MINIMUM DISTRIBUTIONS

**Required Minimum Distributions (RMDs).** The provisions of this Section 8 apply to calendar years beginning on or after January 1, 2003. A Participant's entire interest under the Plan will be distributed, or

begin to be distributed, to the Participant no later than the Participant's Required Beginning Date. All distributions required under this Section 8 will be determined and made in accordance with the regulations under Code  $\S$  401(a)(9) and the minimum distribution incidental benefit requirement of Code  $\S$  401(a)(9)(G). For purposes of applying the required minimum distribution rules under this Section 8, any distribution made in a form other than a lump sum must be made over one of the following periods (or a combination thereof):

- (a) the life of the Participant;
- **(b)** the life of the Participant and a Designated Beneficiary;
- (c) a period certain not extending beyond the life expectancy of the Participant; or
- (d) a period certain not extending beyond the joint and last survivor life expectancy of the Participant and a Designated Beneficiary.
- **8.2 Death of Participant Before Required Distributions Begin.** If the Participant dies before required distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
  - Surviving Spouse as a Designated Beneficiary. If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, the surviving Spouse must take a distribution(s) under the five-year rule described in Section 8.6(a) below or under the life expectancy method. Distributions to the surviving Spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age 70½ (or age 72 if Participant attains age 70½ after December 31, 2019), if later. If such surviving Spouse does not make an election and a distribution must commence, the Plan will make distributions under the life expectancy method.
  - Non-Spousal Designated Beneficiary. If the Participant's surviving Spouse is not the Participant's sole Designated Beneficiary but the Plan segregates each Designated Beneficiary's Account, the Designated Beneficiary may elect to take distributions under the five-year rule described in Section 8.6(a) below or under the life expectancy method. If the non-Spousal Designated Beneficiary(ies) do not elect to commence distributions by December 31 of the calendar year immediately following the calendar year in which the Participant dies, distributions will be made under the life expectancy method.
  - (c) <u>No Designated Beneficiary</u>. If there is no Designated Beneficiary as of the date of the Participant's death who remains a Beneficiary as of September 30 of the year immediately following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
  - **Death of Surviving Spouse.** If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary and the surviving Spouse dies after the Participant but before distributions to the surviving Spouse begin, this Section 8.2 (other than subsection (a)) will apply as if the surviving Spouse were the Participant.

For distributions under this subsection (d), distributions are considered to begin on the date distributions are required to begin to the surviving Spouse under subsection (a) above.

- **Required Minimum Distributions During Participant's Lifetime.** During the Participant's lifetime, the minimum amount that must be distributed for each Distribution Calendar Year is described in this Section 8.3.
  - (a) Amount of Required Minimum Distribution for Each Distribution Calendar Year. During the Participant's lifetime, the minimum amount that will be distributed for each Distribution Calendar Year is the lesser of:
    - the quotient obtained by dividing the Participant's Account Balance by the distribution period set forth in the Uniform Lifetime Table found in Treas. Reg. § 1.401(a)(9)-9, using the Participant's age as of the Participant's birthday in the Distribution Calendar Year; or
    - if the Participant's sole Designated Beneficiary for the Distribution Calendar Year is the Participant's Spouse, the quotient obtained by dividing the Participant's Account Balance by the number in the Joint and Last Survivor Table set forth in Treas. Reg. § 1.401(a)(9)-9, Q&A-3, using the Participant's and Spouse's attained ages as of the Participant's and Spouse's birthdays in the Distribution Calendar Year.
  - **Lifetime Required Minimum Distributions Continue Through Year of Participant's Death.**Required Minimum Distributions will be determined under this subsection (b) beginning with the first Distribution Calendar Year and continuing up to, and including, the Distribution Calendar Year that includes the Participant's date of death.
- 8.4 Required Minimum Distributions After Participant's Death.
  - (a) <u>Death On or After Date Required Distributions Begin</u>. Where the Participant's death occurs on or after his Required Beginning Date and Required Minimum Distributions have already begun, the minimum distributions made to Beneficiaries after the Participant's death must be completed at least as rapidly as under the minimum distribution period in effect at Participant's death.
    - (1) Participant Survived by Designated Beneficiary. If the Participant dies on or after the date required distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the longer of the remaining life expectancy of the Participant or the remaining life expectancy of the Participant's Designated Beneficiary, determined as follows:
      - i. The Participant's remaining life expectancy is calculated in accordance with the Single Life Table found in Treas. Reg. § 1.401(a)(9)-9, Q&A-1, using the age of the Participant in the year of death, reduced by one for each subsequent year.
      - ii. If the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, the remaining life expectancy of the surviving Spouse is calculated using the Single Life Table found in Treas. Reg. § 1.401(a)(9)-9, Q&A-1, for each Distribution

Calendar Year after the year of the Participant's death using the surviving Spouse's age as of the Spouse's birthday in that year. For Distribution Calendar Years after the year of the surviving Spouse's death, the remaining life expectancy of the surviving Spouse is calculated using the age of the surviving Spouse as of the Spouse's birthday in the calendar year of the Spouse's death, reduced by one for each subsequent calendar year.

- iii. If the Participant's surviving Spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining life expectancy is calculated under the Single Life Table using the age of the Designated Beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
- (2) No Designated Beneficiary. If the participant dies on or after the date required distributions begin and there is no Designated Beneficiary as of the Participant's date of death who remains a Designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the Participant's remaining life expectancy under the Single Life Table calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- **(b)** Death Before Date Required Distributions Begin. Where the Participant's death occurs on or after his Required Beginning Date but his or her Required Minimum Distributions have not yet begun, the minimum distributions made to the Participant's Beneficiaries must be made in accordance with the rules in this Section 8.4(b).
  - (1) Participant Survived by Designated Beneficiary. If the Participant dies before the date required distributions begin, and there is a Designated Beneficiary, the minimum amount that will be distributed for each Distribution Calendar Year after the year of the Participant's death is the quotient obtained by dividing the Participant's Account Balance by the remaining life expectancy of the Participant's Designated Beneficiary, determined as provided in subsection 8.4 (a).
  - No Designated Beneficiary. If the Participant dies before the date distributions begin and there is no Designated Beneficiary as of the date of death of the Participant who remains a Designated Beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest must be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
  - (3) <u>Death of Surviving Spouse Before Distributions to Surviving Spouse are Required to Begin.</u> If the Participant dies before the date distributions begin, and the Participant's surviving Spouse is the Participant's sole Designated Beneficiary, and the surviving Spouse dies before distributions are required to begin to the surviving Spouse under Section 8.2(a),

then this subsection 8.4(b) will apply as if the surviving Spouse were the Participant.

### 8.5 **Definitions**.

- (a) <u>Designated Beneficiary</u>. A Beneficiary designated by the Participant (or the Plan), whose life expectancy may be taken into account to calculate minimum distributions, pursuant to Code § 401(a)(9) and Treas. Reg. § 1.401(a)(9)-4.
- (b) <u>Distribution Calendar Year</u>. A calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first Distribution Calendar Year is the calendar year immediately preceding the calendar year that contains the Participant's Required Beginning Date. For distributions beginning after the Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin pursuant to Section 8.2. The Required Minimum Distribution for the Participant's first Distribution Calendar Year will be made on or before the Participant's Required Beginning Date. The Required Minimum Distribution for the Distribution Calendar Years, including the Required Minimum Distribution for the Distribution Calendar Year in which the Participant's Required Beginning Date occurs, will be made on or before December 31 of that Distribution Calendar Year.
- **Life Expectancy.** For purposes of determining a Participant's Required Minimum Distribution amount, life expectancy is computed using one of the following tables, as appropriate: (1) Single Life Table, (2) Uniform Life Table, or (3) Joint and Last Survivor Table found in Treas. Reg. § 1.401(a)(9)-9.
- Account Balance. For purposes of determining a Participant's Required Minimum Distribution, the Participant's Account Balance is determined based on the Account Balance as of the last Valuation Date in the calendar year immediately preceding the Distribution Calendar Year (the "valuation calendar year") increased by the amount of any contributions or forfeitures allocated to the Account Balance as of dates in the calendar year after the Valuation Date and decreased by distributions made in the calendar year after the Valuation Date. The Account Balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the Distribution Calendar Year if distributed or transferred in the valuation calendaryear.
- **Required Beginning Date (RBD).** A Participant's Required Beginning Date under the Plan is April 1 that follows the end of the calendar year in which the later of the following two events occurs:
  - the Participant attains age  $70\frac{1}{2}$  (or age 72 if a Participant attains age  $70\frac{1}{2}$  after December 31, 2019), or
  - (2) the Participant terminates employment. For purposes of the Required Minimum Distribution, a Rehired Annuitant is considered terminated from employment.

A Participant may begin in-service distributions prior to his/her Required Beginning Date only to the extent authorized under Section 7.8 (In-Service Distributions).

## 8.6 **Special Rules**.

- (a) <u>Election to Allow Participants or Beneficiaries to Elect 5-Year Rule</u>. A Participant or Designated Beneficiary may elect whether the life expectancy rule under Section 8.2 or the five-year rule under subsection 8.6(a), no later than the earlier of September 30 of the calendar year in which distribution would be required to begin under Section 8.2 or by September 30 of the calendar year which contains the fifth anniversary of the Participant's (or, if applicable, surviving Spouse's) death.
- **(b)** Forms of Distribution. Unless the Participant's interest is distributed in a lump sum on or before the Required Beginning Date, as of the first Distribution Calendar Year distributions will be made in accordance with Sections 8.2 and 8.4.
- **Waiver of Required Minimum Distributions.** For calendar year 2009, the Required Minimum Distribution rules will not apply. In applying the provisions of this Section 8 for the 2009 Distribution Calendar Year,
  - the Required Beginning Date with respect to any individual shall be determined without regard to this subsection (c) for purposes of applying this paragraph for Distribution Calendar Years after 2009, and
  - required distributions to a beneficiary upon the death of the Participant shall be determined without regard to calendar year 2009.

A Participant or beneficiary who would have been required to receive a Required Minimum Distribution for the 2009 Distribution Calendar Year but for the enactment of Code § 401(a)(9)(H) ("2009 RMD"), may elect whether or not to receive the 2009 RMD (or any portion of such distribution). A distribution of the 2009 RMD or a series of substantially equal distributions (that include the 2009 RMDs) made at least annually and expected to last for the life (or life expectancy) of the participant, the joint lives (or joint life expectancy) of the participant and the participant's designated beneficiary, or for a period of at least 10 years, will be treated as an Eligible Rollover Distribution. However, if all or any portion of a distribution during 2009 is treated as an Eligible Rollover Distribution but would not be so treated if the Required Minimum Distribution requirements under this Section 8 had applied during 2009, such distribution shall not be treated as an Eligible Rollover Distribution for purposes of Code §§ 401(a)(31), 402(f) or 3405(c). (See Notice 2009-82 for transitional rules that apply for purposes of applying the rollover rules to the distribution of 2009 RMDs.).

(d) Treatment of Trust Beneficiaries As Designated Beneficiaries. If a trust is properly named as a Beneficiary under the Plan, the beneficiaries of the trust will be treated as the Designated Beneficiaries of the Participant solely for purposes of determining the distribution period under this Section 8 with respect to the trust's interests in the Participant's vested Account Balance. The beneficiaries of a trust will be treated as Designated Beneficiaries for this purpose only if, during any period during which required minimum distributions are being determined by treating the beneficiaries of the trust as Designated Beneficiaries, the following requirements are met:

- (1) the trust is a valid trust under state law, or would be but for the fact there is no corpus;
- the trust is irrevocable or will, by its terms, become irrevocable upon the death of the Participant;
- the beneficiaries of the trust who are beneficiaries with respect to the trust's interests in the Participant's vested Account Balance are identifiable from the trust instrument; and
- (4) the Plan Administrator receives the documentation described in subsection (e)(1) below.

If the foregoing requirements are satisfied and the Plan Administrator receives such additional information as it may request, the Plan Administrator may treat such beneficiaries of the trust as Designated Beneficiaries.

## (e) <u>Special Rules Applicable to Trust Beneficiaries</u>.

- (1) <u>Information That Must Be Supplied to Plan Administrator.</u>
  - i. Required Minimum Distribution Before Death Where Spouse Is Sole
    Beneficiary. If a Participant designates a trust as the beneficiary of his/her entire
    benefit and the Participant's Spouse is the sole beneficiary of the trust, the
    Participant must provide the information under (A) or (B) below to satisfy the
    information requirements under subsection 8.6(d)(4) above.
    - (A) The Participant must provide to the Plan Administrator a copy of the trust instrument and agree that if the trust instrument is amended at any time in the future, the Participant will, within a reasonable time, provide to the Plan Administrator a copy of each such amendment; or
    - **(B)** The Participant must:
      - (I) provide to the Plan Administrator a list of all of the beneficiaries of the trust (including contingent and remaindermen beneficiaries with a description of the conditions on their entitlement sufficient to establish that the Spouse is the sole beneficiary) for purposes of Code § 401(a)(9);
      - (II) certify that, to the best of the Participant's knowledge, the list under subsection (B)(I) above is correct and complete and that the requirements of subsection 8.6(d) above are satisfied;
      - (III) agree that, if the trust instrument is amended at any time in the future, the Participant will, within a reasonable time, provide to the Plan Administrator corrected certifications to the extent that the amendment changes any information previously certified; and
      - (IV) agree to provide a copy of the trust instrument to the Plan Administrator upon demand.

- ii. Required Minimum Distribution After Death. In order to satisfy the documentation requirement of subsection 8.6(d)(4) for required minimum distributions after the death of the Participant (or Spouse in a case to which U.S. Treas. Reg. § 1.401(a)(9)-3, Q&A-5 applies), the trustee of the trust must satisfy the requirements of subsection (A) or (B) by October 31 of the calendar year immediately following the calendar year in which the Participant died.
  - **(A)** The trustee of the trust must:
    - (I) provide the Plan Administrator with a final list of all beneficiaries of the trust (including contingent and remaindermen beneficiaries with a description of the conditions on their entitlement) as of September 30 of the calendar year following the calendar year of the Participant's death;
    - (II) certify that, to the best of the trustee's knowledge, the list in subsection (B)(I) is correct and complete and that the requirements of subsection 8.6(d) above are satisfied; and
    - (III) agree to provide a copy of the trust instrument to the Plan Administrator upon demand.
  - (B) The trustee of the trust must provide the Plan Administrator with a copy of the actual trust document for the trust that is named as a beneficiary of the Participant under the Plan as of the Participant's date of death.
- Relief for Discrepancy. If required minimum distributions are determined based on the information provided to the Plan Administrator in certifications or trust instruments described in subsection (1) above, the Plan will not fail to satisfy Code § 401(a)(9) merely because the actual terms of the trust instrument are inconsistent with the information in those certifications or trust instruments previously provided to the Plan Administrator, provided the Plan Administrator reasonably relied on the information provided and the required minimum distributions for calendar years after the calendar year in which the discrepancy is discovered are determined based on the actual terms of the trust instrument.
- **8.7 Transitional Rule.** Notwithstanding the other requirements of this Section 8, distribution on behalf of any Employee may be made in accordance with all of the following requirements (regardless of when such distribution commences):
  - (a) The distribution by the Plan is one that would not have disqualified the Plan under Code § 401(a)(9) as in effect prior to amendment by the Deficit Reduction Act of 1984.
  - **(b)** The distribution is in accordance with a method of distribution designated by the Participant whose interest in the Plan is being distributed or, if the Participant is deceased, by a Beneficiary of such Participant.
  - (c) Such designation was in writing, was signed by the Participant or the beneficiary, and was made

- before January 1, 1984.
- (d) The Participant had accrued a benefit under the Plan as of December 31,1983.
- (e) The method of distribution designated by the Participant or the beneficiary specifies the time at which distribution will commence, the period over which distributions will be made, and in the case of any distribution upon the Participant's death, the beneficiaries of the Participant listed in order of priority.

A distribution upon death will not be covered by this transitional rule unless the information in the designation contains the required information described above with respect to the distributions to be made upon the death of the Participant.

For any distribution which commences before January 1, 1984, but continues after December 31, 1983, the Participant, or the Beneficiary, to whom such distribution is being made, will be presumed to have designated the method of distribution under which the distribution is being made if the method of distribution was specified in writing and the distribution satisfies the requirements in subsections (a) - (e) above.

If a designation is revoked any subsequent distribution must satisfy the requirements of Code § 401(a)(9) and the proposed regulations thereunder. If a designation is revoked subsequent to the date distributions are required to begin, the Plan must distribute by the end of the calendar year following the calendar year in which the revocation occurs the total amount not yet distributed which would have been required to have been distributed to satisfy Code § 401(a)(9) and the proposed regulations thereunder, but for the TEFRA § 242(b)(2) election. For calendar years beginning after December 31, 1988, such distributions must meet the minimum distribution incidental benefit requirements. Any changes in the designation will be considered to be a revocation of the designation. However, the mere substitution or addition of another Beneficiary (one not named in the designation) under the designation will not be considered to be a revocation of the designation, so long as such substitution or addition does not alter the period over which distributions are to be made under the designation, directly or indirectly (for example, by altering the relevant measuring life). In the case in which an amount is transferred or rolled over from one plan to another plan, the rules in Treas. Reg. § 1.401(a)(9)-8, Q&A-14 and Q&A-15 shall apply.

### **SECTION 9**

#### **SPOUSAL CONSENT RULES**

- **9.1** Application of Joint and Survivor Annuity Rules. As a governmental plan, the Qualified Joint and Survivor Annuity rules under Code §§ 401(a)(11) and 417 do not apply to the Plan. Notwithstanding the foregoing, the District requires spousal consent for certain designations of Plan benefits.
- **Spousal Consent**. Spousal consent is only required with respect to the distribution of death benefits from the Plan if the Participant designates someone other than the Participant's Spouse as a Beneficiary of more than fifty percent (50%) of the Participant's Account Balance.

A Spousal consent to the Beneficiary designation of more than fifty percent (50%) of Participant's Account to a non-spouse beneficiary must be provided at the time and in the form and manner specified by the Plan Administrator.

<u>Subsequent beneficiary designations submitted to the Plan Administrator in good order revoke all designations filed prior to that date by the same Participant.</u>

Participant's marriage or remarriage revokes any prior Beneficiary designations made by the Participant and a new Beneficiary designation form that complies with the consent requirements of this Section 9.2 (Spousal Consent) must be submitted.

Further, when designating a trust or other entity as a Beneficiary, the time and form of distribution available to such Beneficiaries may be limited by the required minimum distribution rules of Code § 401(a)(9) and Section 8 (RMDs) of the Plan.

### **SECTION 10**

### PLAN ACCOUNTING AND INVESTMENTS

- **Participant Accounts.** The Plan Administrator will establish and maintain a separate Account for each Participant to reflect the Participant's entire interest under the Plan. Each Participant Account established by the Plan Administrator may include one or more of the following separate sub-Accounts:
  - Pre-Tax Salary Deferral Account
  - Matching Contribution Account
  - Rollover Contribution Account
  - Transfer Account

The Plan Administrator may establish other Accounts, as it deems necessary, for the proper administration of the Plan.

- **Valuation of Accounts.** A Participant's portion of the Trust assets is determined as of each Valuation Date under the Plan. The value of a Participant's Account consists of the fair market value of the Participant's share of the Trust assets. The Trustee must value Plan assets at least annually. The Trustee's determination of the value of Trust assets shall be final and conclusive.
  - **Periodic Valuation.** The Plan may elect to value assets on a periodic basis more frequent than annually including daily valuation. The Trustee and the Plan Administrator may adopt reasonable procedures for performing such valuations.
  - **Daily Valuation.** For any period in which the Plan Administrator uses daily valuation, the Plan Administrator may adopt reasonable procedures for performing such valuations. Unless otherwise set forth in the written procedures, a daily valued Plan will have its assets valued at the end of each business day during which the New York Stock Exchange is open. The Plan Administrator has authority to interpret the provisions of this Plan in the context of a daily valuation procedure. This includes, but is not limited to, the determination of the value of the Participant's Account for

- purposes of Participant loans, distribution and consent rights, and corrective distributions.
- **Interim Valuations.** The Plan Administrator may request the Trustee to perform interim valuations.
- **10.3** Adjustments to Participant Accounts. Unless the Plan Administrator adopts other reasonable administrative procedures, as of each Valuation Date under the Plan, each Participant's Account is adjusted in the following manner:
  - (a) <u>Net Income or Loss</u>. A Participant's Account will be adjusted accordingly for any net income or loss from Participant's selected investment vehicles.
  - **(b)** <u>Distributions and Forfeitures from A Participant's Account</u>. A Participant's Account will be reduced by any distributions, forfeitures and other reductions from the Account since the previous Valuation Date.
  - **(c) <u>Dividends.</u>** The Account will be credited with any interest, dividends, distributions received, or credits paid by the Trust for the benefit of the Participant.
  - (d) <u>Contributions and Forfeitures Allocated to A Participant's Account</u>. A Participant's Account will be credited with any contribution, forfeiture or other additions allocated to the Participant since the previous Valuation Date.
  - (e) <u>Directed Accounts</u>.
    - (1) Net income or loss attributable to the investments made by a Directed Account is allocated to such account in a manner that reasonably reflects the investment experience of such Directed Account.
    - Assets held in a Directed Account may be segregated from other investments held in Trust, and when this occurs, allocation of net income or loss shall not result in a Participant being entitled to distribution from the Account that exceeds the value of the Account as of the date of distribution.
  - **Plan Related Fees and Expenses**. A Participant's Account will be reduced by any and all reasonable expenses and fees related to plan administration and investment charges allocable to his or her Account.

- **10.4 Share or Unit Accounting.** The Plan's investment procedures may provide for share or unit accounting to reflect the value of Accounts, if such method is appropriate for the investments allocable to such Accounts.
- **10.5 Suspense Accounts.** The Plan's investment procedures also may provide for special valuation procedures for suspense accounts that are properly established under the Plan.
- **Deferred Compensation Investment Committee.** The DCIC has the full power and authority to invest, manage, and control Plan assets. The DCIC's membership, governance, and powers and duties are set forth in the Bylaws of the Deferred Compensation Investment Committee, as approved by the Board.
- **10.7 Investment Policy Statement.** The DCIC will establish and maintain an Investment Policy Statement (IPS) for the Plan, and will periodically review the policy, as needed. In establishing and maintaining the Plan's IPS, the DCIC will review all pertinent Plan and Participant information and data and may consult with any professional advisors as it deems necessary or appropriate, and will consider matters such as the following:
  - (a) the Plan's objectives and the best method to accomplish those objectives, including the range of investment alternatives that should be available to Participants and their Beneficiaries in order to accomplish those objectives;
  - **(b)** the policies or procedures to be followed by the DCIC in the selection and monitoring of investment alternatives, including the benchmarks and other performance criteria, as well as investment fees and expenses; and
  - **(c)** the liquidity of Plan assets.
- **Investment Manager.** The Board of Directors may appoint one or more investment managers to manage (including the power to acquire and dispose of) any assets of the Plan. Only a person who is a registered investment advisor under the Investment Advisors Act of 1940, a bank or an insurance company qualified to do business in the State of California may be an investment manager under the Plan. The Trustee may be an investment manager.
- 10.9 <u>Investment Funds</u>. The Trustee will establish and maintain investment funds designated by the DCIC or an investment manager designated by the Board of Directors. The assets of each investment fund will be managed by the Trustee or by one or more investment managers, as designated, except for participant-directed investments that permit Participants to direct the investment of some or all of the assets in their Account in specific investments.
- **Participant-Directed Investments.** The Plan permits Participant direction of investments. Each Participant shall have the exclusive right, in accordance with the provisions of the Plan, to direct the investment of all or a portion of the amounts allocated to the separate sub-Accounts of the Participant under the Plan. This Section 10.10 also applies to any Beneficiary or Alternate Payee eligible to direct investments under the Plan.
  - (a) <u>Limits on Participant Investment Direction</u>. The Plan's separate investment procedures may limit Participant direction of investment to specific types of contributions or with respect to specific investment options. If Participant investment direction is limited to specific investment options, it shall be the responsibility of the DCIC to select the investment options available under the Plan for

Participant investment. In no case may Participants direct that investments be made in collectibles, including but not limited to U.S. Government or State-issued gold and silver coins. In addition, investment directions that result in the following will not be followed:

- (1) result in a prohibited transaction;
- (2) cause the assets of the Plan to be maintained outside the jurisdiction of the U.S. courts;
- (3) jeopardize the Plan's tax qualification;
- **(4)** be contrary to the Plan's governing documents;
- (5) cause the assets to be invested in collectibles within the meaning of Code § 408(m);
- **(6)** generate unrelated business taxable income; or
- (7) result (or could result) in a loss exceeding the value of the Participant's Account.
- **(b)** Failure to Direct Investment. The DCIC will designate how a Participant Account shall be invested in the absence of proper affirmative direction from the Participant. The DCIC may designate a default fund under the Plan to deposit contributions to the Trust on behalf of a Participant who has not specified an investment choice under the Plan.

#### **SECTION 11**

### PLAN ADMINISTRATION AND OPERATION

- **Plan Administrator**. The District is the Plan Administrator and has delegated to the General Manager the powers, duties, and responsibilities of the Plan Administrator. The General Manager may delegate, in writing, some or all of the powers, duties and responsibilities of the Plan Administrator position as necessary to adequately operate and maintain the Plan.
- **Role of Plan Administrator.** Any Plan Administrator designated by the District must accept the Plan Administrator role and responsibilities in writing and will serve until his or her successor is appointed. In case of a vacancy in the position of the Plan Administrator, the District will exercise any and all of the powers, authority, duties and discretion as the Plan Administrator pending the filling of the vacancy.
  - (a) Resignation or Removal of Plan Administrator. A Plan Administrator may resign by delivering notice of resignation to the Employer. The Employer may remove a Plan Administrator by delivering notice of removal. If a Plan Administrator resigns or is removed and no new Plan Administrator is designated, the District becomes the Plan Administrator.
- 11.3 <u>Duties, Powers and Responsibilities of the Plan Administrator.</u>
  - **General Duties, Powers and Responsibilities.** The Plan Administrator shall have the responsibility and authority to administer the Plan and shall do so for the exclusive benefit of the Plan Participants and Beneficiaries, and in accordance with the terms of the Plan. The Plan Administrator shall have the authority to appoint or remove any Employee who has administrative

responsibilities under the Plan.

- **Delegation of Duties. Powers and Responsibilities.** The Plan Administrator may delegate its duties, powers or responsibilities to one or more persons. Such delegation must be in writing and accepted by the person or persons receiving the delegation. However, the Employer must agree to any such delegation in writing when made by a delegate of the Plan Administrator. The Plan Administrator has designated the Human Resources Group Manager to carry out the powers, duties and responsibilities of the Plan Administrator.
- **(c)** Specific Plan Administrator Responsibilities. The Plan Administrator has the general responsibility to control and manage the operation of the Plan. More specifically, this responsibility includes, but is not limited to, the following:
  - (1) To communicate with the Trustee and other responsible persons with respect to the crediting of Plan contributions, the disbursement of Plan distributions and other relevant matters;
  - (2) To develop separate procedures, if necessary, consistent with the terms of the Plan to assist in the administration of the Plan, including the adoption of a separate or modified loan policy, procedures for direction of investment by Participants, procedures for determining whether domestic relations orders are QDROs, and procedures for the determination of investment earnings to be allocated to Participants' Accounts;
  - (3) To maintain all records necessary for tax and other administration purposes;
  - (4) To furnish and to file all appropriate notices, reports and other information to Participants, Beneficiaries, the Employer, the Trustee and government agencies (as necessary);
  - (5) To provide information relating to Plan Participants and Beneficiaries;
  - (6) To retain the services of other persons, including, consultants, advisers and others, to assist in the administration of the Plan;
  - (7) To review and decide on administrative claims for benefits under the Plan;
  - (8) To correct any defect or error in the operation of the Plan; and
  - (9) To interpret and enforce the provisions of the Plan, including those related to Plan eligibility, vesting and benefits; to interpret and apply provisions of the Plan in a uniform, nondiscriminatory manner;
  - (10) To act upon instructions from the DCIC to direct the Trustee regarding changes to the Plan's investment options and to execute proxy votes cast by the DCIC;
  - (11) To request and receive input from the DCIC, when reasonable and appropriate, regarding non-investment related Plan features; selection, performance, and retention of the record keeper, investment advisors, service providers, advisors, consultants, and auditor; and Plan amendments; and

(12) To serve as the secretary for the DCIC which includes the following duties: keep and maintain the DCIC meeting minutes, execute documents on behalf of the DCIC, issue formal written correspondence of the DCIC; and provide other administrative support as the DCIC may reasonably delegate to the Plan Administrator.

The Plan Administrator shall coordinate with General Counsel concerning any litigation, compromise, or settlement of claims against the Plans in accordance with MWD Code Section 6433

- **Interpretation of Plan**. The Employer shall have the responsibility and authority, as an express grant of discretionary authority, where necessary, to interpret and resolve ambiguities in this Plan document, provided such interpretation is consistent with the rules of Code § 401(a), and to make discretionary decisions regarding the interpretation of the Plan's terms, and Participant's rights and status under the Plan including, but not limited to, who is eligible to participate under the Plan, and the benefit rights of a Participant or Beneficiary.
- **(e)** Notices and Filings. The Plan Administrator may modify any notice period required by the Plan or designate any office to serve as the recipient of any form or notice that has to be filed under the Plan.
- **Appropriate Adjustments.** Any action taken by the Plan Administrator with respect to 1) the rights or benefits under the Plan or 2) any Participant or Beneficiary shall be revocable by the Plan Administrator as to payments, distributions or deliveries not theretofore made hereunder pursuant to such action. Appropriate adjustments may be made in future payments or distributions to a Participant or Beneficiary.
- **11.4 Immunities.** The provisions of this Section 11.4 shall not be construed to in any way restrict the privileges and immunities that the Employer and its Employees are otherwise entitled to under Government Code Sections 818.8 and 820.2. Furthermore,
  - (a) Neither the District, its Board of Directors, nor its Employees shall be liable: 1) for any action taken or not taken with respect to the Plan including, but not limited to, any action allocated by the foregoing to another person or entity, except for its or their own gross negligence or willful misconduct; or 2) upon any contract, agreement or other instrument made or executed for administration of the Plan. With regards to the District's responsibility and authority to interpret the Plan, unless an interpretation or decision is determined to be arbitrary and capricious, the Plan Administrator will not be held liable for any interpretation of the Plan terms or decision regarding the application of a Plan provision.
  - (b) The District, its Board of Directors, each Employee thereof, specifically including the General Manager, and any other persons to whom the General Manager delegates or assigns any duty with respect to the Plan, may rely and shall be fully protected in acting in good faith upon: 1) the advice of counsel, who may be counsel for the District; 2) the records of the District; 3) the opinion, certificate, valuation, report, recommendation, or determination of an investment advisor, the DCIC or any member thereof, or the District's General Auditor; or, 4) any certificate, statement or other representation made by or any information furnished by a Participant, Beneficiary, Alternate Payee,

DCIC or any member thereof, or investment advisor, concerning any fact required to be determined under any of the provisions of the Plan.

## 11.5 Plan Administration Expenses.

- (a) Reasonable Plan Administration Expenses. All reasonable expenses related to plan administration will be paid from Plan assets, except to the extent the expenses are paid (or reimbursed) by the Employer. For this purpose, Plan expenses include, but are not limited to, all reasonable costs, charges and expenses incurred by the Trustee, DCIC, Plan Sponsor or Plan Administrator in connection with the administration of the Plan and Trust. If liquid assets of the Trust are insufficient to cover the fees of the Trustee or the Plan Administrator, then Trust assets shall be liquidated to the extent necessary for such fees. In the event any part of the Trust becomes subject to tax, all taxes incurred will be paid from the Trust.
- (b) Plan Expense Allocation. The Plan Administrator will allocate plan expenses among the accounts of Plan Participants. The Plan Administrator has authority to allocate these expenses either proportionally based on the value of the Account Balances or pro rata based on the number of Participants in the Plan. The Plan Administrator will determine the proper method for allocating expenses in accordance with such reasonable nondiscriminatory rules as the Plan Administrator deems appropriate under the circumstances. Unless the Plan Administrator decides otherwise, the following expenses will be allocated to the Participant's Account relative to which the expense is incurred: distribution expenses, including those relating to lump sums, installments, QDROs, hardship, in-service and required minimum distributions; loan expenses; participant direction expenses, including brokerage fees; and benefit calculations.
- **Expenses Related to Administration of Former Employee or Surviving Spouse.** If the Plan is making distributions to a former Employee or surviving Spouse, the Plan may charge reasonable Plan administrative expenses to the Account of that former Employee or surviving Spouse, but only if the administrative expenses are on a pro rata basis. Under the pro rata basis, the expenses are based on the amount in each account of a former Employee or surviving Spouse receiving benefits from the Plan. The Plan Administrator may use another reasonable basis for charging the expenses.

## 11.6 <u>Claims Procedure and Determinations</u>.

- (a) Benefit Claims Procedures. The Plan Administrator is authorized to conduct an examination of the relevant facts to determine the merits of a Participant's or Beneficiary's claim for Plan benefits. The Plan Administrator may establish procedures for administering benefit claims. Such benefit claims procedures should provide claimants with a reasonable opportunity to have a full and fair review of a denied claim. Any claims procedure will incorporate the guidelines under this Section 11.6. To the extent any of the time periods specified in this Section 11.6 are amended by law, the time frames specified herein shall automatically be changed in accordance with such law or regulation.
- **(b)** Benefit Claims Determinations. The Plan Administrator, in case of disputes, may make findings of fact and determinations with respect to any benefit claims made under the Plan. Subject to the

provisions of this Section 11.6, such determinations and findings shall be final and conclusive, to the extent permitted by law, as to all interested persons for all purposes of the Plan.

## 11.7 **Qualified Domestic Relations Orders (ODROs)**.

(a) In General. Upon receipt of an order which appears to be a QDRO, the Plan Administrator will notify the Participant involved and each Alternate Payee under the order. The Plan Administrator will determine whether the order is a QDRO and will notify each affected individual of such determination. The Plan Administrator may use the default QDRO procedures set forth in subsection 11.7(h) below or may develop separate QDRO procedures for administering any QDROs submitted under the Plan.

## (b) <u>Definitions Related to ODROs</u>.

- QDRO. A QDRO is a domestic relations order that creates or recognizes the existence of an Alternate Payee's right to receive, or assigns to an Alternate Payee the right to receive, all or a portion of the benefits payable with respect to a Participant under the Plan. (See Code § 414(p).) The QDRO must contain certain information and meet other requirements described in this Section 11.7.
- (2) <u>Domestic Relations Order</u>. A domestic relations order is a judgment, decree, or order (including the approval of a property settlement) that is made pursuant to state domestic relations law (including community property law).
- (3) <u>Alternate Payee</u>. An Alternate Payee must be a Spouse, former Spouse, child, or other dependent of a Participant.
- **Recognition as a QDRO.** To be a QDRO, an order must be a domestic relations order that relates to the provision of child support, alimony payments, or marital property rights for the benefit of an Alternate Payee. The Plan Administrator is not required to determine whether the court or agency issuing the domestic relations order had jurisdiction to issue an order, whether state law is correctly applied in the order, whether service was properly made on the parties, or whether an individual identified in an order as an Alternate Payee is a proper Alternate Payee under state law.

Effective April 6, 2007, a domestic relations order otherwise meeting the requirements to be a QDRO shall not fail to be treated as a QDRO solely because:

- (1) the order is issued after, or revises, another domestic relations order or QDRO; or
- of the time at which the order is issued, including orders issued after the death of the Participant.

Any QDRO described in this Section 11.7 shall be subject to the same requirements and protections which apply to QDROs under Code  $\S$  414(p)(7).

- **(d) Contents of ODRO.** A QDRO must contain the following information:
  - (1) the name and last known mailing address of the Participant and each Alternate Payee;

- (2) the name of each plan to which the order applies;
- (3) the dollar amount or percentage (or the method of determining the amount or percentage) of the benefit to be paid to the Alternate Payee; and
- (4) the number of payments or time period to which the order applies.

## (e) <u>Impermissible ODRO Provisions</u>.

- (1) The order must not require the Plan to provide an Alternate Payee or Participant with any type or form of benefit, or any option, not otherwise provided under the Plan;
- (2) The order must not require the Plan to provide for increased benefits (determined on the basis of actuarial value);
- (3) The order must not require the Plan to pay benefits to an Alternate Payee that are required to be paid to another Alternate Payee under another order previously determined to be a QDRO; and
- (4) The order must not require the Plan to pay benefits to an Alternate Payee in the form of a Qualified Jointand Survivor Annuity for the lives of the Alternate Payee and his or her subsequent Spouse.
- **Immediate Distribution to Alternate Payee.** Even if a Participant is not eligible to receive an immediate distribution from the Plan, an Alternate Payee may receive a QDRO benefit immediately in a lump sum, provided such distribution is consistent with the QDRO provisions.
- **(g) Fee for QDRO Determination.** The Plan Administrator shall condition the making of a QDRO determination on the payment of a fee by the Participant or an Alternate Payee either directly or as a charge against the Participant's Account.
- (h) <u>Default QDRO Procedure</u>. If the Plan Administrator chooses this default QDRO procedure or if the Plan Administrator does not establish a separate QDRO procedure, this subsection 11.7(h) will apply as the procedure the Plan Administrator will use to determine whether a domestic relations order is a QDRO. This default QDRO procedure incorporates the requirements set forth below.
- (i) Access to Information. The Plan Administrator will provide access to Plan and Participant benefit information sufficient for a prospective Alternate Payee to prepare a QDRO. Such information might include the summary plan description, other relevant plan documents, and a statement of the Participant's benefit entitlements. The disclosure of this information is conditioned on the prospective Alternate Payee providing to the Plan Administrator information sufficient to reasonably establish that the disclosure request is being made in connection with a domestic relations order.
  - (1) Notifications to Participant and Alternate Payee. The Plan Administrator will promptly notify the affected Participant and each Alternate Payee named in the domestic relations order of the receipt of the order. The Plan Administrator will send the notification to the address included in the domestic relations order. Along with the notification, the Plan

- Administrator will provide a copy of the Plan's procedures for determining whether a domestic relations order is a QDRO.
- (2) <u>Alternate Payee Representative</u>. The prospective Alternate Payee may designate a representative to receive copies of notices and Plan information that are sent to the Alternate Payee with respect to the domestic relations order.
- (3) Evaluation of Domestic Relations Order. Within a reasonable period of time, the Plan Administrator will evaluate the domestic relations order to determine whether it is a QDRO. A reasonable period will depend on the specific circumstances. The domestic relations order must contain the information described in subsection 11.7(d). If the order is only deficient in a minor respect, the Plan Administrator may supplement information in the order from information within the Plan Administrator's control or through communication with the prospective Alternate Payee.
  - **Separate Accounting.** Upon receipt of a domestic relations order, the Plan Administrator will separately account for and preserve the amounts that would be payable to an Alternate Payee until a determination is made with respect to the status of the order. During the period in which the status of the order is being determined, the Plan Administrator will take whatever steps are necessary to ensure that amounts that would be payable to the Alternate Payee, if the order were a QDRO, are not distributed to the Participant or any other person. The separate accounting requirement may be satisfied, at the Plan Administrator's discretion, by a segregation of the assets that are subject to separate accounting.
  - ii **Separate Accounting Until the End of 18-Month Period.** The Plan Administrator will continue to separately account for amounts that are payable under the QDRO until the end of an 18-month period. The 18-month period will begin on the first date following the Plan's receipt of the order upon which a payment would be required to be made to an Alternate Payee under the order. If, within the 18-month period, the Plan Administrator determines that the order is a QDRO, the Plan Administrator must pay the Alternate Payee in accordance with the terms of the QDRO. If, however, the Plan Administrator determines within the 18-month period that the order is not a QDRO, or, if the status of the order is not resolved by the end of the 18-month period, the Plan Administrator may pay out the amounts otherwise payable under the order to the person or persons who would have been entitled to such amounts if there had been no order. If the order is later determined to be a QDRO, the order will apply only prospectively; that is, the Alternate Payee will be entitled only to amounts payable under the order after the subsequent determination.
  - **Preliminary Review.** The Plan Administrator will perform a preliminary review of the domestic relations order to determine if it is a QDRO. If this preliminary review indicates the order is deficient in some manner, the Plan Administrator will allow

the parties to attempt to correct any deficiency before issuing a final decision on the domestic relations order. The ability to correct is limited to a reasonable period of time.

- **Notification of Determination.** The Plan Administrator will notify in writing the Participant and each Alternate Payee of the Plan Administrator's decision as to whether a domestic relations order is a QDRO. In the case of a determination that an order is not a QDRO, the written notice will contain the following information:
  - (A) references to the Plan provisions on which the Plan Administrator based its decision;
  - (B) an explanation of any time limits that apply to rights available to the parties under the Plan (such as the duration of any protective actions the Plan Administrator will take); and
  - (C) a description of any additional material, information, or modifications necessary for the order to be a QDRO and an explanation of why such material, information, or modifications are necessary.
  - (D) Treatment of Alternate Payee. If an order is accepted as a QDRO, the Plan Administrator will act in accordance with the terms of the QDRO as if it were a part of the Plan. Except as designated otherwise under this subsection 11.7(h)(4)(v), an Alternate Payee will be considered a Beneficiary under the Plan and be afforded the same rights as a Beneficiary. The Plan Administrator will provide any appropriate disclosure information relating to the Plan to the Alternate Payee. In determining the rights of an Alternate Payee, the following rules apply.
  - **(E) Loans.** An Alternate Payee is not permitted to take a loan from the Plan.
  - (F) <u>Death Benefits</u>. If an Alternate Payee dies prior to receiving the entire amount designated under the QDRO, such benefits will be paid in accordance with Section 7.7, treating the Alternate Payee as the Beneficiary. If the Alternate Payee dies without a designated Beneficiary, the benefits will be paid to the Alternate Payee's estate. Any death benefit will be paid in a single sum as soon as administratively feasible after the Alternate Payee's death.
  - **(G)** <u>Direction of Investments</u>. An Alternate Payee has the right to direct the investment of the portion of the Participant's benefit that is segregated for the Alternate Payee's benefit pursuant to a QDRO in the same manner as the Participant.
- **Military Service.** To the extent required under Code § 414(u), an Employee who returns to employment with the Employer following a period of qualified military service will receive any contributions, benefits

and service credit required under Code § 414(u), provided the Employee satisfies all applicable requirements under the Code and regulations. In determining the amount of contributions under Code § 414(u), Plan Compensation will be deemed to be the compensation the Employee would have received during the period while in military service based on the rate of pay the Employee would have received from the Employer but for the absence due to military leave. If the compensation the Employee would have received during the leave is not reasonably certain, Plan Compensation will be equal to the Employee's average compensation from the Employer during the twelve (12) month period immediately preceding the military leave or, if shorter, the Employee's actual period of employment with the Employer.

- **Death Benefits Under Qualified Military Service**. In the case of a Participant who dies while performing qualified military service (as defined in Code § 414(u)), the survivors of the Participant are entitled to any additional benefits (other than benefit accruals relating to the period of qualified military service) provided under the Plan as though the Participant resumed and then terminated employment on account of death. This provision is effective with respect to deaths occurring on or after January 1, 2007.
- (b) <u>Plan Distributions</u>. Notwithstanding the provisions of Section 1.54(c) regarding the treatment of Differential Pay, an individual shall be treated as having been severed from employment during any period the individual is performing service in the Uniformed Services for purposes of receiving a Plan distribution under Code § 401(k)(2)(B)(i)(I). If an individual elects to receive a distribution while on military leave, the individual may not make Salary Deferrals under the Plan during the 6-month period beginning on the date of the distribution.
- (c) Make-Up Contributions. A Participant who is reemployed following a qualified military leave shall have the right to make up any Salary Deferrals to which he/she would have been entitled but for the fact the Participant was on qualified military leave. The Employer will also make any Matching Contributions the Participant would have earned during the period of qualified military leave had the Participant remained employed during such period. The Employer will only be required to make Matching Contributions if the reemployed Participant makes up the underlying contributions that were eligible for the Matching Contributions.

In determining the amount of Make-Up Contributions a Participant may make under this subsection (d), a Participant will be treated as earning Plan Compensation during the period the Participant was on qualified military leave equal to:

- (1) the rate of pay the Participant would have received from the Employer during such period had the Participant not been on qualified military leave, or
- (2) if the Plan Compensation the Participant would have received during such period was not reasonably certain, the Participant's average Plan Compensation during the 12-month period immediately preceding the qualified military leave (or the entire period of employment, if shorter).

If the Employer is required under this subsection (d) to make Employer Contributions for a reemployed Participant, the Employer must make such Employer Contributions not later than

90 days after the date of reemployment or the date the Employer Contributions are otherwise due for the year in which the military service was performed. For Salary Deferrals and After-Tax Employee Contributions, a Participant who is reemployed following a qualified military leave may make up such contributions during the period beginning on the date of reemployment and ending on the earlier of the date that is three times the length of the military service period or 5 years from the date of reemployment. Any required Matching Contributions must be made in the same manner as other Matching Contribution under the Plan following the Participant's contribution of the amounts eligible for the Matching Contributions.

Any make up contributions under this subsection 11.8(c) are subject to Code § 415 Limitation under Section 5.2 (Elective Deferral Dollar Limit) for the year for which the make-up contribution would have been made had the Participant not been on qualified military leave.

#### **SECTION 12**

#### TRUST FUNDS

**12.1 Establishment of Trust.** In conjunction with the establishment of this Plan, a domestic Trust consisting of such sums as shall from time to time be paid to the Trust under the Plan and such earnings, income and appreciation as may accrue thereon will be established pursuant to a written Trust Agreement.

The District may enter into a Trust Agreement with the Trustee providing for the investment of the Trust and prescribing the powers, duties, obligations, and functions of the Trustee with respect to the Plan. The Trust Agreement will form a part of this Plan, and any rights and benefits of any Participant or Beneficiary under the Plan whose Accounts are invested in the Trust will be subject to the terms of the Trust Agreement. However, in the case of any conflict between the Plan and the Trust Agreement, the Plan will govern. The Trustee must be one or more banks or other fiduciaries appointed by the District to hold and administer assets of the Plan.

The Trustee must hold the Plan assets received by it in Trust pursuant to the terms of the Trust Agreement for the exclusive benefit of Participants and their Beneficiaries and defraying the reasonable expenses of administering the Plan. Except as provided in Section 15.2, no assets of the Plan shall inure to the benefit of the Employer prior to the satisfaction of all liabilities with respect to Participants and their Beneficiaries.

The Trustee must hold, manage, administer, value, invest, reinvest and account for the Trust assets in accordance with the terms of the Plan and the Trust Agreement.

In its discretion, the Plan Administrator may recommend that the District terminate the Trustee, appoint new Trustees, and authorize the Plan Administrator to instruct the Trustee transfer assets to a successor Trustee or Trustees.

**12.2 Common/Collective Trusts**. Subject to Plan Administrator's direction, the Trustee may commingle, for investment or administration purposes the assets (or portion of the assets) of the Trust in any common trust fund, collective investment fund, master trust or group trust, but only if the commingled trust satisfies the requirements under IRS Revenue Ruling 81-100 (as modified by Rev. Rul. 2004-67 and Rev. Rul. 2011-1) to be tax exempt.

#### **SECTION 13**

#### PARTICIPANT LOANS

**13.1 Availability of Participant Loans.** In accordance with Code § 72(p), the District may permit Participants to take loans from their vested Account Balance under the Plan.

To receive a Participant loan, a Participant must sign a promissory note along with a pledge or assignment of the portion of the Account Balance used for security on the loan. The loan will be evidenced by a legally enforceable agreement which specifies the amount and term of the loan, and the repayment schedule.

A loan policy shall be established to determine the terms, conditions and restrictions applicable to such loans as well as the time and manner in which loans may be made available. The loan policy will ensure that all loans made under this Plan meet the requirements of Code § 72(p) and its underlying regulations.

- **Must be Available in Reasonably Equivalent Manner.** Participant loans must be made available to Participants in a reasonably equivalent manner. The loan policy may limit the availability of Participant loans to specified events.
- **13.3 Loan Limitations.** A Participant loan may not be made to the extent such loan (when added to the outstanding balance of all other loans made to the Participant) exceeds the lesser of:
  - (a) Fifty-thousand dollars (\$50,000) (reduced by the excess, if any, of the Participant's highest outstanding balance of loans from the Planduring the one-year period ending on the day before the date on which such loan is made, over the Participant's outstanding balance of loans from the Plan as of the date such loan is made) or
  - (b) One-half (½) of the Participant's vested Account Balance, determined as of the Valuation Date coinciding with or immediately preceding such loan, adjusted for any contributions or distributions made since such Valuation Date.

Notwithstanding the above described loan limitation, a Participant may take a loan equal to the greater of \$10,000 or fifty percent (50%) of the Participant's vested Account Balance. However, if a Participant takes a loan in excess of (50%) of the Participant's vested Account Balance, such loan is still subject to the adequate security requirements under Section 13.7 (Adequate Security).

In applying the limitations under this Section 13.3 (Loan Limitations), all plans maintained by the District, including an eligible governmental 457(b) plan, are aggregated and treated as a single plan. In addition, any assignment or pledge of any portion of the Participant's interest in the Plan and any loan, pledge, or assignment with respect to any insurance contract purchased under the Plan will be treated as loan under this Section.

In applying the Loan Limitations described in this Section 13.3, the Participant's Roth Deferral Account and In-Plan Roth Rollover Account will be aggregated with all of the Participant's other Accounts from which loans are permissible.

**Limit on Amount and Number of Loans.** A Participant may not receive a Participant loan of less than \$1,000. Participant may not have more than two Participant loans outstanding at any time.

Notwithstanding the foregoing sentences, the Plan Administrator may amend the minimum loan amount and maximum number of outstanding loans by amending the loan policy.

- 13.5 Participant Must Be Creditworthy. The Plan Administrator may refuse to make a loan to any Participant who is determined to not be creditworthy. For this purpose, a Participant is not creditworthy if, based on the facts and circumstances, it is reasonable to believe that the Participant will not repay the loan. A Participant who has defaulted on a previous loan from the Plan (or on a loan from another plan sponsored by the District) and has not repaid such loan (with accrued interest) will be treated as not creditworthy until such time as the Participant repays the defaulted loan (with accrued interest).
- **13.6** Reasonable Rate of Interest. All Participant loans will be charged a commercially reasonable rate of interest. The interest rate assumptions will be periodically reviewed to ensure the interest rate charged on Participant loans is reasonable.

If a Participant is in military service while he/she has an outstanding Participant loan, the applicable interest charged on such loan during the period while the Participant is in military service will not exceed six percent (6%) per year provided the Participant provides written notice and a copy of his/her call-up or extension orders to the Plan Administrator within 180 days following the Participant's termination or release from military service. For this purpose, military service is as defined in the Soldier's and Sailor's Civil Relief Act of 1940 as modified by the Service members Civil Relief Act of 2003. The Participant may voluntarily waive this six percent (6%) interest limitation and the Plan Administrator may petition the court to retain the original interest rate if the ability to repay is not affected by the Participant's activation to military duty.

- Adequate Security. All Participant loans must be adequately secured. Generally, the Participant's vested Account Balance shall be used as security for a Participant loan provided the outstanding balance of all Participant loans made to such Participant does not exceed: 1) fifty percent (50%) of the Participant's vested Account Balance, or 2) the Participant's vested Account Balance, where the Participant seeks a loan equal to the greater of \$10,000 or fifty percent (50%) of the Participant's vested Account Balance, in either case, as determined immediately after the origination of each loan. The Plan's loan policy shall prescribe rules for defining and obtaining adequate security for loans under the Plan.
- 13.8 Periodic Repayment. A Participant loan must provide for level amortization with payments to be made not less frequently than quarterly. A Participant loan must be repaid in full within a period not exceeding five (5) years from the date the Participant receives the loan from the Plan, unless the loan is for the purchase of the Participant's principal residence, in which case the loan must be repaid within fifteen (15) years or such other period that is commensurate with the repayment period permitted by commercial lenders for similar loans, permissible by the Code § 72(p) and set out in the Plan's loan policy. To the extent a loan is made from a Participant's Roth Deferral Account, the loan repayment requirement must be satisfied separately with respect to that portion of the loan and the portion of the loan from the Participant's other Accounts. The Plan's loan policy shall establish the method loans shall be repaid. The loan policy shall address loan repayments during leave of absence periods in a manner that is consistent with the applicable tax rules.

13.9 Procedures for Loan Default. A Participant will be considered to be in default with respect to a loan if any scheduled repayment with respect to such loan is not made by the end of the calendar quarter following the calendar quarter in which the missed payment was due. Once a loan is in default, the total outstanding balance of the loan will be deemed distributed and reported on an IRS Form 1099-R for the calendar year in which the Participant defaulted on the loan. In accordance with the Code, the deemed distribution of a defaulted loan will not be designated as a qualified distribution from a Roth Deferral Account or In-Plan Roth Rollover Account.

If a Participant defaults on a Participant loan, the Plan may not offset the Participant's Account Balance until the Participant is otherwise entitled to an immediate distribution of the portion of the Account Balance which will be offset and such amount being offset is available as security on the loan, pursuant to Section 13.7 (Adequate Security). For this purpose, a loan default is treated as an immediate distribution event to the extent the law does not prohibit an actual distribution of the type of contributions which would be offset as a result of the loan default). The Participant may repay the outstanding balance of a defaulted loan (including accrued interest through the date of repayment) at any time.

Pending the offset of a Participant's Account Balance following a defaulted loan, the following rules apply to the amount in default.

- (a) Interest continues to accrue on the amount in default until the time of the loan offset or, if earlier, the date the loan repayments are made current or the amount is satisfied with other collateral.
- **(b)** A subsequent offset of the amount in default is not reported as a taxable distribution, except to the extent the taxable portion of the default amount was not previously reported by the Plan as a taxable distribution.
- (c) The post-default accrued interest included in the loan offset is not reported as a taxable distribution at the time of the offset.

#### 13.10 <u>Termination of Employment</u>.

- (a) <u>Outstanding Loan</u>. The loan policy may provide Participant's with an opportunity to continue to repay the loan after termination of employment. The terms and conditions of repayment upon termination of employment shall be established by the Plan Administrator.
- (b) <u>Direct Rollover</u>. Upon termination of employment, a Participant may request a Direct Rollover of the loan note (provided the distribution is an Eligible Rollover Distribution as defined in Section 7.4) to another qualified plan which agrees to accept a Direct Rollover of the loan note. A Participant may not engage in a Direct Rollover of a loan to the extent the Participant has already received a deemed distribution with respect to such loan.
- **13.11** Amendment of Plan to Eliminate Participant Loans. The Plan may be amended at any time to eliminate Participant loans on a prospective basis. However, the elimination of a Participant loan feature may not result in the acceleration of payment of any existing Participant loans, unless the terms of the Participant loan permit such acceleration.

#### **SECTION 14**

#### PLAN AMENDMENT, TERMINATION, MERGERS AND TRANSFERS

#### 14.1 Plan Amendment.

- Authority to Amend. The Employer has the right, at any time and from time to time, to amend the Plan in whole or in part. Any amendment of the Plan by the Employer shall be expressed in an instrument executed by the Employer on the order of its Board of Directors and filed with the Trustee. Further, the Plan Administrator is also authorized, at any time and from time to time, to amend the Plan in whole or in part, to ensure compliance with applicable laws and regulations and further the administration of the Plan. The Plan Administrator may not amend the Plan to effectuate its termination.
  - (1) Amendment Purposes Limited. No Plan amendment may authorize or permit any portion of the assets held under the Plan to be used for or diverted to a purpose other than the exclusive benefit of Participants or Beneficiaries, except to the extent such assets are used to pay taxes or administrative expenses of the Plan. An amendment also may not cause or permit any portion of the assets held under the Plan to revert to or become property of the District.
  - **Method of Amendment**. The Plan Administrator may amend the Plan in writing and without amendment of the MWD Code.
- **Effective Date of Plan Amendments.** If the Plan is restated or amended, such restatement or amendment is generally effective as of the Effective Date of the restatement or amendment except where the context indicates a reference to an earlier Effective Date. The District may designate special effective dates for individual provisions under the Plan.
  - required to comply with IRS guidance or law), the provisions of the Plan as amended generally override the provisions of any prior Plan. However, if the provisions of the Plan as amended are different from the provisions of the District's prior plan and, after the retroactive Effective Date of the Plan as amended, the District operates in compliance with the provisions of the prior plan, the provisions of such prior plan are incorporated into the Plan for purposes of determining whether the District operated the Plan in compliance with its terms, provided operation in compliance with the terms of the prior plan do not violate any qualification requirements under the Code, regulations, or other IRS guidance.
  - (2) Retroactive Effect of PPA, HEART and WRERA Provisions. This Plan is designed to comply with the Code, regulations, and general guidance applicable to qualified retirement plans, including the provisions of the Pension Protection Act of 2006 (PPA), the Heroes Earnings Assistance And Relief Tax Act Of 2008 (HEART Act), and the Worker, Retiree, and Employer Recovery Act of 2008 (WRERA). If this Plan is being restated or amended to comply with the provisions of PPA, HEART and/or WRERA, the Plan contains special effective dates that apply with respect to such provisions. If the Plan is being restated

within the remedial amendment period for retroactive compliance with the PPA, HEART and WRERA provisions, the special effective dates for such provisions (as described below) will apply, even if such special effective dates precede the Effective Date of this restatement. Thus, if the Plan is being restated or amended to comply with PPA, HEART and/or WRERA, and the Effective Date of this restatement or amendment is later than the special effective date applicable to any of the PPA, HEART or WRERA provisions described below, such special effective dates will apply and any prior plan being replaced by this Plan will be considered to have been timely amended for the PPA, HEART and WRERA provisions.

The following provisions contain special effective dates for purposes of complying with the requirements of PPA, HEART and WRERA:

- i. <u>Hardship Distributions</u>. Section 7.8(e) of this Plan allows Hardship distributions to be determined with respect to primary Beneficiaries.
- **ii.** <u>Direct Rollovers By Non-Spouse Beneficiaries</u>. The provisions allowing for direct rollovers by non- Spouse beneficiaries are effective for distributions made on or after January 1, 2007.
- **Direct Rollover of Non-Taxable Amounts.** Effective for taxable years beginning on or after January 1, 2007, the definition of Eligible Rollover Distribution to include the portion of a distribution that is not includible in gross income.
- **iv.** Rollovers to Roth IRA. For distributions occurring on or after January 1, 2008, Participants or Beneficiaries are permitted to rollover a qualified Eligible Rollover Distribution to a Roth IRA.
- v. <u>Distribution Notice Periods</u>. Effective for Plan Years beginning on or after January 1, 2007, the period for providing the Code § 402(f) rollover notice and the period for providing the notice regarding the right to defer receipt of a distribution is increased to 180 days.
- vi. Content of Notice of A Participant's Right to Defer Receipt of A Distribution.

  Effective for Plan Years beginning on or after January 1, 2007, notice relating to a

  Participant's right to defer receipt of a distribution must include a description of the
  consequences of a Participant's decision not to defer the receipt of a distribution.
- vii. Qualified Domestic Relations Orders. The Plan expands the definition of a QDRO effective April 6, 2007 to include modified orders and orders issued after the Participant's death.
- **Penalty-Free Withdrawals for Individuals Called to Active Duty.** Effective September 11, 2001, Qualified Reservist Distribution may be taken from elective deferrals.
- ix. <u>Benefit Accruals for Participants on Qualified Military Service</u>. The HEART Act provisions addressing Participants on qualified military leave are effective for Plan

- Years beginning on or after January 1,2007.
- x. <u>Differential Pay</u>. Effective for years beginning on or after January 1, 2009, Section 1.54(c) of this Plan permits the Employer to include Differential Pay as Total Compensation under the Plan.
- **xi. Waiver of Required Minimum Distributions.** Waiver of the Required Minimum Distribution rules for calendar year 2009 as prescribed under WRERA is available.
- **xii. Final Code § 415 Regulations.** Provisions required by the final Code § 415 regulations are effective for Limitation Years beginning on or after July 1,2007.
- (3) <u>Merged Plans</u>. Except for retroactive application of the provisions under this subsection 14.1(d), if one or more qualified retirement plans have been merged into this Plan, the provisions of the merging plan(s) will remain in full force and effect until the Effective Date of the Plan merger(s).
- **14.2 Plan Termination.** The District may terminate (partial or full termination) this Plan at any time by Board resolution.
  - **Full and Immediate Vesting.** Upon a full or partial termination of the Plan all amounts credited to an affected Participant's Account become one-hundred percent (100%) vested. The Plan Administrator has discretion to determine whether a partial termination has occurred.
  - (b) <u>Distribution Upon Plan Termination</u>. Upon the termination of the Plan, the Plan Administrator shall direct the distribution of Plan assets to Participants in accordance with the provisions under Section 7 (Plan Distribution). For purposes of applying the provisions of this subsection 14.2(b), distribution may be delayed until the District receives a favorable determination letter from the IRS, or an equivalent, as to the qualified status of the Plan upon termination, provided the determination letter request is made within a reasonable period following the termination of the Plan. Until all Plan assets have been distributed from the Plan, the District must amend the Plan in order to comply with current laws and regulations and may take any other actions necessary to retain the qualified status of the Plan.
  - (c) <u>Missing Participants</u>. Upon termination of the Plan, if any Participant cannot be located after a reasonable diligent search as defined in Section 6 (Participant Vesting and Forfeitures), the Plan Administrator may make a direct rollover to an IRA selected by the Plan Administrator. For this purpose, the Plan Administrator will adopt procedures for making automatic rollovers. An Automatic Rollover under this subsection 14.2(c) may be made on behalf of any missing Participant, regardless of the value of his/her vested Account Balance under the Plan.
  - (d) <u>Partial Termination</u>. In determining whether a Plan has experienced a partial termination as described under Code § 411(d)(3), the Plan Administrator will apply the principals set forth under IRS Revenue Ruling 2007-43.
  - **Employee Rights.** The exercise of such right by the District to terminate the 401(k) or 457(b) Plans does not preclude employees or their representatives from either consulting or meeting and

conferring with the Employee Relations Officer, as appropriate, about the practical consequences that the exercise of such power or authority may have on wages, hours, and other terms and conditions of employment subject to MWD Code Section 6102 (Employee Rights).

- Plan Merger or Consolidation. In the event the Plan is merged or consolidated with another plan, each Participant must be entitled to a benefit immediately after such merger or consolidation that is at least equal to the benefit the Participant was entitled to immediately before such merger or consolidation (had the Plan terminated). If the District amends the Plan from one type of Defined Contribution Plan (e.g., a Profit Sharing Plan) into another type of Defined Contribution Plan (e.g., a Money Purchase Plan) will not result in a partial termination or any other event that would require full vesting of some or all Plan Participants.
- 14.4 Transfer of Assets. The Plan may accept a transfer of assets from another qualified retirement plan on behalf of any Employee, even if such Employee is not eligible to receive other contributions under the Plan. If a transfer of assets is made on behalf of an Employee prior to the Employee's becoming a Participant, the Employee shall be treated as a Participant for all purposes with respect to such transferred amount. Any assets transferred to this Plan from another plan must be accompanied by written instructions designating the name of each Employee for whose benefit such amounts are being transferred, the current value of such assets, and the sources from which such amounts are derived. The Plan Administrator will deposit any transferred assets in the appropriate Participant's Transfer Account. The Transfer Account will contain any sub-Accounts necessary to separately track the sources of the transferred assets. Each sub-Account will be treated in the same manner as the corresponding Plan Account.

Any amounts transferred (and any gains attributable to such transferred amounts) continue to be subject to the distribution restrictions applicable to plan assets under the transferor plan. Such amounts may not be distributed for reasons other than death, disability, attainment of Normal Retirement Age, attainment of age 59½, or termination of employment, regardless of any distribution provisions under this Plan that would otherwise permit a distribution prior to such events.

The Plan Administrator may refuse to accept a transfer of assets if the Plan Administrator reasonably believes the transfer: (1) is not being made from a proper qualified plan; (2) could jeopardize the tax-exempt status of the Plan; or (3) could create adverse tax consequences for the Plan or the District. Prior to accepting a transfer of assets, the Plan Administrator may require evidence documenting that the transfer of assets meets the requirements of this Section 14.

- **Trustee's Right to Refuse Transfer.** If the assets to be transferred to the Plan under this Section 14.4 are not susceptible to proper valuation and identification or are of such a nature that their valuation is incompatible with other Plan assets, the Trustee may refuse to accept the transfer of all or any specific asset, or may condition acceptance of the assets on the sale or disposition of any specific asset.
- **Transfer of Plan to Unrelated Employer.** The Employer may not transfer sponsorship of the Plan to an unrelated employer if the transfer is not in connection with a transfer of business assets or operations from the Employer to the unrelated employer.

#### **SECTION 15**

#### **MISCELLANEOUS**

- **15.1 Exclusive Benefit.** Plan assets will not be used for, or diverted to, a purpose other than the exclusive benefit of Participants or Beneficiaries.
- 15.2 Return of Employer Contributions. Upon written request by the Employer, the Trustee must return any Employer Contributions provided that the circumstances and the time frames described below are satisfied. Any Employer Contributions made because of a mistake of fact must be returned to the Employer within one year of the contribution. The Trustee may request the Employer to provide additional information to ensure the amounts may be properly returned. Any amounts returned shall not include earnings, but must be reduced by any losses.
- Participants' Rights Limited. The adoption of this Plan by the District does not give any Participant, Beneficiary, or Employee a right to continued employment with the District and does not affect the District's right to discharge an Employee or Participant at any time. This Plan also does not create any legal or equitable rights in favor of any Participant, Beneficiary, or Employee against the District, Plan Administrator, DCIC, or Trustee. Unless the context indicates otherwise, any amendment to this Plan is not applicable to determine the benefits accrued, or the extent to which such benefits are vested, by a Participant or former Employee whose employment terminated before the effective date of such amendment, except where application of such amendment to the terminated Participant or former Employee is required by statute, regulation or other guidance of general applicability. Where the provisions of the Plan are ambiguous as to the application of an amendment to a terminated Participant or former Employee, the Plan Administrator has the authority to make a final determination on the proper interpretation of the Plan.

#### 15.4 Alienation.

- (a) Except as otherwise provided in subsection (c), no benefit which shall be payable out of the Trust to any person (including a Participant or Beneficiary) shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, or charge, and any attempt to anticipate, alienate, sell, transfer, assign, pledge, encumber, or charge the same shall be void; and no such benefit shall in any manner be liable for, or subject to, the debts, contracts, liabilities, engagements, or torts of any such person, nor shall it be subject to attachment or legal process for or against such person, and the same shall not be recognized by the Trustee, except to such extent as may be required by law.
- (b) In the event a Participant's benefits are garnished or attached by order of any court, the District may bring an action for a declaratory judgment in a court of competent jurisdiction to determine the proper recipient of the benefits to be paid by the Plan. During the pendency of said action, any benefits that become payable shall be paid into the court as they become payable, to be distributed by the court to the recipient it deems proper at the close of said action. Any costs attributable to these actions may be charged to the Participant Account.

- (c) This Section shall not apply to the creation, assignment or recognition of a right to any benefit payable pursuant to a domestic relations order, unless such order is determined to be a QDRO by the Plan Administrator, as defined in Code § 414(p). The Trustee shall be entitled to make distributions to an alternate payee pursuant to a QDRO in accordance with such an Order, without regard to the age or employment status of the Participant.
- 15.5 Receipt and Release for Payments. Any payment to any Participant, Participant's legal representative, Beneficiary, or to any guardian or committee appointed for such Participant or Beneficiary in accordance with the provisions of the Plan, shall, to the extent thereof, be in full satisfaction of all claims hereunder against the Trustee and the District, either of whom may require such Participant, legal representative, Beneficiary, guardian or committee, as a condition precedent to such payment, to execute a receipt and release thereof in such form as shall be determined by the Trustee or District.
- **Determination.** The District, in case of disputes, may make determinations and findings of fact with respect to any matter arising in connection with the administration of the Plan. Such determinations and findings shall be final and conclusive, to the extent permitted by the Plan and by law, as to all interested persons for all purposes of the Plan. The District shall instruct the Plan Administrator and Trustee as to any final determination made hereunder with information as reasonably required to administer the Plan in accordance with the final determination.
- 15.7 No Guarantee Against Loss. The District does not guarantee the Participants' Accounts or investments or any part thereof against loss or depreciation. All persons having any interest in the Accounts or investments shall look solely to such sources for payment with respect to such interest. No action by the District shall be considered to be either an endorsement or guarantee of any investment option, nor shall it be considered to attest the financial soundness or the suitability of any investment for the purpose of meeting future obligations provided for in the Plan.
- **15.8 Gender and Number.** Wherever any words are used herein in the masculine, feminine or neuter gender, they shall be construed as though they were also used in another gender in all cases where they would so apply, and whenever any words are used herein in the singular or plural form, they shall be construed as though they were also used in the other form in all cases where they would so apply.
- **15.9** <u>Use of IRS Compliance Programs</u>. Nothing in this Plan document should be construed to limit the District's ability to correct any tax-qualification error using whatever corrective actions or principals permitted under the IRS's Employee Plans Compliance Resolution System.
- **15.10 Governing Law**. The provisions of this Plan shall be construed, administered, and enforced in accordance with the provisions of applicable Federal Law and, to the extent applicable, the laws of the state of California.
- **15.11 Waiver of Notice.** Any person entitled to a notice under the Plan may waive the right to receive such notice, to the extent such a waiver is not prohibited by law, regulation or other pronouncement.
- **15.12** <u>Use of Electronic Media</u>. The District, Plan Administrator, Trustee and any other designated individual responsible for providing applicable notices or disclosures under the Plan, and any Participant or beneficiary making an election under the Plan may use telephonic or electronic media to satisfy any notice

requirements required by this Plan. Any use of electronic medium under the Plan must comply with the requirements outlined in Treas. Reg. § 1.401(a)-21 or other general guidance concerning the use of telephonic or electronic media. The Plan Administrator also may use telephonic or electronic media to conduct plan transactions such as enrolling participants, making (and changing) salary reduction elections, electing (and changing) investment allocations, applying for Plan loans, and other transactions, to the extent permissible under regulations (or other generally applicable guidance).

- **15.13 Severability of Provisions.** In the event that any provision of this Plan shall be held to be illegal, invalid or unenforceable for any reason, the remaining provisions under the Plan shall be construed as if the illegal, invalid or unenforceable provisions had never been included in the Plan.
- **15.14 Binding Effect.** The Plan, and all actions and decisions made thereunder, shall be binding upon all applicable parties, and their heirs, executors, administrators, successors and assigns.
- **15.15 Headings.** The headings and subheadings of this Plan have been inserted for convenience of reference and are to be ignored in any construction of the provisions hereof.
- **15.16** <u>Uniformity</u>. All provisions of this Plan shall be interpreted and applied in a uniform, nondiscriminatory manner.
- **15.17 Indemnification.** The District will indemnify, defend, and hold harmless the members of the Board of Directors, the General Manager, the Plan Administrator, the DCIC or any member thereof, or any Employees who have administrative responsibility under the Plan with respect to any liability, loss, damage, or expense resulting from any act or omission, except willful misconduct or gross negligence, in their official capacities in the administration of or management of the assets of the Plan, including attorney, accountant and advisory fees and all other expenses reasonably incurred in their defense.

The amended and restated Plan is hereby adopted, effective January 1, 2024.

#### The Metropolitan Water District of Southern California

Ву:	Katano Kassaine
Assistant Gener	al Manager/Chief Financial Officer and
Int	terim Plan Administrator
Date Signed	

	2024 Plan Amendment Summary Charts				
	401(k) Plan				
Sec	tion Subse	ction Summary of Ch	nange	Notes	
Bac	-2 und	Specifies amen	dment and restatement		
1.1	#	Adds Roth Elec Rollover accou	tive Deferral and Roth nts		
1.23	1	Clarifies "Eligib excluded	le Employee" who is not		
1.44	4	Defines Qualifi Distributions (0	ed Birth or Adoption QBOAD)	Secure Act* Optional provision	
1.50	0	Defines "Roth	Deferrals"		
1.52	2	Amends "Salar include Roth D	y Deferrals" definition to eferrals		
2.2	(a)(6)		s ineligible to participate Im work hours and months		
3.2	Pream	ble Adds Roth Defe	errals, specifies effective		
	(a)	Clarifies must o	designate Roth Deferrals		
	(b)	Clarifies deferr applies to Roth	al election change rule Deferrals		
	(c)	Clarifies Catch- consist of Roth	Up Contributions may Deferrals		
3.4			er Contributions to accept lovers, specifies separate		
3.5	(a)-(c)		errals, specifies separate ran leave rules, and		
3.6	(a)-(h)	Adds In-Plan Ro	oth Conversions Accounts,		
5.2	(b)	Adds ordering Correction rule	rule to Excess Deferral		
7.3	Pream		ntary Cash-Out rule nd Pre-Tax Accounts		

7.4	(a)(1)v.	Deletes the \$200 minimum rollover rule	
	(a)(2)	Clarifies direct Roth-to-Roth rollover distribution restriction	
	(b)	Deletes the \$200 minimum rollover rule	
7.8	(e)(1)(ii)(B)	Clarifies all available distributions obtained	
7.8	(e)(2)	Clarifies may receive hardship distributions from Roth Deferrals	
	(g)	Adds Qualified Birth or Adoption Distributions (QBOADs)	Secure Act* Optional provision
	(h)	Adds Coronavirus-Related Distributions (CRDs)	CARES Act* Optional provision Expired 12.31.2020
7.9		Adds the 5-year period of participation rule for Roth Deferral and Rollover Accounts	
8.2	(a)	Increases RBD from age 70 <sup>©</sup> to age 72	Secure Act* Mandatory provision
	(b)	Aligns life expectancy default rule with 457(b) plan for non-spousal designated beneficiaries	Cross ref: 457(b) § 4.3(D)(2)(a)
8.5	(e)(1)	Increases RBD from age 70 <sup>©</sup> to age 72	Secure Act* Mandatory provision
9.2		Clarifies and conforms to spousal consent and beneficiary revocation requirements	Cross ref: 457(b) § 6.1
11.3	(c)(7)	Clarifies Plan Administrator's authority to determine administrative claims	
	Postamble	Clarifies General Counsel's litigation and settlement authority per Adm Code § 6433	Legal procedure
11.4	(a)	Clarifies Plan Administrator's authority to interpret plan	
11.6	(b)	Deletes administrative claim instructions, not applicable to plan	
13.3	Postamble	Clarifies Roth Accounts will be aggregated with Pre-Tax Accounts to compute loan limitations	

<u> </u>			
13.8		Clarifies loan repayments for loans from Roth Deferral Accounts must be satisfied separately	
13.9		Applies loan default rules to Roth Accounts	
401(k) Plan Loan P	olicy Amend	dments 2021 - CARES Act	
Section	Subsection	Summary of Change	Notes
Article XIII		Adds COVID-19 LOAN PROGRAM	CARES Act* Optional Provision Executed 08.01.2021 Expired 12.31.2020
401(k) Plan Restru	cture Amend	dments	
Section	Subsection	Summary of Change	Notes
Background		Updates the amendment and restatement effective date, January 1, 2024	
1.11		Deletes "Custodian" trustee since directed-Trustee will replace; adds DCIC referring to 10.6 description	
1.56		Revises "Trust" definition to accommodate Trust Agreement and incorporate directed-Trustee functions	
1.57		Adds "Trust Agreement" definition to allow Trust Agreement with directed- Trustee	
1.58		Modifies "Trustee" definition to incorporate directed-Trustee functions	
10.6		Describes the "DCIC" referring to bylaws; replaces Plan Administrator with DCIC to align investment duties	
10.7		Adds IPS definition and periodic review; replaces Plan Administrator with DCIC to align IPS duties with DCIC bylaws; moves Common Collective Trusts and Collectibles to Section 12.2	
10.8		Adds authority of Board of Directors to hire an Investment Manager	
10.9		Adds investment funds as it relates to directed-Trustee function	

10.10		Replaces Plan Administrator with DCIC to align SDBA duties	
11.2		Moves indemnification section to Section 15.17	
11.3	(a) and (c)(6)	Removes Plan Administrator authority to hire investment managers and attorneys	
11.3	(c)(10) thru (c)(11	Adds DCIC duties and responsibilities to align with DCIC Bylaws	
12		Deletes "Trust Provisions" and renames Section 12 "Trust Funds"	
12.1		Adds authority to enter Trust Agreement and describes directed-Trustee duties; establishes that Plan Administrator may recommend termination of Trustee	
12.2 thru 12.12		Deletes provisions regarding former in- house, discretionary Trustee duties	
12.2		Relocates Common Collective Trusts and Collectibles from Section 10.7	
14.2	(e)	Reiterates BUs right to meet and confer if plans are terminated	
15.3		Adds DCIC to the MWD parties who do not acquire legal rights upon plan adoption	
15.17		Relocates and unifies indemnification section for all MWD indemnified parties	
		457(b) Plan	
Section	Subsection	Summary of Change	Notes
Back- ground	-2	Specifies amendment and restatement dates	
1.8		Specifies Effective Date	
1.22		Defines "Qualified Birth or Adoption Distribution" (QBOAD)	Secure Act* Optional Provision
1.26		Defines "Roth Elective Deferrals"	
1.28		Amends "Salary Reduction Contribution" definition to include Roth Elective Deferrals	

7-11

3.1	(A)	Adds Roth Elective Deferrals, specifies Effective Date**	
3.6	(A)	Clarifies and conforms to practice of allowing former employees to roll-in	Cross ref: 401(k) § 3.4
	(C)	Amends Rollover Contributions to accept Roth direct rollovers, specifies separate accounting	
3.9	(A)-(C)	Specifies Roth Elective Deferrals rules, designation, separate accounting, distribution ordering rules, and loans, specifies Effective Date, February 18, 2024**	
3.10	(A)-(H)	Adds In-Plan Roth Conversion Accounts, specifies applicable rules	
4.3	(B)(2)(a)	Increases Required Beginning Date (RBD) from age 70 <sup>©</sup> to age 72	Secure Act* Mandatory Provision
	(E)(5)	Increases RBD from age 70 <sup>©</sup> to age 72	Secure Act* Mandatory Provision
4.5	(C)	Reduces in-service distributions minimum age from age 70 <sup>©</sup> to age 59 <sup>©</sup>	Bipartisan Miners Act* Optional Provision Effective 01.01.2020
	(E)	Adds Qualified Birth or Adoption Distributions (QBOADs)	Secure Act* Optional Provision
	(F)	Adds Coronavirus-Related Distributions (CRDs)	CARES Act* Optional Provision Expired 12.31.2020
	(G)	Specifies in-service distributions shall be made pro rata across contribution sources and investment options	Letter of Direction 06.28.2022
4.7	(C)	Specifies separate treatment of Roth and Pre-Tax Accounts for automatic rollover rule purposes	
	(E)(2)	Clarifies direct Roth-to-Roth rollover distribution rules	
5.3	(A)(7)	Clarifies Plan Administrator's authority to determine administrative claims	
	Postamble	Clarifies General Counsel's litigation and settlement authority per Adm Code § 6433(a)	Legal procedure

	6.1		Clarifies and conforms to spousal consent and beneficiary revocation requirements	Cross ref: 401(k) § 9.2
	9.7		Conforms to MWD's indemnification standard per Adm Code § 6815	Legal standard Cross ref: 401(k) § 12.9
457(	b) Plan Loan P	olicy Amend	lments 2021 - CARES Act	
	Section	Subsection	Summary of Change	Notes
	Article XIII		Adds COVID-19 LOAN PROGRAM	CARES Act* Optional Provision Executed 08.01.2021 Expired 12.31.2020
457(	b) Plan Restru	cture Amend	dments	
	1.7		Defines "DCIC"	
	1.34		Modifies "Trust" definition to accommodate Trust Agreement and incorporate directed-Trustee functions	
	1.35		Adds "Trust Agreement" definition to allow agreement with directed-Trustee	
	1.36		Modifies "Trustee" definition to incorporate directed-Trustee functions	
	Art. V		Deletes "Plan Administrator Duties With Respect To Participants' Accounts" and renames Article V. "Plan Administration"	
	5.3		Aligns Plan Administrator powers and duties to the DCIC Bylaws	
	5.4		Reaffirms payment of reasonable expenses to Plan Administrator; bar on additional compensation for DCIC moved to the DCIC Bylaws Article VII, Section 1.	
	5.10		Aligns IPS duties of the DCIC with the DCIC Bylaws	
	Art. VII		Deletes "Trust Provisions" and renames Art. VII "Trust Funds"	
	7.1		Clarifies establishment and parameters of the Trust, affirms directed-Trustee as the trustee; deletes discretionary Trustee functions; establishes that Plan Administrator may recommend termination of Trustee	

7.2	Describes directed-Trustee's core duties, holding and investing funds
7.4	Describes the DCIC and incorporates the DCIC Bylaws
7.5	Aligns IPS functions to the DCIC investment duties; includes considerations
7.6	Adds authority of Board of Directors to hire Investment Manager
7.7	Describes relationship of directed- Trustee to the DCIC
7.8	Aligns duties SDBA related duties to the DCIC investment duties
7.4 thru 7.20	Deletes former, in-house discretionary Trustee duties
8.1(C)	Reiterates BUs right to meet and confer if plans are terminated
9.1	Moves IRS tax levy provision here for payment of participant taxes
9.7	Unifies indemnification section for all MWD indemnified parties

<sup>\*</sup>Deadline to amend plan for Secure Acts 1.0 and 2.0, CARES Act and BiPartisan Miner's Act is March 31, 2027 per IRS Notice 2022-33 (August 3, 2022)

<sup>\*\*</sup>Deadline to amend plan for ROTHs is December 31, 2024 if implemented in 2024.



#### Legal and Claims Committee

# Approve Changes to the Deferred Compensation Plans

Item 7-11 February 13, 2024

## Item 7-ll Establish Committee and Amend Plan Documents

### Subject

Establish an investment committee and amend the Plans and Code

#### Purpose

Delegate investment matters to a committee and conform the Plans and Code to this change <u>and</u> to current law, practices and regulations

#### Recommendation and Fiscal Impact

Approve: (1) changes to the MWD 457(b) and 401(k) deferred compensation plans to establish an investment committee and adopt investment committee bylaws; and (2) amendments necessary to conform the plans and the MWD Administrative Code to the proposed changes, current law, practices, and regulations.

#### Item 7-11

Investment Committee Members

- HR Group Manager
- AGM/CFO or Finance Group Manager
- Treasury and Debt Manager
- AGM/Operations
- General Auditor
- Risk Manager
- One appointee from each bargaining unit (4 total)
- One qualified plan retiree

# Item 7-11 InvestmentRelated Decisions

- Review and update the investment policy
- Evaluate core fund line up
- Determine investment-related features
- Evaluate service provider fees and expenses
- Voting proxy votes

## Item 7-11 Plan Administrator Duties

- Review and update the Plans
- Design communication campaigns
- Procure/monitor plan service providers
- Decide death benefit claims
- Offer non-investment features

## Item 7-11 No Change to Board Role

- Retains ultimate authority
- Delegates its fiduciary duties
- Approves major amendments

# Item 7-11 Plan and Code Amendments

- Conform plans to federal laws
- Establish Roth accounts
- Dissolve current governance structure
- Establish investment committee
- Clarify and update plan rules

## Item 7-11 Board Options

## Option # 1

Approve: (1) changes to the Metropolitan Water District 457(b) and 401(k) deferred compensation plans to establish an investment committee and adopt investment committee bylaws; and (2) amendments necessary to conform the plans and the Metropolitan Administrative Code to the proposed changes, current law, practices, and regulations.

## Option # 1

Do not approve





#### Legal & Claims Committee

## Update on Delta Conveyance Project CEQA Litigation

Item 7b February 13, 2024

### Item 7b Introduction

#### Subject

Update on nine cases challenging the California
 Department of Water Resources' compliance with the
 California Environmental Quality Act and other
 environmental laws regarding its certification of a Final
 Environmental Impact Report for and approval of the
 Delta Conveyance Project

#### Purpose

Provide an initial overview of the parties and causes of action

#### **Next Steps**

Regular updates in written GC reports and committee

## Background

## Project Approval and CEQA Actions Dec. 21, 2023

- DWR certified the Final EIR
- Adopted CEQA findings, a statement of overriding considerations and a Mitigation Monitoring and Reporting Program
- Approved the DCP (Alt 5)
- Filed a Notice of Determination → 30-day statute of limitations

Nine Cases (DWR is the only named defendant)

- City of Stockton v. DWR
- County of Butte v. DWR
- County of Sacramento v. DWR
- County of San Joaquin et al. v. DWR
- Sacramento Area Sewer District v. DWR
- San Francisco Baykeeper, et al. v. DWR
- Sierra Club, et al. v. DWR
- Tulare Lake Basin Water Storage District v. DWR
- South Delta Water Agency & Rudy Mussi Investm't L.P. v. DWR

Sacramento County

> San Joaquin County

#### 34 Plaintiffs

- 13 Environmental NGOs
- 5 Delta Counties
- 5 Fishing NGOs
- 5 In-Delta Agencies
- 2 SWP Contractors
  - Butte County
  - Tulare Lake Basin WSD
- 1 Tribe
- 1 Tribal Environmental NGO
- 1 City (Stockton)
- 1 Delta Landowner

AquAlliance
California Indian Environmental Alliance
California Sportfishing Protection Alliance
California Water Impact Network
Center for Biological Diversity
Central Delta Water Agency
City of Stockton
Contra Costa Water Agency
County of Butte
County of Contra Costa
County of Sacramento
County of San Joaquin
County of Solano
County of Yolo
Environmental Council of Sacramento
Friends of Stone Lakes National Wildlife Refuge
Friends of the River
Golden State Salmon Association
Institute for Fisheries Resources
Local Agencies of the North Delta
North Coast Rivers Alliance
Pacific Coast Federation of Fishermen's Associations
Planning and Conservation League
Restore the Delta
Rudy Mussi Investment L.P.
Sacramento Area Sewer District
Sacramento Audubon Society
San Francisco Baykeeper
Save Our Sandhill Cranes
Shingle Springs Band of Miwok Indians
Sierra Club
South Delta Water Agency
The Bay Institute
Tulare Lake Basin Water Storage District

## Causes of Action

- CEQA
- Delta Reform Act
- Public Trust Doctrine
- Delta and Watershed Protection Acts
- Fully Protected Species Statute (Birds)
- Prop. 9
  - 1982 referendum on Peripheral Canal
- Central Valley Project Act
  - Exceeds DWR's authority to modify the SWP

## Next Steps

### Cases in very early stage

- Cases will be coordinated or consolidated before one judge
- Feb. 16 case management conference in the 8 Sacramento cases
- DWR will prepare the Administrative Record
- Regular updates in the monthly GC Report and at Legal & Claims

